#### ABSTRACT

# Title of dissertation:ADVENTURES ON NETWORKS:<br/>DEGREES AND GAMES

Siddharth Pal

#### Dissertation directed by: Professors A.M. Makowski and R.J. La Dept. of Electrical and Computer Engineering

A network consists of a set of nodes and edges with the edges representing pairwise connections between nodes. Examples of real-world networks include the Internet, the World Wide Web, social networks and transportation networks often modeled as random graphs. In the first half of this thesis, we explore the degree distributions of such random graphs. In homogeneous networks or graphs, the behavior of the (generic) degree of a single node is often thought to reflect the degree distribution of the graph defined as the usual fractions of nodes with given degree. To study this preconceived notion, we introduce a general framework to discuss the conditions under which these two degree distributions coincide asymptotically in large random networks. Although Erdős-Rényi graphs along with other well known random graph models satisfy the aforementioned conditions, we show that there might be homogeneous random graphs for which such a conclusion may fail to hold. A counterexample to this common notion is found in the class of random threshold graphs. An implication of this finding is that random threshold graphs cannot be used as a substitute to the Barabási-Albert model for scale-free network modeling, as proposed in some works.

Since the Barabási-Albert model was proposed, other network growth models were introduced that were shown to generate scale-free networks. We study one such basic network growth model, called the *fitness model*, which captures the inherent attributes of individual nodes through fitness values (drawn from a fitness distribution) that influence network growth. We characterize the tail of the network-wide degree distribution through the fitness distribution and demonstrate that the fitness model is indeed richer than the Barabási-Albert model, in that it is capable of producing power-law degree distributions with varying parameters along with other non-Poisson degree distributions.

In the second half of the thesis, we look at the interactions between nodes in a game-theoretic setting. As an example, these nodes could represent interacting agents making decisions over time while the edges represent the dependence of their payoffs on the decisions taken by other nodes. We study learning rules that could be adopted by the agents so that the entire system of agents reaches a desired operating point in various scenarios motivated by practical concerns facing engineering systems. For our analysis, we abstract out the network and represent the problem in the strategic-form repeated game setting.

We consider two classes of learning rules – a class of better-reply rules and a new class of rules, which we call, the class of monitoring rules. Motivated by practical concerns, we first consider a scenario in which agents revise their actions asynchronously based on delayed payoff information. We prove that, under the better-reply rules (when certain mild assumptions hold), the action profiles played by the agents converge almost surely to a pure-strategy Nash equilibrium (PSNE) with finite expected convergence time in a large class of games called generalized weakly acyclic games (GWAGs). A similar result is shown to hold for the monitoring rules in GWAGs and also in games satisfying a payoff interdependency structure. Secondly, we investigate a scenario in which the payoff information is unreliable, causing agents to make erroneous decisions occasionally. When the agents follow the better-reply rules and the payoff information becomes more accurate over time, we demonstrate the agents will play a PSNE with probability tending to one in GWAGs. Under a similar setting, when the agents follow the monitoring rule, we show that the action profile weakly converges to certain characterizable PSNE(s). Finally, we study a scenario where an agent might erroneously execute an intended action from time to time. Under such a setting, we show that the monitoring rules ensure that the system reaches PSNE(s) which are resilient to deviations by potentially multiple agents.

Adventures on Networks: Degrees and Games

by

### Siddharth Pal

Dissertation submitted to the Faculty of the Graduate School of the University of Maryland, College Park in partial fulfillment of the requirements for the degree of Doctor of Philosophy 2015

Advisory Committee: Professor A.M. Makowski, Chair/Advisor Professor R.J. La, Co-Advisor Professor M. Rotkowitz Professor B. Babadi Professor L. Koralov © Copyright by Siddharth Pal 2015

#### Acknowledgments

I would like to express my sincere gratitude to my advisor Prof. Armand Makowski for his continuous support during my years as a graduate student. I am thankful for the learning experience I have had by sharing detailed discussions with him. I would also like to thank my co-advisor Prof. Richard La for giving me the opportunity to work on interesting and challenging problems. I am grateful to both my advisors for their valuable time and effort. I would like to take the opportunity to acknowledge the support of the following grants: NSF CCF-0830702, NSF CCF-1217997, NSF CCF-0830675 and NIST Contract 70NANB13H012.

My sincere thanks to Prof. Behtash Babadi, Prof. Michael Rotkowitz and Prof. Leonid Koralov for sparing their invaluable time and agreeing to serve on my thesis committee. I am also thankful to the rest of the faculty at the Department of Electrical and Computer Engineering for making the environment intellectually stimulating.

I thank my fellow officemates for all the fun and lively discussions. Also, I thank my friends at the University of Maryland for their support, encouragement and companionship.

Finally, I thank my family and parents for their unconditional love, support, guidance and encouragement throughout my graduate student life.

## Table of Contents

List of	Figures
---------	---------

Ι	Stud	v of Degrees on Networks	1
1	Netv	orks & Degrees	2
2	Degi	ee Distribution of Networks	9
	2.1	A general framework	9
	2.2	The main result	11
	2.3	Concerning Assumption 2.3	13
	2.4	Erdős-Rényi Graphs	15
	2.5	The generic setting	18
	2.6	Random Key Graphs	24
	2.7	Random Geometric Graphs	27
	2.8	Proofs of Propositions 2.1 and 2.2	28
		2.8.1 A proof of Proposition 2.1	30
		2.8.2 A proof of Proposition 2.2	31
	2.9	A proof of Theorem 2.4	32
		2.9.1 Some well known results	32
		2.9.2 The proof	34
	2.10	A proof of Theorem 2.6	39
		2.10.1 Preliminaries	39
		2.10.2 The proof $\ldots$	40
3	A C	unterexample: Random Threshold Graphs	15
	3.1	The model	15
		3.1.1 Applying Proposition 2.1 under exponential fitness	17
	3.2	A proof of Proposition 3.2 – Part I	51
		3.2.1 A reduction step	51
		3.2.2 Order statistics to the rescue	54
	3.3	Asymptotic results for order statistics	57

		3.3.1 A result in one dimension	57
		3.3.2 A result in two dimensions	59
	3.4	A proof of Proposition 3.2 – Part II	62
		3.4.1 The case $d = 0$	63
		3.4.2 The case $d = 1, 2,$ .	64
	3.5	A proof of Proposition 3.6	67
		3.5.1 Preliminaries	67
		3.5.2 Proving Proposition 3.6	69
4	Mor	e on Random Threshold Graphs: Weak Convergence	71
	4.1	Degree distribution – Generic fitness distribution	72
	4.2	The main result	76
	4.3	A roadmap to a proof of Theorem 4.2	77
	4.4	A proof of Theorem 4.2	80
	4.5	A proof of Proposition 4.5 – A reduction step	82
	4.6	A proof of Proposition 4.5 – A decomposition	84
	4.7	A proof of Proposition 4.5 – Taking the limit	88
		4.7.1 A useful intermediary fact	88
		4.7.2 In the limit $\ldots \ldots \ldots$	90
	4.8	A proof of Proposition 4.6	92
5	Degi	ee Distribution in Growth Models	94
	5.1	The model	94
	5.2	The convergence results	97
		5.2.1 The main result $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$ $\ldots$	98
		5.2.2 An alternate expression for the limiting pmf $p_{\xi}$	99
	5.3	Tail behavior	100
	5.4	Special cases when $\xi^{\star} = \infty$	102
		5.4.1 Pareto distribution	103
		5.4.2 Exponential distribution	104
	5.5	A proof of Theorem 5.1 – Preliminaries	105
	5.6	Useful technical facts	107
		5.6.1 A consequence of the Strong Law of Large Numbers	107
		5.6.2 Uniform selection	108
		5.6.3 Limits of certain expectations	109
	5.7	A proof of Theorem 5.1	111
		5.7.1 Establishing the upper bound $(5.52)$	113
		5.7.2 Establishing the lower bound $(5.51)$	114
	5.8	A proof of Proposition 5.10	115
	5.9	A proof of Proposition 5.11	117
	5.10	A proof of Lemma 5.4	121
	5.11	A proof of Lemma 5.6	124

Π	Games on Networks	138
6	Learning in Games	139
	6.1 Summary of Contributions	144
	6.2 Related Literature	146
7	Class of Better reply rules	151
	7.1 Stage and Repeated Games	152
	7.2 Generalized weakly acyclic games	154
	7.3 Proposed Update Rules - Generalized Better Reply Rules	158
	7.4 Convergence under GBRR – Synchronous Update Case	164
	7.5 Convergence under GBRR – Asynchronous Update Case	167
	7.5.1 Model with asynchronous updates and delays	167
	7.5.2 Convergence of action profiles with asynchronous updates a	nd
	delays	172
	7.6 Case with Erroneous Payoff Information	173
	7.6.1 Preliminaries	174
	7.6.2 Main results	178
	7.7 Proof of Theorem 7.5	181
	7.8 Proofs of Lemmas introduced in Section 7.7	184
	7.8.1 Proof of Lemma 7.9	184
	7.8.2 Proof of Lemma 7.10	186
	7.9 Proof of Theorem 7.8	187
	7.10 Proof of Lemma 7.11	189
8	Class of Monitoring Bules	192
0	8.1 Proposed Undate Bule	102
	8.1.1 Action updates	104
	8.1.2 State dynamics	195
	8.2 Convergence under RSEM	107
	8.3 Proof of Theorem 8.3	200
	8.4 Proof of Lemmata for Theorem 8.3	203
	8.4.1 Proof of Lemma 8.5	203
	8.4.2 Proof of Lemma 8.6	205
	8.4.3 Proof of Lemma 8.7	210
	8.5 Case with Erroneous Pavoff Information	210 212
	8.5.1 Proliminarios	· · · 212 919
	$8.5.2$ Main result $\ldots$	212
9	Monitoring Rules under Erroneous Execution	216
	9.1 Resilience of PSNEs	216
	9.2 Proposed Update Rule	217
	9.2.1 Action updates with no errors $-\mathbf{a}(t), t \in \mathbb{N}$	219
	9.2.2 Adopted action profiles $-\tilde{\mathbf{a}}(t), t \in \mathbb{N}$	220
	9.2.3 State dynamics	221

	9.3	Main Result	 							•		. 22	23
	9.4	Proof of Theorem 9.1	 									. 22	26
	9.5	Proof of Lemma $9.2$ .	 									. 23	33
10	Futu	re work										24	41
Bil	oliogr	aphy										$2^2$	43

# List of Figures

7.1	An example of a GWAG that is not a WAG. (a) Game in normal form,
	(b) all possible unilateral deviations and simultaneous deviations 155
7.2	Example – two interacting systems controlled by separate agents 168
7.3	Forward and feedback delays in the system
9.1	State transition diagram in SEM
9.2	An example of a minimum resistance $\mathbf{z}^*$ -tree, $\Gamma^*(\mathbf{z}^*)$ , and the corre-
	sponding Steiner tree $\mathbf{S}(\Gamma^*(\mathbf{z}^*))$

Part I

Study of Degrees on Networks

#### Chapter 1: Networks & Degrees

In the past three decades considerable efforts have been devoted to understanding the rich structure and functions of complex networks, be they technologically engineered, found in nature or generated through social interactions. A popular research direction has been the design of "good" random graph models in the sense of exhibiting key properties found in observed networks – Historically attention was given to the simplest of network properties, namely the degrees of nodes and their distributions.

The discussion invariably starts with the work of Erdős and Rényi [17]: With n nodes and link probability p, the (binomial) Erdős-Rényi graph  $\mathbb{G}(n;p)$  postulates that the  $\frac{n(n-1)}{2}$  potential undirected links between these n nodes are each created with probability p, independently of each other. The degree distribution of a node in Erdős-Rényi graphs is then announced to be "Poisson"-like, the justification going roughly as follows: (i) With  $D_{n,k}(p)$  denoting the degree random variable (rv) of node k in  $\mathbb{G}(n;p)$ , the rvs  $D_{n,1}(p), \ldots, D_{n,n}(p)$  are identically distributed, each distributed according to a binomial rv Bin(n-1;p); (ii) If the link probability scales with n as  $p_n \sim \frac{\lambda}{n}$  for some  $\lambda > 0$ , then Poisson Convergence ensures the distributional

convergence

$$D_{n,1}(p_n) \Longrightarrow_n D \tag{1.1}$$

where the rv D is a Poisson rv with parameter  $\lambda$ .

A rich asymptotic theory has been developed for Erdős-Rényi graphs in the many node regime; see the monographs [7, 13, 27]. However, as more networks have come under investigation, in many cases the data suggest that the degree distribution is not Poisson but displays instead a *power-law* behavior in the following sense: If the network comprises a large number n of nodes and  $N_n(d)$  is the number of nodes with degree d, then the data reveals a behavior of the form

$$\frac{N_n(d)}{n} \simeq C d^{-\alpha} \tag{1.2}$$

for some  $\alpha$  in the range [2, 3] and C > 0 – Such statements are usually left somewhat vague as the range of d is never carefully specified in relation to n; networks where (1.2) was observed are often called *scale-free* networks. On the basis of this observation, it was concluded that Erdős-Rényi graphs cannot model scale-free networks, and new random graph models were sought. The Barabási-Albert model came to prominence as the first random graph model to formally "explain" via the mechanism of preferential attachment the possibility of power law degree distributions in large networks [2].

The statement (1.2) concerns a degree distribution which is computed *network-wide*, whereas the convergence (1.1) addresses the behavior of the (generic) degree of a *single* node, its distribution being identical across nodes. A natural question is then whether this discrepancy can be resolved, at least asymptotically, in large

networks and if so, under what conditions.

Our first contribution lies in exploring this issue in some detail. First we introduce a general framework to investigate this discrepancy in Chapter 2 through a sequence of random graphs { $\mathbb{G}_n$ , n = 1, 2, ...} with increasingly large and unbounded vertex sets. The discussion is carried out under a set of three assumptions, namely

- (i) Weak Homogeneity: For each n = 1, 2, ..., the degree rvs in  $\mathbb{G}_n$  are pairwise equidistributed across nodes – Let  $D_n$  denote the generic degree rv in  $\mathbb{G}_n$ ;
- (ii) Existence of an asymptotic (nodal) degree distribution: There exists an  $\mathbb{N}$ -valued rv D such that

$$D_n \Longrightarrow_n D.$$
 (1.3)

Let  $\boldsymbol{p} = (p(d), d = 0, 1, ...)$  denote the pmf of D; and

 (iii) Asymptotic uncorrelatedness: The degree rvs display a weak form of asymptotic "pairwise independence."

Under the aforementioned assumptions, we show the following result: If  $(p_n(d), d = 0, 1, ...)$  is the empirical degree distribution in  $\mathbb{G}_n$  (with  $p_n(d)$  denoting the fraction of nodes in  $\mathbb{G}_n$  with degree d), then

$$p_n(d) \xrightarrow{P} _n p(d), \quad d = 0, 1, \dots$$
 (1.4)

where the pmf  $\boldsymbol{p} = (p(d), d = 0, 1, ...)$  on  $\mathbb{N}$  is as in (ii) above. Essentially this result gives us a set of necessary conditions for the (empirical) degree distribution to converge in the usual sense (1.4). As we discuss the underlying assumptions in Chapter 2, we see that Erdős-Rényi graphs (under the scaling yielding (1.1)) are readily subsumed in this framework, as are several well-known homogeneous networks of interest in applications. This resolves the discrepancy mentioned earlier in that the appropriate version of (1.4) does hold for both Erdős-Rényi graphs and the Barabási-Albert model.

Next we turn our attention to the belief, mostly unsubstantiated, that in homogeneous graphs the convergence (1.3) of the generic degree distribution might *automatically* imply the convergence (1.4) of the empirical degree distribution. In Chapter 2 we show in homogeneous graphs that weak asymptotic uncorrelatedness (assumption (iii)) is necessary to ensure convergence of the empirical degree distribution in the usual sense even when the (nodal) degree distribution converges according to (1.3). This brings us to the question – Are there counterexamples in the class of homogeneous random graphs of any significant interest for which convergence (1.3) of the generic degree distribution take place while the usual convergence of the empirical degree distribution (1.4) does not hold? A counterexample described in Chapter 3 is found in the class of random threshold graphs.

We motivate the counterexample by going back to the Barabási-Albert network model. After the Barabási-Albert network model was proposed, researchers began to wonder whether there were mechanisms other than preferential attachment which could also lead to power law degree distribution. Caldarelli et al. [9] proposed a homogeneous random graph model called the random threshold model based on the "good-get-richer" mechanism. They argued that in many scenarios, the degree information about every node might not be readily available (as assumed in the Barabási-Albert network model). Instead, two nodes might form a connection if they are *mutually beneficial* to each other depending on their intrinsic properties (maybe friendship, interaction strength, attractiveness, etc.). The intrisic property of a node is modeled as a *fitness value* drawn from a fitness distribution – A connection is said to form between any two nodes if the sum of their individual fitnesses exceeds a certain threshold. With an exponential fitness distribution, when the threshold is scaled appropriately, they argued that the empirical degree distribution is power-law in the limit of large graph size.

For the setting just described, although (1.3) is known to take place [22] with the power tail behavior

$$p(d) \sim d^{-2} \quad (d \to \infty),$$

we show that (1.4) fails to hold [Proposition 3.3]. One implication of this finding is that random threshold graphs cannot be used as an alternative scale-free model to the Barabási-Albert model (see below) as claimed by the authors [9,37]. Indeed, only the convergence (1.4) has meaning in the preferential attachment model while (1.3) is meaningless, with the situation being reversed for random threshold graphs. In other words, the two models cannot be compared in terms of their degree distributions! This also highlights the fact that *even* in homogeneous graphs, there are noteworthy situations when the network-wide degree distribution and the nodal degree distribution may capture vastly different information. We take the discussion one step further by showing in Chapter 4 that the empirical degree distribution actually converges in a weaker sense. Echoing the modeling concern in the Barabási-Albert network model posed by Caldarelli et al. [9], Ghadge et al. [23] argued that often degree information might not be readily available, and that the inherent quality of each node captured through a fitness variable (drawn from a fitness distribution in an i.i.d. fashion) should instead be the primary driver of network growth. Through simulations, the authors were able to show a wide range of achievable degree distributions including power-law for fitness lognormally distributed. In Chapter 5, we study this model, often called the *fitness model*, in great detail. We investigate convergence of the empirical degree distribution in the following expected sense

$$\mathbb{E}\left[\frac{N_n(d)}{n}\right] \longrightarrow_n p(d), \ d = 0, 1, \dots$$
(1.5)

where the pmf (p(d), d = 0, 1, ...) on N is the empirical degree distribution in the limit of large graph size. The convergence (1.5) is a weaker form of convergence than (1.4). However, if the convergence (1.4) were to hold, then (1.5) must also hold with the same limiting pmf. Our results indicate that if the fitness distribution is bounded, the tail of the (limiting) empirical degree distribution shows *geometric* decay, i.e., roughly speaking

$$p(d+1) \simeq C e^{-\beta d}, \ d = 0, 1, \dots$$
 (1.6)

for some  $\beta > 0$  and C > 0. Thus, if the fitness distribution is bounded, we cannot have a power-law behavior. Conversely, for fitness distribution with infinite support, we prove that the tail of the (limiting) empirical degree distribution cannot have a geometric decay. We consider two special cases – (i) When fitness is pareto distributed we show that the asymptotic empirical degree distribution  $\{p(d), d = 0, 1, ...\}$  is indeed power-law with a parameter depending on the fitness distribution. By appropriately choosing the fitness distribution, any power-law degree distribution of parameter greater than two is shown to be achievable. (ii) On the other hand, with exponentially distributed fitness we show that a power-law behavior does not emerge, implying that unbounded fitness distribution would not necessarily lead to a power-law behavior. The fitness model is therefore richer than the Barabási-Albert model in the sense that it is capable of producing a variety of tail behavior along with power-law distributions of different parameters.

This portion of the thesis is organized as follows: In Chapter 2 we introduce a general framework for studying the degree distributions of networks modeled as random graphs. We give necessary and sufficient conditions for homogeneous graphs under which the empirical degree distribution will converge and be identical to the asymptotic nodal degree distribution. We also consider well known examples of random graph models that satisfy these conditions. In Chapter 3 we investigate a specific counterexample in the class of random threshold graphs, where the convergence (1.4) of the empirical degree distribution does not hold. In Chapter 4 we study the degree distribution of random threshold graphs in further detail and show that the empirical degree distribution actually converges in a weaker sense. Finally in Chapter 5 we investigate the degree distribution of the fitness model and study its tail behavior.

#### Chapter 2: Degree Distribution of Networks

In this chapter we first introduce a general framework for studying various degree distributions of networks modeled as random graphs, namely the degree distribution of a particular node and the (empirical) degree distribution of the graph defined as the fraction of nodes with given degree. We obtain necessary and sufficient conditions in the large graph regime under which the network-wide degree distribution exists and is identical to the nodal degree distribution. In the later part of this chapter, we consider well known examples of random graph models that satisfy these conditions.

#### 2.1 A general framework

We are given a sequence of random graphs { $\mathbb{G}_n$ , n = 2, 3, ...} defined on the probability triple  $(\Omega, \mathcal{F}, \mathbb{P})$  with the following structure: Fix n = 2, 3, ... With  $V_n$ a finite and non-empty set, the random graph  $\mathbb{G}_n$  is an ordered pair  $(V_n, \mathbb{E}_n)$  defined on the set of nodes  $V_n$  with random edge set  $\mathbb{E}_n \subseteq V_n \times V_n$ . The edge set  $\mathbb{E}_n$  is equivalently determined by a set of {0,1}-valued edge variables { $\chi_n(k, \ell), k, \ell \in V_n$ } – Thus,  $\chi_n(k, \ell) = 1$  (resp.  $\chi_n(k, \ell) = 0$ ) if there is an edge (resp. no edge) from node k to node  $\ell$ , so that

$$\mathbb{E}_n = \{ (k, \ell) \in V_n \times V_n : \chi_n(k, \ell) = 1 \}.$$

There is no loss in generality in taking  $V_n = \{1, ..., k_n\}$  for some positive integer  $k_n$ . In most cases of interest  $V_n = \{1, ..., n\}$  so that  $k_n = n$ .

We assume  $\mathbb{G}_n$  to be an *undirected* graph with *no self-loops*. This amounts to

$$\chi_n(k,k) = 0$$
 and  $\chi_n(k,\ell) = \chi_n(\ell,k), \quad k,\ell \in V_n$ 

Under these conditions the edge set  $\mathbb{E}_n$  is a symmetric subset of  $V_n \times V_n$  because there is an edge from node k to node  $\ell$  (i.e.,  $\chi_n(k, \ell) = 1$ ) if and only if there is an edge from node  $\ell$  to node k (i.e.,  $\chi_n(\ell, k) = 1$ ).

For each k in  $V_n$ , the degree of node k in  $\mathbb{G}_n$  is the rv  $D_{n,k}$  given by

$$D_{n,k} = \sum_{\ell \in V_n} \chi_n(k,\ell).$$
(2.1)

For each  $d = 0, 1, \ldots$ , the rv  $N_n(d)$  defined by

$$N_n(d) = \sum_{k \in V_n} \mathbf{1} \left[ D_{n,k} = d \right]$$

counts the number of nodes in  $V_n$  which have degree d in  $\mathbb{G}_n$ . The fraction of nodes in  $V_n$  with degree d in  $\mathbb{G}_n$  is then given by

$$\frac{N_n(d)}{|V_n|} = \frac{1}{|V_n|} \sum_{k \in V_n} \mathbf{1} \left[ D_{n,k} = d \right].$$

This defines the pmf-valued rv

$$\left(\frac{N_n(d)}{|V_n|}, \ d = 0, 1, \ldots\right)$$

which takes its values in the space of pmfs on  $\mathbb{N}$  with support contained in  $V_n \cup \{0\}$ .

As we focus on limiting results for n large, we assume the sets  $\{V_n, n = 1, 2, ...\}$ to grow unboundedly large with n, namely  $\lim_{n\to\infty} |V_n| = \infty$ . Concretely, we assume the sequence  $n \to k_n$  to be monotone increasing with  $\lim_{n\to\infty} k_n = \infty$ .

#### 2.2 The main result

With  $\lim_{n\to\infty} |V_n| = \infty$ , we seek sufficient conditions to ensure the convergence

$$\frac{N_n(d)}{|V_n|} \xrightarrow{P} {}_n p(d), \quad d = 0, 1, \dots$$
(2.2)

for some pmf  $\boldsymbol{p} = (p(d), d = 0, 1, ...)$  on N. A set of assumptions to that effect is presented next.

Assumption 2.1. (Weak Homogeneity) For each n = 2, 3, ..., the degree rvs in  $\mathbb{G}_n$  are pairwise equidistributed in the sense that

Note that (2.3) necessarily implies that the degree rvs in  $\mathbb{G}_n$  are equidistributed with

$$D_{n,k} =_{st} D_{n,1}, \quad k \in V_n.$$

Assumption 2.2. (Existence of an asymptotic (nodal) degree distribution) Under Assumption 2.1, there exists an  $\mathbb{N}$ -valued rv D such that

$$D_{n,1} \Longrightarrow_n D.$$
 (2.5)

Let  $\mathbf{p} = (p(d), d = 0, 1, ...)$  denote the pmf of the limiting rv D.

Assumption 2.2 can be rephrased as

$$\lim_{n \to \infty} \mathbb{P}[D_{n,1} = d] = p(d), \quad d = 0, 1, \dots$$
(2.6)

Assumption 2.3. (Asymptotic uncorrelatedness) Under Assumption 2.1, for each d = 0, 1, ..., the rvs  $\mathbf{1} [D_{n,1} = d]$  and  $\mathbf{1} [D_{n,2} = d]$  are asymptotically uncorrelated in the sense that

$$\lim_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1} = d \right], \mathbf{1} \left[ D_{n,2} = d \right] \right] = 0.$$
 (2.7)

Assumption 2.3 amounts to the convergence statement

$$\lim_{n \to \infty} \left( \mathbb{P} \left[ D_{n,1} = d, D_{n,2} = d \right] - \mathbb{P} \left[ D_{n,1} = d \right] \mathbb{P} \left[ D_{n,2} = d \right] \right) = 0$$
(2.8)

for each d = 0, 1, ... As will become apparent shortly, Assumption 2.2 and Assumption 2.3 will always be used in combination with Assumption 2.1. The main result of this chapter can now be given.

Proposition 2.1. Under Assumptions 2.1-2.3, we have

$$\frac{N_n(d)}{|V_n|} \xrightarrow{P} {}_n p(d), \quad d = 0, 1, \dots$$
(2.9)

where the pmf  $\mathbf{p} = (p(d), d = 0, 1, ...)$  is postulated in Assumption 2.2.

The proof of Proposition 2.1 mimics the classical proof of the Weak Law of Large Numbers, and is provided in Section 2.8. A careful inspection of the arguments given there shows that the following partial converse also holds; see Section 2.8 for details. **Proposition 2.2.** Assume Assumptions 2.1-2.2 to hold. If for some d = 0, 1, ..., we have

$$\frac{N_n(d)}{|V_n|} \xrightarrow{P} {}_n L(d) \tag{2.10}$$

for some constant L(d) in  $\mathbb{R}$ , then we necessarily have L(d) = p(d) where the pmf  $\mathbf{p} = (p(d), d = 0, 1, ...)$  is the one postulated in Assumption 2.2, and the limit

$$C(d) \equiv \lim_{n \to \infty} \operatorname{Cov}[\mathbf{1}[D_{n,1} = d], \mathbf{1}[D_{n,2} = d]]$$
(2.11)

must exist with C(d) = 0.

This converse has the following consequence to be used later: Under Assumptions 2.1-2.2, whenever we have

$$\lim_{n \to \infty} \operatorname{Cov}[\mathbf{1}[D_{n,1} = d], \mathbf{1}[D_{n,2} = d]] > 0, \qquad (2.12)$$

then the conclusion (2.9) cannot hold.

Finally, under Assumption 2.1, the convergence

$$\frac{N_n(d)}{|V_n|} \xrightarrow{P} _n L(d)$$

for some d = 0, 1, ... with some constant L(d) in  $\mathbb{R}$  necessarily implies

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1} = d\right] = L(d)$$

by bounded convergence. This shows the necessity of Assumption 2.2 for (2.9) to hold.

#### 2.3 Concerning Assumption 2.3

Assumption 2.3 is implied by the following assumption which is easier to check in some cases; see Section 2.5 for some examples in a commonly occurring setting. Assumption 2.4. (Pairwise asymptotic independence) Under Assumptions 2.1 and 2.2, the degree rvs  $D_{n,1}$  and  $D_{n,2}$  are asymptotically independent in the sense that

$$(D_{n,1}, D_{n,2}) \Longrightarrow_n (D_1, D_2) \tag{2.13}$$

where  $D_1$  and  $D_2$  are independent  $\mathbb{N}$ -valued rvs, each distributed according to the pmf  $\boldsymbol{p}$  postulated in Assumption 2.2.

Assumption 2.4 can be rephrased as

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1} = d, D_{n,2} = d'\right] = p(d)p(d'), \quad d, d' = 0, 1, \dots$$
(2.14)

Assumption 2.3 does not require the joint convergence (2.13) to hold. However, if (2.13) is known to hold (with *no further* characterization of the limit), then under Assumption 2.2 it is easy to check that (2.8) is equivalent to the independence of the *binary* rvs  $\mathbf{1}$  [ $D_1 = d$ ] and  $\mathbf{1}$  [ $D_2 = d$ ] for each d = 0, 1, ...: Indeed, the existence of the limit (2.13) implies

$$\lim_{n \to \infty} \mathbb{P} \left[ D_{n,1} = d, D_{n,2} = d \right] = \mathbb{P} \left[ D_1 = d, D_2 = d \right]$$

and

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,j} = d\right] = \mathbb{P}\left[D_j = d\right], \quad j = 1, 2.$$

The condition (2.8) is now equivalent to

$$\mathbb{P}[D_1 = d, D_2 = d] = p(d)p(d) = \mathbb{P}[D_1 = d]\mathbb{P}[D_2 = d], \qquad (2.15)$$

and states the independence of the *binary* rvs  $\mathbf{1} [D_1 = d]$  and  $\mathbf{1} [D_2 = d]$ . It should be pointed out that the lack of independence of the rvs  $D_1$  and  $D_2$  does not preclude the possibility that the rvs  $\mathbf{1} [D_1 = d]$  and  $\mathbf{1} [D_2 = d]$  are independent – Indeed it is possible for (2.15) to hold even for all d = 0, 1, ... without the rvs  $D_1$  and  $D_2$  being independent.

#### 2.4 Erdős-Rényi Graphs

We first consider the popular Erdős-Rényi random graph model, and examine whether Assumptions 2.1-2.3 are satisfied.

Consider an undirected graph of n nodes with link probability p – The Erdős-Rényi graph  $\mathbb{G}(n;p)$  on the vertex set  $V_n := \{1, 2, ..., n\}$  postulates that the  $\frac{n(n-1)}{2}$ potential undirected links between these n nodes are each created with probability p, independently of each other.

The links can be modelled by a collection of i.i.d. Bernoulli rvs  $\{B_{ij}(p), i, j = 1, 2, ..., n\}$  with parameter p; with the restriction that  $B_{ij}(p) = B_{ji}(p)$  for distinct i, j = 1, 2, ..., n and  $B_{ii}(p) = 0$  for i = 1, 2, ..., n. For i = 1, 2, ..., n and  $0 , let <math>D_{n,i}(p)$  denote the degree of node i in  $\mathbb{G}(n; p)$ . Clearly, we have

$$D_{n,i}(p) = \sum_{j=1, j \neq i}^{n} B_{ij}(p)$$

and Assumption 2.1 is satisfied given the homogeneity of the model. It is well known that when p is scaled according to  $p^* : \mathbb{N}_0 \to [0, 1]$  given by

$$p_n^* \sim \frac{\lambda}{n},\tag{2.16}$$

the degree rvs converge to the Poisson pmf  $p_{\lambda} = (p_{\lambda}(d), d = 0, 1, ...)$  with parameter  $\lambda$  given by

$$p_{\lambda}(d) = \frac{\lambda^d}{d!} e^{-\lambda}, \ d = 0, 1, \dots$$
 (2.17)

Hence Assumption 2.2 is also satisfied with  $p = p_{\lambda}$ . It is also relatively straightforward to show that the degree rvs become pairwise asymptotically independent. For fixed  $n = 1, 2, ..., \ell, k = 0, 1, ..., n - 1$  and 0 , we have

$$\mathbb{P}\left[D_{n,1}(p) = k, D_{n,2}(p) = \ell\right] \\
= \mathbb{P}\left[\sum_{j=2}^{n} B_{1j}(p) = k, \sum_{j=1, j \neq 2}^{n} B_{2j}(p) = \ell\right] \\
= \mathbb{P}\left[\sum_{j=3}^{n} B_{1j}(p) = k, \sum_{j=3}^{n} B_{2j}(p) = \ell, B_{12}(p) = 0\right] \\
+ \mathbb{P}\left[\sum_{j=3}^{n} B_{1j}(p) = k - 1, \sum_{j=3}^{n} B_{2j}(p) = \ell - 1, B_{12}(p) = 1\right].$$
(2.18)

For the second term in (2.18), we have the bound

$$\mathbb{P}\left[\sum_{j=3}^{n} B_{1j}(p) = k - 1, \sum_{j=3}^{n} B_{2j}(p) = \ell - 1, B_{12}(p) = 1\right] \leq \mathbb{P}\left[B_{12}(p) = 1\right].$$

However under the scaling (2.16), we have

$$\lim_{n \to \infty} \mathbb{P} \left[ B_{12}(p_n^*) = 1 \right] = 0, \tag{2.19}$$

which leads to

$$\lim_{n \to \infty} \mathbb{P}\left[\sum_{j=3}^{n} B_{1j}(p_n^*) = k - 1, \sum_{j=3}^{n} B_{2j}(p_n^*) = \ell - 1, B_{12}(p_n^*) = 1\right] = 0.$$
(2.20)

For each  $n = 2, 3, \ldots$  and 0 , set

$$D_{n,i}^*(p) = \sum_{j=3}^n B_{ij}(p), \ i = 1, 2.$$

For the first term in (2.18), using the fact that the Bernoulli rvs are assumed

to be i.i.d., we obtain

$$\mathbb{P}\left[\sum_{j=3}^{n} B_{1j}(p) = k, \sum_{j=3}^{n} B_{2j}(p) = \ell, B_{12}(p) = 0\right]$$
  
=  $\mathbb{P}\left[\sum_{j=3}^{n} B_{1j}(p) = k\right] \mathbb{P}\left[\sum_{j=3}^{n} B_{2j}(p) = \ell\right] \mathbb{P}\left[B_{12}(p) = 0\right]$   
=  $\mathbb{P}\left[D_{n,1}^{*}(p) = k\right] \mathbb{P}\left[D_{n,2}^{*}(p) = \ell\right] \mathbb{P}\left[B_{12}(p) = 0\right].$  (2.21)

The following reduction step will simplify calculations.

**Lemma 2.3.** Fix  $n = 2, 3, \ldots$  and  $0 \le p \le 1$ . For each  $d = 0, 1, \ldots, n-1$ , we have

$$|\mathbb{P}[D_{n,i}(p) = d] - \mathbb{P}[D_{n,i}^*(p) = d]| \le 2\mathbb{P}[B_{12}(p) = 1], \ i = 1, 2.$$
(2.22)

**Proof.** Fix  $n = 2, 3, \ldots$  and  $0 \leq p \leq 1$ . For i = 1, 2, we observe that

$$\mathbb{P}\left[D_{n,i}(p) = 0\right] = \mathbb{P}\left[D_{n,i}^{*}(p) + B_{12}(p) = 0\right]$$
$$= \mathbb{P}\left[D_{n,i}^{*}(p) = 0, B_{12}(p) = 0\right]$$
$$= \mathbb{P}\left[D_{n,i}^{*}(p) = 0\right] - \mathbb{P}\left[D_{n,i}^{*}(p) = 0, B_{12}(p) = 1\right]$$
(2.23)

and the bound (2.22) follows for d = 0.

For each  $d = 1, 2, \ldots, n - 1$ , and i = 1, 2, we observe that

$$\mathbb{P}\left[D_{n,i}(p) = d\right]$$
  
=  $\mathbb{P}\left[D_{n,i}^{*}(p) + B_{12}(p) = d\right]$   
=  $\mathbb{P}\left[D_{n,i}^{*}(p) = d, B_{12}(p) = 0\right] + \mathbb{P}\left[D_{n,i}^{*}(p) = d - 1, B_{12}(p) = 1\right]$   
=  $\mathbb{P}\left[D_{n,i}^{*}(p) = d\right] - \mathbb{P}\left[D_{n,i}^{*}(p) = d, B_{12}(p) = 1\right] + \mathbb{P}\left[D_{n,i}^{*}(p) = d - 1, B_{12}(p) = 1\right]$ 

and the bound (2.22) follows.

Under the scaling (2.16), using Lemma 2.3 and (2.19), we conclude that

$$D_{n,j}^*(p_n^*) \Longrightarrow_n \operatorname{Poi}(\lambda), \ j = 1, 2.$$

In other words

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,j}^{*}(p_{n}^{*}) = d\right] = \frac{\lambda^{d}}{d!}e^{-\lambda}, \qquad d = 0, 1, \dots$$

$$j = 1, 2.$$
(2.24)

Returning to (2.21), we conclude from (2.24) (along with (2.19)) that

$$\lim_{n \to \infty} \mathbb{P}\left[\sum_{j=3}^{n} B_{1j}(p_n^*) = k, \sum_{j=3}^{n} B_{2j}(p_n^*) = \ell, B_{12}(p_n^*) = 0\right]$$
$$= \lim_{n \to \infty} \mathbb{P}\left[D_{n,1}^*(p) = k\right] \mathbb{P}\left[D_{n,2}^*(p) = \ell\right] \mathbb{P}\left[B_{12}(p) = 0\right]$$
$$= p_{\lambda}(k)p_{\lambda}(\ell)$$
(2.25)

where  $\boldsymbol{p}_{\lambda} = (p_{\lambda}(d), \ d = 0, 1, ...)$  is the Poisson pmf with parameter  $\lambda$  as defined in (2.17). Putting together (2.20) and (2.25) in (2.18), we obtain the desired result

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(p_n^*) = k, D_{n,2}(p_n^*) = \ell\right] = p_{\lambda}(k)p_{\lambda}(\ell).$$

This shows that the stronger Assumption 2.4 holds. Thus, we have a setting where the Assumptions described in Section 2.2 hold and the empirical degree distribution converges as announced in Proposition 2.1.

#### 2.5 The generic setting

In many situations of interest the sequence of random graphs { $\mathbb{G}_n$ ,  $n = 1, 2, \ldots$ } arises in the following natural manner: Given is an underlying parametric family of random graphs, say

$$\{\mathbb{G}(n;\alpha), \ n=2,3,\ldots\}, \quad \alpha \in A \subseteq \mathbb{R}^p \tag{2.26}$$

where A is some parameter set and p is the dimension of the parameter space. With  $\alpha$  in A, for each  $n = 2, 3, \ldots$ , the random graph  $\mathbb{G}(n; \alpha)$  is a random graph on  $V_n$  whose statistics depend on the parameter  $\alpha$ .

For  $\alpha$  in A, we define a set of  $\{0, 1\}$ -valued edge variables  $\{\chi_n(k, \ell; \alpha), k, \ell \in V_n\}$ corresponding to the graph  $\mathbb{G}(n; \alpha)$ . Thus,  $\chi_n(k, \ell; \alpha) = 1$  (resp.  $\chi_n(k, \ell; \alpha) = 0$ ) if there is an edge (resp. no edge) between node k and node  $\ell$ . Since the graph  $\mathbb{G}(n; \alpha)$  is an undirected graph with no self-loops, we must have

$$\chi_n(k,k;\alpha) = 0$$
 and  $\chi_n(k,\ell;\alpha) = \chi_n(\ell,k;\alpha), \qquad k,\ell \in V_n$ 

For each k in  $V_n$ , let  $D_{n,k}(\alpha)$  denote the degree of node k in  $\mathbb{G}(n; \alpha)$ . For each  $d = 0, 1, \ldots$ , the rv  $N_n(d; \alpha)$  defined by

$$N_n(d;\alpha) = \sum_{k=1}^n \mathbf{1} \left[ D_{n,k}(\alpha) = d \right]$$

counts the number of nodes in  $\{1, \ldots, n\}$  which have degree d in  $\mathbb{G}(n; \alpha)$ . The fraction of nodes in  $\{1, \ldots, n\}$  with degree d in  $\mathbb{G}(n; \alpha)$  is then given by

$$p_n(d;\alpha) = \frac{N_n(d;\alpha)}{n}$$

The next assumption imposes a probabilistic structure on the edge rvs.

**Property 2.1.** For all  $\alpha$  in A and  $n = 2, 3, ..., the set of rvs <math>\begin{cases} k \neq \ell \\ \chi_n(k, \ell; \alpha), \\ k, \ell \in V_n \end{cases}$  constitutes an exchangeable family.

Property 2.1 implies that the rvs  $\{D_{n,k}(\alpha), k \in V_n\}$  also constitute an exchangeable family. Therefore in  $\mathbb{G}(n; \alpha)$  there is no ambiguity as to what is the (nodal) degree distribution because *all* nodes have the same degree distribution, namely that of the rv  $D_{n,1}(\alpha)$ . Further, Property 2.1 guarantees that for each  $\alpha$  in A, and all  $k, \ell$  in  $V_n$  with  $k \neq \ell$ , it holds that  $(D_{n,k}(\alpha), D_{n,\ell}(\alpha)) =_{st} (D_{n,1}(\alpha), D_{n,2}(\alpha))$ .

We construct the collection  $\{\mathbb{G}_n, n = 2, 3, ...\}$  by setting

$$\mathbb{G}_n = \mathbb{G}(n; \alpha_n^*), \quad n = 2, 3, \dots$$
(2.27)

where the scaling  $\alpha^* : \mathbb{N}_0 \to A$  is the (usually unique) scaling which ensures the convergence

$$D_{n,1}(\alpha_n^*) \Longrightarrow_n D \tag{2.28}$$

for some non-degenerate  $\mathbb{N}$ -valued rv D. This scaling is often the critical scaling associated with the emergence of a maximal component. Under these circumstances, Assumptions 2.1 and 2.2 are automatically satisfied, and only Assumption 2.3 needs to be verified.

The example of Erdős-Rényi graphs was given earlier. In the following sections we consider additional examples of random graph models routinely discussed in the literature. The setting outlined above is used as it applies to these examples : With  $\lambda > 0$ ,

- 1. Random key graphs  $\mathbb{K}(n; K, P)$   $(K < P \text{ in } \mathbb{N}_0)$  with scalings  $K^*, P^* : \mathbb{N}_0 \to \mathbb{N}_0$ given by  $\frac{(K_n^*)^2}{P_n^*} \sim \frac{\lambda}{n}$  [38]; and
- 2. Geometric random graphs  $\mathbb{G}(n; \alpha)$  on a unit square  $(\alpha > 0)$  with scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  given by  $\pi(\alpha_n^*)^2 \sim \frac{\lambda}{n}$  [34].

In each case, it is a simple matter to check that D in (2.28) is a Poisson rv with parameter  $\lambda$ . We show that the stronger Assumption 2.4 actually holds in both cases, as was the case for Erdős-Rényi graphs.

We continue the discussion by imposing additional structure on the generic setting in order to find conditions under which the stronger Assumption 2.4 might hold. Fix integers  $k, \ell = 0, 1, ...$  and  $\alpha$  in A. For each n = 2, 3, ..., such that  $\max(k, \ell) + 2 \leq n$ , we have the decomposition

$$\mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell\right] = \mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 0\right] + \mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 1\right].$$
(2.29)

The second term in (2.29) satisfies the following bound

$$\mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 1\right] \leq \mathbb{P}\left[\chi_n(1,2;\alpha) = 1\right].$$
(2.30)

Together (2.29) and (2.30) suggest the following assumption, namely

**Property 2.2.** Under the scaling  $\alpha^* : \mathbb{N}_0 \to A$  satisfying (2.9), it holds that

$$\lim_{n \to \infty} \mathbb{P}\left[\chi_n(1, 2; \alpha_n^*) = 1\right] = 0.$$

If the above-mentioned property is satisfied, then the second term in (2.29) can be disregarded in the limit of large n under the appropriate scaling. Let  $\mathcal{N}_{n,i}(\alpha)$ denote the neighbor set of node i in  $V_n$  in the graph  $\mathbb{G}(n; \alpha)$ , i.e.,

$$\mathcal{N}_{n,i}(\alpha) = \{ j \in V_n : \chi_n(i,j;\alpha) = 1 \}, \ i \in V_n.$$

We can further decompose the first term in (2.29) as

$$\mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 0\right]$$
$$= \mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 0, \mathcal{N}_{n,1}(\alpha) \cap \mathcal{N}_{n,2}(\alpha) = \varnothing\right]$$
$$+ \mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 0, \mathcal{N}_{n,1}(\alpha) \cap \mathcal{N}_{n,2}(\alpha) \neq \varnothing\right]. \quad (2.31)$$

Under Property 2.1, we bound the second term in (2.31) as

$$\mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 0, \mathcal{N}_{n,1}(\alpha) \cap \mathcal{N}_{n,2}(\alpha) \neq \emptyset\right]$$

$$\leq \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) \cap \mathcal{N}_{n,2}(\alpha) \neq \emptyset\right]$$

$$= \mathbb{P}\left[\chi_n(1,j;\alpha) = 1, \chi_n(2,j;\alpha) = 1 \text{ for some } j = 3, \dots, n\right]$$

$$\leq \sum_{j=3}^n \mathbb{P}\left[\chi_n(1,j;\alpha) = 1, \chi_n(2,j;\alpha) = 1\right]$$

$$= (n-2)\mathbb{P}\left[\chi_n(1,3;\alpha) = 1, \chi_n(2,3;\alpha) = 1\right]$$
(2.32)

with the help of a simple union bound. This leads naturally to the next assumption, namely

**Property 2.3.** Under the scaling  $\alpha^* : \mathbb{N}_0 \to A$  satisfying (2.9), it holds that

$$\lim_{n \to \infty} n \mathbb{P} \left[ \chi_n(1,3;\alpha_n^*) = 1, \chi_n(2,3;\alpha_n^*) = 1 \right] = 0.$$

Therefore, if Properties 2.1-2.3 are satisfied, then for each  $k, \ell = 0, 1, \ldots$ , we have

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^*) = k, D_{n,2}(\alpha_n^*) = \ell\right]$$

$$= \lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^*) = k, D_{n,2}(\alpha_n^*) = \ell, \chi_n(1,2;\alpha_n^*) = 0, \mathcal{N}_{n,1}(\alpha_n^*) \cap \mathcal{N}_{n,2}(\alpha_n^*) \neq \varnothing\right]$$
(2.33)

with the understanding that if one of the limits exists, so does the other, and the limiting values coincide.

As we will see, the presence of the events  $[\chi_n(1,2;\alpha) = 0]$  and  $[\mathcal{N}_{n,1}(\alpha) \cap \mathcal{N}_{n,2}(\alpha) \neq \emptyset]$  simplifies the calculations that follow. With integers  $k, \ell = 0, 1, ...$  pick n = 2, 3, ... such that  $k + \ell + 2 \leq n$ , and define the set

$$\Upsilon_n(\ell,k) = \left\{ (S,T) \mid S, T \subseteq V_n \setminus \{1,2\}, S \cap T = \emptyset, |S| = k, |T| = \ell \right\}.$$

Under Property 2.1, for each  $\alpha$  in A, we have

$$\mathbb{P}\left[D_{n,1}(\alpha) = k, D_{n,2}(\alpha) = \ell, \chi_n(1,2;\alpha) = 0, \mathcal{N}_{n,1}(\alpha) \cap \mathcal{N}_{n,2}(\alpha) \neq \emptyset\right]$$
$$= \sum_{(S,T)\in\Upsilon_n(\ell,k)} \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S, \mathcal{N}_{n,2}(\alpha) = T\right]$$
$$= \binom{n-2}{\ell+k} \binom{\ell+k}{k} \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0\right]$$
(2.34)

where we have set  $S_0 = \{3, ..., k+2\}$  and  $T_0 = \{k+3, ..., k+\ell+2\}.$ 

Note that

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0\right]$$

$$= \mathbb{P}\left[\begin{array}{c} \chi_n(1,2;\alpha) = 0 \\ \chi_n(1,s;\alpha) = 1, \chi_n(2,s;\alpha) = 0, s \in S_0 \\ \chi_n(1,t;\alpha) = 0, \chi_n(2,t;\alpha) = 1, t \in T_0 \\ \chi_n(1,r;\alpha) = 0, \chi_n(2,r;\alpha) = 0, r \in (S_0 \cup T_0 \cup \{1,2\})^c \end{array}\right].$$
(2.35)

It is easy to see that we cannot proceed further without imposing additional structure on the underlying random graph model  $\mathbb{G}(n; \alpha)$ . Therefore, we shall instead consider specific examples of random graph models.

#### 2.6 Random Key Graphs

In this section we introduce a class of random graphs that belong to the class of random intersection graphs; it is naturally associated with the random key predistribution scheme of Eschenauer and Gligor [19] in the context of wireless sensor networks. These random graphs have also appeared recently in application areas like clustering analysis [24, 25] and collaborative filtering in recommender systems [29].

Consider *n* nodes and a pool of *P* keys. Each node is assigned a set of *K* distinct keys which are selected at random from the pool of *P* keys (with K < P). Two nodes form a connection if their key rings have at least one key in common. The resulting notion of adjacency defines the *random key graph*  $\mathbb{K}(n; \alpha)$  on the vertex set  $V_n := \{1, 2, \ldots, n\}$ , with the parameter  $\alpha \equiv (K, P)$ . Clearly the parameter space *A* is here given by

$$A \equiv \{ (K, P) : K, P \in \mathbb{N}_0, K < P \}.$$

Conditions on n, K and P have been sought under which the random key graph  $\mathbb{K}(n; \alpha)$  exhibits a non-trivial degree distribution. In [38], the authors show that if the parameters  $K^*, P^* : \mathbb{N}_0 \to \mathbb{N}_0$  are scaled such that

$$\frac{(K_n^*)^2}{P_n^*} \sim \frac{\lambda}{n},\tag{2.36}$$

the asymptotic nodal degree distribution is the Poisson pmf  $p_{\lambda} = \{p_{\lambda}(d), d = 0, 1, ...\}$  with parameter  $\lambda$ . Under this scaling we shall show that weak asymptotic independence in the sense of (2.8) holds for the random key graphs. In fact, we shall prove something stronger – the pairwise asymptotic independence of the degree rvs

in the usual sense.

Throughout it is convenient to assume that the keys are labeled  $1, \ldots, P$ . For each node  $i = 1, \ldots, n$ , let  $K_{n,i}(\alpha)$  denote the random set of K distinct keys assigned to node i. Let  $\mathcal{P}_K$  denote the collection of all subsets of  $\{1, \ldots, P\}$  which contain exactly K elements. The rvs  $K_{n,1}(\alpha), \ldots, K_{n,n}(\alpha)$  are assumed to be i.i.d. rvs, each of which is distributed uniformly over  $\mathcal{P}_K$  according to

$$\mathbb{P}[K_{n,i}(\alpha) = S] = \binom{P}{K}^{-1}, \quad S \in \mathcal{P}_K.$$

With this notation, distinct nodes i, j = 1, 2, ..., n are seen to be adjacent if,

$$K_{n,i}(\alpha) \cap K_{n,j}(\alpha) \neq \emptyset.$$

It is easy to check that

$$\mathbb{P}\left[K_{n,i}(\alpha) \cap K_{n,j}(\alpha) = \emptyset\right] = q(\alpha)$$

with

$$q(\alpha) = \begin{cases} 0 & \text{if } P < 2K \\ \frac{\binom{P-K}{K}}{\binom{P}{K}} & \text{if } 2K \leq P. \end{cases}$$
(2.37)

Expression (2.37) is a simple consequence of the fact that

$$\mathbb{P}\left[S \cap K_{n,i}(\alpha) = \emptyset\right] = \begin{cases} 0 & \text{if } |S| > P - K\\ \frac{\binom{P-|S|}{K}}{\binom{P}{K}} & \text{if } |S| \leq P - K \end{cases}$$

for any subset S of  $\{1, \ldots, P\}$ . Thus, with

$$\chi_n(i,j;\alpha) = \mathbf{1} \left[ K_{n,i}(\alpha) \cap K_{n,j}(\alpha) \neq \emptyset \right]$$

under the generic setting, the probability of edge occurrence in  $\mathbb{K}(n; \alpha)$  is given by

$$\mathbb{P}\left[\chi_n(i,j;\alpha) = 1\right] = 1 - q(\alpha).$$
(2.38)
For each i in  $V_n$ , it is plain that

$$D_{n,i}(\alpha) = \sum_{j \in V_n, j \neq i} \mathbf{1} \left[ K_{n,i}(\alpha) \cap K_{n,j}(\alpha) \neq \emptyset \right].$$

Now we consider a sequence of random key graphs { $\mathbb{K}(n; \alpha_n^*)$ , n = 2, 3, ...} with  $\alpha_n^* \equiv (K_n^*, P_n^*)$  such that (2.36) is satisfied.

**Theorem 2.4.** For the class of random key graphs, under the scaling  $\alpha^* : \mathbb{N}_0 \to A$  satisfying (2.36), we have

$$\lim_{n \to \infty} \mathbb{P} \left[ D_{n,1}(\alpha_n^*) = k, D_{n,2}(\alpha_n^*) = \ell \right] = p_{\lambda}(k) p_{\lambda}(\ell), \quad k, \ell = 0, 1, \dots$$
(2.39)

where the pmf  $\mathbf{p}_{\lambda} = (p_{\lambda}(d), d = 0, 1, ...)$  is the Poisson pmf on  $\mathbb{N}$  with parameter  $\lambda$ .

Theorem 2.4 is established in Section 2.9, and yields a stronger form of independence compared to what is required for Assumption 2.3. Assumptions 2.1-2.3 therefore hold for the random key graphs under the aforementioned scaling. This leads to the following corollary.

**Corollary 2.5.** For the class of random key graphs, under the scaling  $\alpha^* : \mathbb{N}_0 \to A$  satisfying (2.36), we have

$$\frac{N_n(d;\alpha_n^*)}{|V_n|} \xrightarrow{P} p_\lambda(d), \quad d = 0, 1, \dots$$
(2.40)

where the pmf  $\mathbf{p}_{\lambda} = (p_{\lambda}(d), d = 0, 1, ...)$  is the Poisson pmf on  $\mathbb{N}$  with parameter  $\lambda$ .

#### 2.7 Random Geometric Graphs

Next, we introduce a class of random graphs which is often considered to be a relevant model for (ad-hoc) wireless sensor networks [20, 35]: Consider n nodes which are assumed to be placed uniformly at random on the square  $[0, 1]^2$ . With set of nodes  $V_n = \{1, \ldots, n\}$ , let the position of node i in  $V_n$  be denoted as  $\mathbf{Z}_i = (X_i, Y_i)$ . Distinct nodes  $i, j = 1, 2, \ldots, n$  are said to be adjacent if the distance between them is less than some  $\alpha > 0$ . If  $d : \mathbb{R}^2 \times \mathbb{R}^2 \to \mathbb{R}$  denotes the (Euclidean) distance on  $\mathbb{R}^2$ , distinct nodes i and j are then adjacent if

$$d(\boldsymbol{Z}_i, \boldsymbol{Z}_j) \leqslant \alpha.$$

According to the notation developed in Section 2.5, with distinct nodes  $i, j = 1, \ldots, n$ , we have

$$\chi_n(i,j;\alpha) = \mathbf{1} \left[ d(\mathbf{Z}_i, \mathbf{Z}_j) \leqslant \alpha \right].$$

This defines the random geometric graph  $\mathbb{G}(n; \alpha)$  on the vertex set  $V_n$ . This time, for each *i* in  $V_n$ , we have

$$D_{n,i}(\alpha) = \sum_{j \in V_n, j \neq i} \mathbf{1} \left[ d(\mathbf{Z}_i, \mathbf{Z}_j) \leqslant \alpha \right].$$

Under the scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  such that

$$\pi(\alpha_n^*)^2 \sim \frac{\lambda}{n},\tag{2.41}$$

the asymptotic nodal degree distribution can be shown to be the Poisson pmf  $p_{\lambda} = \{p_{\lambda}(d), d = 0, 1, ...\}$  on  $\mathbb{N}$  with parameter  $\lambda$ . Under this scaling we will show that weak asymptotic independence holds in the sense of (2.8). As was done for random

key graphs, we prove the stronger pairwise asymptotic independence of the degree rvs in the usual sense.

**Theorem 2.6.** For the class of random geometric graphs, under the scaling  $\alpha^*$ :  $\mathbb{N}_0 \to \mathbb{R}_+$  satisfying (2.41), we have

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^*) = k, D_{n,2}(\alpha_n^*) = \ell\right] = p_\lambda(k)p_\lambda(\ell), \quad k, \ell \in \mathbb{N},$$
(2.42)

where the pmf  $\mathbf{p}_{\lambda} = (p_{\lambda}(d), d = 0, 1, ...)$  is the Poisson pmf on  $\mathbb{N}$  with parameter  $\lambda$ .

We prove Theorem 2.6 in Section 2.10. This result is analogous to Theorem 2.4 for random key graphs, and here as well, implies convergence of the empirical degree distribution.

**Corollary 2.7.** For the class of random geometric graphs, under the scaling  $\alpha^*$ :  $\mathbb{N}_0 \to \mathbb{R}_+$  satisfying (2.41), we have

$$\frac{N_n(d;\alpha_n^*)}{|V_n|} \xrightarrow{P} p_\lambda(d), \quad d = 0, 1, \dots$$
(2.43)

where the pmf  $\mathbf{p}_{\lambda} = (p_{\lambda}(d), d = 0, 1, ...)$  is the Poisson pmf on  $\mathbb{N}$  with parameter  $\lambda$ .

### 2.8 Proofs of Propositions 2.1 and 2.2

We begin with a preliminary technical lemma.

Lemma 2.8. If Assumptions 2.1-2.2 hold, then we have

$$\lim_{n \to \infty} \mathbb{E}\left[ \left| \frac{N_n(d)}{|V_n|} - p(d) \right|^2 \right] = \lim_{n \to \infty} \operatorname{Cov}[\mathbf{1}[D_{n,1} = d], \mathbf{1}[D_{n,2} = d]]$$
(2.44)

for each d = 0, 1, ..., with the understanding that if one of the limits exists, so does the other and the limiting values coincide.

**Proof.** Fix n = 2, 3, ... and d = 0, 1, ... By standard properties of the variance, we note that

$$\mathbb{E}\left[\left|\frac{N_n(d)}{|V_n|} - p(d)\right|^2\right] = \operatorname{Var}\left[\frac{N_n(d)}{|V_n|}\right] + \left(\mathbb{E}\left[\frac{N_n(d)}{|V_n|}\right] - p(d)\right)^2 \\
= \frac{\operatorname{Var}\left[N_n(d)\right]}{|V_n|^2} + \left(\mathbb{E}\left[\frac{N_n(d)}{|V_n|}\right] - p(d)\right)^2. \quad (2.45)$$

Proceeding in the usual manner, we use the definition of the rv  $N_n(d)$  to obtain the expressions

$$\mathbb{E}\left[N_n(d)\right] = \sum_{k \in V_n} \mathbb{P}\left[D_{n,k} = d\right]$$

and

$$\operatorname{Var}\left[N_{n}(d)\right] = \sum_{k \in V_{n}} \operatorname{Var}\left[\mathbf{1}\left[D_{n,k} = d\right]\right] + \sum_{k,\ell \in V_{n}: k \neq \ell} \operatorname{Cov}\left[\mathbf{1}\left[D_{n,k} = d\right], \mathbf{1}\left[D_{n,\ell} = d\right]\right]$$

by the binary nature of the involved rvs. Under Assumption 2.1, these expressions imply

$$\mathbb{E}\left[\frac{N_n(d)}{|V_n|}\right] = \frac{\mathbb{E}\left[N_n(d)\right]}{|V_n|} = \mathbb{P}\left[D_{n,1} = d\right]$$
(2.46)

and

$$\operatorname{Var}[N_n(d)] = |V_n| \operatorname{Var}[\mathbf{1}[D_{n,1} = d]] + |V_n|(|V_n| - 1) \cdot \operatorname{Cov}[\mathbf{1}[D_{n,1} = d], \mathbf{1}[D_{n,2} = d]],$$

respectively. Collecting terms we then conclude that

$$\mathbb{E}\left[\left|\frac{N_{n}(d)}{|V_{n}|} - p(d)\right|^{2}\right] = \left(\mathbb{P}\left[D_{n,1} = d\right] - p(d)\right)^{2} + \frac{\operatorname{Var}\left[\mathbf{1}\left[D_{n,1} = d\right]\right]}{|V_{n}|} + \frac{|V_{n}| - 1}{|V_{n}|} \cdot \operatorname{Cov}\left[\mathbf{1}\left[D_{n,1} = d\right], \mathbf{1}\left[D_{n,2} = d\right]\right].(2.47)$$

Let n go to infinity in (2.47): Assumption 2.2 implies

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1} = d\right] = p(d), \tag{2.48}$$

whereas

$$\lim_{n \to \infty} \frac{\operatorname{Var} \left[ \mathbf{1} \left[ D_{n,1} = d \right] \right]}{|V_n|} = 0$$

always holds, and the equivalence of the limits at (2.44) immediately follows.

It is worth pointing out that under Assumptions 2.1-2.2, the arguments just given also show that

$$\liminf_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1} = d \right], \mathbf{1} \left[ D_{n,2} = d \right] \right] \ge 0.$$
(2.49)

# 2.8.1 A proof of Proposition 2.1

Fix  $d = 0, 1, \ldots$  With  $\varepsilon > 0$ , Tchebychev's inequality gives

$$\mathbb{P}\left[\left|\frac{N_n(d)}{|V_n|} - p(d)\right| > \varepsilon\right] \leqslant \frac{1}{\varepsilon^2} \cdot \mathbb{E}\left[\left|\frac{N_n(d)}{|V_n|} - p(d)\right|^2\right], \quad n = 2, 3, \dots$$
(2.50)

and the convergence (2.43) will be established if we show

$$\lim_{n \to \infty} \mathbb{E}\left[ \left| \frac{N_n(d)}{|V_n|} - p(d) \right|^2 \right] = 0.$$

By virtue of Lemma 2.8 this is equivalent to having

$$\lim_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1} = d \right], \mathbf{1} \left[ D_{n,2} = d \right] \right] = 0,$$

and the proof of Proposition 2.1 is now complete since this limiting condition coincides with the enforced Assumption 2.3.

## 2.8.2 A proof of Proposition 2.2

Fix  $d = 0, 1, \ldots$ , and assume that the convergence

$$\frac{N_n(d)}{|V_n|} \xrightarrow{P} {}_n L(d) \tag{2.51}$$

indeed takes place for some constant L(d) in  $\mathbb{R}$ . Then, it is also the case that

$$\lim_{n \to \infty} \mathbb{E}\left[\frac{N_n(d)}{|V_n|}\right] = L(d)$$
(2.52)

by the Bounded Convergence Theorem. The relation (2.46) (seen earlier to hold under Assumption 2.1) and Assumption 2.2 together imply

$$\lim_{n \to \infty} \mathbb{E}\left[\frac{N_n(d)}{|V_n|}\right] = \lim_{n \to \infty} \mathbb{P}\left[D_{n,1} = d\right] = p(d).$$
(2.53)

Comparing (2.52) and (2.53) yields L(d) = p(d).

Note that (2.51) (necessarily with L(d) = p(d) under the assumed conditions) occurs if and only if

$$\frac{N_n(d)}{|V_n|} \xrightarrow{L^2} {}_n p(d).$$
(2.54)

This is because convergence in probability and  $L^2$ -convergence are equivalent for uniformly bounded rvs. The latter being equivalent to

$$\lim_{n \to \infty} \mathbb{E}\left[ \left| \frac{N_n(d)}{|V_n|} - p(d) \right|^2 \right] = 0,$$

we get

$$\lim_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1} = d \right], \mathbf{1} \left[ D_{n,2} = d \right] \right] = 0$$

by a final appeal to Lemma 2.8. This completes the proof of Proposition 2.2.

# 2.9 A proof of Theorem 2.4

Before commencing the proof we state a few preliminary results.

## 2.9.1 Some well known results

We state two lemmas which present simple bounds. The details of the first lemma can be found in [38].

**Lemma 2.9.** For positive integers K, L and P such that  $K + L \leq P$ , we have

$$\left(1 - \frac{L}{P - K}\right)^{K} \leqslant \frac{\binom{P - L}{K}}{\binom{P}{K}} \leqslant \left(1 - \frac{L}{P}\right)^{K}.$$
(2.55)

This lemma directly leads to the following bounds.

**Lemma 2.10.** With positive integers K, L and P such that  $K + L \leq P$ , we have

$$1 - e^{-\frac{LK}{P}} \leqslant 1 - \frac{\binom{P-L}{K}}{\binom{P}{K}} \leqslant \frac{LK}{P-K}.$$
(2.56)

**Proof.** Lemma 2.9 yields the bounds

$$1 - e^{-\frac{LK}{P}} \leq 1 - \frac{\binom{P-L}{K}}{\binom{P}{K}} \leq 1 - \left(1 - \frac{L}{P-K}\right)^{K}.$$

The upper bound in (2.56) follows by noting that

$$1 - \left(1 - \frac{L}{P - K}\right)^K = \int_{1 - \frac{L}{P - K}}^1 K t^{K - 1} dt \le \frac{LK}{P - K}$$

It is worth mentioning that the proof given above is based on a proof given in [38, p. 2988].

**Corollary 2.11.** For any scaling  $P, K : \mathbb{N}_0 \to \mathbb{N}_0$  and any sequence  $L : \mathbb{N}_0 \to \mathbb{N}_0$ such that  $K_n + L_n < P_n$  for all n = 1, 2, ..., it holds that

$$1 - \frac{\binom{P_n - L_n}{K_n}}{\binom{P_n}{K_n}} \sim \frac{L_n K_n}{P_n} \tag{2.57}$$

if and only if

$$\lim_{n \to \infty} \frac{L_n K_n}{P_n} = 0. \tag{2.58}$$

**Proof.** From Lemma 2.10, we obtain

$$1 - e^{-\frac{L_n K_n}{P_n}} \leq 1 - \frac{\binom{P_n - L_n}{K_n}}{\binom{P_n}{K_n}} \leq \frac{L_n K_n}{P_n - K_n}, \ n = 1, 2, \dots$$
(2.59)

provided  $K_n + L_n > P_n$ . Multiply (2.59) by  $\frac{P_n}{L_n K_n}$  and let *n* go to infinity in the resulting set of inequalities. Under (2.58), we get

$$\lim_{n \to \infty} \frac{P_n}{L_n K_n} \cdot \left(1 - e^{-\frac{L_n K_n}{P_n}}\right) = 1$$

from the elementary fact  $\lim_{t\to 0^+} \frac{1-e^{-t}}{t} = 1$ , while

$$\lim_{n \to \infty} \frac{P_n}{L_n K_n} \cdot \frac{L_n K_n}{P_n - K_n} = \lim_{n \to \infty} \frac{P_n}{P_n - K_n} = 1$$

by virtue of (2.58) (as it implies  $\lim_{n\to\infty} \frac{K_n}{P_n} = 0$ ). The asymptotic equivalence (2.57) follows.

Fix a positive integer c. Under the scaling (2.36), there exists  $n_0(c)$  such that,

$$(c+1)K_n^* < P_n^*$$

for all  $n \ge n_0(c)$  and Corollary 2.11 yields

$$1 - \frac{\binom{P_n^* - cK_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}} \sim \frac{c(K_n^*)^2}{P_n^*}.$$
 (2.60)

# 2.9.2 The proof

Previously in Section 2.5, we examined a generic method for establishing the asymptotic independence of degree rvs required by Assumption 2.4. Before we can proceed further, we first need to show that Properties 2.1-2.3 are satisfied for random key graphs.

It is clear that Property 2.1 holds for the model. For n = 2, 3, ... and P, K > 0such that  $2K \leq P$ , we have from (2.38),

$$\mathbb{P}\left[\chi_n(1,2;\alpha)=1\right] = 1 - q(\alpha) = 1 - \binom{P-K}{K}$$
(2.61)

where  $\alpha \equiv (K, P)$ . Under the scaling  $\alpha^* : \mathbb{N}_0 \to A$  satisfying (2.36), we have

$$\lim_{n \to \infty} \mathbb{P}\left[\chi_n(1,2;\alpha_n^*) = 1\right] = \lim_{n \to \infty} \mathbb{P}\left[K_{n,1}(\alpha_n^*) \cap K_{n,2}(\alpha_n^*) \neq \emptyset\right] = 0$$
(2.62)

as we use (2.60) (with c = 1). This implies that Property 2.2 is satisfied under this scaling. Next, we show that Property 2.3 also holds under the aforementioned scaling. For  $\alpha \equiv (K, P)$  in A such that  $2K \leq P$ , and  $n = 3, 4, \ldots$ , we find

$$n\mathbb{P}\left[\chi_{n}(1,3;\alpha) = 1, \chi_{n}(2,3;\alpha) = 1\right]$$
$$= n\mathbb{P}\left[K_{n,1}(\alpha) \cap K_{n,3}(\alpha) \neq \emptyset, K_{n,2}(\alpha) \cap K_{n,3}(\alpha) \neq \emptyset\right]$$
$$= n\mathbb{E}\left[\left(\mathbb{P}\left[K_{n,1}(\alpha) \cap S \neq \emptyset\right]^{2}\right)_{S=K_{n,3}(\alpha)}\right] = n\left(1 - \frac{\binom{P-K}{K}}{\binom{P}{K}}\right)^{2}.$$

Under the scaling satisfying (2.36), by setting c = 1 in (2.60) we conclude that Property 2.3 is indeed satisfied.

Since Properties 2.1-2.3 hold for random key graphs, the equivalence (2.33) holds. Therefore, continuing from (2.35), for  $\alpha$  in A, all  $k, \ell = 0, 1, ...$  and n = 2, 3, ... such that  $n > k + \ell + 2$ , we find

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0\right]$$

$$= \mathbb{P}\left[\begin{array}{c} \chi_n(1,2;\alpha) = 0 \\ \chi_n(1,s;\alpha) = 1, \chi_n(2,s;\alpha) = 0, s \in S_0 \\ \chi_n(1,t;\alpha) = 0, \chi_n(2,t;\alpha) = 1, t \in T_0 \\ \chi_n(1,r;\alpha) = 0, \chi_n(2,r;\alpha) = 0, r \in (S_0 \cup T_0 \cup \{1,2\})^c \end{array}\right]$$

$$= \mathbb{P}\left[\begin{array}{c} K_{n,1}(\alpha) \cap K_{n,2}(\alpha) = \varnothing \\ K_{n,1}(\alpha) \cap K_{n,s}(\alpha) \neq \varnothing, K_{n,2}(\alpha) \cap K_{n,s}(\alpha) = \varnothing, s \in S_0 \\ K_{n,1}(\alpha) \cap K_{n,t}(\alpha) = \varnothing, K_{n,2}(\alpha) \cap K_{n,t}(\alpha) \neq \varnothing, t \in T_0 \\ K_{n,1}(\alpha) \cap K_{n,r}(\alpha) = \varnothing, K_{n,2}(\alpha) \cap K_{n,r}(\alpha) = \varnothing, r \in (S_0 \cup T_0 \cup \{1,2\})^c \end{array}\right]$$

$$(2.63)$$

with  $S_0 = \{3, \dots, k+2\}$  and  $T_0 = \{k+3, \dots, k+\ell+2\}.$ 

To help deal with (2.63), with R and S given in  $\mathcal{P}_K$ , we define the events

$$E_{1,n}(R,S;\alpha) = [R \cap K_{n,s}(\alpha) \neq \emptyset, S \cap K_{n,s}(\alpha) = \emptyset, \ s \in S_0],$$
$$E_{2,n}(R,S;\alpha) = [R \cap K_{n,t}(\alpha) = \emptyset, S \cap K_{n,t}(\alpha) \neq \emptyset, \ t \in T_0],$$

and

$$E_{3,n}(R,S;\alpha) = \left[R \cap K_{n,r}(\alpha) = \emptyset, S \cap K_{n,r}(\alpha) = \emptyset, r \in V_n \setminus (\{1,2\} \cup S_0 \cup T_0)\right].$$

Expressing (2.63) in terms of the events defined above, we see that

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0\right]$$

$$= \mathbb{E}\left[\mathbf{1}\left[K_{n,1}(\alpha) \cap K_{n,2}(\alpha) = \varnothing\right]$$

$$\times \mathbb{P}\left[E_{1,n}(R, S; \alpha) \cap E_{2,n}(R, S; \alpha) \cap E_{3,n}(R, S; \alpha) \middle| K_{n,1}(\alpha), K_{n,2}(\alpha)\right]_{K_{n,1}(\alpha) = R, K_{n,2}(\alpha) = S}\right]$$

$$= \mathbb{E}\left[\mathbf{1}\left[K_{n,1}(\alpha) \cap K_{n,2}(\alpha) = \varnothing\right]$$

$$\times \left(\mathbb{P}\left[E_{1,n}(R, S; \alpha)\right] \mathbb{P}\left[E_{2,n}(R, S; \alpha)\right] \mathbb{P}\left[E_{3,n}(R, S; \alpha)\right]\right)_{R=K_{n,1}(\alpha), S=K_{n,2}(\alpha)}\right].$$
(2.64)

For R and S in  $\mathcal{P}_K$  such that  $R \cap S = \emptyset$ , and  $\alpha \equiv (K, P)$  in A such that  $3K \leq P$ , we obtain

$$\mathbb{P}\left[E_{1,n}(R,S;\alpha)\right] = \mathbb{P}\left[R \cap K_{n,j}(\alpha) \neq \emptyset, S \cap K_{n,j}(\alpha) = \emptyset, \ j \in S_0\right]$$

$$= \mathbb{P}\left[R \cap K_{n,3}(\alpha) \neq \emptyset, S \cap K_{n,3}(\alpha) = \emptyset\right]^k$$

$$= \left(\mathbb{P}\left[S \cap K_{n,3}(\alpha) = \emptyset\right] - \mathbb{P}\left[R \cap K_{n,3}(\alpha) = \emptyset, S \cap K_{n,3}(\alpha) = \emptyset\right]\right)^k$$

$$= \left(\frac{\binom{P-K}{K}}{\binom{P-K}{K}} - \frac{\binom{P-2K}{K}}{\binom{P}{K}}\right)^k$$

$$= \left[\frac{\binom{P-K}{K}}{\binom{P-K}{K}} \left(1 - \frac{\binom{P-2K}{K}}{\binom{P-K}{K}}\right)\right]^k, \qquad (2.65)$$

and

$$\mathbb{P}\left[E_{2,n}(R,S;\alpha)\right] = \left[\frac{\binom{P-K}{K}}{\binom{P}{K}} \left(1 - \frac{\binom{P-2K}{K}}{\binom{P-K}{K}}\right)\right]^{\ell}.$$
(2.66)

Similarly, we obtain

$$\mathbb{P}\left[E_{3,n}(R,S;\alpha)\right] = \mathbb{P}\left[R \cap K_{n,3}(\alpha) = \emptyset, S \cap K_{n,3}(\alpha) = \emptyset\right]^{n-2-k-\ell}$$
$$= \left(\frac{\binom{P-2K}{K}}{\binom{P}{K}}\right)^{n-2-k-\ell}.$$
(2.67)

Substituting (2.65), (2.66) and (2.67) into (2.64) yields

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0\right]$$

$$= \left[\frac{\binom{P-K}{K}}{\binom{P}{K}} \left(1 - \frac{\binom{P-2K}{K}}{\binom{P--K}{K}}\right)\right]^{k+\ell} \left(\frac{\binom{P-2K}{K}}{\binom{P}{K}}\right)^{n-2-k-\ell} \mathbb{P}\left[K_{n,1}(\alpha) \cap K_{n,2}(\alpha) = \varnothing\right].$$
(2.68)

Next, we study the asymptotics of the individual terms in (2.68) under the scaling satisfying (2.36): Investigating the first term in (2.68), we observe

$$\frac{\binom{P_n^* - K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}} \left( 1 - \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^* - K_n^*}{K_n^*}} \right) \\
= \left( 1 - \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^* - K_n^*}{K_n^*}} \right) - \left( 1 - \frac{\binom{P_n^* - K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}} \right) \left( 1 - \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^* - K_n^*}{K_n^*}} \right).$$
(2.69)

From equation (2.60) (with c = 1), we obtain

$$\lim_{n \to \infty} n \left( 1 - \frac{\binom{P_n^* - K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}} \right) = \lambda$$
(2.70)

and

$$\lim_{n \to \infty} n \left( 1 - \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^* - K_n^*}{K_n^*}} \right) = \lambda$$
(2.71)

where the latter follows by substituting  $P_n = P_n^* - K_n^*$ ,  $K_n = K_n^*$  in (2.60) and using the fact that  $\lim_{n\to\infty} \frac{K_n^*}{P_n^*} = 0$ . Expression (2.70) and (2.71) when substituted in (2.69) yields

$$\lim_{n \to \infty} n \left[ \frac{\binom{P_n^* - K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}} \left( 1 - \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^* - K_n^*}{K_n^*}} \right) \right] = \lambda.$$
(2.72)

Next, we state a simple result to be used later.

**Lemma 2.12.** If for a sequence  $a : \mathbb{N} \to \mathbb{R}$  there exists  $a^*$  in  $\mathbb{R}$  such that

$$\lim_{n \to \infty} n a_n = a^*,$$

then

$$\lim_{n \to \infty} (1+a_n)^{n+b} = e^{a^*}, \ b \in \mathbb{R}.$$

From (2.60) (with c = 2), under the scaling satisfying (2.36), we obtain

$$\lim_{n \to \infty} n \left( 1 - \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}} \right) = 2\lambda.$$

$$(2.73)$$

Returning to (2.67), we can write

$$\left(\frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}}\right)^{n-2-k-\ell} = \left[1 - \left(1 - \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}}\right)\right]^{n-2-k-\ell}$$

and Lemma 2.12 now yields

$$\lim_{n \to \infty} \left( \frac{\binom{P_n^* - 2K_n^*}{K_n^*}}{\binom{P_n^*}{K_n^*}} \right)^{n-2-k-\ell} = e^{-2\lambda}$$
(2.74)

with the help of (2.73). Applying the asymptotic results (2.62), (2.72) and (2.74), we get from (2.68) that

$$\lim_{n \to \infty} n^{k+\ell} \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha_n^*) = S_0, \mathcal{N}_{n,2}(\alpha_n^*) = T_0\right] = \lambda^{k+\ell} e^{-2\lambda}.$$

This last relation readily leads to

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^*) = k, D_{n,2}(\alpha_n^*) = \ell\right] = \left(\frac{\lambda^k}{k!}e^{-\lambda}\right) \left(\frac{\lambda^\ell}{\ell!}e^{-\lambda}\right)$$
$$= p_\lambda(k)p_\lambda(\ell).$$

upon using (2.33) and (2.34).

### 2.10 A proof of Theorem 2.6

First, we define a number of events to be used in the proof of Theorem 2.6, and compute their asymptotic probabilities under the appropriate scaling  $\alpha^*$ .

# 2.10.1 Preliminaries

Fix  $n = 2, 3, \ldots$  and  $0 < \alpha < \frac{1}{2}$ . We find it useful to define the set of points

$$A_n(\alpha) = \{ (X, Y) \mid \alpha < X < 1 - \alpha, \ \alpha < Y < 1 - \alpha \}.$$

For each i in  $V_n$ , we find it useful to define the event

$$E_{n,i}(\alpha) = \left[ (X_i, Y_i) \in A_n(\alpha) \right].$$

It is easy to see that

$$\mathbb{P}\left[E_{n,i}(\alpha)\right] = (1 - 2\alpha)^2. \tag{2.75}$$

On this event a particular node is at least distance  $\alpha$  away from the borders of the square  $[0, 1]^2$ . Next we define the event where a pair of nodes (say nodes 1 and 2) are at least distance  $2\alpha$  apart from each other, namely

$$E_n(\alpha) = \left[ d(\boldsymbol{Z}_1, \boldsymbol{Z}_2) > 2\alpha \right].$$

Using simple arguments we obtain the following lower bound

$$\mathbb{P}\left[\tilde{E}_{n}(\alpha)\right] = \mathbb{P}\left[d(\boldsymbol{Z}_{1}, \boldsymbol{Z}_{2}) > 2\alpha\right]$$
$$= \mathbb{E}\left[\mathbb{P}\left[d(\boldsymbol{Z}_{1}, \boldsymbol{Z}_{2}) > 2\alpha \mid \boldsymbol{Z}_{1}\right]\right]$$
$$\geq 1 - 4\pi\alpha^{2}.$$
(2.76)

For  $0 < \alpha < \frac{1}{2}$ , define the event  $E_n(\alpha) = E_{n,1}(\alpha) \cap E_{n,2}(\alpha) \cap \tilde{E}_n(\alpha)$ . Under the scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  satisfying (2.41), we obtain

$$\lim_{n \to \infty} \mathbb{P}\left[E_n(\alpha_n^*)\right] = 1 \tag{2.77}$$

since

$$\lim_{n \to \infty} \mathbb{P}\left[E_{n,j}(\alpha_n^*)\right] = 1, \ j = 1, 2, \text{ and}$$
$$\lim_{n \to \infty} \mathbb{P}\left[\tilde{E}_n(\alpha_n^*)\right] = 1.$$

# 2.10.2 The proof

First, we note that random geometric graphs satisfy Property 2.1. For 0 <  $\alpha < \frac{1}{2}$  and  $n = 2, 3, \ldots$ , it is plain that

$$\mathbb{P}\left[\chi_n(1,2;\alpha)=1\right] = \mathbb{P}\left[|\boldsymbol{Z}_1 - \boldsymbol{Z}_2| \leq \alpha\right] \leq \pi \alpha^2$$

and Property 2.2 is therefore satisfied under the scaling specified in (2.41). Again, for  $\alpha$  in A and  $n = 2, 3, \ldots$ , we have

$$n\mathbb{P}\left[\chi_n(1,3;\alpha) = 1, \chi_n(2,3;\alpha) = 1\right] = n\mathbb{P}\left[|\boldsymbol{Z}_1 - \boldsymbol{Z}_3| \leq \alpha, |\boldsymbol{Z}_2 - \boldsymbol{Z}_3| \leq \alpha\right]$$
$$= n\mathbb{E}\left[\left(\mathbb{P}\left[|\boldsymbol{Z}_1 - \mathbf{x}| \leq \alpha, |\boldsymbol{Z}_2 - \mathbf{x}| \leq \alpha\right]\right)_{\mathbf{x}=\boldsymbol{Z}_3}\right]$$
$$= n\mathbb{E}\left[\left(\mathbb{P}\left[|\boldsymbol{Z}_1 - \mathbf{x}| \leq \alpha\right]^2\right)_{\mathbf{x}=\boldsymbol{Z}_3}\right]$$
$$\leq n \cdot (\pi \alpha^2)^2$$

and Property 2.3 is satisfied under the scaling  $\alpha^*$ . Properties 2.1-2.3 being satisfied, the equivalence (2.33) is valid for random geometric graphs. Continuing from (2.35), for  $0 < \alpha < \frac{1}{2}$ , all  $k, \ell = 0, 1, ...$  and n = 2, 3, ... such that  $n > k + \ell + 2$ , we get

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_{0}, \mathcal{N}_{n,2}(\alpha) = T_{0}\right] \\
= \mathbb{P}\left[\begin{array}{c} \chi_{n}(1,2;\alpha) = 0 \\ \chi_{n}(1,s;\alpha) = 1, \chi_{n}(2,s;\alpha) = 0, s \in S_{0} \\ \chi_{n}(1,t;\alpha) = 0, \chi_{n}(2,t;\alpha) = 1, t \in T_{0} \\ \chi_{n}(1,r;\alpha) = 0, \chi_{n}(2,r;\alpha) = 0, r \in (S_{0} \cup T_{0} \cup \{1,2\})^{c} \end{array}\right] \\
= \mathbb{P}\left[\begin{array}{c} |\mathbf{Z}_{1} - \mathbf{Z}_{2}| > \alpha \\ |\mathbf{Z}_{1} - \mathbf{Z}_{s}| \leq \alpha, |\mathbf{Z}_{2} - \mathbf{Z}_{s}| > \alpha, s \in S_{0} \\ |\mathbf{Z}_{1} - \mathbf{Z}_{t}| > \alpha, |\mathbf{Z}_{2} - \mathbf{Z}_{t}| \leq \alpha, t \in T_{0} \\ |\mathbf{Z}_{1} - \mathbf{Z}_{r}| > \alpha, |\mathbf{Z}_{2} - \mathbf{Z}_{r}| > \alpha, r \in (S_{0} \cup T_{0} \cup \{1,2\})^{c} \end{array}\right] (2.78)$$

with  $S_0 = \{3, \dots, k+2\}$  and  $T_0 = \{k+3, \dots, k+\ell+2\}$  as before.

To help deal with (2.78), we define the events

$$E_{1,n}'(\mathbf{x}, \mathbf{y}; \alpha) = [|\mathbf{x} - \mathbf{Z}_s| \leq \alpha, |\mathbf{y} - \mathbf{Z}_s| > \alpha, s \in S_0],$$
  

$$E_{2,n}'(\mathbf{x}, \mathbf{y}; \alpha) = [|\mathbf{x} - \mathbf{Z}_t| > \alpha, |\mathbf{y} - \mathbf{Z}_t| \leq \alpha, t \in T_0],$$
  
and

$$E'_{3,n}(\mathbf{x}, \mathbf{y}; \alpha) = \left[ |\mathbf{y} - \mathbf{Z}_r| > \alpha, |\mathbf{y} - \mathbf{Z}_r| > \alpha, \ r \in (S_0 \cup T_0 \cup \{1, 2\})^c \right]$$
(2.79)

for  $\mathbf{x}, \mathbf{y}$  in  $(0, 1)^2$ . We decompose (2.78) further by writing

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0\right]$$

$$= \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0, E_n(\alpha)\right] + \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0, E_n(\alpha)^c\right].$$
(2.80)

Writing the first term in (2.80) in terms of the events defined above, we obtain

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0, E_n(\alpha)\right]$$
$$= \mathbb{E}\left[\mathbf{1}\left[E_n(\alpha)\right] \left(\mathbb{P}\left[E'_{1,n}(\mathbf{x}, \mathbf{y}; \alpha)\right] \mathbb{P}\left[E'_{2,n}(\mathbf{x}, \mathbf{y}; \alpha)\right] \mathbb{P}\left[E'_{3,n}(\mathbf{x}, \mathbf{y}; \alpha)\right]\right)_{\mathbf{x} = \mathbf{Z}_{1, \mathbf{y} = \mathbf{Z}_{2}}\right].$$
(2.81)

Observe that under the event  $E_{n,1}(\alpha)$ , we can draw a circle of radius  $\alpha$  centered at node 1 which is completely contained within the square  $[0, 1]^2$ , implying that the probability of an edge forming between node 1 and any node s in  $S_0$  is  $\pi \alpha^2$ . Also, under the event  $\tilde{E}_n(\alpha)$  any node connected to node 1 cannot be connected to node 2. Thus, for  $\mathbf{x}, \mathbf{y}$  in  $A_n(\alpha)$  such that  $d(\mathbf{x}, \mathbf{y}) > 2\alpha$ , we have

$$\mathbb{P}\left[E_{1,n}'(\mathbf{x}, \mathbf{y}; \alpha)\right] = \mathbb{P}\left[|\mathbf{x} - \mathbf{Z}_{s}| \leq \alpha, |\mathbf{y} - \mathbf{Z}_{s}| > \alpha, s \in S_{0}\right]$$
$$= \mathbb{P}\left[|\mathbf{x} - \mathbf{Z}_{3}| \leq \alpha, |\mathbf{y} - \mathbf{Z}_{3}| > \alpha\right]^{k}$$
$$= \left(\pi \alpha^{2}\right)^{k}$$
(2.82)

and

$$\mathbb{P}\left[E_{2,n}'(\mathbf{x},\mathbf{y};\alpha)\right] = \left(\pi\alpha^2\right)^{\ell}.$$
(2.83)

Under the event  $E_n(\alpha)$ , circles of radius  $\alpha$  centred at  $\mathbf{x}$  and  $\mathbf{y}$  do not intersect. Therefore the probability that a particular node does not connect to either node 1 or node 2 is given by  $(1 - 2\pi\alpha^2)$ . Hence, for  $\mathbf{x}, \mathbf{y}$  in  $A_n(\alpha)$  such that  $d(\mathbf{x}, \mathbf{y}) > 2\alpha$ , we have

$$\mathbb{P}\left[E_{3,n}'(\mathbf{x}, \mathbf{y}; \alpha)\right] = \mathbb{P}\left[|\mathbf{y} - \mathbf{Z}_{r}| > \alpha, |\mathbf{y} - \mathbf{Z}_{r}| > \alpha, r \in (S_{0} \cup T_{0} \cup \{1, 2\})^{c}\right]$$
$$= \mathbb{P}\left[|\mathbf{y} - \mathbf{Z}_{k+\ell+3}| > \alpha, |\mathbf{y} - \mathbf{Z}_{k+\ell+3}| > \alpha\right]^{n-2-k-\ell}$$
$$= (1 - 2\pi\alpha^{2})^{n-2-k-\ell}.$$
(2.84)

Furthermore, for all  $\mathbf{x}, \mathbf{y}$  in  $[0, 1]^2$ , we have the following upper bound

$$\mathbb{P}\left[E_{1,n}'(\mathbf{x}, \mathbf{y}; \alpha)\right] = \mathbb{P}\left[|\mathbf{x} - \mathbf{Z}_{s}| \leq \alpha, |\mathbf{y} - \mathbf{Z}_{s}| > \alpha, s \in S_{0}\right]$$
$$\leq \mathbb{P}\left[|\mathbf{x} - \mathbf{Z}_{s}| \leq \alpha, s \in S_{0}\right]$$
$$\leq \left(\pi \alpha^{2}\right)^{k}$$
(2.85)

and

$$\mathbb{P}\left[E_{2,n}'(\mathbf{x},\mathbf{y};\alpha)\right] \leqslant \left(\pi\alpha^2\right)^{\ell}.$$
(2.86)

Using (2.82),(2.83) and (2.84) on (2.81), we obtain

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0, E_n(\alpha)\right] = (\pi \alpha^2)^{k+\ell} (1 - 2\pi \alpha^2)^{n-2-k-\ell} \mathbb{P}\left[E_n(\alpha)\right].$$
(2.87)

Under the scaling  $\alpha^*$  satisfying (2.41), we obtain

$$\lim_{n \to \infty} n^{k+\ell} \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha_n^*) = S_0, \mathcal{N}_{n,2}(\alpha_n^*) = T_0, E_n(\alpha_n^*)\right] = \lambda^{k+\ell} e^{-2\lambda}$$
(2.88)

upon using Lemma 2.12 and (2.77). The bounds (2.85) and (2.86) allow us to bound the second term in (2.80) as

$$\mathbb{P}\left[\mathcal{N}_{n,1}(\alpha) = S_0, \mathcal{N}_{n,2}(\alpha) = T_0, E_n(\alpha)^c\right]$$
  
=  $\mathbb{E}\left[\mathbf{1}\left[E_n(\alpha)^c\right] \left(\mathbb{P}\left[E'_{1,n}(\mathbf{x}, \mathbf{y}; \alpha)\right] \mathbb{P}\left[E'_{2,n}(\mathbf{x}, \mathbf{y}; \alpha)\right] \mathbb{P}\left[E'_{3,n}(\mathbf{x}, \mathbf{y}; \alpha)\right]\right)_{\mathbf{x}=\mathbf{Z}_{1}, \mathbf{y}=\mathbf{Z}_{2}}\right]$   
 $\leq \mathbb{E}\left[\mathbf{1}\left[E_n(\alpha)^c\right] \left(\mathbb{P}\left[E'_{1,n}(\mathbf{x}, \mathbf{y}; \alpha)\right] \mathbb{P}\left[E'_{2,n}(\mathbf{x}, \mathbf{y}; \alpha)\right]\right)_{\mathbf{x}=\mathbf{Z}_{1}, \mathbf{y}=\mathbf{Z}_{2}}\right]$   
 $\leq (\pi \alpha^2)^{k+\ell} \mathbb{P}\left[E_n(\alpha)^c\right],$ 

whence

$$\lim_{n \to \infty} n^{k+\ell} \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha_n^*) = S_0, \mathcal{N}_{n,2}(\alpha_n^*) = T_0, E_n(\alpha_n^*)^c\right] = 0$$
(2.89)

by virtue of (2.77). Returning to (2.80) and using (2.88) and (2.89), we obtain

$$\lim_{n \to \infty} n^{k+\ell} \mathbb{P}\left[\mathcal{N}_{n,1}(\alpha_n^*) = S_0, \mathcal{N}_{n,2}(\alpha_n^*) = T_0\right] = \lambda^{k+\ell} e^{-2\lambda}.$$
(2.90)

Substituting into (2.34) and using the equivalence step (2.33), we obtain

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^*) = k, D_{n,2}(\alpha_n^*) = \ell\right] = \left(\frac{\lambda^k}{k!}e^{-\lambda}\right) \left(\frac{\lambda^\ell}{\ell!}e^{-\lambda}\right)$$
$$= p_{\lambda}(k)p_{\lambda}(\ell).$$

# Chapter 3: A Counterexample: Random Threshold Graphs

In the previous chapter, we introduced a generic framework for studying the degree distributions of random graphs. We identified a set of necessary and sufficient conditions under which the network-wide (empirical) degree distribution coincides with the nodal degree distribution in the large graph limit. We also demonstrated three instances of homogeneous graphs where these conditions are satisfied, namely Erdős-Rényi graphs, random key graphs and random geometric graphs.

In the present chapter, we give a counterexample to show that *even* in homogeneous graphs, the empirical degree distribution and the nodal degree distribution may capture vastly different information. This counterexample is found in the class of random threshold graphs, where the empirical degree distribution does not converge in the usual sense even though the asymptotic nodal degree distribution exists.

### 3.1 The model

As discussed earlier in Chapter 1, Caldarelli et al. [9] proposed the class of random threshold graphs as capable of achieving scale-free degree distribution without the notion of preferential attachment [2]. It belongs to the broader class of *hidden variable* models where connections are formed on the basis of *fitness variables*  associated with individual nodes. The random threshold graph model is based on the notion that connections between nodes are driven by mutual benefit based on intrinsic attributes, and is realized as follows – Two nodes form a connection if the sum of their fitness variables exceeds a certain threshold.

We now formally introduce the model: Let  $\{\xi, \xi_k, k = 1, 2, ...\}$  denote a collection of i.i.d.  $\mathbb{R}_+$ -valued rvs defined on the probability triple  $(\Omega, \mathcal{F}, \mathbb{P})$ , each distributed according to a given (probability) distribution function  $F : \mathbb{R} \to [0, 1]$ . <sup>1</sup> With  $\xi$  acting as a generic representative for this sequence of i.i.d. rvs, we have

$$\mathbb{P}\left[\xi \leqslant x\right] = F(x), \quad x \in \mathbb{R}.$$

At minimum we assume that F is a continuous function on  $\mathbb R$  with

$$F(x) = 0, \quad x \le 0.$$

Once F is specified, random thresholds graphs are characterized by two parameters, namely a positive integer n and a threshold value  $\alpha > 0$ . Specifically, the network comprises n nodes, labelled k = 1, ..., n, and to each node k we assign a *fitness* variable (or weight)  $\xi_k$  which measures its importance or rank. For distinct i, j = 1, ..., n, nodes i and j are declared to be adjacent if

$$\xi_i + \xi_j > \alpha, \tag{3.1}$$

i.e.,

$$\chi_n(i,j;\alpha) = \mathbf{1} \left[ \xi_i + \xi_j > \alpha \right]$$

<sup>&</sup>lt;sup>1</sup>What we call here a probability distribution function is also called a cumulative distribution function in other literatures.

according to the notation developed in Section 2.1. The adjacency notion (3.1) defines the random threshold graph  $\mathbb{T}(n; \alpha)$  on the set of vertices  $V_n = \{1, \ldots, n\}$ .

For each  $n = 1, 2, \ldots$ , and each  $\alpha > 0$ , we set

$$D_{n,k}(\alpha) = \sum_{\ell=1, \ \ell \neq k}^{n} \mathbf{1} \left[ \xi_k + \xi_\ell > \alpha \right], \quad k = 1, \dots, n$$

so that  $D_{n,k}(\alpha)$  is the degree of node k in  $\mathbb{T}(n; \alpha)$ . Under the enforced assumptions, the rvs  $D_{n,1}(\alpha), \ldots, D_{n,n}(\alpha)$  are exchangeable, thus equidistributed. Furthermore, the edge rvs  $\begin{cases} \chi_n(i,j;\alpha), & i \neq j \\ i,j \in V_n \end{cases}$  constitute an exchangeable family, i.e., Property 2.1 is satisfied.

#### 3.1.1 Applying Proposition 2.1 under exponential fitness

From now on we focus on the special case when  $\xi$  is exponentially distributed with parameter  $\lambda > 0$ , written  $\xi \sim \text{Exp}(\lambda)$ , that is

$$\mathbb{P}\left[\xi \leqslant x\right] = 1 - e^{-\lambda x^+}, \quad x \in \mathbb{R}.$$
(3.2)

Here we use the standard notation  $x^+ = \max(x, 0)$  for x in  $\mathbb{R}$ . Other distributions could be considered to develop counterexamples to Proposition 2.1. However, the exponential distribution was selected for two main reasons: This case was considered in [9,22,37] to show that scale-free networks can be generated through the fitnessbased mechanism used in random threshold graphs; more on that later. Moreover, calculations are greatly simplified in the exponential case.

Fujihara et al. [22, Example 1, p. 366] showed that under the scaling  $\alpha^*$ :

 $\mathbb{N}_0 \to \mathbb{R}_+$  given by

$$\alpha_n^{\star} = \lambda^{-1} \log n, \quad n = 2, 3, \dots$$
(3.3)

with the understanding that

$$\mathbb{G}_n = \mathbb{T}(n; \alpha_n^*), \quad n = 2, 3, \dots$$

we have the following distributional convergence  $D_{n,1}(\alpha_n^*) \Longrightarrow_n D$  where the Nvalued rv D is a conditionally Poisson rv with pmf  $\mathbf{p}_{\text{Fuj}} = (p_{\text{Fuj}}(d), d = 0, 1, ...)$ given by

$$p_{\mathrm{Fuj}}(d) = \mathbb{P}\left[D = d\right] = \mathbb{E}\left[\frac{(e^{\lambda\xi})^d}{d!}e^{-e^{\lambda\xi}}\right], \quad d = 0, 1, \dots$$
(3.4)

Therefore, Assumption 2.2 holds with

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^*) = d\right] = p_{\mathrm{Fuj}}(d), \quad d = 0, 1, \dots$$
(3.5)

Hence, we are in the generic setting of Section 2.5. One way to prove Assumption 2.4 would be to first show that Properties 2.1-2.3 hold. While we have already argued that Property 2.1 is satisfied, the following result states that in the present context Property 2.2 holds whereas Property 2.3 does not.

**Proposition 3.1.** Consider the class of random threshold graphs with  $\xi \sim Exp(\lambda)$ for some  $\lambda > 0$ . Under the scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  given by (3.3), Property 2.2 holds whereas Property 2.3 does not. **Proof.** For a fixed  $\alpha > 0$ , and  $n = 2, 3, \ldots$ , we get

$$\mathbb{P}\left[\xi_{1}+\xi_{2}>\alpha\right] = \mathbb{E}\left[\left(\mathbb{P}\left[\xi_{2}>\alpha-t\right]\right)_{t=\xi_{1}}\right]$$
$$= \mathbb{E}\left[\mathbf{1}\left[\xi_{1}>\alpha\right]\right] + \mathbb{E}\left[\mathbf{1}\left[\xi_{1}\leqslant\alpha\right]\left(\mathbb{P}\left[\xi_{2}>\alpha-t\right]\right)_{t=\xi_{1}}\right]$$
$$= \mathbb{P}\left[\xi_{1}>\alpha\right] + \mathbb{E}\left[\mathbf{1}\left[\xi_{1}\leqslant\alpha\right]e^{-\lambda(\alpha-\xi_{1})}\right]$$
$$= e^{-\lambda\alpha}\left[1 + \mathbb{E}\left[\mathbf{1}\left[\xi_{1}\leqslant\alpha\right]e^{\lambda\xi_{1}}\right]\right]$$
$$= e^{-\lambda\alpha}\left(1 + \alpha\lambda\right)$$

by easy calculation, and Property 2.2 clearly holds under the scaling  $\alpha^*$  satisfying (3.3).

For a fixed  $\alpha > 0$ , and  $n = 3, 4, \ldots$ , we now see that

$$\mathbb{P}\left[\xi_{1} + \xi_{3} > \alpha, \xi_{2} + \xi_{3} > \alpha\right]$$

$$= \mathbb{E}\left[\mathbf{1}\left[\xi_{3} > \alpha\right]\right] + \mathbb{E}\left[\mathbf{1}\left[\xi_{3} \leqslant \alpha\right] \left(\mathbb{P}\left[\xi_{1} > \alpha - t\right]^{2}\right)_{t=\xi_{3}}\right]$$

$$= \mathbb{P}\left[\xi_{3} > \alpha\right] + \mathbb{E}\left[\mathbf{1}\left[\xi_{3} \leqslant \alpha\right] e^{-2\lambda(\alpha - \xi_{3})}\right]$$

$$= e^{-\lambda\alpha} + e^{-2\lambda\alpha} \cdot \mathbb{E}\left[\mathbf{1}\left[\xi_{3} \leqslant \alpha\right] e^{2\lambda\xi_{3}}\right]$$

$$= e^{-\lambda\alpha} + e^{-2\lambda\alpha} \int_{0}^{\alpha} \lambda e^{-\lambda y} e^{2\lambda y} dy$$

$$= e^{-\lambda\alpha} \left(2 - e^{-\lambda\alpha}\right).$$

Under the scaling  $\alpha^*$  given by (3.3), we conclude

$$\lim_{n \to \infty} n \mathbb{P}\left[\xi_1 + \xi_3 > \alpha_n^*, \xi_2 + \xi_3 > \alpha_n^*\right] = 2,$$

so that Property 2.3 does not hold.

Therefore the approach formulated in Section 2.5 using the step (2.33) is not applicable here. Hence, we are left with no choice but to directly test the validity of the weaker Assumption 2.3. The remainder of the chapter is devoted to showing that the weaker Assumption 2.3 actually fails.

**Proposition 3.2.** Consider the class of random threshold graphs with  $\xi \sim \text{Exp}(\lambda)$ for some  $\lambda > 0$ . Under the scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  given by (3.3), the limit

$$C(d) \equiv \lim_{n \to \infty} \operatorname{Cov}[\mathbf{1} \left[ D_{n,1}(\alpha_n^{\star}) = d \right], \mathbf{1} \left[ D_{n,2}(\alpha_n^{\star}) = d \right]]$$
(3.6)

exists with C(d) > 0 for each  $d = 0, 1, \ldots$ 

The specific values for (3.6) are omitted here but are computed during the proof of Proposition 3.2 given from Section 3.2 to Section 3.4. For instance, we show that

$$C(0) = \mathbb{E}\left[e^{-\max(e^{\lambda\xi_1}, e^{\lambda\xi_2})}\right] - \mathbb{E}\left[e^{-(e^{\lambda\xi_1} + e^{\lambda\xi_2})}\right] > 0.$$

The expression of C(d) for arbitrary  $d \neq 0$  is rather cumbersome and is not shown at this time. However, the fact that C(d) > 0 on the entire range suffices to establish the desired counterexample via the observation following Proposition 2.2.

**Proposition 3.3.** Consider the class of random threshold graphs with  $\xi \sim \text{Exp}(\lambda)$ for some  $\lambda > 0$ . Under the scaling  $\theta^* : \mathbb{N}_0 \to \mathbb{R}_+$  given by (3.3), each of the sequence of rvs

$$\left\{\frac{1}{n}\sum_{k=1}^{n}\mathbf{1}\left[D_{n,k}(\theta_{n}^{\star})=d\right], \ n=2,3,\ldots\right\}, \ d=0,1,\ldots$$
(3.7)

does not converge in probability to a constant.

The fact that the convergence (2.43) fails to occur in the context of random threshold graphs is significant for the following reason: Caldarelli et al. [9,37] have proposed this subclass of hidden variable models as an alternative scale-free model to the preferential attachment model of Barabási and Albert [2]. The evidence behind this proposal lies in the provable power-law behavior [22, Example 1, p. 366]

$$p_{\mathrm{Fuj}}(d) \sim d^{-2} \quad (d \to \infty).$$
 (3.8)

However, a meaningful comparison between the two models would have required at minimum the validity of the convergence

$$\frac{1}{n}\sum_{k=1}^{n}\mathbf{1}\left[D_{n,k}(\alpha_{n}^{\star})=d\right] \xrightarrow{P} p_{\mathrm{Fuj}}(d), \quad d=0,1,\ldots$$

As we now know through Proposition 3.3, this fails to happen, and the two models cannot be meaningfully compared as already explained in the introductory section.

# 3.2 A proof of Proposition 3.2 – Part I

We begin with an easy observation.

## 3.2.1 A reduction step

For every n = 2, 3, ... and  $\alpha > 0$ , note the decomposition

$$D_{n,j}(\alpha) = \mathbf{1} \left[ \xi_1 + \xi_2 > \alpha \right] + D_{n,j}^{\star}(\alpha), \quad j = 1, 2$$
(3.9)

where we have set

$$D_{n,j}^{\star}(\alpha) = \sum_{k=3}^{n} \mathbf{1} \left[ \xi_j + \xi_k > \alpha \right].$$

Evaluating the limit (3.6) can be simplified through an easy reduction step which we now develop.

**Lemma 3.4.** Fix n = 2, 3, ... and  $\alpha > 0$ . For each d = 0, 1, ..., n - 2, we have

$$\left|\mathbb{P}\left[D_{n,j}(\alpha)=d\right]-\mathbb{P}\left[D_{n,j}^{\star}(\alpha)=d\right]\right| \leq 2\mathbb{P}\left[\xi_{1}+\xi_{2}>\alpha\right], \quad j=1,2$$
(3.10)

and

$$\left|\mathbb{P}\left[D_{n,1}(\alpha) = d, D_{n,2}(\alpha) = d\right] - \mathbb{P}\left[D_{n,1}^{\star}(\alpha) = d, D_{n,2}^{\star}(\alpha) = d\right]\right| \leq 2\mathbb{P}\left[\xi_1 + \xi_2 > \alpha\right].$$
(3.11)

**Proof.** The first part of the Lemma can be proved similarly to that of Lemma 2.3. To prove the second part, fix n = 2, 3, ... and  $\alpha > 0$ . Firstly for d = 0, using the decomposition (3.9) we find

$$\mathbb{P}\left[D_{n,1}(\alpha) = 0, D_{n,2}(\alpha) = 0\right]$$
  
=  $\mathbb{P}\left[D_{n,1}^{*}(\alpha) = 0, D_{n,2}^{*}(\alpha) = 0, \xi_{1} + \xi_{2} \leq \alpha\right]$   
=  $\mathbb{P}\left[D_{n,1}^{*}(\alpha) = 0, D_{n,2}^{*}(\alpha) = 0\right] - \mathbb{P}\left[D_{n,1}^{*}(\alpha) = 0, D_{n,2}^{*}(\alpha) = 0, \xi_{1} + \xi_{2} > \alpha\right]$  (3.12)

and the bound (3.11) holds. For each d = 1, 2, ..., n - 2, we have

$$\mathbb{P}\left[D_{n,1}(\alpha) = d, D_{n,2}(\alpha) = d\right] \\
= \mathbb{P}\left[\mathbf{1}\left[\xi_{1} + \xi_{2} > \alpha\right] + D_{n,1}^{\star}(\alpha) = d, \mathbf{1}\left[\xi_{1} + \xi_{2} > \alpha\right] + D_{n,2}^{\star}(\alpha) = d\right] \\
= \mathbb{P}\left[\xi_{1} + \xi_{2} \leq \alpha, D_{n,1}^{\star}(\alpha) = d, D_{n,2}^{\star}(\alpha) = d\right] \\
+ \mathbb{P}\left[\xi_{1} + \xi_{2} > \alpha, D_{n,1}^{\star}(\alpha) = d - 1, D_{n,2}^{\star}(\alpha) = d - 1\right] \\
= \mathbb{P}\left[\xi_{1} + \xi_{2} > \alpha, D_{n,1}^{\star}(\alpha) = d - 1, D_{n,2}^{\star}(\alpha) = d - 1\right] \\
- \mathbb{P}\left[\xi_{1} + \xi_{2} > \alpha, D_{n,1}^{\star}(\alpha) = d, D_{n,2}^{\star}(\alpha) = d\right] + \mathbb{P}\left[D_{n,1}^{\star}(\alpha) = d, D_{n,2}^{\star}(\alpha) = d\right] \\$$
(3.13)

and the bound (3.11) follows by combining (3.12) and (3.13).

This simple fact leads to the following reduction step when evaluating the limit at (3.6): With d = 0, 1, ... held fixed, for each n = 2, 3, ... we substitute  $\alpha$  by  $\alpha_n^{\star}$  in the bound (3.11) according to (3.3), and let n go to infinity in the resulting inequality. Since  $\lim_{n\to\infty} \alpha_n^{\star} = \infty$ , we conclude that

$$\lim_{n \to \infty} \left| \mathbb{P}\left[ D_{n,1}(\alpha_n^{\star}) = d, D_{n,2}(\alpha_n^{\star}) = d \right] - \mathbb{P}\left[ D_{n,1}^{\star}(\alpha_n^{\star}) = d, D_{n,2}^{\star}(\alpha_n^{\star}) = d \right] \right| = 0,$$

whence

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^{\star}) = d, D_{n,2}(\alpha_n^{\star}) = d\right] = \lim_{n \to \infty} \mathbb{P}\left[D_{n,1}^{\star}(\alpha_n^{\star}) = d, D_{n,2}^{\star}(\alpha_n^{\star}) = d\right] \quad (3.14)$$

provided either limit exists. The same argument applied to the bounds (3.10) readily yields

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,j}(\alpha_n^{\star}) = d\right] = \lim_{n \to \infty} \mathbb{P}\left[D_{n,j}^{\star}(\alpha_n^{\star}) = d\right] = p_{\mathrm{Fuj}}(d), \quad j = 1, 2$$
(3.15)

in light of (3.5). It follows from (3.15) and (3.14) that

$$C(d) = \lim_{n \to \infty} \operatorname{Cov}\left[\mathbf{1}\left[D_{n,1}^{\star}(\alpha_n^{\star}) = d\right], \mathbf{1}\left[D_{n,2}^{\star}(\alpha_n^{\star}) = d\right]\right]$$
(3.16)

provided the limits at (3.14) exist.

### 3.2.2 Order statistics to the rescue

When evaluating (3.16), it will be convenient to introduce a second collection of  $\mathbb{R}_+$ -valued rvs { $\eta_{\ell}$ ,  $\ell = 1, 2, ...$ }. We assume that the rvs { $\eta_{\ell}$ ,  $\ell = 1, 2, ...$ } are also i.i.d. rvs, each of which is exponentially distributed with parameter  $\lambda > 0$ . The two collections { $\xi, \xi_k, k = 1, 2, ...$ } and { $\eta_{\ell}, \ell = 1, 2, ...$ } are assumed to be mutually independent. For each integer p = 2, 3, ..., let  $\eta_{p,1}, ..., \eta_{p,p}$  denote the values of the rvs  $\eta_1, ..., \eta_p$  arranged in increasing order, namely  $\eta_{p,1} \leq ... \leq \eta_{p,p}$ , with a lexicographic tiebreaker when needed. The rvs  $\eta_{p,1}, ..., \eta_{p,p}$  are known as the order statistics associated with the collection  $\eta_1, ..., \eta_p$ , and the rvs  $\eta_{p,1}$  and  $\eta_{p,p}$  are the minimum and maximum of the rvs  $\eta_1, ..., \eta_p$ , respectively [11, 16].

To evaluate (3.16), we start with the following observation: Fix d = 0, 1, ...and take n = 2, 3, ... such that d < n - 2. Under the enforced i.i.d. assumptions, for each  $\alpha > 0$  we get

$$\left( D_{n,1}^{\star}(\alpha), D_{n,2}^{\star}(\alpha) \right) = \left( \sum_{k=3}^{n} \mathbf{1} \left[ \xi_{1} + \xi_{k} > \alpha \right], \sum_{k=3}^{n} \mathbf{1} \left[ \xi_{2} + \xi_{k} > \alpha \right] \right)$$

$$= _{st} \left( \sum_{\ell=1}^{n-2} \mathbf{1} \left[ \xi_{1} + \eta_{\ell} > \alpha \right], \sum_{\ell=1}^{n-2} \mathbf{1} \left[ \xi_{2} + \eta_{\ell} > \alpha \right] \right)$$

$$= \left( \sum_{\ell=1}^{n-2} \mathbf{1} \left[ \xi_{1} + \eta_{n-2,\ell} > \alpha \right], \sum_{\ell=1}^{n-2} \mathbf{1} \left[ \xi_{2} + \eta_{n-2,\ell} > \alpha \right] \right).$$
(3.17)

With this notation the probabilities of interest can now be expressed as

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha) = d\right] = \mathbb{P}\left[\sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_1 + \eta_{n-2,\ell} > \alpha\right] = d\right]$$
(3.18)

and

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha) = d, D_{n,2}^{\star}(\alpha) = d\right] = \mathbb{P}\left[\sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{1} + \eta_{n-2,\ell} > \alpha\right] = d, \sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{2} + \eta_{n-2,\ell} > \alpha\right] = d\right].$$
(3.19)

Two different cases arise when applying these facts: With d = 0 we find

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha) = 0\right] = \mathbb{P}\left[\sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{1} + \eta_{n-2,\ell} > \alpha\right] = 0\right]$$
$$= \mathbb{P}\left[\xi_{1} + \eta_{\ell} \leqslant \alpha, \ \ell = 1, \dots, n-2\right]$$
$$= \mathbb{P}\left[\xi_{1} + \eta_{n-2,n-2} \leqslant \alpha\right], \qquad (3.20)$$

and

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha) = 0, D_{n,2}^{\star}(\alpha) = 0\right] \\
= \mathbb{P}\left[\sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{1} + \eta_{n-2,\ell} > \alpha\right] = 0, \sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{2} + \eta_{n-2,\ell} > \alpha\right] = 0\right] \\
= \mathbb{P}\left[\xi_{1} + \eta_{\ell} \leqslant \alpha, \xi_{2} + \eta_{\ell} \leqslant \alpha, \ \ell = 1, \dots, n-2\right] \\
= \mathbb{P}\left[\xi_{1} + \eta_{n-2,n-2} \leqslant \alpha, \xi_{2} + \eta_{n-2,n-2} \leqslant \alpha\right] \\
= \mathbb{P}\left[\max(\xi_{1}, \xi_{2}) + \eta_{n-2,n-2} \leqslant \alpha\right].$$
(3.21)

For the case  $d = 1, 2, \ldots$ , we introduce the index

$$t_n(d) = n - 2 - d. (3.22)$$

Under the enforced independence assumptions we now have

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha) = d\right] = \mathbb{P}\left[\sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{1} + \eta_{n-2,\ell} > \alpha\right] = d\right]$$
$$= \mathbb{P}\left[\xi_{1} + \eta_{n-2,t_{n}(d)} \leqslant \alpha < \xi_{1} + \eta_{n-2,t_{n}(d)+1}\right]$$
$$= \mathbb{P}\left[\alpha - \eta_{n-2,t_{n}(d)+1} < \xi_{1} \leqslant \alpha - \eta_{n-2,t_{n}(d)}\right]$$
$$= \mathbb{E}\left[e^{-\lambda(\alpha - \eta_{n-2,t_{n}(d)+1})^{+}} - e^{-\lambda(\alpha - \eta_{n-2,t_{n}(d)})^{+}}\right], \qquad (3.23)$$

and

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha) = d, D_{n,2}^{\star}(\alpha) = d\right] \\
= \mathbb{P}\left[\sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{1} + \eta_{n-2,\ell} > \alpha\right] = d, \sum_{\ell=1}^{n-2} \mathbf{1}\left[\xi_{2} + \eta_{n-2,\ell} > \alpha\right] = d\right] \\
= \mathbb{P}\left[\xi_{j} + \eta_{n-2,t_{n}(d)} \leqslant \alpha < \xi_{j} + \eta_{n-2,t_{n}(d)+1}, \ j = 1, 2\right] \\
= \mathbb{P}\left[\alpha - \eta_{n-2,t_{n}(d)+1} < \xi_{j} \leqslant \alpha - \eta_{n-2,t_{n}(d)}, \ j = 1, 2\right] \\
= \mathbb{E}\left[\left(e^{-\lambda(\alpha - \eta_{n-2,t_{n}(d)+1})^{+}} - e^{-\lambda(\alpha - \eta_{n-2,t_{n}(d)})^{+}}\right)^{2}\right].$$
(3.24)

In particular we note that

$$\operatorname{Cov}\left[\mathbf{1}\left[D_{n,1}^{\star}(\alpha)=d\right],\mathbf{1}\left[D_{n,2}^{\star}(\alpha)=d\right]\right]$$
$$=\mathbb{P}\left[D_{n,1}^{\star}(\alpha)=d, D_{n,2}^{\star}(\alpha)=d\right]-\mathbb{P}\left[D_{n,1}^{\star}(\alpha)=d\right]\mathbb{P}\left[D_{n,2}^{\star}(\alpha)=d\right]$$
$$=\operatorname{Var}\left[e^{-\lambda(\alpha-\eta_{n-2,t_{n}(d)+1})^{+}}-e^{-\lambda(\alpha-\eta_{n-2,t_{n}(d)})^{+}}\right] \ge 0.$$
(3.25)

We need to evaluate the quantities (3.20), (3.21), (3.23) and (3.24), and explore their asymptotic behavior for large n when  $\alpha$  is replaced by  $\alpha_n^{\star}$  in these expressions.

#### 3.3 Asymptotic results for order statistics

In order to carry the program outlined above we need to develop some simple facts concerning the asymptotic theory of order statistics. The notation and definitions are the ones introduced in Section 3.2.2. In carrying out this asymptotic analysis, we shall rely on the following facts: Recall that for each s = 0, 1, ..., we have

$$\binom{p}{s} \sim \frac{p^s}{s!} \quad (p \to \infty), \tag{3.26}$$

and for any sequence  $a: \mathbb{N}_0 \to \mathbb{R}_+$ , it holds that

$$\lim_{p \to \infty} \left( 1 - \frac{u}{p} \right)^{a_p} = e^{-au}, \quad u > 0$$
(3.27)

whenever

$$\lim_{p \to \infty} \frac{a_p}{p} = a > 0. \tag{3.28}$$

## 3.3.1 A result in one dimension

Fix p = 1, 2, ... and t = 1, ..., p - 1. It is well known [11, p. 9] that the probability distribution function of the rv  $\eta_{p,t}$  is given by

$$\mathbb{P}\left[\eta_{p,t} \leqslant x\right] = \sum_{r=t}^{p} {p \choose r} F(x)^{r} (1 - F(x))^{p-r}$$
$$= \sum_{r=t}^{p} {p \choose r} \left(1 - e^{-\lambda x}\right)^{r} e^{-\lambda(p-r)x}, \quad x \ge 0.$$
(3.29)

For each  $s = 0, 1, \ldots$ , we introduce the mapping  $G_s : \mathbb{R} \to \mathbb{R}_+$  defined by

$$G_s(x_s) = \left(\sum_{m=0}^s \frac{e^{-mx_s}}{m!}\right) G_0(x_s), \quad x_s \in \mathbb{R}$$
(3.30)

where  $G_0 : \mathbb{R} \to \mathbb{R}_+$  is the well-known *Gumbel* distribution given by

$$G_0(x_0) = e^{-e^{-x_0}}, \quad x_0 \in \mathbb{R}.$$
 (3.31)

Using well-known stochastic monotonicity properties of Poisson rvs (with respect to their mean parameter), it is easy to check that  $G_s : \mathbb{R} \to \mathbb{R}_+$  is indeed a probability distribution function. Let  $\Lambda_s$  denote any  $\mathbb{R}$ -valued rv which is distributed according to  $G_s$ , i.e.,

$$\mathbb{P}\left[\Lambda_s \leqslant x_s\right] = G_s(x_s), \quad x_s \in \mathbb{R}.$$
(3.32)

In fact one could interpret (3.30) as

$$\mathbb{P}\left[\Lambda_s \leqslant x_s\right] = \mathbb{P}\left[\operatorname{Poi}(e^{-x_s}) \leqslant s\right], \quad x_s \in \mathbb{R}$$

where  $\text{Poi}(\lambda)$  is a generic Poisson rv with parameter  $\lambda > 0$ .

**Lemma 3.5.** For each s = 0, 1, ... we have

$$\lambda \left( \eta_{p,p-s} - \alpha_p^\star \right) \Longrightarrow_p \Lambda_s. \tag{3.33}$$

Some comments before giving a proof:

With s = 0, Lemma 3.5 gives the distributional convergence  $\lambda(\eta_{p,p} - \alpha_p^{\star}) \Longrightarrow_p \Lambda_0$  which expresses the well-known membership of exponential distributions in the maximal domain of attraction of the Gumbel distribution [16]. For future use note that (3.33) is equivalent to

$$\lim_{p \to \infty} \mathbb{P}\left[\lambda\left(\eta_{p,p-s} - \alpha_p^{\star}\right) \leqslant x_s\right] = G_s(x_s), \quad x_s \in \mathbb{R}$$
(3.34)

since every point in  $\mathbb{R}$  is a point of continuity for  $G_s$ .

**Proof.** Pick s = 0, 1, ... and  $x_s$  arbitrary in  $\mathbb{R}$ . With p > s and t = p - s such that  $x_s + \log p > 0$ , we obtain from (3.29) that

$$\mathbb{P}\left[\lambda\left(\eta_{p,p-s}-\alpha_{p}^{\star}\right)\leqslant x_{s}\right] = \mathbb{P}\left[\eta_{p,p-s}\leqslant\lambda^{-1}\left(x_{s}+\log p\right)\right] \\
= \sum_{r=p-s}^{p} \binom{p}{r} \left(1-\frac{e^{-x_{s}}}{p}\right)^{r} \frac{e^{-(p-r)x_{s}}}{p^{p-r}} \\
= \sum_{q=0}^{s} \binom{p}{p-s+q} \left(1-\frac{e^{-x_{s}}}{p}\right)^{p-s+q} \frac{e^{-(s-q)x_{s}}}{p^{s-q}} \\
= \sum_{m=0}^{s} \binom{p}{p-m} \left(1-\frac{e^{-x_{s}}}{p}\right)^{p-m} \frac{e^{-mx_{s}}}{p^{m}} \\
= \sum_{m=0}^{s} \frac{\binom{p}{m}}{p^{m}} \left(1-\frac{e^{-x_{s}}}{p}\right)^{p-m} e^{-mx_{s}}.$$
(3.35)

Let p go to infinity in this last expression: For each m = 0, 1, ..., we get  $\frac{\binom{p}{m}}{p^m} \sim \frac{1}{m!} \quad (p \to \infty) \quad \text{and} \quad \lim_{p \to \infty} \left(1 - \frac{e^{-x_s}}{p}\right)^{p-m} = e^{-e^{-x_s}} = G_0(x_s), \quad x_s \in \mathbb{R}$ 

as we make use of (3.26) and (3.27)-(3.28), respectively. Therefore, (3.34) holds, and this completes the proof.

It is easy to see that  $G_s \leq_{st} G_{s-1}$ , or equivalently,  $\Lambda_s \leq_{st} \Lambda_{s-1}$  for all  $s = 1, 2, \ldots$  where  $\leq_{st}$  denotes the usual stochastic ordering [36, Chap. 8] – It suffices to note that for all  $s = 1, 2, \ldots$ , we have  $\eta_{p,p-s} \leq_{st} \eta_{p,p-(s-1)}$  for all p > s. This can also be checked analytically through the expression (3.30).

#### 3.3.2 A result in two dimensions

To state the key asymptotic result we introduce some additional notation: For all t = 0, 1, ..., let the mapping  $H_t : \mathbb{R}_+ \to \mathbb{R}_+$  be given by

$$H_t(a) = \int_a^\infty v^t e^{-v} dv, \quad a \ge 0.$$

Then, for each s = 1, 2, ..., we define the mapping  $J_s : \mathbb{R}^2 \to \mathbb{R}_+$  by

$$J_{s}(x_{s}, x_{s-1}) = \frac{1}{s!} \left( \int_{e^{-\min(x_{s}, x_{s-1})}}^{\infty} v^{s} e^{-v} dv - e^{-sx_{s-1}} \cdot e^{-e^{-\min(x_{s}, x_{s-1})}} \right)$$
$$= \frac{1}{s!} \left( H_{s}(e^{-\min(x_{s}, x_{s-1})}) - e^{-sx_{s-1}} \cdot e^{-e^{-\min(x_{s}, x_{s-1})}} \right), \quad x_{s}, x_{s-1} \in \mathbb{R}.$$
(3.36)

**Proposition 3.6.** For each  $s = 1, 2, \ldots$ , we have

$$\lim_{p \to \infty} \mathbb{P} \left[ \lambda(\eta_{p,p-s} - \alpha_p^{\star}) \leqslant x_s, \lambda(\eta_{p,p-(s-1)} - \alpha_p^{\star}) \leqslant x_{s-1} \right]$$
$$= J_s(x_s, x_{s-1}), \quad x_s, x_{s-1} \in \mathbb{R}$$
(3.37)

where the mapping  $J_s : \mathbb{R}^2 \to \mathbb{R}_+$  is given by (3.36).

Some comments concerning this result before giving a proof in Section 3.5: First, given the convergence (3.37) it is easy to check that  $\mathbb{R}^2 \to [0,1] : (x_s, x_{s-1}) \to J_s(x_s, x_{s-1})$  is a *bona fide* probability distribution on  $\mathbb{R}^2$  with  $\lim_{x_s, x_{s-1} \to \infty} J_s(x_s, x_{s-1}) = 1$ ,

 $\lim_{x_s\to\infty} J_s(x_s, x_{s-1}) = 0 \text{ (with } x_{s-1} \text{ held fixed) and } \lim_{x_{s-1}\to\infty} J_s(x_s, x_{s-1}) = 0$ (with  $x_s$  held fixed).

Next, we turn to extracting the marginal distributions from (3.37): Upon setting  $x_{s-1} = \infty$  (resp.  $x_s = \infty$ ) in (3.37) with  $x_s$  (resp.  $x_{s-1}$ ) arbitrary but held fixed, we get

$$\lim_{p \to \infty} \mathbb{P}\left[\lambda(\eta_{p,p-s} - \alpha_p^{\star}) \leqslant x_s\right] = \frac{1}{s!} H_s(e^{-x_s}), \quad x_s \in \mathbb{R}$$
(3.38)

and

$$\lim_{p \to \infty} \mathbb{P} \left[ \lambda(\eta_{p,p-(s-1)} - \alpha_p^{\star}) \leqslant x_{s-1} \right] \\ = \frac{1}{s!} \left( H_s(e^{-x_{s-1}}) - e^{-sx_{s-1}} \cdot e^{-e^{-x_{s-1}}} \right), \quad x_{s-1} \in \mathbb{R}.$$
(3.39)

As we return to Lemma 3.5, e.g., see (3.34), it is natural to wonder whether the marginalization of the two-dimensional result can be reconciled analytically with the one-dimensional convergence obtained earlier. In other words, is it indeed the case that the relations

$$G_s(x_s) = \frac{1}{s!} H_s(e^{-x_s}), \quad x_s \in \mathbb{R}$$
(3.40)

and

$$G_{s-1}(x_{s-1}) = \frac{1}{s!} \left( H_s(e^{-x_{s-1}}) - e^{-sx_{s-1}} \cdot e^{-e^{-x_{s-1}}} \right), \quad x_{s-1} \in \mathbb{R}$$
(3.41)

hold.

We argue as follows: For t = 1, 2, ..., integration by parts yields

$$H_t(a) = a^t e^{-a} + t H_{t-1}(a), \quad a \ge 0$$
(3.42)

with boundary value  $H_0(a) = e^{-a}$ . Iterating on this relation, we readily check that

$$H_t(a) = t! \left(\sum_{r=0}^t \frac{a^r}{r!}\right) e^{-a}, \qquad a \ge 0$$

$$t = 0, 1, \dots$$
(3.43)

Using (3.43) it is plain that

$$\frac{1}{s!}H_s(e^{-x_s}) = G_s(x_s), \quad x_s \in \mathbb{R}$$

as we recall (3.30), and this shows that (3.40) indeed holds. Next, as we turn to (3.41), use (3.42) (with t = s and  $a = e^{-x_{s-1}}$ ) and note that

$$\frac{1}{s!} \left( H_s(e^{-x_{s-1}}) - e^{-sx_{s-1}} \cdot e^{-e^{-x_{s-1}}} \right)$$

$$= \frac{1}{s!} \left( e^{-sx_{s-1}} \cdot e^{-e^{-x_{s-1}}} + sH_{s-1}(e^{-x_{s-1}}) - e^{-sx_{s-1}} \cdot e^{-e^{-x_{s-1}}} \right)$$

$$= \frac{1}{(s-1)!} H_{s-1}(e^{-x_{s-1}}) = G_{s-1}(x_{s-1}), \quad x_{s-1} \in \mathbb{R}$$
(3.44)
where we again relied on (3.43). This shows that (3.41) holds.

Therefore, as we combine Lemma 3.5 and Proposition 3.6, the following conclusion emerges.

**Corollary 3.7.** For any given s = 1, 2, ..., there exist a pair  $(\Lambda_s, \Lambda_{s-1})$  of  $\mathbb{R}$ -valued rvs such that

$$\left(\lambda\left(\eta_{p,p-s}-\alpha_{p}^{\star}\right),\lambda\left(\eta_{p,p-(s-1)}-\alpha_{p}^{\star}\right)\right)\Longrightarrow_{p}\left(\Lambda_{s},\Lambda_{s-1}\right)$$
(3.45)

with  $(\Lambda_s, \Lambda_{s-1})$  jointly distributed according to  $J_s$ , and  $\Lambda_s$  and  $\Lambda_{s-1}$ , each distributed according to  $G_s$  and  $G_{s-1}$ , respectively.

In other words,

$$\mathbb{P}\left[\Lambda_s \leqslant x_s, \Lambda_{s-1} \leqslant x_{s-1}\right] = J_s(x_s, x_{s-1}), \quad x_s, x_{s-1} \in \mathbb{R}$$

with marginals

$$\mathbb{P}\left[\Lambda_s \leqslant x_s\right] = G_s(x_s) \quad \text{and} \quad \mathbb{P}\left[\Lambda_{s-1} \leqslant x_{s-1}\right] = G_{s-1}(x_{s-1}), \quad x_s, x_{s-1} \in \mathbb{R}.$$

# 3.4 A proof of Proposition 3.2 - Part II

The notation is that of Section 3.2.2 as we return to the expressions (3.20), (3.21), (3.23) and (3.24) obtained there. With d = 0, 1, ... held fixed, for each n = 2, 3, ... we substitute  $\alpha$  by  $\alpha_n^*$  in these expressions according to (3.3), and let n go to infinity in the resulting inequality.

# 3.4.1 The case d = 0

For each n = 3, 4, ..., with the aforementioned substitution, we rewrite (3.20) and (3.21) as

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha_{n}^{\star})=0\right]=\mathbb{P}\left[\lambda(\eta_{n-2,n-2}-\alpha_{n}^{\star})\leqslant-\lambda\xi_{1}\right]$$

and

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha_{n}^{\star})=0, D_{n,2}^{\star}(\alpha_{n}^{\star})=0\right] = \mathbb{P}\left[\max(\xi_{1},\xi_{2})+\eta_{n-2,n-2}\leqslant\alpha_{n}^{\star}\right]$$
$$=\mathbb{P}\left[\lambda(\eta_{n-2,n-2}-\alpha_{n}^{\star})\leqslant-\lambda\max(\xi_{1},\xi_{2})\right].$$

By Lemma 3.5 (and comments following it), it is now plain that

$$p_{\mathrm{Fuj}}(0) = \lim_{n \to \infty} \mathbb{P}\left[D_{n,1}^{\star}(\alpha_n^{\star}) = 0\right] = \mathbb{P}\left[\Lambda_0 \leqslant -\lambda\xi_1\right] = \mathbb{E}\left[e^{-e^{\lambda\xi_1}}\right]$$

and

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}^{\star}(\alpha_n^{\star}) = 0, D_{n,2}^{\star}(\alpha_n^{\star}) = 0\right] = \mathbb{P}\left[\Lambda_0 \leqslant -\lambda \max(\xi_1, \xi_2)\right] = \mathbb{E}\left[e^{-e^{\lambda \max(\xi_1, \xi_2)}}\right],$$

respectively, where  $\Lambda_0$  is any rv which is distributed according to the Gumbel distribution (3.31). Collecting these facts and using the reduction step discussed in Section 3.2, we find

$$C(0) = \lim_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1}(\alpha_n^{\star}) = 0 \right], \mathbf{1} \left[ D_{n,2}(\alpha_n^{\star}) = 0 \right] \right]$$
$$= \lim_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1}^{\star}(\alpha_n^{\star}) = 0 \right], \mathbf{1} \left[ D_{n,2}^{\star}(\alpha_n^{\star}) = 0 \right] \right]$$
$$= \mathbb{E} \left[ e^{-e^{\lambda \max(\xi_1, \xi_2)}} \right] - \mathbb{E} \left[ e^{-e^{\lambda \xi_1}} \right] \mathbb{E} \left[ e^{-e^{\lambda \xi_2}} \right]$$
$$= \mathbb{E} \left[ e^{-e^{\lambda \max(\xi_1, \xi_2)}} \right] - \mathbb{E} \left[ e^{-e^{\lambda \xi_1}} \cdot e^{-e^{\lambda \xi_2}} \right].$$

Note that

$$C(0) = \mathbb{E}\left[e^{-\max(e^{\lambda\xi_1}, e^{\lambda\xi_2})}\right] - \mathbb{E}\left[e^{-(e^{\lambda\xi_1} + e^{\lambda\xi_2})}\right] > 0$$

since  $\max(e^{\lambda\xi_1}, e^{\lambda\xi_2}) < e^{\lambda\xi_1} + e^{\lambda\xi_2}$  a.s.

3.4.2 The case d = 1, 2, ...

Pick n = 3, 4, ... such that d < n - 2. Under the aforementioned substitution, we can rewrite (3.23) and (3.24) as

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha_n^{\star}) = d\right] = \mathbb{E}\left[e^{-\lambda(\alpha_n^{\star} - \eta_{n-2,t_n(d)+1})^+} - e^{-\lambda(\alpha_n^{\star} - \eta_{n-2,t_n(d)})^+}\right]$$
(3.46)

and

$$\mathbb{P}\left[D_{n,1}^{\star}(\alpha_{n}^{\star}) = d, D_{n,2}^{\star}(\alpha_{n}^{\star}) = d\right] = \mathbb{E}\left[\left(e^{-\lambda(\alpha_{n}^{\star} - \eta_{n-2,t_{n}(d)+1})^{+}} - e^{-\lambda(\alpha_{n}^{\star} - \eta_{n-2,t_{n}(d)})^{+}}\right)^{2}\right]3.47\right)$$

Applying Corollary 3.7 (with p = n - 2 and s = d) we conclude that

$$\left(\lambda\left(\eta_{n-2,t_n(d)} - \alpha_{n-2}^{\star}\right), \lambda\left(\eta_{n-2,t_n(d)+1} - \alpha_{n-2}^{\star}\right)\right) \Longrightarrow_n (\Lambda_d, \Lambda_{d-1}), \tag{3.48}$$

whence

$$\left(\lambda\left(\eta_{n-2,t_n(d)} - \alpha_n^{\star}\right), \lambda\left(\eta_{n-2,t_n(d)+1} - \alpha_n^{\star}\right)\right) \Longrightarrow_n (\Lambda_d, \Lambda_{d-1}) \tag{3.49}$$

by standard facts concerning weak convergence since  $\lim_{n\to\infty} (\alpha_n^{\star} - \alpha_{n-2}^{\star}) = 0$ . It immediately follows by the Continuous Mapping Theorem for weak convergence that

$$\left(\lambda\left(\alpha_{n-2}^{\star}-\eta_{n-2,t_n(d)}\right)^+,\lambda\left(\alpha_{n-2}^{\star}-\eta_{n-2,t_n(d)+1}\right)^+\right)\Longrightarrow_n\left((-\Lambda_d)^+,(-\Lambda_{d-1})^+\right).$$
(3.50)

Applying the Continuous Mapping Theorem once more we find that

$$e^{-\lambda(\alpha_n^{\star}-\eta_{n-2,t_n(d)+1})^+} - e^{-\lambda(\alpha_n^{\star}-\eta_{n-2,t_n(d)})^+} \Longrightarrow_n e^{-(-\Lambda_{d-1})^+} - e^{-(-\Lambda_d)^+}.$$
 (3.51)

Let n go to infinity in (3.46) and (3.47): Making use of the convergence (3.51) we conclude by the Bounded Convergence Theorem that

$$\lim_{n \to \infty} \mathbb{E} \left[ \left( e^{-\lambda (\alpha_n^{\star} - \eta_{n-2,t_n(d)+1})^+} - e^{-\lambda (\alpha_n^{\star} - \eta_{n-2,t_n(d)})^+} \right)^a \right] \\ = \mathbb{E} \left[ \left( e^{-(-\Lambda_{d-1})^+} - e^{-(-\Lambda_d)^+} \right)^a \right], \quad a = 1, 2.$$
(3.52)

This is made possible with the help of the obvious bounds

$$\left| e^{-\lambda(\alpha_n^* - \eta_{n-2,t_n(d)+1})^+} - e^{-\lambda(\alpha_n^* - \eta_{n-2,t_n(d)})^+} \right| \le 1, \quad n = 3, 4, \dots$$

From (3.25) we conclude that

$$\lim_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1}^{\star}(\alpha_n^{\star}) = d \right], \mathbf{1} \left[ D_{n,2}^{\star}(\alpha_n^{\star}) = d \right] \right]$$
$$= \lim_{n \to \infty} \operatorname{Var} \left[ e^{-\lambda(\alpha_n^{\star} - \eta_{n-2,t_n(d)+1})^+} - e^{-\lambda(\alpha_n^{\star} - \eta_{n-2,t_n(d)})^+} \right]$$
$$= \operatorname{Var} \left[ e^{-(-\Lambda_{d-1})^+} - e^{-(-\Lambda_d)^+} \right], \qquad (3.53)$$

and the reduction step via (3.16) finally leads to

$$C(d) = \lim_{n \to \infty} \operatorname{Cov} \left[ \mathbf{1} \left[ D_{n,1}(\alpha_n^{\star}) = d \right], \mathbf{1} \left[ D_{n,2}(\alpha_n^{\star}) = d \right] \right]$$
  
=  $\operatorname{Var} \left[ e^{-(-\Lambda_{d-1})^+} - e^{-(-\Lambda_d)^+} \right].$  (3.54)

Note that C(d) > 0 as the variance of the non-degenerate rv  $e^{-(-\Lambda_{d-1})^+} - e^{-(-\Lambda_d)^+}$ .

Now, a sanity check: Let n go to infinity in (3.46). The arguments given earlier also yield

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}^{\star}(\alpha_n^{\star}) = d\right] = \mathbb{E}\left[e^{-(-\Lambda_{d-1})^+} - e^{-(-\Lambda_d)^+}\right].$$

Standard calculations give

$$\mathbb{E}\left[e^{-(-\Lambda_d)^+}\right] = \int_0^\infty \mathbb{P}\left[e^{-(-\Lambda_d)^+} > t\right] dt$$
  

$$= \int_0^\infty \mathbb{P}\left[(-\Lambda_d)^+ < -\log t\right] dt$$
  

$$= \int_0^1 \mathbb{P}\left[(-\Lambda_d)^+ < -\log t\right] dt$$
  

$$= \int_0^1 \mathbb{P}\left[\Lambda_d > 0, 0 < -\log t\right] dt + \int_0^1 \mathbb{P}\left[\Lambda_d \le 0, -\Lambda_d < -\log t\right] dt.$$
  

$$= \int_0^1 \mathbb{P}\left[\log t < \Lambda_d\right] dt,$$
(3.55)

and

$$\mathbb{E}\left[e^{-(-\Lambda_{d-1})^{+}}\right] = \int_{0}^{1} \mathbb{P}\left[\log t < \Lambda_{d-1}\right] dt \qquad (3.56)$$

by similar arguments. Therefore,

$$\mathbb{E}\left[e^{-(-\Lambda_{d-1})^{+}}\right] - \mathbb{E}\left[e^{-(-\Lambda_{d})^{+}}\right] = \int_{0}^{1} \left(\mathbb{P}\left[\log t < \Lambda_{d-1}\right] - \mathbb{P}\left[\log t < \Lambda_{d}\right]\right) dt$$
$$= \int_{0}^{1} \left(\mathbb{P}\left[\Lambda_{d} \le \log t\right] - \mathbb{P}\left[\Lambda_{d-1} \le \log t\right]\right) dt$$
$$= \int_{0}^{1} \left(G_{d}(\log t) - G_{d-1}(\log t)\right) dt$$
$$= \int_{0}^{\infty} \left(G_{d}(-x) - G_{d-1}(-x)\right) e^{-x} dx \quad [t = e^{-x}]$$
$$= \int_{0}^{\infty} \frac{(e^{x})^{d}}{d!} e^{-e^{x}} e^{-x} dx$$
$$= p_{\mathrm{Fuj}}(d) \tag{3.57}$$

as it should be in view of (3.15) when combined with (3.5).

As a by product it follows that the pmf  $\boldsymbol{p}_{\mathrm{Fuj}}$  admits the multiple representations

$$p_{\rm Fuj}(d) = \mathbb{E}\left[e^{-(-\Lambda_{d-1})^+} - e^{-(-\Lambda_d)^+}\right] = \mathbb{E}\left[\frac{(e^{\lambda\xi})^d}{d!}e^{-e^{\lambda\xi}}\right], \quad d = 0, 1, \dots$$

# 3.5 A proof of Proposition 3.6

# 3.5.1 Preliminaries

Fix p = 2, 3, ... and t = 1, ..., p - 1. It is also well known [11, p. 11] that the joint probability distribution function of the pair  $(\eta_{p,t}, \eta_{p,t+1})$  admits a probability density function  $f_{p,t,t+1} : \mathbb{R}_+ \to \mathbb{R}_+$  given by

$$f_{p,t,t+1}(x_t, x_{t+1}) = \begin{cases} p(p-1) \binom{p-2}{t-1} \left(1 - e^{-\lambda x_t}\right)^{t-1} e^{-\lambda (p-(t+1))x_{t+1}} \lambda e^{-\lambda x_t} \lambda e^{-\lambda x_{t+1}} & \text{if } 0 \leq x_t \leq x_{t+1} \\ \\ 0 & \text{otherwise.} \end{cases}$$

Therefore, since  $\eta_{p,t} \leq \eta_{p,t+1}$ , with arbitrary  $x_t$  and  $x_{t+1}$  in  $\mathbb{R}_+$ , elementary calculations yield

$$\mathbb{P}\left[\eta_{p,t} \leqslant x_{t}, \eta_{p,t+1} \leqslant x_{t+1}\right] \\
= \mathbb{P}\left[\eta_{p,t} \leqslant \min(x_{t}, x_{t+1}), \eta_{p,t+1} \leqslant x_{t+1}\right] \\
= \int_{0}^{\min(x_{t}, x_{t+1})} \left(\int_{y_{t}}^{x_{t+1}} f_{p,t,t+1}(y_{t}, y_{t+1}) dy_{t+1}\right) dy_{t} \\
= p(p-1) \binom{p-2}{t-1} \int_{0}^{\min(x_{t}, x_{t+1})} \left(\int_{y_{t}}^{x_{t+1}} \lambda e^{-\lambda(p-t)y_{t+1}} dy_{t+1}\right) \left(1 - e^{-\lambda y_{t}}\right)^{t-1} \lambda e^{-\lambda y_{t}} dy_{t} \\
= \frac{p(p-1)}{p-t} \binom{p-2}{t-1} \int_{0}^{\min(x_{t}, x_{t+1})} \left(1 - e^{-\lambda y_{t}}\right)^{t-1} \lambda e^{-\lambda y_{t}} \left(e^{-\lambda(p-t)y_{t}} - e^{-\lambda(p-t)x_{t+1}}\right) dy_{t} \\
= t \binom{p}{t} \int_{0}^{\lambda \min(x_{t}, x_{t+1})} \left(1 - e^{-y_{t}}\right)^{t-1} e^{-y_{t}} \left(e^{-(p-t)y_{t}} - e^{-\lambda(p-t)x_{t+1}}\right) dy_{t} \\
= t \binom{p}{t} \left(I_{t}(\lambda \min(x_{t}, x_{t+1}); p-t) - e^{-\lambda(p-t)x_{t+1}} \cdot I_{t}(\lambda \min(x_{t}, x_{t+1}); 0)\right) \quad (3.58)$$

as we use the notation

$$a \ge 0$$
$$I_t(a;r) = \int_0^a \left(1 - e^{-y}\right)^{t-1} e^{-(r+1)y} dy, \quad r = 0, 1, \dots$$
$$t = 1, 2, \dots$$

We begin with an intermediary result that will help us in investigating the asymptotics of the relevant pair of order statistics.

**Lemma 3.8.** Fix s = 0, 1, ... and x arbitrary in  $\mathbb{R}$ . For each r = 0, 1, ..., it holds that

$$\lim_{p \to \infty} p^{r+1} I_{p-s} \left( (x + \log p)^+; r \right) = \int_{e^{-x}}^{\infty} v^r e^{-v} dv.$$
(3.59)

**Proof.** Throughout the proof the integer r = 0, 1, ... is held fixed. For  $a \ge 0$  and positive integer t, elementary calculus gives

$$I_{t}(a;r) = \int_{0}^{a} \left(1 - e^{-y}\right)^{t-1} e^{-(r+1)y} dy$$
  
= 
$$\int_{0}^{a} \left(1 - \frac{e^{z}}{e^{a}}\right)^{t-1} \left(\frac{e^{z}}{e^{a}}\right)^{r+1} dz \quad [z = a - y]$$
  
= 
$$e^{-a} \int_{1}^{e^{a}} \left(1 - \frac{u}{e^{a}}\right)^{t-1} \left(\frac{u}{e^{a}}\right)^{r} du \quad [u = e^{z}].$$
(3.60)

Now pick s = 0, 1, ... and x arbitrary in  $\mathbb{R}$ , and take p sufficiently large so that p > s and  $x + \log p > 0$ . Then, with t = p - s and  $a = x + \log p$ , the last relation (3.60) becomes

$$I_{p-s}(x + \log p; r) = \frac{e^{-x}}{p} \int_{1}^{pe^{x}} \left(1 - \frac{u}{pe^{x}}\right)^{p-s-1} \left(\frac{u}{pe^{x}}\right)^{r} du$$
$$= \frac{1}{p^{r+1}} \int_{e^{-x}}^{p} \left(1 - \frac{v}{p}\right)^{p-s-1} v^{r} dv \quad [v = e^{-x}u].$$
(3.61)

On the range  $0 \leq v \leq p$ , the bounds

$$0 \leqslant \left(1 - \frac{v}{p}\right)^{p-s-1} \leqslant e^{-\frac{p-s-1}{p}v} \leqslant 1$$

hold uniformly in p, while

$$\lim_{p \to \infty} \left( 1 - \frac{v}{p} \right)^{p-s-1} = e^{-v}, \quad v \ge 0$$

as we again make use of (3.27)-(3.28). Using the Bounded Convergence Theorem, we readily conclude by standard arguments that

$$\lim_{p \to \infty} \int_{e^{-x}}^{p} \left( 1 - \frac{v}{p} \right)^{p-s-1} v^{r} dv = \int_{e^{-x}}^{\infty} v^{r} e^{-v} dv$$
(3.62)

since the  $r^{th}$  moment of Exp(1) is finite and given by  $\int_0^\infty v^r e^{-v} dv = r!$ . This completes the proof of (3.59).

## 3.5.2 Proving Proposition 3.6

Fix positive integers p and t such that t < p, and pick  $x_t$  and  $x_{t+1}$  arbitrary in  $\mathbb{R}$ . Take p large enough so that such  $\min(x_t, x_{t+1}) + \log p > 0$ . Under such conditions, using (3.58) we find that

$$\mathbb{P}\left[\lambda(\eta_{p,t} - \alpha_{p}^{\star}) \leq x_{t}, \lambda(\eta_{p,t+1} - \alpha_{p}^{\star}) \leq x_{t+1}\right] \\
= \mathbb{P}\left[\eta_{p,t} \leq \frac{x_{t} + \log p}{\lambda}, \eta_{p,t+1} \leq \frac{x_{t+1} + \log p}{\lambda}\right] \\
= t\binom{p}{t} \left(I_{t}(\min(x_{t}, x_{t+1}) + \log p; p - t) - e^{-(p-t)(x_{t+1} + \log p)} \cdot I_{t}(\min(x_{t}, x_{t+1}) + \log p; 0)\right) \\
= t\binom{p}{t} \left(I_{t}(\min(x_{t}, x_{t+1}) + \log p; p - t) - \frac{e^{-(p-t)x_{t+1}}}{p^{p-t}} \cdot I_{t}(\min(x_{t}, x_{t+1}) + \log p; 0)\right).$$

Now pick s = 1, 2, ..., and  $x_s$  and  $x_{s-1}$  arbitrary in  $\mathbb{R}$ . With positive integer p sufficiently large so that p > s and  $\min(x_s, x_{s-1}) + \log p > 0$ , we use the last relation

with t = p - s,  $x_t = x_s$  and  $x_{t+1} = x_{s-1}$ . This gives

$$\mathbb{P}\left[\lambda(\eta_{p,p-s} - \alpha_p^{\star}) \leqslant x_s, \lambda(\eta_{p,p-s+1} - \alpha_p^{\star}) \leqslant x_{s-1}\right] \\ = (p-s)\binom{p}{s} \left(I_{p-s}(\min(x_s, x_{s-1}) + \log p; s) - \frac{e^{-sx_{s-1}}}{p^s} \cdot I_{p-s}(\min(x_s, x_{s-1}) + \log p; 0)\right).$$

Let p go to infinity in this last expression: Lemma 3.8 yields

$$\lim_{p \to \infty} p^{s+1} I_{p-s} \left( (\min(x_s, x_{s-1}) + \log p)^+; s \right) = \int_{e^{-\min(x_s, x_{s-1})}}^{\infty} v^s e^{-v} dv$$
(3.63)

and

$$\lim_{p \to \infty} p I_{p-s} \left( (\min(x_s, x_{s-1}) + \log p)^+; 0 \right) = \int_{e^{-\min(x_s, x_{s-1})}}^{\infty} e^{-v} dv = e^{-e^{-\min(x_s, x_{s-1})}}.$$
(3.64)

Next, noting

$$(p-s)\binom{p}{s} \sim \frac{p^{s+1}}{s!} \quad (p \to \infty)$$

by virtue of (3.26), we easily conclude from (3.63) that

$$\lim_{p \to \infty} (p-s) {p \choose s} I_{p-s}(\min(x_s, x_{s-1}) + \log p; s)$$
  
= 
$$\lim_{p \to \infty} \frac{(p-s) {p \choose s}}{p^{s+1}} \cdot p^{s+1} I_{p-s}(\min(x_s, x_{s-1}) + \log p; s)$$
  
= 
$$\frac{1}{s!} \int_{e^{-\min(x_s, x_{s-1})}}^{\infty} v^s e^{-v} dv,$$
 (3.65)

while (3.64) gives

$$\lim_{p \to \infty} (p-s) {p \choose s} \frac{e^{-sx_{s-1}}}{p^s} \cdot I_{p-s} (\min(x_s, x_{s-1}) + \log p; 0)$$
  
= 
$$\lim_{p \to \infty} \frac{(p-s) {p \choose s}}{p^{s+1}} e^{-sx_{s-1}} \cdot pI_{p-s} (\min(x_s, x_{s-1}) + \log p; 0)$$
  
= 
$$\frac{1}{s!} e^{-sx_{s-1}} \cdot e^{-e^{-\min(x_s, x_{s-1})}}.$$
 (3.66)

Collecting these last two convergence statements we readily obtain the convergence (3.37) with (3.36).

## Chapter 4: More on Random Threshold Graphs: Weak Convergence

In Chapter 3, we considered a specific counterexample in the class of homogeneous random graphs: Let  $\{\xi, \xi_k, k = 1, 2, ...\}$  represent an i.i.d. collection of  $\mathbb{R}_+$ -valued rvs with  $\xi_k$  being a fitness rv associated with node k denoting its importance or rank. With n nodes and threshold  $\alpha$ , the random threshold graph  $\mathbb{T}(n, \alpha)$ postulates that two distinct nodes i and j form a connection iff  $\xi_i + \xi_j > \alpha$ . With  $\xi$ exponentially distributed and the threshold  $\alpha$  scaled as

$$\alpha_n^* = \frac{1}{\lambda} \log n, \ n = 1, 2, \dots$$
(4.1)

the following distributional convergence in the sequence of graphs  $\{\mathbb{T}(n; \alpha_n^*), n = 1, 2, ...\}$ [22] is known to take place

$$D_{n,1}(\alpha_n^*) \Longrightarrow_n D, \tag{4.2}$$

where the rv D has a power-tail. However, in Chapter 3 we concluded that when  $\xi$  is exponentially distributed, the usual convergence (2.9) of the empirical degree distribution fails to take place. Following the discussions in the previous chapter 3, two questions naturally emerge:

1. While in the previous chapter we considered the special case of  $\xi$  exponentially distributed, there are other regimes under which interesting asymptotic behaviour of the degree distribution occurs in the sense of (4.2). An obvious question is whether the assertions made in the previous chapter can be extended to these more general fitness distributions.

2. Under the regime where the empirical degree distribution does not converge in the usual sense (which we now know includes the case of  $\xi$  exponentially distributed), are there weaker forms of convergence that could be shown to hold?

In what follows, we show that under certain conditions the empirical degree distribution actually converges weakly: For each d = 0, 1, ..., there exists a nondegenerate [0, 1]-valued rv  $\Pi(d)$  such that

$$\frac{N_n(d;\alpha_n^{\star})}{n} \Longrightarrow_n \Pi(d) \tag{4.3}$$

where the scaling  $\alpha^{\star} : \mathbb{N}_0 \to \mathbb{R}_+$  is the one yielding a non-trivial degree distribution in the sense of (4.2), and  $\frac{N_n(d;\alpha_n^{\star})}{n}$  is the fraction of nodes with degree d in  $\mathbb{T}(n;\alpha_n^{\star})$ . The non-degeneracy of the rv  $\Pi(d)$  for each  $d = 0, 1, \ldots$ , ascertains that the empirical degree distribution cannot converge in the sense of (2.9).

#### 4.1 Degree distribution – Generic fitness distribution

The setting is that of Section 3.1: Let  $\{\xi, \xi_k, k = 1, 2, ...\}$  denote a collection of i.i.d.  $\mathbb{R}_+$ -valued rvs defined on the probability triple  $(\Omega, \mathcal{F}, \mathbb{P})$ , each distributed according to a given (probability) distribution function  $F : \mathbb{R} \to [0, 1]$ . With  $\xi$  acting as a generic representative for this sequence of i.i.d. rvs, we have

$$\mathbb{P}\left[\xi \leqslant x\right] = F(x), \quad x \in \mathbb{R}.$$

As mentioned in the previous chapter, we assume that F is a continuous function on  $\mathbb{R}$  with

$$F(x) = 0, \quad x \leqslant 0. \tag{4.4}$$

Recall that with *n* number of nodes and threshold  $\alpha > 0$ , distinct nodes i, j = 1, 2, ..., n are said to be adjacent if  $\xi_i + \xi_j > \alpha$ . For each k = 1, 2, ..., n, and threshold  $\alpha > 0$  the degree of node k in  $\mathbb{T}(n; \alpha)$  is the rv  $D_{n,k}(\alpha)$  given by

$$D_{n,k}(\alpha) := \sum_{\ell=1, \ell \neq k}^{n} \mathbf{1} \left[ \xi_k + \xi_\ell > \alpha \right].$$

Under the enforced assumptions on the rvs  $\xi_1, \ldots, \xi_n$ , the rv  $D_{n,k}(\alpha)$  is a Binomial rv Bin $(n-1; 1 - F(\alpha - \xi_k))$  conditioned on  $\xi_k$ . Throughout we make the following assumption.

Assumption 4.1. There exists a scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  with the property

$$\lim_{n \to \infty} \alpha_n^\star = \infty, \tag{4.5}$$

such that

$$\lim_{n \to \infty} n \left( 1 - F(\alpha_n^{\star} - x) \right) = \lambda(x), \quad x \ge 0$$
(4.6)

for some non-identically zero mapping  $\lambda : \mathbb{R}_+ \to \mathbb{R}_+$ .

The mapping  $\lambda : \mathbb{R}_+ \to \mathbb{R}_+$  is necessarily non-decreasing. The following result overlaps with a similar result by Fujihara et al. [22, Thm. 2, p. 362]:

**Proposition 4.1.** Under Assumption 4.1, there exists an  $\mathbb{N}$ -valued rv D such that

$$D_n(\alpha_n^{\star}) \Longrightarrow_n D.$$
 (4.7)

The rv D is conditionally Poisson with pmf given by

$$\mathbb{P}\left[D=d\right] = \mathbb{E}\left[\frac{\lambda(\xi)^d}{d!}e^{-\lambda(\xi)}\right], \quad d=0,1,\dots$$
(4.8)

**Proof.** Fix  $n = 2, 3, ..., \alpha > 0$  and z in  $\mathbb{R}$ . Standard pre-conditioning arguments yield

$$\mathbb{E}\left[z^{D_n(\alpha)}\right] = \mathbb{E}\left[\prod_{\ell=2}^n z^{\mathbf{1}[\xi_1 + \xi_\ell > \alpha]}\right] = \mathbb{E}\left[\left(\mathbb{E}\left[z^{\mathbf{1}[x + \xi > \alpha]}\right]_{x = \xi_1}\right)^{n-1}\right]$$

where

$$\mathbb{E}\left[z^{\mathbf{1}[x+\xi>\alpha]}\right] = \mathbb{P}\left[x+\xi\leqslant\alpha\right] + \mathbb{P}\left[x+\xi>\alpha\right]z$$
$$= 1-(1-z)\mathbb{P}\left[x+\xi>\alpha\right]$$
$$= 1-(1-z)\left(1-F(\alpha-x)\right), \quad x\in\mathbb{R}.$$
(4.9)

Consequently, as we use (4.4), we get

$$\mathbb{E}\left[z^{D_{n}(\alpha)}\right] = \mathbb{E}\left[\left(1 - (1 - z)\left(1 - F(\alpha - \xi)\right)\right)^{n-1}\right]$$
$$= \mathbb{P}\left[\xi > \alpha\right] z^{n-1} + \mathbb{E}\left[\mathbf{1}\left[\xi \le \alpha\right]\left(1 - (1 - z)\left(1 - F(\alpha - \xi)\right)\right)^{n-1}\right].10\right]$$

For each n = 1, 2, ..., replace  $\alpha$  by  $\alpha_n^{\star}$  in (4.10) according to the scaling  $\alpha^{\star} : \mathbb{N}_0 \to \mathbb{R}_+$ . Let n go to infinity in the resulting equality when  $|z| \leq 1$ : It is plain that  $\lim_{n\to\infty} \mathbb{P}\left[\xi > \alpha_n^{\star}\right] z^{n-1} = 0$ , while

$$\lim_{n \to \infty} \left( 1 - (1 - z) \left( 1 - F(\alpha_n^{\star} - \xi) \right) \right)^{n-1} = e^{-(1 - z)\lambda(\xi)}$$

by standard arguments as we note that

$$\lim_{n \to \infty} n(1-z) \left(1 - F(\alpha_n^{\star} - \xi)\right) = (1-z)\lambda(\xi)$$

under Assumption 4.1. Invoking the Bounded Convergence Theorem we conclude that

$$\lim_{n \to \infty} \mathbb{E}\left[z^{D_n(\alpha_n^{\star})}\right] = \mathbb{E}\left[e^{-(1-z)\lambda(\xi)}\right], \quad |z| \leq 1$$

and the desired conclusion follows upon noting that the right-hand side is the pgf of the pmf (4.8).

In what follows we use the standard notation  $x^+ = \max(x, 0)$  for x in  $\mathbb{R}$ . Assumption 4.1 holds in a number of interesting cases: Consider the case discussed in the previous chapter where  $\xi$  is exponentially distributed with parameter  $\lambda > 0$ , i.e.,

$$\mathbb{P}\left[\xi > x\right] = e^{-\lambda x^+}, \quad x \in \mathbb{R}.$$

Assumption 4.1 holds with

$$\lambda(x) = e^{\lambda x}, \quad x \ge 0$$

upon taking

$$\alpha_n^{\star} = \lambda^{-1} \log n, \quad n = 1, 2, \dots$$

The rv  $\xi$  is said to be a Pareto rv with parameters  $\nu > 0$  and a > 0, if

$$\mathbb{P}\left[\xi > x\right] = \left(\frac{a}{a+x^+}\right)^{\nu}, \quad x \in \mathbb{R}.$$

Assumption 4.1 holds if we take

$$\alpha_n^\star = a n^{\frac{1}{\nu}}, \quad n = 1, 2, \dots$$

in which case

$$\lambda(x) = 1, \quad x \ge 0.$$

### 4.2 The main result

Fix n = 2, 3, ... and  $\alpha > 0$ . For each d = 0, 1, ..., the rv  $N_n(d; \alpha)$  defined by

$$N_n(d;\alpha) = \sum_{k=1}^n \mathbf{1} \left[ D_{n,k}(\alpha) = d \right]$$

counts the number of nodes in  $\{1, \ldots, n\}$  which have degree d in  $\mathbb{T}(n; \alpha)$ . The fraction of nodes in  $\{1, \ldots, n\}$  with degree d in  $\mathbb{T}(n; \alpha)$  is then given by

$$p_n(d;\alpha) = \frac{N_n(d;\alpha)}{n}.$$

We now state the main result of the chapter.

**Theorem 4.2.** Assume Assumption 4.1 to hold. Then, for each d = 0, 1, ..., there exists a non-degenerate [0, 1]-valued rv  $\Pi(d)$  such that

$$\frac{N_n(d;\alpha_n^{\star})}{n} \Longrightarrow_n \Pi(d) \tag{4.11}$$

where the scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  is the one postulated in Assumption 4.1

In the process of proving Theorem 4.2 we will show that for each d = 0, 1, ...,

$$\mathbb{E}\left[\Pi(d)\right] = \mathbb{P}\left[D = d\right] \quad \text{with} \quad \operatorname{Var}\left[\Pi(d)\right] > 0. \tag{4.12}$$

In other words, the rv  $\Pi(d)$  is never a degenerate rv with the following consequence.

**Corollary 4.3.** Assume Assumption 4.1 to hold. Then, for each d = 0, 1, ..., the sequence

$$\left\{\frac{N_n(d;\alpha_n^\star)}{n}, \ n=1,2,\ldots\right\}$$

cannot converge in probability, i.e., there exists no constant L(d) such that

$$\frac{N_n(d;\alpha_n^{\star})}{n} \xrightarrow{P} {}_n L(d).$$
(4.13)

Proposition 3.3 is a special case of the corollary stated above.

#### 4.3 A roadmap to a proof of Theorem 4.2

The remainder of the chapter is concerned with a proof of Theorem 4.2. The technical approach is rooted in the method of moments, and articulated through several intermediary results, the first one being a multi-dimensional version of Proposition 4.1.

**Proposition 4.4.** Assume Assumption 4.1 to hold. For each r = 1, 2, ..., there exists an  $\mathbb{N}^r$ -valued rv  $(D_1, \ldots, D_r)$  such that

$$(D_{n,1}(\alpha_n^{\star}),\ldots,D_{n,r}(\alpha_n^{\star})) \Longrightarrow_n (D_1,\ldots,D_r).$$

$$(4.14)$$

The limiting rvs  $D_1, \ldots, D_r$  are exchangeable, but not independent, each being distributed according to the limiting rv D whose existence is established in Proposition 4.1.

Proposition 4.4 is a simple consequence of Proposition 4.5 discussed next, but first some notation: For each r = 1, 2, ...,let  $\xi_{r,1}, ..., \xi_{r,r}$  denote the values of the fitness rvs  $\xi_1, ..., \xi_r$  arranged in increasing order, namely  $\xi_{r,1} \leq ... \leq \xi_{r,r}$ , with a lexicographic tiebreaker when needed. Thus, the rvs  $\xi_{r,1}, ..., \xi_{r,r}$  are the *order statistics* associated with the collection  $\xi_1, ..., \xi_r$ ; the rvs  $\xi_{r,1}$  and  $\xi_{r,r}$  are simply the minimum and maximum of the rvs  $\xi_1, ..., \xi_r$ , respectively [11]. In what follows, the random permutation  $\beta : \{1, \ldots, r\} \rightarrow \{1, \ldots, r\}$  arranges the rvs  $\xi_1, \ldots, \xi_r$  in increasing order, i.e.,

$$\xi_{r,s} = \xi_{\beta(s)}, \quad s = 1, \dots, r$$

(under the lexicographic tiebreaker) – Note that  $\beta$  is determined by the rvs  $\xi_1, \ldots, \xi_r$ and is uniformly distributed over the group of permutations of  $\{1, \ldots, r\}$ . Finally, with the notation introduced so far, write

$$G_{r}(z_{1},...,z_{r}) = \mathbb{E}\left[e^{-\sum_{t=1}^{r}(1-z_{\beta(t)})\left(\prod_{s=t+1}^{r}z_{\beta(s)}\right)\lambda(\xi_{r,t})}\right], \qquad \begin{array}{c} 0 \leq z_{s} \leq 1, \\ s = 1,...,r. \end{array}$$
(4.15)

**Proposition 4.5.** Assume Assumption 4.1 to hold. For each r = 1, 2, ..., we have

$$\lim_{n \to \infty} \mathbb{E}\left[\prod_{s=1}^{r} z_s^{D_{n,s}(\alpha_n^{\star})}\right] = G_r(z_1, \dots, z_r)$$
(4.16)

for all  $z_1, \ldots, z_r$  in  $\mathbb{R}$  satisfying

$$0 \leqslant z_s \leqslant 1, \quad s = 1, \dots, r. \tag{4.17}$$

This result is established in several steps which are presented across Section 4.5, Section 4.6 and Section 4.7. However, Proposition 4.5 does imply Proposition 4.4 by the usual arguments: Indeed, by the Bounded Convergence Theorem we get

$$\lim_{z_s \uparrow 1, s=1,\dots,r} G_r(z_1,\dots,z_r) = 1$$

where the convergence is taken from inside  $[0,1)^r$ , and the mapping  $G_r : [0,1]^r \to \mathbb{R}$ is therefore continuous at the point  $(1,\ldots,1)$ . This fact, coupled with the convergence (4.16), suffices to reach the conclusion that  $G_r$  is an *r*-dimensional pgf. Thus, there exists an  $\mathbb{N}^r$ -valued rv, denoted  $(D_1, \ldots, D_r)$ , such that

$$\mathbb{E}\left[\prod_{s=1}^{r} z_s^{D_s}\right] = G_r(z_1, \dots, z_r), \qquad \begin{array}{c} 0 \leqslant z_s \leqslant 1, \\ s = 1, \dots, r. \end{array}$$
(4.18)

and the convergence (4.14) follows; see details in [21, p. 431]. This completes the proof of Proposition 4.4. It is also plain that the rvs  $D_1, \ldots, D_r$  are not independent; see below.

The next step establishes the requisite convergence of the moments; its proof is available in Section 4.8.

**Proposition 4.6.** Assume Assumption 4.1 to hold. For each r = 1, 2, ..., we have

$$\lim_{n \to \infty} \mathbb{E}\left[\left(\frac{N_n(d; \alpha_n^{\star})}{n}\right)^r\right] = \mathbb{P}\left[D_1 = d, \dots, D_r = d\right], \quad d = 0, 1, \dots$$
(4.19)

where the  $\mathbb{N}^r$ -valued rv  $(D_1, \ldots, D_r)$  is the limiting rv shown to exist in Proposition 4.4.

The following information is easily obtained by combining (4.15) and (4.18): As expected, we get back Proposition 4.1 by looking at the case r = 1. For r = 2, we read

$$\mathbb{E}\left[z_1^{D_1} z_2^{D_2}\right] = \mathbb{E}\left[e^{-(1-z_{\beta(1)})z_{\beta(2)}\lambda(\xi_{2,1}) - (1-z_{\beta(2)})\lambda(\xi_{2,2})}\right], \quad 0 \le z_1, z_2 \le 1.$$
(4.20)

By bounded convergence it follows from Proposition 4.6 that the first half of (4.12) holds with

$$Var[\Pi(d)] = \mathbb{P}[D_1 = d, D_2 = d] - \mathbb{P}[D_1 = d] \mathbb{P}[D_2 = d].$$

It is now straightforward to argue that  $Var[\Pi(d)] > 0$ . Rewriting (4.20), we obtain

$$\mathbb{E}\left[z_{1}^{D_{1}}z_{2}^{D_{2}}\right] = \mathbb{E}\left[e^{-(1-z_{\beta(1)})z_{\beta(2)}\lambda(\xi_{2,1})-(1-z_{\beta(2)})\lambda(\xi_{2,2})}\right]$$
$$= \mathbb{E}\left[e^{-(1-z_{\beta(1)})\lambda(\xi_{2,1})-(1-z_{\beta(2)})\lambda(\xi_{2,2})+(1-z_{\beta(1)})(1-z_{\beta(2)})\lambda(\xi_{2,1})}\right]$$
$$= \mathbb{E}\left[e^{-(1-z_{\beta(1)})\lambda(\xi_{\beta(1)})-(1-z_{\beta(2)})\lambda(\xi_{\beta(2)})+(1-z_{\beta(1)})(1-z_{\beta(2)})\lambda(\xi_{2,1})}\right]$$
$$= \mathbb{E}\left[e^{-(1-z_{1})\lambda(\xi_{1})-(1-z_{2})\lambda(\xi_{2})+(1-z_{1})(1-z_{2})\lambda(\xi_{2,1})}\right]$$
(4.21)

for  $0 \leq z_1, z_2 \leq 1$ . Collecting a single term corresponding to  $z_1^d z_2^d$  from (4.21), we obtain the lower bound

$$\mathbb{P}\left[D_1 = d, D_2 = d\right] \ge \mathbb{E}\left[\frac{\lambda(\xi_1)^d}{d!} \frac{\lambda(\xi_2)^d}{d!} e^{-\lambda(\xi_1) - \lambda(\xi_2)} e^{\lambda(\min(\xi_1, \xi_2))}\right]$$
(4.22)

for each  $d = 0, 1, \ldots$  Therefore for each  $d = 0, 1, \ldots$ , using (4.22) and (4.8), we obtain the desired lower bound

$$\operatorname{Var}[\Pi(d)] = \mathbb{P}\left[D_1 = d, D_2 = d\right] - \mathbb{P}\left[D_1 = d\right] \mathbb{P}\left[D_2 = d\right]$$
$$\geq \mathbb{E}\left[\frac{\lambda(\xi_1)^d}{d!} \frac{\lambda(\xi_2)^d}{d!} e^{-\lambda(\xi_1) - \lambda(\xi_2)} \left(e^{\lambda(\min(\xi_1, \xi_2))} - 1\right)\right]$$
$$> 0,$$

where the last step follows from Assumption 4.1.

## 4.4 A proof of Theorem 4.2

Equipped with Propositions 4.4 and 4.6 we can now provide a proof of Theorem 4.2: Fix  $d = 0, 1, \ldots$  Proposition 4.6 suggests that we consider the mapping  $\phi_d$ :  $\mathbb{R} \to \mathbb{C}$  given by

$$\phi_d(t) = 1 + \sum_{r=1}^{\infty} \frac{(it)^r}{r!} \mathbb{P}\left[D_1 = d, \dots, D_r = d\right], \quad t \in \mathbb{R}.$$

This definition is well posed with  $\phi_d(t)$  always an element of  $\mathbb C$  since

$$1 + \sum_{r=1}^{\infty} \frac{|t|^r}{r!} \mathbb{P}\left[D_1 = d, \dots, D_r = d\right] \le 1 + \sum_{r=1}^{\infty} \frac{|t|^r}{r!} = e^{|t|}, \quad t \in \mathbb{R}.$$

In particular, the mapping  $\phi_d : \mathbb{R} \to \mathbb{C}$  is analytic on  $\mathbb{R}$ , hence continuous at t = 0.

Now, for each  $n = 2, 3, ..., \text{let } \phi_{d,n} : \mathbb{R} \to \mathbb{C}$  denote the characteristic function of the rv  $\frac{N_n(d;\alpha_n^*)}{n}$ , i.e.,

$$\phi_{d,n}(t) = \mathbb{E}\left[e^{it\frac{N_n(d;\alpha_n^*)}{n}}\right], \quad t \in \mathbb{R}.$$

The obvious bounds

$$0 \leqslant \frac{N_n(d;\alpha_n^\star)}{n} \leqslant 1, \quad n = 2, 3, \dots$$
(4.23)

can be used to validate the series expansion

$$\phi_{d,n}(t) = 1 + \sum_{r=1}^{\infty} \frac{(it)^r}{r!} \mathbb{E}\left[\left(\frac{N_n(d;\alpha_n^{\star})}{n}\right)^r\right], \quad t \in \mathbb{R}$$

since here as well we have

$$1 + \sum_{r=1}^{\infty} \frac{|t|^r}{r!} \mathbb{E}\left[\left(\frac{N_n(d;\alpha_n^{\star})}{n}\right)^r\right] \le 1 + \sum_{r=1}^{\infty} \frac{|t|^r}{r!} = e^{|t|}, \quad t \in \mathbb{R}.$$

In view of these remarks, for each n = 2, 3, ... and t in  $\mathbb{R}$ , we can write

$$\phi_{d,n}(t) - \phi_d(t) = \sum_{r=1}^{\infty} \frac{(it)^r}{r!} \left( \mathbb{E}\left[ \left( \frac{N_n(d; \alpha_n^{\star})}{n} \right)^r \right] - \mathbb{P}\left[ D_1 = d, \dots, D_r = d \right] \right).$$

Picking a positive integer R, we get

$$\begin{aligned} |\phi_{d,n}(t) - \phi_d(t)| \\ &= \sum_{r=1}^R \frac{|t|^r}{r!} \left| \mathbb{E}\left[ \left( \frac{N_n(d; \alpha_n^\star)}{n} \right)^r \right] - \mathbb{P}\left[ D_1 = d, \dots, D_r = d \right] \right| + 2 \sum_{r=R+1}^\infty \frac{|t|^r}{r!}. \end{aligned}$$

With arbitrary  $\varepsilon > 0$  there exists a positive integer  $R^{\star}(\varepsilon, t)$  such that

$$\sum_{r=R+1}^\infty \frac{|t|^r}{r!} \leqslant \varepsilon, \quad R \geqslant R^\star(\varepsilon,t),$$

and on that range we obtain

$$\limsup_{n \to \infty} |\phi_{d,n}(t) - \phi_d(t)| = \lim_{n \to \infty} \sum_{r=1}^R \frac{|t|^r}{r!} \left| \mathbb{E} \left[ \left( \frac{N_n(d; \alpha_n^{\star})}{n} \right)^r \right] - \mathbb{P} \left[ D_1 = d, \dots, D_r = d \right] \right| + 2\varepsilon$$

$$= 2\varepsilon$$

$$(4.24)$$

upon invoking Proposition 4.6. Since  $\varepsilon > 0$  is arbitrary, we conclude that  $\lim_{n\to\infty} \phi_{d,n}(t) = \phi_d(t)$ .

The mapping  $\phi_d : \mathbb{R} \to \mathbb{C}$  being continuous at t = 0, it follows by a standard result on weak convergence due to Lévy [21, p. 431] that the distributional convergence (4.11) takes place. The distribution of the limiting rv  $\Pi(d)$  is determined in terms of its characteristic function through

$$\phi_d(t) = \mathbb{E}\left[e^{it\Pi(d)}\right], \quad t \in \mathbb{R}$$

4.5 A proof of Proposition 4.5 – A reduction step

Throughout this section the integer r = 1, 2, ... and the parameter  $\alpha > 0$  are being held fixed. Pick n > r. Thus, for each k = 1, ..., r, we find

$$D_{n,k}(\alpha) = \sum_{\ell=1, \ \ell \neq k}^{r} \mathbf{1} \left[ \xi_k + \xi_\ell > \alpha \right] + \widetilde{D}_{n,k}^{(r)}(\alpha)$$

with

$$\widetilde{D}_{n,k}^{(r)}(\alpha) = \sum_{\ell=r+1}^{n} \mathbf{1} \left[ \xi_k + \xi_\ell > \alpha \right].$$

Because the scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  satisfies  $\lim_{n\to\infty} \alpha_n^* = \infty$ , it is plain that

$$\lim_{n \to \infty} \max\left(\sum_{\ell=1, \ \ell \neq k}^{r} \mathbf{1}\left[\xi_k + \xi_\ell > \alpha_n^{\star}\right], \ k = 1, \dots, r\right) = 0 \quad a.s.$$

and (4.14) takes place if and only if

$$(\widetilde{D}_{n,1}^{(r)}(\alpha_n^{\star}),\ldots,\widetilde{D}_{n,r}^{(r)}(\alpha_n^{\star})) \Longrightarrow_n (D_1,\ldots,D_r).$$
(4.25)

The remainder of the proof consists in establishing that (4.25) holds. This will be done by showing that the joint pgfs converge to the joint pgf of an  $\mathbb{N}^r$ -valued rv over a suitable range.

To do so, our first step is to evaluate the pgfs. Pick  $z_1, \ldots, z_r$  in  $\mathbb{R}$ . Under the enforced independence assumptions, it is plain that

$$\mathbb{E}\left[\prod_{s=1}^{r} z_{s}^{\tilde{D}_{n,s}^{(r)}(\alpha)}\right] = \mathbb{E}\left[\prod_{s=1}^{r} z_{s}^{\sum_{\ell=r+1}^{n} \mathbf{1}[\xi_{s}+\xi_{\ell}>\alpha]}\right] \\
= \mathbb{E}\left[\prod_{s=1}^{r} \prod_{\ell=r+1}^{n} z_{s}^{\mathbf{1}[\xi_{s}+\xi_{\ell}>\alpha]}\right] \\
= \mathbb{E}\left[\prod_{\ell=r+1}^{n} \prod_{s=1}^{r} z_{s}^{\mathbf{1}[\xi_{s}+\xi_{\ell}>\alpha]}\right] \\
= \mathbb{E}\left[\mathbb{E}\left[\prod_{\ell=r+1}^{n} \prod_{s=1}^{r} z_{s}^{\mathbf{1}[\xi_{s}+\xi_{\ell}>\alpha]}|\xi_{1},\ldots,\xi_{r}\right]\right] \\
= \mathbb{E}\left[\mathbb{E}\left[\mathbb{E}\left[\prod_{\ell=r+1}^{n} \prod_{s=1}^{r} z_{s}^{\mathbf{1}[x_{s}+\xi_{\ell}>\alpha]}\right]_{x_{1}=\xi_{1},\ldots,x_{r}=\xi_{r}}\right].$$
(4.26)

With arbitrary  $x_1, \ldots, x_r$  in  $\mathbb{R}_+$ , we get

$$\mathbb{E}\left[\prod_{\ell=r+1}^{n}\prod_{s=1}^{r}z_{s}^{\mathbf{1}[x_{s}+\xi_{\ell}>\alpha]}\right] = \prod_{\ell=r+1}^{n}\mathbb{E}\left[\prod_{s=1}^{r}z_{s}^{\mathbf{1}[x_{s}+\xi_{\ell}>\alpha]}\right]$$
$$= F_{r}(\alpha; z_{1}, \dots, z_{r}; x_{1}, \dots, x_{r})^{n-r}$$

where we have set

$$F_r(\alpha; z_1, \dots, z_r; x_1, \dots, x_r) = \mathbb{E}\left[\prod_{s=1}^r z_s^{\mathbf{1}[x_s + \xi > \alpha]}\right]$$
$$= \mathbb{E}\left[\prod_{s=1}^r \left(\mathbf{1}\left[x_s + \xi > \alpha\right] z_s + \mathbf{1}\left[x_s + \xi \le \alpha\right]\right)\right] 4.27$$

In short, we have

$$\mathbb{E}\left[\prod_{s=1}^{r} z_s^{\tilde{D}_{n,s}^{(r)}(\alpha)}\right] = \mathbb{E}\left[F_r(\alpha; z_1, \dots, z_r; \xi_1, \dots, \xi_r)^{n-r}\right],\tag{4.28}$$

and the desired result (4.16) will hold if we show that

$$\lim_{n \to \infty} \mathbb{E}\left[F_r(\alpha; z_1, \dots, z_r; \xi_1, \dots, \xi_r)^{n-r}\right] = G_r(z_1, \dots, z_r)$$
(4.29)

for all  $z_1, \ldots, z_r$  in  $\mathbb{R}$  which satisfy (4.17).

# 4.6 A proof of Proposition 4.5 – A decomposition

To further analyze this last expression, with  $x_1, \ldots, x_r$  in  $\mathbb{R}_+$ , we introduce the index set

$$\mathcal{S}(\alpha; x_1, \ldots, x_r) = \{s = 1, \ldots, r : x_s > \alpha\}.$$

There are two possibilities which we now explore in turn: Either  $S(\alpha; x_1, \ldots, x_r)$  is empty or it is not, leading to a natural decomposition expressed through Lemmas 4.7 and 4.8.

**Lemma 4.7.** With  $x_1, \ldots, x_r$  in  $\mathbb{R}$ , whenever  $\mathcal{S}(\alpha; x_1, \ldots, x_r)$  is non-empty, we have

$$F_r(\alpha; z_1, \dots, z_r; x_1, \dots, x_r) = \left(\prod_{s \in \mathcal{S}(\alpha; x_1, \dots, x_r)} z_s\right) \cdot \mathbb{E}\left[\prod_{s \notin \mathcal{S}(\alpha; x_1, \dots, x_r)} (1 - (1 - z_s) \mathbf{1} [x_s + \xi > \alpha])\right] (4.30)$$

for all  $z_1, \ldots, z_r$  in  $\mathbb{R}$ .

**Proof.** Pick arbitrary  $x_1, \ldots, x_r$  in  $\mathbb{R}_+$  with non-empty  $\mathcal{S}(\alpha; x_1, \ldots, x_r)$ . For all  $z_1, \ldots, z_r$  in  $\mathbb{R}$ , it is easy to check by direct inspection from the expression (4.27) that (4.30) holds since

$$1 - (1 - z_s) \mathbf{1} \left[ x_s + \xi > \alpha \right] = z_s$$

whenever s belongs to  $\mathcal{S}(\alpha; x_1, \ldots, x_r)$ .

As an immediate consequence of (4.30) we have the inequality

$$x_1, \dots, x_r \in \mathbb{R}_+$$

$$0 \leq F_r(\alpha; z_1, \dots, z_r; x_1, \dots, x_r) \leq 1, \quad \text{with} \quad (4.31)$$

$$|\mathcal{S}(\alpha; x_1, \dots, x_r)| > 0$$

for all  $z_1, \ldots, z_r$  in  $\mathbb{R}$  in the range

$$|z_s| \leqslant 1, \quad s = 1, \dots, r. \tag{4.32}$$

This is because, it is always the case that

$$|1 - (1 - z_s) \mathbf{1} [x_s + \xi > \alpha]| \leq 1 \quad \text{if } |z_s| \leq 1.$$

We now turn to the case when the index set  $S(\alpha; x_1, \ldots, x_r)$  is empty, a fact characterized by the conditions

$$x_s \leqslant \alpha, \quad s = 1, \dots, r. \tag{4.33}$$

It will be convenient to arrange the values  $x_1, \ldots, x_r$  in increasing order, say  $x_{(1)} \leq x_{(2)} \leq \ldots \leq x_{(r)}$ , say with a lexicographic tiebreaker. Let  $\beta$  be any permutation of  $\{1, \ldots, r\}$  such that  $x_{(s)} = x_{\beta(s)}$  for all  $s = 1, \ldots, r$  – Obviously this permutation depends on  $x_1, \ldots, x_r$ . In what follows we shall use the convention  $x_{(0)} = -\infty$  and  $x_{(r+1)} = \infty$ .

**Lemma 4.8.** With  $x_1, \ldots, x_r$  in  $\mathbb{R}$ , whenever  $\mathcal{S}(\alpha; x_1, \ldots, x_r)$  is empty, we have

$$F_r(\alpha; z_1, \dots, z_r; x_1, \dots, x_r) = \sum_{t=0}^r \left( \prod_{s=t+1}^r z_{\beta(s)} \right) \cdot \left( F(\alpha - x_{(t)}) - F(\alpha - x_{(t+1)}) \right) (4.34)$$

for all  $z_1, \ldots, z_r$  in  $\mathbb{R}$ . In this expression the product of an empty number of factors is set to unity by convention.

**Proof.** In what follows, the values  $z_1, \ldots, z_r$  in  $\mathbb{R}$  are held fixed. Fix  $x_1, \ldots, x_r$  in  $\mathbb{R}$  and  $\alpha > 0$ . We define the events

$$A_{r|t}(x_1, \dots, x_r; \alpha) = \left[ x_{(t)} + \xi \leqslant \alpha < x_{(t+1)} + \xi \right], \quad t = 0, 1, \dots, r.$$

Under the enforced conventions, we have

$$A_{r|0}(x_1, \dots, x_r; \alpha) = [\alpha < x_{(1)} + \xi] \text{ and } A_{r|r}(x_1, \dots, x_r; \alpha) = [x_{(r)} + \xi \leqslant \alpha].$$

When  $\mathcal{S}(\alpha; x_1, \ldots, x_r)$  is empty, these r+1 events  $A_{r|0}(x_1, \ldots, x_r; \alpha), \ldots, A_{r|r}(x_1, \ldots, x_r; \alpha)$ are mutually exclusive and form a partition of the sample space, so that

$$F_{r}(\alpha; z_{1}, \dots, z_{r}; x_{1}, \dots, x_{r}) = \sum_{t=0}^{r} \mathbb{E} \left[ \mathbf{1} \left[ A_{r|t}(x_{1}, \dots, x_{r}; \alpha) \right] \prod_{s=1}^{r} \left( \mathbf{1} \left[ x_{s} + \xi > \alpha \right] z_{s} + \mathbf{1} \left[ x_{s} + \xi \leqslant \alpha \right] \right) \right] (4.35)$$

(i) On the event  $A_{r|0}(x_1, \ldots, x_r; \alpha)$ , we have  $\alpha < x_{(1)} + \xi$ , thus  $\alpha < x_s + \xi$  for all  $s = 1, \ldots, r$ , so that

$$\prod_{s=1}^{r} (1 - (1 - z_s) \mathbf{1} [x_s + \xi > \alpha]) = \prod_{s=1}^{r} z_s.$$

It then follows that

$$\mathbb{E}\left[\mathbf{1}\left[A_{r\mid0}(x_{1},\ldots,x_{r};\alpha)\right]\prod_{s=1}^{r}\left(1-(1-z_{s})\mathbf{1}\left[x_{s}+\xi>\alpha\right]\right)\right]$$
$$=\left(\prod_{s=1}^{r}z_{s}\right)\cdot\mathbb{P}\left[\alpha< x_{(1)}+\xi\right]$$
$$=\left(\prod_{s=1}^{r}z_{s}\right)\cdot\left(1-F(\alpha-x_{(1)})\right).$$
(4.36)

(ii) With t = 1, ..., r-1, on the event  $A_{r|t}(x_1, ..., x_r; \alpha)$  it holds that  $x_{(1)} + \xi \leq$ 

 $\alpha, \ldots, x_{(t)} + \xi \leq \alpha$  and  $\alpha < x_{(t+1)} + \xi, \ldots, \alpha < x_{(r)} + \xi$ . Therefore,

$$\prod_{s=1}^{r} \left( 1 - \left( 1 - z_s \right) \mathbf{1} \left[ x_s + \xi > \alpha \right] \right) = \left( \prod_{s=t+1}^{r} z_{\beta(s)} \right),$$

and we readily conclude to

$$\mathbb{E}\left[\mathbf{1}\left[A_{r|t}(x_{1},\ldots,x_{r};\alpha)\right]\prod_{s=1}^{r}\left(1-(1-z_{s})\mathbf{1}\left[x_{s}+\xi>\alpha\right]\right)\right]$$

$$=\left(\prod_{s=t+1}^{r}z_{\beta(s)}\right)\cdot\mathbb{P}\left[x_{(t)}+\xi\leqslant\alpha< x_{(t+1)}+\xi\right]$$

$$=\left(\prod_{s=t+1}^{r}z_{\beta(s)}\right)\cdot\left(F(\alpha-x_{(t)})-F(\alpha-x_{(t+1)})\right).$$
(4.37)

(iii) Finally, on the event  $A_{r|r}(x_1, \ldots, x_r; \alpha)$ ,  $x_{(r)} + \xi \leq \alpha$ , thus  $x_s + \xi \leq \alpha$  for

all  $s = 1, \ldots, r$ , so that

$$\prod_{s=1}^{r} (1 - (1 - z_s) \mathbf{1} [x_s + \xi > \alpha]) = 1,$$

whence

$$\mathbb{E}\left[\mathbf{1}\left[A_{r|r}(x_1,\ldots,x_r;\alpha)\right]\prod_{s=1}^r \left(1-(1-z_s)\mathbf{1}\left[x_s+\xi>\alpha\right]\right)\right] = \mathbb{P}\left[x_{(r)}+\xi\leqslant\alpha\right]$$
$$= F(\alpha-x_{(r)}).(4.38)$$

To complete the proof we substitute (4.36), (4.37) and (4.38) into (4.35), and recall that  $F(\alpha - x_{(0)}) = 1$  and  $F(\alpha - x_{(r+1)}) = 0$  under the enforced conventions.

#### 4.7 A proof of Proposition 4.5 – Taking the limit

In order to establish the convergence (4.25) we return to the expression (4.28) for the joint pgf of the relevant rvs. Fix n = 2, 3, ... with r < n, and replace  $\alpha$  by  $\alpha_n^{\star}$  according to the scaling  $\alpha^{\star} : \mathbb{N}_0 \to \mathbb{R}_+$  appearing in Assumption 4.1.

## 4.7.1 A useful intermediary fact

For arbitrary  $\alpha > 0$ , consider  $x_1, \ldots, x_r$  in  $\mathbb{R}$  and  $z_1, \ldots, z_r$  in  $\mathbb{R}$ . In what follows it will be convenient to define

$$\Lambda_r(\alpha; z_1, \ldots, z_r; x_1, \ldots, x_r) = 1 - F_r(\alpha; z_1, \ldots, z_r; x_1, \ldots, x_r)$$

so that

$$F_r(\alpha; z_1, \dots, z_r; x_1, \dots, x_r) = 1 - \Lambda_r(\alpha; z_1, \dots, z_r; x_1, \dots, x_r).$$
(4.39)

Whenever  $\mathcal{S}(\alpha; x_1, \ldots, x_r)$  is empty, Lemma 4.8 gives

$$\Lambda_{r}(\alpha; z_{1}, \dots, z_{r}; x_{1}, \dots, x_{r}) = 1 - \sum_{t=0}^{r} \left( \prod_{s=t+1}^{r} z_{\beta(s)} \right) \cdot \left( F(\alpha - x_{(t)}) - F(\alpha - x_{(t+1)}) \right) \\
= -\sum_{t=0}^{r-1} \left( \prod_{s=t+1}^{r} z_{\beta(s)} \right) \cdot \left( F(\alpha - x_{(t)}) - F(\alpha - x_{(t+1)}) + \left( 1 - F(\alpha - x_{(r)}) \right) \right) 4.40$$

Now fix n = 2, 3, ... with r < n, and replace  $\alpha$  by  $\alpha_n^{\star}$  in (4.40) according to the scaling  $\alpha^{\star} : \mathbb{N}_0 \to \mathbb{R}_+$  appearing in Assumption 4.1: We get

$$\lim_{n \to \infty} n \left( 1 - F(\alpha_n^{\star} - x_{(r)}) \right) = \lambda(x_{(r)})$$

$$\lim_{n \to \infty} n \left( F(\alpha_n^{\star} - x_{(t)}) - F(\alpha_n^{\star} - x_{(t+1)}) \right)$$

$$= \lim_{n \to \infty} n \left( \left( 1 - F(\alpha_n^{\star} - x_{(t+1)}) - \left( 1 - F(\alpha_n^{\star} - x_{(t)}) \right) \right) \right)$$

$$= \begin{cases} \lambda(x_{(1)}) & \text{if } t = 0 \\ \lambda(x_{(1)}) - \lambda(x_{(t)}) & \text{if } t = 1, \dots, r - 1. \end{cases}$$
(4.41)

As a result, with  $\mathcal{S}(\alpha; x_1, \ldots, x_r)$  empty, we have

$$\lim_{n \to \infty} n\Lambda_r(\alpha_n^{\star}; z_1, \dots, z_r; x_1, \dots, x_r) \\
= -\lambda(x_{(1)}) \left(\prod_{s=1}^r z_s\right) - \sum_{t=1}^{r-1} \left(\prod_{s=t+1}^r z_{\beta(s)}\right) \left(\lambda(x_{(t+1)}) - \lambda(x_{(t)})\right) + \lambda(x_{(r)}) \\
= -\sum_{t=1}^r \lambda(x_{(t)}) \left(\prod_{s=t}^r z_{\beta(s)} - \prod_{s=t+1}^r z_{\beta(s)}\right) \\
= \sum_{t=1}^r \lambda(x_{(t)}) (1 - z_{\beta(t)}) \prod_{s=t+1}^r z_{\beta(s)},$$
(4.42)

and the conclusion

$$\lim_{n \to \infty} F_r(\alpha_n^*; z_1, \dots, z_r; x_1, \dots, x_r)^{n-r}$$

$$= \lim_{n \to \infty} \left( 1 - \Lambda_r(\alpha_n^*; z_1, \dots, z_r; x_1, \dots, x_r) \right)^{n-r}$$

$$= e^{-\sum_{t=1}^r \lambda(x_{(t)})(1-z_{\beta(t)}) \prod_{s=t+1}^r z_{\beta(s)}}$$
(4.43)

follows by standard arguments.

and

# 4.7.2 In the limit

Pick  $z_1, \ldots, z_r$  in  $\mathbb{R}$  such that (4.17) holds. For each  $n = 2, 3, \ldots$  with r < n, the decomposition

$$\mathbb{E}\left[F_{r}(\alpha_{n}^{\star};z_{1},\ldots,z_{r};\xi_{1},\ldots,\xi_{r})^{n-r}\right]$$

$$=\mathbb{E}\left[\mathbf{1}\left[|\mathcal{S}(\alpha_{n}^{\star};\xi_{1},\ldots,\xi_{r})|>0\right]F_{r}(\alpha_{n}^{\star};z_{1},\ldots,z_{r};\xi_{1},\ldots,\xi_{r})^{n-r}\right]$$

$$+\mathbb{E}\left[\mathbf{1}\left[|\mathcal{S}(\alpha_{n}^{\star};\xi_{1},\ldots,\xi_{r})|=0\right]F_{r}(\alpha_{n}^{\star};z_{1},\ldots,z_{r};\xi_{1},\ldots,\xi_{r})^{n-r}\right] (4.44)$$

holds. Because  $\lim_{n\to\infty}\alpha_n^\star=\infty,$  it follows that

$$\lim_{n \to \infty} \mathbb{P}\left[ |\mathcal{S}(\alpha_n^{\star}; \xi_1, \dots, \xi_r)| = 0 \right] = \lim_{n \to \infty} \mathbb{P}\left[ \xi_1 \leqslant \alpha_n^{\star}, \dots, \xi_r \leqslant \alpha_n^{\star} \right] = 1,$$

and the inequality (4.31) readily yields

$$\lim_{n \to \infty} \mathbb{E} \left[ \mathbf{1} \left[ |\mathcal{S}(\alpha_n^\star; \xi_1, \dots, \xi_r)| > 0 \right] F_r(\alpha_n^\star; z_1, \dots, z_r; \xi_1, \dots, \xi_r)^{n-r} \right] = 0$$
(4.45)

since the condition (4.17) is more restrictive than (4.32).

Next,

$$\mathbb{E}\left[\mathbf{1}\left[|\mathcal{S}(\alpha_n^{\star};\xi_1,\ldots,\xi_r)|=0\right]F_r(\alpha_n^{\star};z_1,\ldots,z_r;\xi_1,\ldots,\xi_r)^{n-r}\right]$$
$$=\mathbb{E}\left[\mathbf{1}\left[|\mathcal{S}(\alpha_n^{\star};\xi_1,\ldots,\xi_r)|=0\right]\left(1-\Lambda_r(\alpha_n^{\star};z_1,\ldots,z_r;\xi_1,\ldots,\xi_r)\right)^{n-r}\right](4.46)$$

with (4.40) yielding

$$\Lambda_r(\alpha_n^{\star}; z_1, \dots, z_r; \xi_1, \dots, \xi_r) = \frac{1 - F(\alpha_n^{\star} - \xi_{r,r}) - \sum_{t=0}^{r-1} \left(\prod_{s=t+1}^r z_{\beta(s)}\right) \cdot \left(F(\alpha_n^{\star} - \xi_{r,t}) - F(\alpha_n^{\star} - \xi_{r,t+1})\right) (4.47)$$

Here the order statistics  $\xi_{r,1}, \ldots, \xi_{r,r}$  associated with  $\xi_1, \ldots, \xi_r$  were introduced in the statement of Proposition 4.5, together with the random permutation  $\beta : \{1, \ldots, r\} \rightarrow \{1, \ldots, r\}$ .

It is plain that

$$0 \leqslant \left(\prod_{s=t+1}^{r} z_{\beta(s)}\right) \leqslant 1, \quad t = 0, \dots, r-1$$

under the condition (4.17), and direct inspection gives

$$0 \leq \Lambda_r(\alpha_n^{\star}; z_1, \dots, z_r; \xi_1, \dots, \xi_r) \leq 1$$

so that

$$\left|F_r(\alpha_n^{\star}; z_1, \dots, z_r; \xi_1, \dots, \xi_r)^{n-r}\right| \leq 1.$$

Therefore, with the fact  $\lim_{n\to\infty} \mathbf{1} \left[ |\mathcal{S}(\alpha_n^{\star}; \xi_1, \dots, \xi_r)| = 0 \right] = 1$  noted earlier, we see that the convergence

$$\lim_{n \to \infty} \mathbf{1} \left[ \left| \mathcal{S}(\alpha_n^{\star}; \xi_1, \dots, \xi_r) \right| = 0 \right] F_r(\alpha_n^{\star}; z_1, \dots, z_r; \xi_1, \dots, \xi_r)^{n-r} \\ = e^{-\sum_{t=1}^r \lambda(\xi_{r,t})(1-z_{\beta(t)}) \prod_{s=t+1}^r z_{\beta(s)}}$$
(4.48)

takes place boundedly. By the Bounded Convergence Theorem it follows that

$$\lim_{n \to \infty} \mathbb{E} \left[ \mathbf{1} \left[ |\mathcal{S}(\alpha_n^\star; \xi_1, \dots, \xi_r)| = 0 \right] F_r(\alpha_n^\star; z_1, \dots, z_r; \xi_1, \dots, \xi_r)^{n-r} \right]$$
$$= \mathbb{E} \left[ e^{-\sum_{t=1}^r \lambda(\xi_{r,t})(1-z_{\beta(t)}) \prod_{s=t+1}^r z_{\beta(s)}} \right]$$
$$= G_r(z_1, \dots, z_r)$$
(4.49)

Collecting (4.45) and (4.49), and using (4.44) we conclude that (4.29) holds on the range (4.17).

## 4.8 A proof of Proposition 4.6

Fix  $\alpha > 0$ , d = 0, 1, ... and r = 1, 2, ... With n = r, r + 1, ..., let  $\mathcal{P}_{n,r}$  denote the collection of all *ordered* arrangements of r distinct elements drawn from the set  $\{1, ..., n\}$ . Any such arrangement can be viewed as a one-to-one mapping  $\pi : \{1, ..., r\} \rightarrow \{1, ..., n\}$ .

We begin by noting that we can always write

$$N_n(d;\alpha)^r = \left(\sum_{k=1}^n \mathbf{1} \left[ D_{n,k}(\alpha) = d \right] \right)^r$$
$$= \sum_{\pi \in \mathcal{P}_{n,r}} \mathbf{1} \left[ D_{n,\pi(1)}(\alpha) = d \right] \dots \mathbf{1} \left[ D_{n,\pi(r)}(\alpha) = d \right] + R_{n,r}(d;\alpha) (4.50)$$

where the correction term  $R_{n,r}(d;\alpha)$  is a sum comprising  $\{0,1\}$ -valued rvs. Since the correction term  $R_{n,r}(d;\alpha)$  is a sum of exactly  $n^r - |\mathcal{P}_{n,r}|$  terms, with each term bounded by 1, we have the upper bound

$$R_{n,r}(d;\alpha) \leqslant n^r - |\mathcal{P}_{n,r}|.$$
(4.51)

Therefore, taking expectations, we obtain

$$\mathbb{E}\left[N_{n}(d;\alpha)^{r}\right]$$

$$= \sum_{\pi\in\mathcal{P}_{n,r}}\mathbb{E}\left[\mathbf{1}\left[D_{n,\pi(1)}(\alpha)=d\right]\dots\mathbf{1}\left[D_{n,\pi(r)}(\alpha)=d\right]\right] + \mathbb{E}\left[R_{n,r}(d;\alpha)\right]$$

$$= |\mathcal{P}_{n,r}|\cdot\mathbb{P}\left[D_{n,1}(\alpha)=d,\dots,D_{n,r}(\alpha)=d\right] + \mathbb{E}\left[R_{n,r}(d;\alpha)\right]$$
(4.52)

as we make use of the fact that the rvs  $D_{n,1}(\alpha), \ldots, D_{n,n}(\alpha)$  are exchangeable. As a result,

$$\mathbb{E}\left[\left(\frac{N_n(d;\alpha)}{n}\right)^r\right]$$
  
=  $\frac{|\mathcal{P}_{n,r}|}{n^r} \cdot \mathbb{P}\left[D_{n,1}(\alpha) = d, \dots, D_{n,r}(\alpha) = d\right] + \mathbb{E}\left[\frac{R_{n,r}(d;\alpha)}{n^r}\right],$  (4.53)

and the bound (4.51) implies

$$\mathbb{E}\left[\frac{R_{n,r}(d;\alpha)}{n^r}\right] \leqslant 1 - \frac{|\mathcal{P}_{n,r}|}{n^r} \tag{4.54}$$

with

$$|\mathcal{P}_{n,r}| = n(n-1)\dots(n-r+1).$$

Now consider the scaling  $\alpha^* : \mathbb{N}_0 \to \mathbb{R}_+$  whose existence is assumed in Assumption 4.1. For each  $n = r, r + 1, \ldots$  replace  $\alpha$  by  $\alpha_n^*$  in (4.53) according to this scaling and let n go to infinity in the resulting relation: First, we note that

$$\lim_{n \to \infty} \frac{|\mathcal{P}_{n,r}|}{n^r} = 1$$

by direct inspection, so that

$$\lim_{n \to \infty} \mathbb{E}\left[\frac{R_{n,r}(d; \alpha_n^{\star})}{n^r}\right] = 0$$

as we make use of (4.54). It follows that

$$\lim_{n \to \infty} \mathbb{E}\left[\left(\frac{N_n(d; \alpha_n^{\star})}{n}\right)^r\right] = \lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^{\star}) = d, \dots, D_{n,r}(\alpha_n^{\star}) = d\right]$$

with the understanding that if one of the limits exists, so does the other and their value coincide. The latter exists since by Proposition 4.4 we have

$$\lim_{n \to \infty} \mathbb{P}\left[D_{n,1}(\alpha_n^{\star}) = d, \dots, D_{n,r}(\alpha_n^{\star}) = d\right] = \mathbb{P}\left[D_1 = d, \dots, D_r = d\right]$$

where the rvs  $D_1, \ldots, D_r$  are the limiting rvs appearing in the convergence (4.14). It is now plain that (4.19) holds and the proof of Proposition 4.6 is now complete.

## Chapter 5: Degree Distribution in Growth Models

As indicated in Chapter 1, we are interested in growth models that can explain power-law behavior in real-world networks. There we mentioned a number of existing models that implement preferential attachment on the basis of degree and fitness information combined in various ways. Existing works consider special cases where power-law behavior is observed under these models, but do not give a satisfactory account of what conditions are required (e.g., on the fitness distribution) to obtain power-law behavior. We consider arguably the simplest of these models, called the *fitness model* and analyze its degree distribution. Our main motivation is to explore how the fitness distribution affects the empirical degree distribution, if at all, and under what conditions power-law behavior could be obtained.

## 5.1 The model

A word on notation: Here we consider the sequence of graphs indexed by t instead of n (as was done in the previous chapters). This notation is natural when considering growth models where there is a notion of adding nodes and edges over time.

The fitness-based random graph model is defined by means of two collections

of rvs, namely  $\{U, U_t, t = 0, 1, ...\}$  and  $\{\xi, \xi_t, t = 0, 1, ...\}$ , defined on some probability triple  $(\Omega, \mathcal{F}, \mathbb{P})$  – All probabilistic statements are made with respect to the probability measure  $\mathbb{P}$ , whose expectation operator is denoted by  $\mathbb{E}$ .

The discussion is carried out under the following set of assumptions:

- (i) The rvs  $\{U, U_t, t = 0, 1, ...\}$  and  $\{\xi, \xi_t, t = 0, 1, ...\}$  are mutually independent;
- (ii) The rvs  $\{U, U_t, t = 0, 1, ...\}$  are i.i.d. rvs, each of which is uniformly distributed on the interval (0, 1); and
- (iii) The rvs  $\{\xi, \xi_t, t = 0, 1, ...\}$  are i.i.d.  $\mathbb{R}_+$ -valued rvs. Throughout we assume the non-degeneracy condition

$$\mathbb{P}\left[\xi=0\right]=0,\tag{5.1}$$

as well as the finite mean condition

$$0 < \mathbb{E}\left[\xi\right] < \infty. \tag{5.2}$$

Under these assumptions we can always select (as we do from now on) the probability triple  $(\Omega, \mathcal{F}, \mathbb{P})$ , and the rvs  $\{U, U_t, t = 0, 1, ...\}$  and  $\{\xi, \xi_t, t = 0, 1, ...\}$ defined on it as mappings  $\Omega \to \mathbb{R}$  which simultaneously satisfy the conditions

$$\xi > 0, \xi_t > 0$$
 and  $0 < U, U_t < 1, t = 0, 1, \dots$ 

We shall find it convenient to write

$$\Xi_{-1} \equiv 0, \ \Xi_t = \sum_{s=0}^t \xi_s, \quad t = 0, 1, \dots$$
 (5.3)

and

$$\xi^t = (\xi_0, \xi_1, \dots, \xi_t), \quad t = 0, 1, \dots$$
 (5.4)

We shall also make use of the filtration  $\{\mathcal{F}_t, t = 0, 1, ...\}$  on  $\Omega$  given by

$$\mathcal{F}_t = \sigma (U_s, \xi_s, s = 0, 1, \dots, t), \quad t = 0, 1, \dots$$

We now formally define the sequence of (undirected) random graphs { $\mathbb{G}_t$ ,  $t = 0, 1, \ldots$ } studied in this chapter: For each  $t = 0, 1, \ldots$ , the random graph  $\mathbb{G}_t$  has vertex set  $V_t = \{0, 1, \ldots, t\}$  and random edge set  $\mathbb{E}_t \subseteq V_t \times V_t$ . As this is a growth model, imagine there being an initial node, labelled node 0, present in the system at time t = 0, with new nodes, labelled  $t = 1, 2, \ldots$ , arriving one at a time, say at times  $t = 1, 2, \ldots$ . The definition is a recursive one, starting with the initial random graph  $\mathbb{G}_0 = (V_0, \mathbb{E}_0)$  where  $V_0 = \{0\}$  and  $\mathbb{E}_0 = \emptyset$ . With  $t = 0, 1, \ldots$ , once the random graphs  $\mathbb{G}_0, \ldots, \mathbb{G}_t$  have been defined, we can generate  $\mathbb{G}_{t+1}$  from  $\mathbb{G}_t$  by introducing a new vertex not in  $V_t$  (which we label t + 1), and connecting it to the node  $S_{t+1}$ randomly selected in  $V_t$  according to

$$S_{t+1} = s \quad \text{if} \quad \frac{\Xi_{s-1}}{\Xi_t} < U_{t+1} \leqslant \frac{\Xi_s}{\Xi_t}, \quad s \in V_t.$$
(5.5)

Only the fitness levels of the nodes already present in  $V_t$  matter in determining the likelihood of the node to which node t + 1 will attach. The newly created link between nodes t + 1 and  $S_{t+1}$  is interpreted as an *undirected* link, so that

$$\mathbb{E}_{t+1} = \mathbb{E}_t \cup \{ (t+1, S_{t+1}), (S_{t+1}, t+1) \}.$$

Although the rvs  $S_1, \ldots, S_t$  are *not* mutually independent, they are condition-

ally mutually independent given  $\xi^t$ . It is a simple matter to check that

$$\mathbb{P}[S_1 = r_1, \dots, S_t = r_t, S_{t+1} = r_{t+1} | \mathcal{F}_t] = \prod_{\tau=1}^{t+1} \mathbb{P}[S_\tau = r_\tau | \mathcal{F}_t]$$
(5.6)

with arbitrary  $r_1$  in  $V_1, \ldots, r_{t+1}$  in  $V_{t+1}$ , where

$$\mathbb{P}\left[S_{\tau} = r_{\tau} | \mathcal{F}_t\right] = \frac{\xi_{r_{\tau}}}{\Xi_{\tau-1}}, \quad \tau = 1, \dots, t+1$$
(5.7)

Fix t = 1, 2, ... For each s = 0, 1, ..., t, let the rv  $D_t(s)$  denote the degree of node s in  $\mathbb{G}_t$ . By construction the rv  $D_t(s)$  can be expressed as

$$D_t(s) = 1 + \sum_{r=s+1}^t \mathbf{1} \left[ S_r = s \right]$$
(5.8)

with possible values  $1, \ldots, t - s + 1$ . With  $d = 1, 2, \ldots$ , the number  $N_t(d)$  of nodes in  $\mathbb{G}_t$  with degree d is then the rv given by

$$N_t(d) = \sum_{s=0}^t \mathbf{1} \left[ D_t(s) = d \right],$$
(5.9)

and

$$\frac{N_t(d)}{t+1} = \frac{1}{t+1} \sum_{s=0}^t \mathbf{1} \left[ D_t(s) = d \right]$$

is the fraction of nodes in  $\mathbb{G}_t$  whose degree is d.

## 5.2 The convergence results

Before stating the main result of the chapter, we introduce the pmf  $p_{\xi} = (p_{\xi}(d), d = 1, 2, ...)$  on  $\mathbb{N}_0$  defined by

$$p_{\xi}(d+1) = \mathbb{E}\left[\frac{\Lambda(\xi)^d}{d!}e^{-\Lambda(\xi)}\right], \quad d = 0, 1, \dots$$
(5.10)
where the rv  $\Lambda(\xi)$  is given by

$$\Lambda(\xi) = \frac{\xi}{\mathbb{E}\left[\xi\right]} \log\left(\frac{1}{U}\right). \tag{5.11}$$

The pmf  $p_{\xi}$  can be interpreted as the pmf of the rv  $1 + Z(\xi)$  where the rv  $Z(\xi)$  is a conditionally Poisson rv with random parameter  $\Lambda(\xi)$ , namely

$$\mathbb{P}\left[Z(\xi) = d\right] = \mathbb{E}\left[\frac{\Lambda(\xi)^d}{d!}e^{-\Lambda(\xi)}\right], \quad d = 0, 1, \dots$$

#### 5.2.1 The main result

With this notation we have the following result.

**Theorem 5.1.** Under the foregoing assumptions, we have the convergence

$$\lim_{t \to \infty} \mathbb{E}\left[\frac{N_t(d)}{t+1}\right] = p_{\xi}(d), \quad d = 1, 2, \dots$$
(5.12)

A proof of Theorem 5.1 is given in Sections 5.5-5.9. The pmf  $p_{\xi}$  depends on  $\xi$  only through normalization to its mean, namely  $\frac{\xi}{\mathbb{E}[\xi]}$ , since

$$\Lambda(\xi) = \Lambda\left(\frac{\xi}{\mathbb{E}\left[\xi\right]}\right).$$

This implies

$$p_{\xi}(d) = p_{\frac{\xi}{\mathbb{E}[\xi]}}(d), \quad d = 1, 2, \dots$$

as should be expected from the form (7.4) of the link creation probabilities.

For each  $t = 0, 1, ..., \text{let } \nu_t$  denote a rv which is uniformly distributed over the edge set  $V_t$  and independent of  $\mathcal{F}_t$ , thus of  $\mathbb{G}_t$ . It is plain that the representation

$$\mathbb{E}\left[\frac{N_t(d)}{t+1}\right] = \mathbb{P}\left[D_t(\nu_t) = d\right], \quad d = 1, 2, \dots$$
(5.13)

holds, and Theorem 5.1 can be given the following probabilistic form.

Theorem 5.2. Under the foregoing assumptions, we have the convergence

$$\lim_{t \to \infty} \mathbb{P}\left[D_t(\nu_t) = d\right] = p_{\xi}(d), \quad d = 1, 2, \dots$$
(5.14)

More compactly,

$$D_t(\nu_t) \Longrightarrow_t D(\xi) \tag{5.15}$$

where  $D(\xi)$  is an  $\mathbb{N}_0$ -valued rv distributed according to the pmf  $p_{\xi}$ .

# 5.2.2 An alternate expression for the limiting pmf $p_{\xi}$

The pmf  $p_{\xi}$  admits an alternate expression which will yield insights into its tail behavior.

#### Proposition 5.3. It holds that

$$p_{\xi}(d) = \mathbb{E}\left[\frac{\mathbb{E}\left[\xi\right]}{\mathbb{E}\left[\xi\right] + \xi} \cdot \left(\frac{\xi}{\mathbb{E}\left[\xi\right] + \xi}\right)^{d-1}\right], \quad d = 1, 2, \dots$$
(5.16)

In other words, the pmf  $p_{\xi}$  can also be viewed as the pmf of a conditionally geometric rv on  $\mathbb{N}_0$  with random parameter  $R(\xi)$  given by

$$R(\xi) = \frac{\xi}{\mathbb{E}\left[\xi\right] + \xi} \tag{5.17}$$

as we note from (5.16)-(5.17) that

$$p_{\xi}(d) = \mathbb{E}\left[ (1 - R(\xi)) R(\xi)^{d-1} \right], \quad d = 1, 2, \dots$$
 (5.18)

**Proof.** In view of the expressions (5.10)-(5.11) and (5.16)-(5.17), there is no loss of generality in assuming  $\mathbb{E}[\xi] = 1$  (as we do from now on in this proof) – Just replace  $\xi$  by  $\frac{\xi}{\mathbb{E}[\xi]}$ , in which case the quantities  $\Lambda(\xi)$  and  $R(\xi)$  become

$$\Lambda(\xi) = \xi \log\left(\frac{1}{U}\right)$$
 and  $R(\xi) = \frac{\xi}{1+\xi}$ ,

respectively. Thus,  $e^{-\Lambda(\xi)} = U^{\xi}$  and for each  $d = 0, 1, \ldots$ , we use (5.10) to obtain

$$p_{\xi}(d+1) = \mathbb{E}\left[\frac{\xi^{d}}{d!}\left(\log\left(\frac{1}{U}\right)\right)^{d}U^{\xi}\right]$$
$$= (-1)^{d} \cdot \mathbb{E}\left[\frac{\xi^{d}}{d!} \cdot \mathbb{E}\left[U^{\xi}(\log U)^{d}|\xi\right]\right]$$
$$= (-1)^{d} \cdot \mathbb{E}\left[\frac{\xi^{d}}{d!} \cdot \mathbb{E}\left[U^{t}(\log U)^{d}\right]_{t=\xi}\right]$$
(5.19)

by the independence of the rvs  $\xi$  and U, where we note that

$$\mathbb{E}\left[U^t(\log U)^d\right] = \int_0^1 x^t(\log x)^d dx = (-1)^d \frac{d!}{(1+t)^{d+1}}, \quad t > 0$$

This last fact follows by repeated integration by parts; details are left to the interested reader. Substituting into (5.19) we get

$$p_{\xi}(d+1) = (-1)^{d} \cdot \mathbb{E}\left[\frac{\xi^{d}}{d!} \cdot (-1)^{d} \frac{d!}{(1+\xi)^{d+1}}\right] = \mathbb{E}\left[\frac{\xi^{d}}{(1+\xi)^{d+1}}\right]$$

as desired.

## 5.3 Tail behavior

The tail behavior of the pmf  $p_{\xi}$  depends on the distributional properties of the rv  $\xi$ , with a key role being played by the quantity  $\xi^{\star}$  given by

$$\xi^{\star} = \inf\left(x \ge 0: \ \mathbb{P}\left[\xi \le x\right] = 1\right) \tag{5.20}$$

with the customary understanding that  $\xi^* = \infty$  if the defining set in (5.20) is empty. Under (5.2) we necessarily have  $\xi^* > 0$  (possibly infinite).

The next result provides some important information regarding the tail behavior of the pmf  $p_{\xi}$ . Of particular interest in the discussion is the bounded mapping  $g_{\xi} : [0, \infty] \to \mathbb{R}$  given by

$$g_{\xi}(x) = \begin{cases} \frac{x}{\mathbb{E}[\xi]+x} & \text{if } x \in \mathbb{R}_+ \\ \\ 1 & \text{if } x = \infty, \end{cases}$$

the value at  $x = \infty$  being determined by continuity.

Lemma 5.4. Under the foregoing assumptions, it is always the case that

$$\lim_{d \to \infty} p_{\xi}(d)^{\frac{1}{d}} = g_{\xi}(\xi^{\star}) \tag{5.21}$$

with

$$g_{\xi}(\xi^{\star}) = \begin{cases} \frac{\xi^{\star}}{\mathbb{E}[\xi] + \xi^{\star}} & \text{if } 0 < \xi^{\star} < \infty \\\\ 1 & \text{if } \xi^{\star} = \infty. \end{cases}$$

The proof of Lemma 5.4 is given in Section 5.10. When  $\xi^*$  is finite, then  $g_{\xi}(\xi^*) < 1$  so that

$$\lim_{d \to \infty} \frac{1}{d} \log p_{\xi}(d) = \log \left( \frac{\xi^{\star}}{\mathbb{E}\left[\xi\right] + \xi^{\star}} \right) < 0.$$
(5.22)

This indicates a *geometric* decay for  $p_{\xi}$  in the following sense: For  $\epsilon$  in (0, 1) sufficiently small so that  $(1 + \delta)g_{\xi}(\xi^{\star}) < 1$  where we have set  $\epsilon = \log(1 + \delta)$ , the

convergence (5.22) implies

$$-\epsilon < \frac{1}{d}\log p_{\xi}(d) - \log g(\xi^*) < \epsilon \tag{5.23}$$

or

$$\left(\frac{g_{\xi}(\xi^{\star})}{(1+\delta)}\right)^d < p_{\xi}(d) < ((1+\delta)g_{\xi}(\xi^{\star}))^d$$

for all d = 1, 2, ... sufficiently large (and determined by  $\delta$ ). In fact, direct inspection of (5.17) yields

$$p_{\xi}(d) = \mathbb{E}\left[\left(1 - R(\xi)\right) R(\xi)^{d-1}\right]$$

$$\leq \mathbb{E}\left[\left(1 - R(\xi)\right) R(\xi^{\star})^{d-1}\right]$$

$$= \left(1 - \mathbb{E}\left[R(\xi)\right]\right) \cdot g_{\xi}(\xi^{\star})^{d-1}$$

$$= \left(\frac{1 - \mathbb{E}\left[R(\xi)\right]}{g_{\xi}(\xi^{\star})}\right) \cdot g_{\xi}(\xi^{\star})^{d} \qquad (5.24)$$

for all d = 1, 2, ...

# 5.4 Special cases when $\xi^* = \infty$

When  $\xi^*$  is infinite, geometric decay is not possible anymore, and many types of tail behavior are possible for the pmf  $p_{\xi}$  as we now illustrate with two special cases. The probability distribution function of the rv  $\xi$  is assumed to admit a probability density function  $f_{\xi} : \mathbb{R}_+ \to \mathbb{R}_+$ . For each  $d = 0, 1, \ldots$ , the expression (5.16) becomes

$$p(d+1) = \mathbb{E}\left[\left(1 - \frac{1}{1+\xi}\right)^{d} \frac{1}{1+\xi}\right] \\ = \int_{0}^{\infty} \left(1 - \frac{1}{1+t}\right)^{d} \frac{1}{1+t} \cdot f_{\xi}(t) dt \\ = \int_{0}^{1} \frac{(1-s)^{d}}{s} \cdot f_{\xi}\left(\frac{1-s}{s}\right) ds \quad [s = \frac{1}{1+t} \text{ and } t = \frac{1-s}{s}].$$
(5.25)

#### 5.4.1 Pareto distribution

We say that the rv  $\xi$  has a power law if

$$\mathbb{P}\left[\xi \leqslant t\right] = 1 - \left(\frac{a}{a+\alpha}\right)^{\alpha}, \quad t \ge 0$$
(5.26)

for some a > 0 and  $\alpha > 0$ . Its probability density function is given by

$$f_{\xi}(t) = \alpha a^{\alpha} (a+t)^{-(\alpha+1)}, \quad t \ge 0$$
 (5.27)

and the first moment is easily computed to be

$$\mathbb{E}\left[\xi\right] = \frac{a}{(\alpha - 1)^+}.$$

The requirement that  $\mathbb{E}[\xi]$  be finite is equivalent to  $\alpha > 1$ , in which case  $\mathbb{E}[\xi] = a(\alpha - 1)^{-1}$ . Thus,  $\mathbb{E}[\xi] = 1$  amounts to  $\alpha > 1$  and  $a = \alpha - 1$ .

Fix d = 0, 1, ... If we insert (5.27) into (5.25) we obtain by elementary calculations that

$$p_{\xi}(d+1) = \alpha a^{\alpha} \int_{0}^{1} (1-s)^{d} s^{\alpha} (1+(a-1)s)^{-(\alpha+1)} ds$$
$$= \alpha a^{\alpha} d^{-(\alpha+1)} \int_{0}^{d} \left(1-\frac{y}{d}\right)^{d} y^{\alpha} \left(1+(a-1)\frac{y}{d}\right)^{-(\alpha+1)} dy \quad [s = \frac{y}{d}]5.28)$$

By the Bounded Convergence Theorem it is now plain that

$$\lim_{d \to \infty} \frac{p_{\xi}(d+1)}{\alpha a^{\alpha} d^{-(\alpha+1)}} = \int_0^\infty y^{\alpha} e^{-y} dy,$$

and the following result follows.

**Lemma 5.5.** When  $\xi$  is distributed according to (5.26) with  $\alpha > 1$  and  $a = \alpha - 1$ , then the asymptotic equivalence

$$p_{\xi}(d+1) \sim C(\alpha)\alpha a^{\alpha} d^{-(\alpha+1)}$$
 with  $C(\alpha) = \int_0^\infty y^{\alpha} e^{-y} dy$ 

holds.

Note that  $C(\alpha)$  is finite for all  $\alpha \ge 1$ .

#### 5.4.2 Exponential distribution

The rv  $\xi$  is an exponentially distributed rv with unit mean if

$$\mathbb{P}\left[\xi \leqslant t\right] = 1 - e^{-t}, \quad t \ge 0, \tag{5.29}$$

in which case its probability density function is given by

$$f_{\xi}(t) = e^{-t}, \quad t \ge 0.$$

For each d = 0, 1, ..., with (5.25) as starting point, elementary calculations now give

$$p_{\xi}(d+1) = \int_{0}^{1} (1-s)^{d} \frac{e^{-\frac{1-s}{s}}}{s} ds$$
$$= e \int_{0}^{1} (1-s)^{d} \frac{e^{-\frac{1}{s}}}{s} ds.$$
(5.30)

**Lemma 5.6.** When  $\xi$  is distributed according to (5.29), then the asymptotic equivalence

$$p_{\xi}(d+1) \sim \sqrt{\frac{\pi}{e^2}} \frac{e^{-2\sqrt{d}}}{\sqrt[4]{d}}$$

holds.

A discussion of Lemma 5.6 can be found in Section 5.11.

#### 5.5 A proof of Theorem 5.1 – Preliminaries

Pick  $d = 0, 1, \ldots$  Theorem 5.1 is concerned with the existence and value of the limit

$$\lim_{t \to \infty} \frac{1}{t+1} \sum_{s=0}^{t} \mathbb{P}\left[D_t(s) = d+1\right].$$

Fix t = 1, 2, ... and s = 0, ..., t. As we recall the definition (5.8) of the degree rv  $D_t(s)$ , we start by asking how can the rv  $D_t(s)$  achieve the value d + 1. To avoid trivial situations of limited interest we assume that t - s + 1 > d + 1 or equivalently, t - s > d. Note that

$$\mathbb{P}\left[D_t(s) = d+1\right] = \mathbb{P}\left[1 + \sum_{r=s+1}^t \mathbf{1}\left[S_r = s\right] = d+1\right]$$
$$= \mathbb{P}\left[\sum_{r=s+1}^t \mathbf{1}\left[S_r = s\right] = d\right].$$
(5.31)

Thus, the event  $D_t(s) = d + 1$  corresponds to the following situation: Amongst the t-s nodes arriving at time  $s + 1, \ldots, t$ , exactly d arrivals attach themselves to node s, while the remaining t-s-d arrivals attach themselves to a node other than node s. This observation naturally leads to considering the set  $\mathcal{P}_d[s+1,t]$  of partitions of  $\{s+1,\ldots,t\}$  into two sets of size d and t-s-d, respectively. Thus,  $\mathcal{P}_d[s+1,t]$  is given by

$$\mathcal{P}_d[s+1,t] = \left\{ (A,B): A, B \subseteq \{s+1,\dots,t\}, \begin{array}{c} |A| = d, |B| = t-s-d, \\ A \cap B = \emptyset \end{array} \right\}$$

With this notation we get

$$\mathbb{P}\left[\sum_{r=s+1}^{t} \mathbf{1}\left[S_r=s\right] = d\right]$$
$$= \sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \mathbb{P}\left[S_r=s, \ s\in A, \ S_r\neq s, \ s\in B\right].$$
(5.32)

For a given pair (A, B) in  $\mathcal{P}_d[s+1, t]$ , a standard preconditioning argument

yields

$$\mathbb{P}\left[S_{r} = s, \ s \in A, \ S_{r} \neq s, \ s \in B\right]$$

$$= \mathbb{E}\left[\mathbb{P}\left[S_{r} = s, \ s \in A, \ S_{r} \neq s, \ s \in B|\xi^{t}\right]\right]$$

$$= \mathbb{E}\left[\prod_{r \in A} \left(\frac{\xi_{s}}{\Xi_{r-1}}\right) \cdot \prod_{r \in B} \left(1 - \frac{\xi_{s}}{\Xi_{r-1}}\right)\right]$$

$$= \mathbb{E}\left[\prod_{r \in A} \left(\frac{\xi}{\xi + \Xi_{r-2}}\right) \cdot \prod_{r \in B} \left(1 - \frac{\xi}{\xi + \Xi_{r-2}}\right)\right]$$
(5.33)

with the following justifications: The relation (5.33) is a consequence of the fact that the rvs  $S_1, \ldots, S_t$  are conditionally mutually independent given  $\xi^t$ . The equality (5.34) takes advantage of the fact that

$$(\xi_s, \Xi_{r-1}, r = s + 1, \dots, t) =_{st} (\xi, \xi + \Xi_{r-2}, r = s + 1, \dots, t)$$

under the enforced i.i.d. assumptions on the fitness variables.

Next, elementary calculations show that

$$\prod_{r \in A} \left( \frac{\xi}{\xi + \Xi_{r-2}} \right) \cdot \prod_{r \in B} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right)$$

$$= \prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right) \cdot \prod_{r \in A} \left( \frac{\frac{\xi}{\xi + \Xi_{r-2}}}{1 - \frac{\xi}{\xi + \Xi_{r-2}}} \right)$$

$$= \prod_{r \in A} \left( \frac{\xi}{\Xi_{r-2}} \right) \cdot \prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right)$$

$$= \xi^d \prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right) \cdot \prod_{r \in A} \left( \frac{1}{\Xi_{r-2}} \right).$$
(5.35)

since |A| = d.

Therefore, we can write

$$\mathbb{P}\left[\sum_{r=s+1}^{t} \mathbf{1}\left[S_{r}=s\right] = d\right]$$

$$= \sum_{(A,B)\in\mathcal{P}_{d}[s+1,t]} \mathbb{E}\left[\xi^{d}\prod_{r=s+1}^{t}\left(1-\frac{\xi}{\xi+\Xi_{r-2}}\right)\cdot\left(\prod_{r\in A}\frac{1}{\Xi_{r-2}}\right)\right]$$

$$= \mathbb{E}\left[\xi^{d}\prod_{r=s+1}^{t}\left(1-\frac{\xi}{\xi+\Xi_{r-2}}\right)\cdot\left(\sum_{(A,B)\in\mathcal{P}_{d}[s+1,t]}\left(\prod_{r\in A}\frac{1}{\Xi_{r-2}}\right)\right)\right]. \quad (5.36)$$

In the next section we present some useful preliminary results that will be used in the main proof.

# 5.6 Useful technical facts

Before giving a proof of Theorem 5.1 we present three useful technical facts to be used in the course of the discussion.

## 5.6.1 A consequence of the Strong Law of Large Numbers

We begin with an easy consequence of the Strong Law of Large Numbers.

**Lemma 5.7.** Assume  $\mathbb{E}[\xi] = 1$ . For arbitrary  $\alpha$  and  $\varepsilon$  in the unit interval (0, 1) we have

$$\lim_{t \to \infty} \mathbb{P}\left[B_t(\alpha, \varepsilon)\right] = 1 \tag{5.37}$$

where

$$B_t(\alpha,\varepsilon) = \bigcap_{r=\lceil \alpha t \rceil}^t \left[ \left| \frac{\Xi_r}{r+1} - 1 \right| \le \varepsilon \right], \quad t = 0, 1, \dots$$
 (5.38)

**Proof.** Consider the event

$$C = \left[\lim_{t \to \infty} \frac{\Xi_t}{t+1} = 1\right],$$

and for every  $\varepsilon$  in (0,1) note the inclusion

$$C \subset \bigcup_{t=0}^{\infty} \cap_{r=t}^{\infty} \left[ \left| \frac{\Xi_r}{r+1} - 1 \right| \leq \varepsilon \right].$$

Therefore,

$$\mathbb{P}\left[C\right] \leqslant \mathbb{P}\left[\bigcup_{t=0}^{\infty} \bigcap_{r=t}^{\infty} \left[\left|\frac{\Xi_r}{r+1} - 1\right| \leqslant \varepsilon\right]\right] = \lim_{t \to \infty} \mathbb{P}\left[\bigcap_{r=t}^{\infty} \left[\left|\frac{\Xi_r}{r+1} - 1\right| \leqslant \varepsilon\right]\right]$$

by the usual monotonicity argument. By the Strong Law of Large Numbers we have  $\mathbb{P}[C] = 1$ , whence

$$\lim_{t \to \infty} \mathbb{P}\left[ \bigcap_{r=t}^{\infty} \left[ \left| \frac{\Xi_r}{r+1} - 1 \right| \leqslant \varepsilon \right] \right] = 1.$$

Applying this fact along the subsequence  $t \to \lceil \alpha t \rceil$ , we get

$$\lim_{t \to \infty} \mathbb{P}\left[ \bigcap_{r=\lceil \alpha t \rceil}^{\infty} \left[ \left| \frac{\Xi_r}{r+1} - 1 \right| \leqslant \varepsilon \right] \right] = 1$$

and the desired result follows by virtue of the inclusion

$$\bigcap_{r=\lceil \alpha t \rceil}^{\infty} \left[ \left| \frac{\Xi_r}{r+1} - 1 \right| \leq \varepsilon \right] \subseteq B_t(\alpha, \varepsilon), \quad t = 0, 1, \dots$$

## 5.6.2 Uniform selection

Consider  $\alpha$  in the unit interval (0, 1). For each t = 1, 2, ..., let  $\nu_{\alpha,t}$  denote a rv which is uniformly distributed over the set  $\{ [\alpha t], [\alpha t] + 1, ..., t \}$ . The following fact is elementary and given here for easy reference.

**Lemma 5.8.** For each  $\alpha$  in (0, 1), it holds

$$\frac{\nu_{\alpha,t}}{t} \Longrightarrow_t U_\alpha \tag{5.39}$$

where the rv  $U_{\alpha}$  is uniformly distributed over the interval  $(\alpha, 1)$ .

**Proof.** Fix  $\alpha$  in (0,1). With b > 0, elementary facts concerning geometric series yield

$$\mathbb{E}\left[e^{-b\frac{\nu_{\alpha,t}}{t}}\right] = \frac{1}{t - \lceil \alpha t \rceil + 1} \sum_{r = \lceil \alpha t \rceil}^{t} e^{-\frac{br}{t}} = \frac{1}{t - \lceil \alpha t \rceil + 1} \cdot \frac{e^{-\frac{b\lceil \alpha t \rceil}{t}} - e^{-\frac{b(t+1)}{t}}}{1 - e^{-\frac{b}{t}}}$$

Therefore, since

$$\lim_{t \to \infty} \left( 1 - e^{-\frac{b}{t}} \right) \cdot \frac{t}{b} = 1 \quad \text{and} \quad \lim_{t \to \infty} \frac{t}{t - \left[\alpha t\right] + 1} = \left( 1 - \alpha \right)^{-1},$$

it is easy to check that

$$\lim_{t \to \infty} \mathbb{E}\left[e^{-b\frac{\nu_{\alpha,t}}{t}}\right] = \frac{e^{-b\alpha} - e^{-b}}{b(1-\alpha)}.$$
(5.40)

The desired conclusion (5.39) follows by standard arguments upon noting that the right handside of (5.40) is the Laplace transform of  $U_{\alpha}$ .

#### 5.6.3 Limits of certain expectations

Pick  $\alpha$  in the unit interval (0, 1), and take  $\lambda, \gamma > 0$ . To simplify the presentation here and elsewhere, we write

$$J(d;\alpha,\lambda,\gamma) = \frac{\xi^d}{d!} \cdot U_{\alpha}^{\lambda\xi} \left( \log\left(\frac{1}{U_{\alpha}}\right) \right)^d \mathbf{1} \left[ \xi \leqslant \gamma \right], \quad d = 0, 1, \dots$$

with the rv  $U_{\alpha}$  being uniformly distributed on the open interval  $(\alpha, 1)$  and independent of  $\xi$ . The expected values

$$T(d;\alpha,\lambda,\gamma) = \mathbb{E}\left[J(d;\alpha,\lambda,\gamma)\right], \quad d = 0,1,\dots$$
(5.41)

will play a crucial role in the proof of Theorem 5.1.

Fix  $d = 0, 1, \ldots$  Note that  $J(d; \alpha, \lambda, \gamma) \ge 0$  so that the expected value at (5.41) is always well defined, possibly infinite. However, the bounds  $\alpha < U_{\alpha} < 1$ imply

$$0 \leq U_{\alpha}^{\lambda\xi} \left( \log \left( \frac{1}{U_{\alpha}} \right) \right)^{d} \leq \left( \log \alpha^{-1} \right)^{d}.$$
(5.42)

while it is plain that

$$0 \leqslant \frac{\xi^d}{d!} \mathbf{1} \left[ \xi \leqslant \gamma \right] \leqslant \frac{\gamma^d}{d!} \mathbf{1} \left[ \xi \leqslant \gamma \right].$$
(5.43)

Combining these bounds we conclude that

$$0 \leqslant J(d; \alpha, \lambda, \gamma) \leqslant \frac{\gamma^d}{d!} \left( \log \alpha^{-1} \right)^d, \tag{5.44}$$

and  $T(d; \alpha, \lambda, \gamma)$  is in fact finite with

$$0 \leqslant T(d; \alpha, \lambda, \gamma) \leqslant \frac{\gamma^d}{d!} \left(\log \alpha^{-1}\right)^d.$$
(5.45)

**Lemma 5.9.** Fix  $d = 0, 1, \ldots$  With  $\alpha$  in the unit interval (0, 1), and  $\lambda, \gamma > 0$ , it holds that

$$\lim_{\lambda \to 1} T(d; \alpha, \lambda, \gamma) = T(d; \alpha, 1, \gamma),$$
(5.46)

$$\lim_{\gamma \to \infty} T(d; \alpha, 1, \gamma) = \mathbb{E}\left[\frac{\xi^d}{d!} \cdot U^{\xi}_{\alpha} \left(\log\left(\frac{1}{U_{\alpha}}\right)\right)^d\right]$$
(5.47)

and

$$\lim_{\alpha \downarrow 0} \mathbb{E}\left[\frac{\xi^d}{d!} \cdot U^{\xi}_{\alpha} \left(\log\left(\frac{1}{U_{\alpha}}\right)\right)^d\right] = p_{\xi}(d+1).$$
(5.48)

**Proof.** Fix  $d = 0, 1, \ldots$  The deterministic bound (5.44) being uniform in  $\lambda$ , the conclusion (5.46) is immediate by the Bounded Convergence Theorem. For each  $\alpha$  in (0, 1) the validity of (5.47) follows by the Monotone Convergence Theorem because  $T(d; \alpha, \lambda, \gamma)$  is non-negative for each  $\gamma \ge 0$  and is non-decreasing as  $\gamma \to \infty$ .

We finally turn to (5.48): For each  $\alpha$  in (0, 1) note that

$$\frac{\xi^d}{d!} \cdot U^{\xi}_{\alpha} \left( \log\left(\frac{1}{U_{\alpha}}\right) \right)^d = \frac{1}{d!} \left( \xi \log\left(\frac{1}{U_{\alpha}}\right) \right)^d e^{-\xi \log\left(\frac{1}{U_{\alpha}}\right)} \leqslant 1$$
(5.49)

since this term corresponds to a Poisson pmf with parameter  $\xi \log \left(\frac{1}{U_{\alpha}}\right)$  evaluated at d. Obviously, we have  $U_{\alpha} \Longrightarrow_{\alpha} U$  (when  $\alpha$  is driven to zero), and the Continuous Mapping Theorem yields

$$\frac{\xi^d}{d!} \cdot U^{\xi}_{\alpha} \left( \log \left( \frac{1}{U_{\alpha}} \right) \right)^d \Longrightarrow_{\alpha} \frac{\xi^d}{d!} \cdot U^{\xi} \left( \log \left( \frac{1}{U} \right) \right)^d$$

The convergence (5.48) is now a consequence of the Bounded Convergence Theorem by virtue of the bound (5.49) (which is uniform in  $\alpha$ ).

In particular, it follows that

$$\lim_{\alpha \downarrow 0} \left( \lim_{\gamma \to \infty} \left( \lim_{\lambda \to 1} T(d; \alpha, \lambda, \gamma) \right) \right) \\
= \lim_{\alpha \downarrow 0} \left( \lim_{\gamma \to \infty} T(d; \alpha, 1, \gamma) \right) \\
= \lim_{\alpha \downarrow 0} \left( \mathbb{E} \left[ \frac{\xi^d}{d!} \cdot U^{\xi}_{\alpha} \left( \log \left( \frac{1}{U_{\alpha}} \right) \right)^d \right] \right) \\
= p_{\xi}(d+1), \quad d = 0, 1, \dots$$
(5.50)

#### 5.7 A proof of Theorem 5.1

We shall establish Theorem 5.1 in the equivalent form given in Theorem 5.2. To do so, fix d = 0, 1, ... Under the assumptions of Theorem 5.1 we shall now show that the convergence statements

$$\limsup_{t \to \infty} \mathbb{P}\left[D_t(\nu_t) = d + 1\right] \le p_{\xi}(d + 1)$$
(5.51)

and

$$p_{\xi}(d+1) \leq \liminf_{t \to \infty} \mathbb{P}\left[D_t(\nu_t) = d+1\right]$$
(5.52)

hold where for each t = 0, 1, ..., the rv  $\nu_t$  is uniformly distributed over the edge set  $V_t$  and independent of  $\mathcal{F}_t$ , thus of  $\mathbb{G}_t$ .

Pick  $\alpha$  and  $\varepsilon$  in the unit interval (0, 1), and  $\gamma > 0$ . With  $t = 1, 2, \ldots$  sufficiently large so that  $t - \lceil \alpha t \rceil \ge d$ , consider the event

$$B_t(\alpha, \varepsilon; \gamma) = B_t(\alpha, \varepsilon) \cap [\xi \leq \gamma].$$

with  $B_t(\alpha, \varepsilon)$  defined at (5.38).

Keeping (5.13) in mind, consider the decomposition

$$\mathbb{E}\left[N_t(d+1)\right] = \mathbb{E}\left[\sum_{s=0}^t \mathbf{1}\left[D_t(s) = d+1\right]\right]$$
$$= \sum_{s=0}^{\lceil \alpha t \rceil - 1} \mathbb{P}\left[D_t(s) = d+1\right] + \sum_{s=\lceil \alpha t \rceil}^t \mathbb{P}\left[D_t(s) = d+1\right]$$
$$= \sum_{s=0}^{\lceil \alpha t \rceil - 1} \mathbb{P}\left[D_t(s) = d+1\right] + T_t(d; \alpha, \varepsilon, \gamma)$$
$$+ \sum_{s=\lceil \alpha t \rceil}^t \mathbb{P}\left[\left[D_t(s) = d+1\right] \cap B_t(\alpha, \varepsilon; \gamma)^c\right]$$
(5.53)

where we have set

$$T_t(d; \alpha, \varepsilon, \gamma) = \sum_{s=\lceil \alpha t \rceil}^t \mathbb{P}\left[ [D_t(s) = d+1] \cap B_t(\alpha, \varepsilon; \gamma) \right]$$
(5.54)

for notational convenience

#### 5.7.1 Establishing the upper bound (5.52)

The first and third terms in (5.53) are easily upper bounded. Specifically, we get

$$\sum_{s=0}^{\lceil \alpha t \rceil - 1} \mathbb{P}\left[D_t(s) = d + 1\right] \leqslant \lceil \alpha t \rceil$$
(5.55)

and

$$\sum_{s=\lceil \alpha t \rceil}^{t} \mathbb{P}\left[\left[D_{t}(s) = d+1\right] \cap B_{t}(\alpha, \varepsilon; \gamma)^{c}\right] \leq (t - \lceil \alpha t \rceil + 1) \mathbb{P}\left[B_{t}(\alpha, \varepsilon; \gamma)^{c}\right]$$
$$\leq (t - \lceil \alpha t \rceil + 1) \left(\mathbb{P}\left[B_{t}(\alpha, \varepsilon)^{c}\right] + \mathbb{P}\left[\gamma < \xi\right]\right)$$
$$\leq (t + 1) \left(\mathbb{P}\left[B_{t}(\alpha, \varepsilon)^{c}\right] + \mathbb{P}\left[\gamma < \xi\right]\right). (5.56)$$

Therefore,

$$\limsup_{t \to \infty} \left( \frac{1}{t+1} \sum_{s=0}^{\lceil \alpha t \rceil - 1} \mathbb{P}\left[ D_t(s) = d + 1 \right] \right) \leq \alpha$$

and

$$\limsup_{t \to \infty} \left( \frac{1}{t+1} \sum_{s=\lceil \alpha t \rceil}^{t} \mathbb{P}\left[ [D_t(s) = d+1] \cap B_t(\alpha, \varepsilon; \gamma)^c \right] \right)$$
  
$$\leqslant \mathbb{P}\left[ \gamma < \xi \right] + \limsup_{t \to \infty} \mathbb{P}\left[ B_t(\alpha, \varepsilon)^c \right]$$
  
$$= \mathbb{P}\left[ \gamma < \xi \right]$$
(5.57)

as we invoke Lemma 5.7. We readily conclude from (5.13) that

$$\limsup_{t \to \infty} \mathbb{P}\left[D_t(\nu_t) = d+1\right] \le \alpha + \limsup_{t \to \infty} \left(\frac{T_t(d; \alpha, \varepsilon, \gamma)}{t+1}\right) + \mathbb{P}\left[\gamma < \xi\right].$$
(5.58)

Most of the technical work that remains consists in identifying the limiting term in this last inequality; a proof can be found in Section 5.8. **Proposition 5.10.** Under the assumptions of Theorem 5.1, we have

$$\limsup_{t \to \infty} \left( \frac{T_t(d; \alpha, \varepsilon, \gamma)}{t + 1 - [\alpha t]} \right) \le (1 - \varepsilon)^{-d} \cdot T\left(d; \alpha, (1 + \varepsilon)^{-1}, \gamma\right)$$
(5.59)

Collecting (5.58) and (5.59) we find that

$$\limsup_{t \to \infty} \mathbb{P}\left[D_t(\nu_t) = d + 1\right]$$
  
$$\leq \alpha + (1 - \alpha)(1 - \varepsilon)^{-d} \cdot T\left(d; \alpha, (1 + \varepsilon)^{-1}, \gamma\right) + \mathbb{P}\left[\gamma < \xi\right]$$
(5.60)

The left handside does not depend on either of the parameters  $\alpha$ ,  $\varepsilon$  or  $\gamma$ . Therefore, in (5.60) let  $\varepsilon$  go to zero,  $\gamma$  go to infinity and  $\alpha$  go to zero in that order, and Lemma 5.9 leads to (5.51).

## 5.7.2 Establishing the lower bound (5.51)

This time, neglecting the first and last terms in (5.53) (which are non-negative), we get

$$\liminf_{t \to \infty} \left( \frac{T_t(d; \alpha, \varepsilon, \gamma)}{t+1} \right) \leq \liminf_{t \to \infty} \mathbb{P}\left[ D_t(\nu_t) = d+1 \right]$$
(5.61)

by arguments similar to those used earlier for deriving the upper bound. This time we need to show the following analog of Proposition 5.10; a proof is available in Section 5.9.

Proposition 5.11. Under the assumptions of Theorem 5.1, we have

$$\liminf_{t \to \infty} \left( \frac{T_t(d; \alpha, \varepsilon, \gamma)}{t + 1 - \lceil \alpha t \rceil} \right) \ge (1 + \varepsilon)^{-d} \cdot T\left(d; \alpha, (1 - \varepsilon)^{-1}, \gamma\right)$$
(5.62)

It now follows from (5.61) and (5.62) that

$$(1-\alpha)(1+\varepsilon)^{-d} \cdot T\left(d; \alpha, (1-\varepsilon)^{-1}, \gamma\right) \leq \liminf_{t \to \infty} \mathbb{P}\left[D_t(\nu_t) = d+1\right]$$
(5.63)

Here as well the left handside does not depend on either of the parameters  $\alpha$ ,  $\varepsilon$  or  $\gamma$ . Therefore, in (5.63) let  $\varepsilon$  go to zero,  $\gamma$  go to infinity and  $\alpha$  go to zero in that order. The validity of (5.52) is now a straightforward consequence of Lemma 5.9.

### 5.8 A proof of Proposition 5.10

Fix  $d = 0, 1, \ldots$  Fix  $\alpha$  and  $\epsilon$  in the unit interval (0, 1) and  $\gamma > 0$ . For ease of exposition we recollect some notation: For each  $d = 0, 1, \ldots, t = 1, 2, \ldots, \alpha$  and  $\epsilon$  in (0, 1) and  $\gamma > 0$ , recall that we had set

$$T_t(d; \alpha, \varepsilon, \gamma) = \sum_{s=\lceil \alpha t \rceil}^t \mathbb{P}\left[ \left[ D_t(s) = d + 1 \right] \cap B_t(\alpha, \varepsilon; \gamma) \right]$$

where the event

$$B_t(\alpha,\epsilon;\gamma) = B_t(\alpha,\epsilon) \cap [\xi \leqslant \gamma]$$

with

$$B_t(\alpha,\varepsilon) = \bigcap_{r=\lceil \alpha t \rceil}^t \left[ \left| \frac{\Xi_r}{r+1} - 1 \right| \le \varepsilon \right].$$

Fix  $t = 1, 2, \ldots$  and  $s = \lceil \alpha t \rceil, \ldots, t$ . From (5.36), we have

$$\mathbb{P}\left[D_t(s) = d+1\right] = \mathbb{E}\left[\xi^d \prod_{r=s+1}^t \left(1 - \frac{\xi}{\xi + \Xi_{r-2}}\right) \cdot \left(\sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \left(\prod_{r\in A} \frac{1}{\Xi_{r-2}}\right)\right)\right].$$
(5.64)

Under the event  $B_t(\alpha, \epsilon; \gamma)$ , we have the following upper bound

$$\prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right) \leq \prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\gamma + (r-2)(1+\epsilon)} \right)$$
$$\leq \prod_{r=s+1}^{t} e^{-\frac{\xi}{\gamma + (r-2)(1+\epsilon)}}$$
$$= e^{-\frac{\xi}{1+\epsilon} \sum_{r=s}^{t-1} \frac{\gamma}{1+\epsilon} + (r-1)}.$$
(5.65)

By observing that

$$\sum_{r=s}^{t-1} \frac{1}{\frac{\gamma}{1+\epsilon} + (r-1)} \ge \int_s^t \frac{1}{\frac{\gamma}{1+\epsilon} + (p-1)} dp = \log\left(\frac{t-1+\frac{\gamma}{1+\epsilon}}{s-1+\frac{\gamma}{1+\epsilon}}\right)$$

we refine the bound (5.65) as follows

$$\prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right) \leqslant e^{-\frac{\xi}{(1+\epsilon)} \log\left(\frac{t-1+\frac{\gamma}{1+\epsilon}}{s-1+\frac{\gamma}{1+\epsilon}}\right)} = \left(\frac{s-1+\frac{\gamma}{1+\epsilon}}{t-1+\frac{\gamma}{1+\epsilon}}\right)^{\frac{\xi}{(1+\epsilon)}}.$$
(5.66)

Upper bounding the final term in (5.64) on the set  $B_t(\alpha, \epsilon; \gamma)$ , we get

$$\sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \left(\prod_{r\in A} \frac{1}{\Xi_{r-2}}\right) \leqslant \sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \left(\prod_{r\in A} \frac{1}{(1-\epsilon)(r-2)}\right)$$
$$= \frac{1}{(1-\epsilon)^d} \sum_{(A,B)\in\mathcal{P}_d[s,t-1]} \left(\prod_{r\in A} \frac{1}{r-1}\right)$$
$$\leqslant \frac{1}{(1-\epsilon)^d} \cdot \frac{1}{d!} \left(\sum_{r=s}^{t-1} \frac{1}{r-1}\right)^d$$
$$\leqslant \frac{1}{(1-\epsilon)^d} \cdot \frac{1}{d!} \left(\int_s^t \frac{1}{p-1} dp\right)^d$$
$$= \frac{1}{(1-\epsilon)^d} \cdot \frac{1}{d!} \left(\log\left(\frac{t-1}{s-1}\right)\right)^d.$$
(5.67)

Substituting the upper bounds (5.66) and (5.67) in (5.64), we obtain

$$\mathbb{P}\left[\left[D_t(s) = d+1\right] \cap B_t(\alpha, \epsilon; \gamma)\right]$$

$$\leqslant \frac{1}{(1-\epsilon)^d} \mathbb{E}\left[\frac{\xi^d}{d!} \left(\frac{s-1+\frac{\gamma}{1+\epsilon}}{t-1+\frac{\gamma}{1+\epsilon}}\right)^{\frac{\xi}{(1+\epsilon)}} \left(\log\left(\frac{t-1}{s-1}\right)\right)^d \mathbf{1}\left[\xi < \gamma\right]\right]. \quad (5.68)$$

For  $c_0 > 1$ ,  $t > \frac{c_0}{\alpha}$  and  $s = \lceil \alpha t \rceil, \ldots, t$ , we have the upper bound

$$\frac{\xi^{d}}{d!} \left( \frac{s-1+\frac{\gamma}{1+\epsilon}}{t-1+\frac{\gamma}{1+\epsilon}} \right)^{\frac{\xi}{(1+\epsilon)}} \left( \log\left(\frac{t-1}{s-1}\right) \right)^{d} \mathbf{1} \left[ \xi < \gamma \right] \leq \frac{\gamma^{d}}{d!} \left( \log\left(\frac{t-1}{\alpha t-1}\right) \right)^{d} = \frac{\gamma^{d}}{d!} \left( \log\left(\frac{1-\frac{1}{t}}{\alpha-\frac{1}{t}}\right) \right)^{d} \leq \frac{\gamma^{d}}{d!} \left( \log\left(\frac{1-\frac{1}{t}}{\alpha-\frac{1}{c_{0}}}\right) \right)^{d}.$$
(5.69)

Substituting (5.68) in the expression for  $T_t(d; \alpha, \epsilon, \gamma)$ , we obtain

$$T_t(d;\alpha,\epsilon,\gamma) \leqslant \frac{1}{(1-\epsilon)^d} \mathbb{E}\left[\frac{\xi^d}{d!} \left(\frac{\nu_{\alpha,t}-1+\frac{\gamma}{1+\epsilon}}{t-1+\frac{\gamma}{1+\epsilon}}\right)^{\frac{\xi}{(1+\epsilon)}} \left(\log\left(\frac{t-1}{\nu_{\alpha,t}-1}\right)\right)^d \mathbf{1}\left[\xi < \gamma\right]\right].$$
(5.70)

Allowing t to go to infinity in (5.70), we get the desired result using the Bounded Convergence theorem (by virtue of the upper bound (5.69)) and Lemma 5.8.

## 5.9 A proof of Proposition 5.11

Fix  $d = 0, 1, \ldots$  Fix  $\alpha$  and  $\epsilon$  in the unit interval (0, 1) and  $\gamma > 0$ . For each  $t = 1, 2, \ldots$  and  $s = [\alpha t], \ldots, t$ , we have

$$\mathbb{P}\left[D_t(s) = d+1\right] = \mathbb{E}\left[\xi^d \prod_{r=s+1}^t \left(1 - \frac{\xi}{\xi + \Xi_{r-2}}\right) \cdot \left(\sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \left(\prod_{r\in A} \frac{1}{\Xi_{r-2}}\right)\right)\right].$$
(5.71)

Under the event  $B_t(\alpha, \epsilon; \gamma)$ , we have the lower bound

$$\prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right) \ge \prod_{r=s}^{t-1} \left( 1 - \frac{\xi}{\xi + (1 - \epsilon)(r - 1)} \right)$$
$$= \prod_{r=s}^{t-1} e^{-\frac{\xi}{\xi + (1 - \epsilon)(r - 1)} - \Psi\left(\frac{\xi}{\xi + (1 - \epsilon)(r - 1)}\right)}$$
$$= e^{-\sum_{r=s}^{t-1} \frac{\xi}{\xi + (1 - \epsilon)(r - 1)} - \sum_{r=s}^{t-1} \Psi\left(\frac{\xi}{\xi + (1 - \epsilon)(r - 1)}\right)}$$
$$\ge e^{-\frac{\xi}{1 - \epsilon} \sum_{r=s}^{t-1} \frac{1}{r-1} - \sum_{r=s}^{t-1} \Psi\left(\frac{\xi}{\xi + (1 - \epsilon)(r - 1)}\right)}$$
(5.72)

where we have set

$$\Psi(x) = \int_0^x \frac{t}{1-t} dt, \ 0 \le x < 1.$$

The first term in (5.72) can be bounded as follows

$$\sum_{r=s}^{t-1} \frac{1}{r-1} \leqslant \int_{s}^{t-1} \frac{1}{r-2} = \log\left(\frac{t-2}{s-2}\right).$$
(5.73)

Before bounding the second term in (5.72) we note that for  $0 \le x \le y < 1$ ,

$$\frac{1}{2} \leqslant \frac{\Psi(x)}{x^2} \leqslant \frac{\Psi(y)}{y^2} < \infty.$$
(5.74)

In view of the above fact for sufficiently large t we would like to uniformly bound the sequence of rvs

$$\left\{\frac{\Psi\left(\frac{\xi}{\xi+(1-\epsilon)(r-1)}\right)}{\left(\frac{\xi}{\xi+(1-\epsilon)(r-1)}\right)^2}, \ r=\left\lceil\alpha t\right\rceil, \dots, t\right\}.$$

Since  $\frac{\xi}{\xi+(1-\epsilon)(r-1)}$  is monotonically increasing in  $\xi$  and decreasing in r, on the set  $[\xi < \gamma]$  it is sufficient to ensure  $\frac{\gamma}{\gamma+(1-\epsilon)(\alpha t-1)}$  is strictly less than one for all t being considered. Under the condition  $t > \frac{\gamma}{(1-\epsilon)\alpha} + 1$ ,

$$\frac{\gamma}{\gamma + (1 - \epsilon)(\alpha t - 1)} < \frac{1}{2}$$

which implies the uniform bound

$$\frac{\Psi\left(\frac{\xi}{\xi+(1-\epsilon)(r-1)}\right)}{\left(\frac{\xi}{\xi+(1-\epsilon)(r-1)}\right)^2} < \frac{\Psi\left(\frac{1}{2}\right)}{\left(\frac{1}{2}\right)^2}, \qquad r = \lceil \alpha t \rceil, \dots, t$$

$$t > \frac{\gamma}{(1-\epsilon)\alpha}$$
(5.75)

by virtue of the fact (5.74). Using the bound (5.75), on the set  $[\xi < \gamma]$  we have the upper bound

$$\begin{split} \sum_{r=s}^{t-1} \Psi\left(\frac{\xi}{\xi + (1-\epsilon)(r-1)}\right) &= \sum_{r=s}^{t-1} \frac{\Psi\left(\frac{\xi}{\xi + (1-\epsilon)r}\right)}{\left(\frac{\xi}{\xi + (1-\epsilon)(r-1)}\right)^2} \left(\frac{\xi}{\xi + (1-\epsilon)(r-1)}\right)^2 \\ &\leq \frac{\Psi\left(\frac{1}{2}\right)}{\left(\frac{1}{2}\right)^2} \sum_{r=s}^{t-1} \left(\frac{\xi}{\xi + (1-\epsilon)(r-1)}\right)^2 \\ &\leq 4\Psi\left(\frac{1}{2}\right) \frac{\gamma^2}{(1-\epsilon)^2} \sum_{r=s}^{t-1} \frac{1}{(r-1)^2} \\ &\leq 4\Psi\left(\frac{1}{2}\right) \frac{\gamma^2}{(1-\epsilon)^2} \int_s^{t-1} \frac{1}{p^2} dp \\ &\leq 4\Psi\left(\frac{1}{2}\right) \frac{\gamma^2}{(1-\epsilon)^2} \frac{1-\alpha}{\alpha t}, \end{split}$$
(5.76)

where the last step follows by noting that  $s \ge \alpha t$ . The bounds (5.73) and (5.76) when substituted in (5.72) yields the lower bound

$$\prod_{r=s+1}^{t} \left( 1 - \frac{\xi}{\xi + \Xi_{r-2}} \right) \ge e^{-4\Psi\left(\frac{1}{2}\right)\frac{\gamma^2}{(1-\epsilon)^2}\frac{1-\alpha}{\alpha t}} \mathbb{E}\left[ \left(\frac{s-2}{t-2}\right)^{\frac{\xi}{(1-\epsilon)}} \right]$$
(5.77)

on the set  $B_t(\alpha, \epsilon; \gamma)$ . The final term inside the expectation in (5.71) can be lower bounded as

$$\sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \left(\prod_{r\in A} \frac{1}{\Xi_{r-2}}\right) \ge \sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \left(\prod_{r\in A} \frac{1}{(1+\epsilon)(r-2)}\right)$$
$$= \frac{1}{(1+\epsilon)^d} \left(\sum_{(A,B)\in\mathcal{P}_d[s,t-1]} \left(\prod_{r\in A} \frac{1}{r-1}\right)\right)$$
(5.78)

on the set  $B_t(\alpha, \epsilon; \gamma)$ . Before we proceed further, we note that

$$\left(\sum_{r=s}^{t-1} \frac{1}{r-1}\right)^d = d! \left(\sum_{(A,B)\in\mathcal{P}_d[s,t-1]} \left(\prod_{r\in A} \frac{1}{r-1}\right)\right) + R(s,t)$$
(5.79)

where R(s,t) is an error term that we would like to bound. Observe that the number of terms in R(s,t) is  $((t-1-s)^d - {t-1-s \choose d}d!)$  where each term is at most  $\frac{1}{s^d}$ . Therefore for  $s = [\alpha t], \ldots, t$ , we have the upper bound

$$R(s,t) \leq \left[ (t-1-s)^d - d! \binom{t-1-s}{d} \right] \frac{1}{(\alpha t)^d}$$
$$\leq \frac{1}{\alpha^d} \left( \frac{(t-1-s)^d - d! \binom{t-1-s}{d}}{t^d} \right).$$
(5.80)

Also, we have the lower bound

$$\left(\sum_{r=s}^{t-1} \frac{1}{r-1}\right)^{d} \ge \left(\int_{s}^{t} \frac{1}{p-1} dp\right)^{d}$$
$$\ge \left(\log\left(\frac{t-1}{s-1}\right)\right)^{d}$$
(5.81)

Substituting the bounds (5.80) and (5.81) in (5.78), we obtain

$$\sum_{(A,B)\in\mathcal{P}_d[s+1,t]} \left(\prod_{r\in A} \frac{1}{\Xi_{r-2}}\right)$$
  
$$\geqslant \frac{1}{(1+\epsilon)^d} \cdot \frac{1}{d!} \left[ \left( \log\left(\frac{t-1}{s-1}\right) \right)^d - \frac{1}{\alpha^d} \left(\frac{(t-1-s)^d - d!\binom{t-1-s}{d}}{t^d} \right) \right] \quad (5.82)$$

using the expression (5.79) on the set  $B_t(\alpha, \epsilon; \gamma)$ .

The lower bounds (5.77) and (5.82) when substituted in (5.71) yields

$$\mathbb{P}\left[\left[D_{t}(s)=d+1\right]\cap B_{t}(\alpha,\epsilon;\gamma)\right] \geq \frac{1}{(1+\epsilon)^{d}}e^{-4\Psi\left(\frac{1}{2}\right)\frac{\gamma^{2}}{(1-\epsilon)^{2}}\frac{1-\alpha}{\alpha t}}$$
$$\times \mathbb{E}\left[\frac{\xi^{d}}{d!}\left(\frac{s-2}{t-2}\right)^{\frac{\xi}{(1-\epsilon)}}\left[\left(\log\left(\frac{t-1}{s-1}\right)\right)^{d}-\frac{1}{\alpha^{d}}\left(\frac{(t-1-s)^{d}-d!\binom{t-1-s}{d}}{t^{d}}\right)\right]\mathbf{1}\left[\xi<\gamma\right]\right]$$
(5.83)

Substituting the bound (5.83) in the expression of  $T_t(d; \alpha, \epsilon, \gamma)$  and using Lemma 5.8 and the following limit

$$\left(\frac{(t-1-\nu_{\alpha,t})^d-d!\binom{t-1-\nu_{\alpha,t}}{d}}{t^d}\right) \longrightarrow_t 0, \ d=0,1,\ldots.$$

we get the desired result by virtue of the Bounded Convergence theorem.

#### 5.10 A proof of Lemma 5.4

Let  $\mu$  denote the probability measure on  $(\mathbb{R}_+, \mathcal{B}(\mathbb{R}_+))$  induced by the rv  $\xi$ , i.e.,

$$\mu(B) = \mathbb{P}\left[\xi \in B\right], \quad B \in \mathcal{B}(\mathbb{R}_+).$$

With  $\mu$  we associate another measure  $\nu$  on  $(\mathbb{R}_+, \mathcal{B}(\mathbb{R}_+))$  which is absolutely continuous with respect to  $\mu$ , its Radon-Nikodym derivative being given by

$$\frac{d\nu}{d\mu}(x) = \frac{\mathbb{E}\left[\xi\right]}{\mathbb{E}\left[\xi\right] + x}, \quad x \ge 0.$$

The measure  $\nu$  is finite with

$$0 < \nu(\mathbb{R}_+) = \int_0^\infty \left(\frac{\mathbb{E}\left[\xi\right]}{\mathbb{E}\left[\xi\right] + x}\right) d\mu(x) \le 1.$$

Thus, while  $\mu$  is a probability measure, the positive measure  $\nu$  will be a subprobability measure on  $(\mathbb{R}_+, \mathcal{B}(\mathbb{R}_+))$ . The measures  $\mu$  and  $\nu$  are mutually absolutely continuous, so that with B in  $\mathcal{B}(\mathbb{R}_+)$  we have  $\mu(B) = 0$  if and only if  $\nu(B) = 0$ .

Fix  $d = 0, 1, \ldots$  We note that

$$p_{\xi}(d+1) = \mathbb{E}\left[\left(1-R(\xi)\right)R(\xi)^{d}\right]$$
$$= \int_{0}^{\infty} \frac{\mathbb{E}\left[\xi\right]}{\mathbb{E}\left[\xi\right]+x} \cdot \left(\frac{x}{\mathbb{E}\left[\xi\right]+x}\right)^{d} d\mu(x)$$
$$= \int_{0}^{\infty} g_{\xi}(x)^{d} d\nu(x)$$
(5.84)

We can now rewrite (5.84) more compactly as

$$\left(p_{\xi}(d+1)\right)^{\frac{1}{d}} = \|g_{\xi}\|_{L^{d}(\mathbb{R}_{+};\nu)}$$
(5.85)

where  $\|\cdot\|_{L^d(\mathbb{R}_+;\nu)}$  denotes the usual prenorm on the linear space  $L^d(\nu, \mathbb{R}_+)$  of all Borel measurable functions  $\mathbb{R}_+ \to \mathbb{R}$  which are *d*-integrable with respect to  $\nu$ . Standard arguments based on Hölder's inequality can be used to show the convexity of the mapping

$$\mathbb{R}_+ \to \mathbb{R} : a \to \log\left(\frac{\int_0^\infty |g_{\xi}(x)|^a d\nu(x)}{\nu(\mathbb{R}_+)}\right),$$

and the mapping

$$(0,\infty) \to \mathbb{R} : a \to \frac{1}{a} \log\left(\frac{\int_0^\infty |g_{\xi}(x)|^a d\nu(x)}{\nu(\mathbb{R}_+)}\right)$$

is therefore non-decreasing. In particular for 0 < a < b we have

$$\left(\frac{\int_0^\infty |g_{\xi}(x)|^a d\nu(x)}{\nu(\mathbb{R}_+)}\right)^{\frac{1}{a}} \leqslant \left(\frac{\int_0^\infty |g_{\xi}(x)|^b d\nu(x)}{\nu(\mathbb{R}_+)}\right)^{\frac{1}{b}},$$

or equivalently,

$$\frac{\left(\int_{0}^{\infty}|g_{\xi}(x)|^{a}d\nu(x)\right)^{\frac{1}{a}}}{\left(\int_{0}^{\infty}|g_{\xi}(x)|^{b}d\nu(x)\right)^{\frac{1}{b}}} \leqslant \nu(\mathbb{R}_{+})^{\frac{1}{a}-\frac{1}{b}}.$$

Because  $\nu(\mathbb{R}_+) \leq 1$ , the mapping

$$(0,\infty) \to \mathbb{R}_+ : a \to \left(\int_0^\infty |g_{\xi}(x)|^a d\nu(x)\right)^{\frac{1}{a}}$$

is also non-decreasing, and the limit

$$\lim_{a \to \infty} \left( \int_0^\infty |g_{\xi}(x)|^a d\nu(x) \right)^{\frac{1}{a}}$$

therefore exists. It is easy to see that

$$\lim_{a \to \infty} \left( \int_0^\infty |g_{\xi}(x)|^a d\nu(x) \right)^{\frac{1}{a}} = \|g_{\xi}\|_{L^\infty(\mathbb{R}_+;\nu)}$$

with

$$||g_{\xi}||_{L^{\infty}(\mathbb{R}_{+};\nu)} = \nu - \text{Ess sup}(|g_{\xi}(x)|: x \ge 0)$$
  
=  $\inf (a \in \mathbb{R}_{+}: \nu\{x \ge 0: |g_{\xi}(x)| > a\} = 0)$   
=  $\inf (a \in \mathbb{R}_{+}: \mu\{x \ge 0: |g_{\xi}(x)| > a\} = 0)$  (5.86)

where the last step follows from the fact noted earlier that the measures  $\mu$  and  $\nu$ are mutually absolutely continuous (in which case  $\nu\{x \ge 0 : |g_{\xi}(x)| > a\} = 0$  if and only if  $\mu\{x \ge 0 : |g_{\xi}(x)| > a\} = 0$ . From the definition of  $g_{\xi}$  it is plain that

whence

$$\mu\{x \ge 0: |g_{\xi}(x)| > a\} = \begin{cases} \mathbb{P}\left[\xi > \frac{a}{1-a} \cdot \mathbb{E}\left[\xi\right]\right] & \text{if } 0 \le a < 1\\ \\ 0 & \text{if } 1 \le a. \end{cases}$$
(5.88)

Next, note that the mapping  $g_{\xi}: [0,\infty] \to [0,1]$  admits an inverse mapping  $g_{\xi}^{-1}: [0,1] \to [0,\infty]$  given by

$$g_{\xi}^{-1}(a) = \begin{cases} \frac{a}{1-a} \mathbb{E}\left[\xi\right] & \text{if } 0 \le a < 1 \\\\\\\infty & \text{if } a = 1. \end{cases}$$

Therefore, using the obvious convention

$$\mathbb{P}\left[\xi > \frac{a}{1-a} \cdot \mathbb{E}\left[\xi\right]\right] = 0, \quad \text{if } a = 1,$$

we readily conclude

$$\inf \left(a \in \mathbb{R}_{+} : \mu\{x \ge 0 : |g_{\xi}(x)| > a\} = 0\right)$$
  
= 
$$\inf \left(a \in [0,1] : \mathbb{P}\left[\xi > \frac{a}{1-a} \cdot \mathbb{E}\left[\xi\right]\right] = 0\right)$$
  
= 
$$\inf \left(a \in [0,1] : \mathbb{P}\left[\xi > g_{\xi}^{-1}(a)\right] = 0\right)$$
  
= 
$$g_{\xi} \left(\inf \left(x \in [0,\infty] : \mathbb{P}\left[\xi > x\right] = 0\right)\right) = g_{\xi}(\xi^{\star})$$
(5.89)

as we make use of the definition (5.20) of  $\xi^*$ . The desired conclusion (5.21) readily follows.

## 5.11 A proof of Lemma 5.6

Fix d = 0, 1, ... From (5.30) we have

$$p_{\xi}(d+1) = e \int_0^1 (1-s)^d \frac{e^{-\frac{1}{s}}}{s} ds.$$
 (5.90)

We shall find it helpful to write

$$I_d(s) = (1-s)^d \frac{e^{-\frac{1}{s}}}{s}, \quad s \ge 0$$

with the understanding that  $I_d(0) = 0$  by the usual continuity argument since  $\lim_{s\downarrow 0} e^{-\frac{1}{s}} = 0$  much faster than s. With

$$J_d(s) = \log I_d(s) = d \log(1-s) - \log s - \frac{1}{s}, \quad 0 < s < 1,$$

we have

$$I_d(s) = e^{J_d(s)}, \quad 0 < s < 1.$$

Note that

$$J_d(s)' = -d(1-s)^{-1} - s^{-1} + s^{-2} = -\frac{d}{1-s} + \frac{1-s}{s^2}, \quad 0 < s < 1.$$

Therefore,  $J_d(s)' = 0$  is equivalent to

$$\frac{1-s}{s^2} = \frac{d}{1-s},$$

an equation with a unique solution  $s^{\star}(d)$  in the interval (0,1) given by

$$s^{\star}(d) = \frac{1}{1 + \sqrt{d}}.$$

It is easy to check that  $s \to J_d(s)$  is increasing on the interval  $(0, s^*(d))$  and decreasing on the interval  $(s^*(d), 1)$ .

For future reference we note that

$$1 - s^{\star}(d) = 1 - \frac{1}{1 + \sqrt{d}} = \frac{\sqrt{d}}{1 + \sqrt{d}}$$

and

$$\frac{1}{1-s^{\star}(d)} = \frac{1+\sqrt{d}}{\sqrt{d}}.$$

It is plain that  $s^{\star}(d)$  maximizes  $J_d(s)$ , hence  $I_d(s)$ , on the interval [0, 1], namely

$$I_d(s) \leq I_d(s^{\star}(d)), \quad 0 \leq s \leq 1$$

so that

$$\int_0^1 I_d(s) ds \leqslant I_d(s^{\star}(d)).$$

We have

$$J_d(s^{\star}(d)) = d \log(1 - s^{\star}(d)) - \log s^{\star}(d) - \frac{1}{s^{\star}(d)}$$
$$= d \log\left(\frac{\sqrt{d}}{1 + \sqrt{d}}\right) - \log\left(\frac{1}{1 + \sqrt{d}}\right) - \left(1 + \sqrt{d}\right)$$
$$= d \log\left(\frac{\sqrt{d}}{1 + \sqrt{d}}\right) + \log\left(1 + \sqrt{d}\right) - \left(1 + \sqrt{d}\right), \quad (5.91)$$

so that

$$eI_d(s^{\star}(d)) = ee^{J_d(s^{\star}(d))}$$

$$= e\left(\frac{\sqrt{d}}{1+\sqrt{d}}\right)^d \left(1+\sqrt{d}\right) \cdot e^{-(1+\sqrt{d})}$$

$$= \left(\frac{\sqrt{d}}{1+\sqrt{d}}\right)^d \left(1+\sqrt{d}\right) \cdot e^{-\sqrt{d}}.$$
(5.92)

An easy induction argument shows the following fact: For all k = 1, 2, ..., we have

$$\frac{d^k}{ds^k}J_d(s) = -(k-1)!d(1-s)^{-k} + (-1)^k(k-1)!s^{-k} - (-1)^kk!s^{-(k+1)}, \quad 0 < s < 1.$$

Now fix  $k = 1, 2, \ldots$  It is easy to check that

$$\begin{pmatrix} \frac{d^k}{ds^k} J_d(s) \end{pmatrix}_{s=s^*(d)}$$

$$= -(k-1)! d(1-s^*(d))^{-k} + (-1)^k (k-1)! s^*(d)^{-k} - (-1)^k k! s^*(d)^{-(k+1)}$$

$$= -(k-1)! d\left(\frac{1+\sqrt{d}}{\sqrt{d}}\right)^k + (-1)^k (k-1)! \left(1+\sqrt{d}\right)^k - (-1)^k k! \left(1+\sqrt{d}\right)^{k+1}.$$

If we set

$$a_k(d) = \frac{1}{k!} \cdot \left(\frac{d^k}{ds^k} J_d(s)\right)_{s=s^\star(d)},$$

then

$$a_{k}(d) = -\frac{d}{k} \left( \frac{1+\sqrt{d}}{\sqrt{d}} \right)^{k} + (-1)^{k} \frac{1}{k} \left( 1+\sqrt{d} \right)^{k} - (-1)^{k} \left( 1+\sqrt{d} \right)^{k+1}$$
$$= \left( -\frac{d^{1-\frac{k}{2}}}{k} + \frac{(-1)^{k}}{k} - (-1)^{k} \left( 1+\sqrt{d} \right)^{k} \right) \left( 1+\sqrt{d} \right)^{k}.$$
(5.93)

We note that  $a_k(d) < 0$  if and only if

$$-\frac{d^{1-\frac{k}{2}}}{k} + \frac{(-1)^k}{k} - (-1)^k \left(1 + \sqrt{d}\right) < 0,$$

a condition equivalent to

$$-d^{1-\frac{k}{2}} + (-1)^k < (-1)^k k \left(1 + \sqrt{d}\right).$$
(5.94)

When k is even, say k = 2p for p = 1, 2, ..., this condition becomes

$$1 - d^{1-p} < 2p\left(1 + \sqrt{d}\right),$$

and is clearly satisfied.

When k is odd, say k = 2p + 1 for p = 1, 2, ..., then condition (5.94) becomes

$$(2p+1)\left(1+\sqrt{d}\right) < d^{-p+\frac{1}{2}}+1,$$

and is never satisfied. The discussion can be summarized as follows.

**Fact 5.12.** For  $k \ge 2$ , we always have  $(-1)^k a_k(d) < 0$ .

We now turn to the question as to whether the power series

$$\sum_{k=2} a_k(d)(s-s^\star(d))^k$$

is convergent on the interval (0, 1).

To answer this question, fix k = 2, 3, ... and consider

$$\frac{a_{k+1}(d)}{a_k(d)} = \frac{\left(-\frac{d^{1-\frac{k+1}{2}}}{k+1} + \frac{(-1)^{k+1}}{k+1} - (-1)^{k+1}\left(1 + \sqrt{d}\right)\right)\left(1 + \sqrt{d}\right)^{k+1}}{\left(-\frac{d^{1-\frac{k}{2}}}{k} + \frac{(-1)^k}{k} - (-1)^k\left(1 + \sqrt{d}\right)\right)\left(1 + \sqrt{d}\right)^k} \\ = \frac{\left(-\frac{d^{1-\frac{k+1}{2}}}{k+1} + \frac{(-1)^{k+1}}{k+1} - (-1)^{k+1}\left(1 + \sqrt{d}\right)\right)}{\left(-\frac{d^{1-\frac{k}{2}}}{k} + \frac{(-1)^k}{k} - (-1)^k\left(1 + \sqrt{d}\right)\right)} \cdot \left(1 + \sqrt{d}\right)} \\ = \frac{\left(-\frac{d^{-\frac{k-1}{2}}}{k+1} + \frac{(-1)^{k+1}}{k+1} - (-1)^{k+1}\left(1 + \sqrt{d}\right)\right)}{\left(-\frac{d^{-\frac{k-2}{2}}}{k} + \frac{(-1)^k}{k} - (-1)^k\left(1 + \sqrt{d}\right)\right)} \cdot \left(1 + \sqrt{d}\right)} \\ = \frac{k}{k+1} \frac{\left(-d^{-\frac{k-2}{2}} + (-1)^{k+1} - (-1)^{k+1}(k+1)\left(1 + \sqrt{d}\right)\right)}{\left(-d^{-\frac{k-2}{2}} + (-1)^k - (-1)^kk\left(1 + \sqrt{d}\right)\right)} \cdot \left(1 + \sqrt{d}\right)} \\ = \frac{k}{k+1} \frac{\left(-(-1)^k d^{-\frac{k-1}{2}} - 1 + (k+1)\left(1 + \sqrt{d}\right)\right)}{\left(-(-1)^k d^{-\frac{k-2}{2}} + 1 - k\left(1 + \sqrt{d}\right)\right)} \cdot \left(1 + \sqrt{d}\right). \quad (5.95)$$

It is now plain that

$$\lim_{k \to \infty} \left| \frac{a_{k+1}(d)}{a_k(d)} \right| = 1 + \sqrt{d}.$$

Therefore, with s in (0, 1) we have

$$\lim_{k \to \infty} \left| \frac{a_{k+1}(d)(s - s^{\star}(d))^{k+1}}{a_k(d)(s - s^{\star}(d))^k} \right| = \left( \lim_{k \to \infty} \left| \frac{a_{k+1}(d)}{a_k(d)} \right| \right) \cdot |s - s^{\star}(d)| = |s - s^{\star}(d)| \cdot \left( 1 + \sqrt{d} \right).$$
(5.96)

Is it possible for some s in (0, 1) to have

$$|s - s^{\star}(d)| \cdot \left(1 + \sqrt{d}\right) < 1?$$

This last condition is equivalent to having

$$-\frac{1}{1+\sqrt{d}} < s - s^{\star}(d) < \frac{1}{1+\sqrt{d}},$$

or equivalently,

$$-s^{\star}(d) < s - s^{\star}(d) < s^{\star}(d).$$

In other words, the condition becomes

$$0 < s < 2s^{\star}(d) = \frac{2}{1 + \sqrt{d}}.$$

In particular, the interval  $(0, 2s^{\star}(d))$  where  $J_d(s)$  admits a Taylor series representation shrinks with increasing d.

Pick  $\lambda$  in the unit interval [0, 1]. Note that

$$(1 \pm \lambda)s^{\star}(d) = \frac{1 \pm \lambda}{1 + \sqrt{d}}$$

so that

$$1 - (1 \pm \lambda)s^{\star}(d) = \frac{\mp \lambda + \sqrt{d}}{1 + \sqrt{d}}.$$

Therefore,

$$J_d((1 \pm \lambda)s^{\star}(d)) = d \log(1 - (1 \pm \lambda)s^{\star}(d)) - \log((1 \pm \lambda)s^{\star}(d)) - \frac{1}{(1 \pm \lambda)s^{\star}(d)}$$
$$= d \log\left(\frac{\mp \lambda + \sqrt{d}}{1 + \sqrt{d}}\right) - \log\left(\frac{1 \pm \lambda}{1 + \sqrt{d}}\right) - \frac{1 + \sqrt{d}}{1 \pm \lambda}, \qquad (5.97)$$

whence

$$I_d((1\pm\lambda)s^{\star}(d)) = \left(\frac{\mp\lambda+\sqrt{d}}{1+\sqrt{d}}\right)^d \cdot \left(\frac{1+\sqrt{d}}{1\pm\lambda}\right) \cdot e^{-\frac{1+\sqrt{d}}{1\pm\lambda}}.$$
 (5.98)

Next, with  $\alpha$  in [0, 1) and  $\beta$  in [0, 1], we conclude that

$$\frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))} = \frac{\left(\frac{-\beta+\sqrt{d}}{1+\sqrt{d}}\right)^d \cdot \left(\frac{1+\sqrt{d}}{1+\beta}\right) \cdot e^{-\frac{1+\sqrt{d}}{1+\beta}}}{\left(\frac{\alpha+\sqrt{d}}{1+\sqrt{d}}\right)^d \cdot \left(\frac{1+\sqrt{d}}{1-\alpha}\right) \cdot e^{-\frac{1+\sqrt{d}}{1-\alpha}}} \\
= \left(\frac{-\beta+\sqrt{d}}{\alpha+\sqrt{d}}\right)^d \cdot \frac{1-\alpha}{1+\beta} \cdot e^{-\frac{1+\sqrt{d}}{1+\beta}+\frac{1+\sqrt{d}}{1-\alpha}}.$$

Note that

$$\frac{1+\sqrt{d}}{1+\beta} - \frac{1+\sqrt{d}}{1-\alpha} = -C(\alpha,\beta) \cdot \left(1+\sqrt{d}\right)$$

where we have set

$$C(\alpha, \beta) = \frac{\alpha + \beta}{(1 - \alpha)(1 + \beta)},$$

whence

$$\frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))} = \left(\frac{-\beta + \sqrt{d}}{\alpha + \sqrt{d}}\right)^d \cdot \frac{1-\alpha}{1+\beta} \cdot e^{C(\alpha,\beta) \cdot (1+\sqrt{d})}.$$

Next, write

$$\left(\frac{-\beta + \sqrt{d}}{\alpha + \sqrt{d}}\right)^{d} = \left(1 - \frac{\alpha + \beta}{\alpha + \sqrt{d}}\right)^{d}$$
$$= e^{-d\left(\frac{\alpha + \beta}{\alpha + \sqrt{d}} + \Psi\left(\frac{\alpha + \beta}{\alpha + \sqrt{d}}\right)\right)}$$
(5.99)

where we have set

$$\Psi(x) = \int_0^x \frac{t}{1-t} dt, \ 0 \le x < 1.$$

$$d\Psi\left(\frac{\alpha+\beta}{\alpha+\sqrt{d}}\right) = d\left(\frac{\alpha+\beta}{\alpha+\sqrt{d}}\right)^2 r(d)$$
(5.100)

as we set

•

$$r(d) = \frac{\Psi\left(\frac{\alpha+\beta}{\alpha+\sqrt{d}}\right)}{\left(\frac{\alpha+\beta}{\alpha+\sqrt{d}}\right)^2} = \frac{1}{2} \left(1+o(1)\right).$$

It is plain that

$$\lim_{d \to \infty} d\left(\frac{\alpha + \beta}{\alpha + \sqrt{d}}\right)^2 = (\alpha + \beta)^2,$$

whence

$$\lim_{d \to \infty} d\Psi\left(\frac{\alpha+\beta}{\alpha+\sqrt{d}}\right) = \frac{(\alpha+\beta)^2}{2}.$$

Thus,

$$d\left(\frac{\alpha+\beta}{\alpha+\sqrt{d}}+\Psi\left(\frac{\alpha+\beta}{\alpha+\sqrt{d}}\right)\right)$$
$$=\frac{(\alpha+\beta)d}{\alpha+\sqrt{d}}+\frac{(\alpha+\beta)^2}{2}\left(1+o(1)\right)$$
(5.101)

and collecting we conclude that

$$\frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))} = e^{-\frac{(\alpha+\beta)^2}{\alpha+\sqrt{d}}} \cdot e^{-\frac{(\alpha+\beta)^2}{2}(1+o(1))} \cdot \frac{1-\alpha}{1+\beta} \cdot e^{C(\alpha,\beta)(1+\sqrt{d})} \\
= e^{-\frac{(\alpha+\beta)d}{\alpha+\sqrt{d}}} \cdot e^{C(\alpha,\beta)\sqrt{d}} \cdot \frac{1-\alpha}{1+\beta} e^{-\left(-C(\alpha,\beta)+\frac{(\alpha+\beta)^2}{2}(1+o(1))\right)}$$
(5.102)

However,

$$\lim_{d \to \infty} \frac{1}{\sqrt{d}} \left( \frac{(\alpha + \beta)d}{\alpha + \sqrt{d}} - C(\alpha, \beta)\sqrt{d} \right) = \alpha + \beta - C(\alpha, \beta)$$
$$= ((1 - \alpha)(1 + \beta) - 1)C(\alpha, \beta)$$
$$= (\beta(1 - \alpha) - \alpha)C(\alpha, \beta).$$
(5.103)

In other words,

$$\frac{(\alpha+\beta)d}{\alpha+\sqrt{d}} - C(\alpha,\beta)\sqrt{d} \sim (\beta(1-\alpha)-\alpha)C(\alpha,\beta)\sqrt{d}$$

and the following conclusion follows.

**Lemma 5.13.** With  $\alpha$  and  $\beta$  in (0, 1] such that

$$\frac{\alpha}{1-\alpha} < \beta,$$

we have

$$I_d((1+\beta)s^{\star}(d)) = o\left(I_d((1-\alpha)s^{\star}(d))\right)$$

in the sense that

$$\lim_{d \to \infty} \frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))} \cdot e^{(\beta(1-\alpha)-\alpha+o(1))C(\alpha,\beta)\sqrt{d}} = K(\alpha,\beta)$$

where

$$K(\alpha,\beta) = \frac{1-\alpha}{1+\beta} \cdot e^{-\left(\frac{(\alpha+\beta)^2}{2} - C(\alpha,\beta)\right)}.$$

In other words

$$\frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))} \sim K(\alpha,\beta) \cdot e^{-(\beta(1-\alpha)-\alpha+o(1))C(\alpha,\beta)\sqrt{d}}$$

We exploit this fact as follows: With  $\alpha$  and  $\beta$  in (0, 1] such that  $\alpha < (1 - \alpha)\beta$ , Lemma 5.13 ensures that for all d sufficiently large, the inequality

$$5I_d((1+\beta)s^*(d)) < I_d((1-\alpha)s^*(d))$$

holds. However, recall that the mapping  $s \to I_d(s)$  is monotone increasing on  $(0, s^*(d))$  and monotone decreasing on  $(s^*(d), 1)$ , with a maximum at  $s = s^*(d)$ . Therefore, with d sufficiently large, there exists t(d) to the right of  $s^*(d)$  in the interval  $(s^*(d), (1 + \beta)s^*(d))$  which depends on  $\alpha, \beta$  and d such that

$$I_d(t(d)) = I_d((1-\alpha)s^*(d)).$$

It is now plain that

$$\int_{(1+\beta)s^{\star}(d)}^{1} I_d(s) ds \leq (1 - (1+\beta)s^{\star}(d)) \cdot I_d((1+\beta)s^{\star}(d))$$

while

$$(t(d) - (1 - \alpha)s^{\star}(d)) \cdot I_d((1 - \alpha)s^{\star}(d)) \leqslant \int_0^{(1 + \beta)s^{\star}(d)} I_d(s)ds.$$

Therefore,

$$\frac{\int_{(1+\beta)s^{\star}(d)}^{1} I_d(s)ds}{\int_{0}^{(1+\beta)s^{\star}(d)} I_d(s)ds} \leq \frac{(1-(1+\beta)s^{\star}(d))}{(t(d)-(1-\alpha)s^{\star}(d))} \cdot \frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))}$$

As we note that

$$t(d) - (1 - \alpha)s^{\star}(d) > s^{\star}(d) - (1 - \alpha)s^{\star}(d) = \alpha s^{\star}(d)$$

by construction, it follows that

$$\frac{\int_{(1+\beta)s^{\star}(d)}^{1} I_d(s) ds}{\int_{0}^{(1+\beta)s^{\star}(d)} I_d(s) ds} \leqslant \frac{1}{\alpha s^{\star}(d)} \cdot \frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))} \\
= \frac{1+\sqrt{d}}{\alpha} \cdot \frac{I_d((1+\beta)s^{\star}(d))}{I_d((1-\alpha)s^{\star}(d))}$$
(5.104)

**Lemma 5.14.** For every  $\beta$  in [0, 1], we have

$$\lim_{d \to \infty} \frac{\int_{(1+\beta)s^{\star}(d)}^{1} I_d(s) ds}{\int_{0}^{(1+\beta)s^{\star}(d)} I_d(s) ds} = 0$$

and

$$\int_0^1 I_d(s) ds \sim \int_0^{(1+\beta)s^{\star}(d)} I_d(s) ds.$$

Pick d = 1, 2, ... Pick  $\alpha$  and  $\beta$  in (0, 1). Recall that for s in the interval  $((1 - \alpha)s^{\star}(d), (1 + \beta)s^{\star}(d))$ , we have

$$J_{d}(s) - J_{d}(s^{\star}(d)) = \sum_{k=2}^{\infty} a_{k}(d) (s - s^{\star})^{k}$$
  
$$= -\left(1 + \sqrt{d}\right)^{3} (s - s^{\star}(d))^{2} + \sum_{k=3}^{\infty} a_{k}(d) (s - s^{\star}(d))^{k}$$
  
$$= -\left(1 + \sqrt{d}\right)^{3} (s - s^{\star}(d))^{2} \left(1 - \sum_{k=3}^{\infty} \frac{a_{k}(d)}{(1 + \sqrt{d})^{3}} (s - s^{\star}(d))^{k-2}\right)$$
  
$$= -\left(1 + \sqrt{d}\right)^{3} (s - s^{\star}(d))^{2} (1 - h_{d}(s - s^{\star}(d)))$$
(5.105)
where we have set

$$h_d(s) = \sum_{k=3}^{\infty} \frac{a_k(d)}{\left(1 + \sqrt{d}\right)^3} \cdot s^{k-2}, \quad |s| < s^*(d)$$

Therefore,

$$\int_{(1-\alpha)s^{\star}(d)}^{(1+\beta)s^{\star}(d)} I_d(s) ds = e^{J_d(s^{\star}(d))} \int_{(1-\alpha)s^{\star}(d)}^{(1+\beta)s^{\star}(d)} e^{J_d(s) - J_d(s^{\star}(d))} ds$$
(5.106)

$$= I_d(s^*(d)) \int_{(1-\alpha)s^*(d)}^{(1+\beta)s^*(d)} e^{-(1+\sqrt{d})^3(s-s^*(d))^2(1-h_d(s-s^*(d)))} ds$$
  

$$= I_d(s^*(d)) \int_{-\alpha s^*(d)}^{\beta s^*(d)} e^{-(1+\sqrt{d})^3 y^2(1-h_d(y))} dy \quad [y = s - s^*(d)]$$
  

$$= s^*(d) I_d(s^*(d)) \int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})x^2(1-h_d(s^*(d)x))} dx \quad [y = s^*(\overline{a})td]$$

Thus, with x such that |x| < 1, we obtain

$$|h_{d}(s^{\star}(d)x)| = \left| \sum_{k=3}^{\infty} \frac{a_{k}(d)}{(1+\sqrt{d})^{3}} (s^{\star}(d)x)^{k-2} \right|$$

$$\leq \sum_{k=3}^{\infty} \frac{|a_{k}(d)|}{(1+\sqrt{d})^{3}} s^{\star}(d)^{k-2} |x|^{k-2}$$

$$= \sum_{k=3}^{\infty} |a_{k}(d)| s^{\star}(d)^{k+1} |x|^{k-2}$$
(5.108)

with

$$|a_{k}(d)|s^{\star}(d)^{k+1} = \left| \left( -\frac{d^{1-\frac{k}{2}}}{k} + \frac{(-1)^{k}}{k} - (-1)^{k} \left(1 + \sqrt{d}\right)^{k} \right| s^{\star}(d)^{k+1} \\ \leq 1 + \frac{1 + d^{1-\frac{k}{2}}}{k} \cdot s^{\star}(d) \\ \leq 1 + \frac{1 + \sqrt{d}}{k\sqrt{d}} \cdot s^{\star}(d) \\ = 1 + \frac{1}{k\sqrt{d}}, \quad k = 3, 4, \dots$$
(5.109)

As a result,

$$|h_{d}(s^{\star}(d)x)| \leq \sum_{k=3}^{\infty} \left(1 + \frac{1}{k\sqrt{d}}\right) |x|^{k-2}$$
$$\leq 2\sum_{k=3}^{\infty} |x|^{k-2}$$
$$= \frac{2|x|}{1-|x|}, \quad d = 1, 2, \dots$$
(5.110)

Hence, with  $\gamma = \max(\alpha, \beta)$ , we get

$$\sup\left(|h_d(s^{\star}(d)x)|, \ -\alpha \leqslant x \leqslant \beta\right) \leqslant \frac{2\gamma}{1-\gamma}$$
(5.111)

uniformly in d. It follows then for each  $\lambda$  in (0, 1),

$$\sup\left(|h_d(s^{\star}(d)x)|, \ -\alpha \leqslant x \leqslant \beta\right) \leqslant \lambda$$

uniformly in d whenever

$$\gamma < \frac{\lambda}{2+\lambda} := \gamma^{\star}(\lambda)$$

with  $\gamma^{\star}(\lambda)$  in the unit interval (0, 1).

Thus, fix  $\lambda$  in (0,1). With  $\gamma$  in  $(0,\gamma^{\star}(\lambda))$ , whenever  $\gamma = \max(\alpha,\beta)$ , we have

$$\int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})x^2(1-h_d(s^*(d)x))} dx \leq \int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})(1-\lambda)x^2} dx$$

and

$$\int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})(1+\lambda)x^2} dx \leq \int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})x^2(1-h_d(s^{\star}(d)x))} dx$$

uniformly in d.

We will make use of these bounds together with the following asymptotics.

**Lemma 5.15.** For arbitrary  $\alpha > 0$  and  $\beta > 0$  it holds that

$$\int_{-\alpha}^{\beta} e^{-\theta x^2} dx \sim \sqrt{\frac{\pi}{\theta}}$$

as  $\theta$  goes to infinity.

**Proof.** Fix  $\theta > 0$ . We get

$$\int_{-\alpha}^{\beta} e^{-\theta x^{2}} dx = \frac{1}{\sqrt{2\theta}} \int_{-\alpha\sqrt{2\theta}}^{\beta\sqrt{2\theta}} e^{-\frac{y^{2}}{2}} dy \quad [y = \sqrt{2\theta}x]$$
$$= \sqrt{\frac{\pi}{\theta}} \left( \Phi(\beta\sqrt{2\theta}) - \Phi(-\alpha\sqrt{2\theta}) \right)$$
$$= \sqrt{\frac{\pi}{\theta}} \left( \Phi(\beta\sqrt{2\theta}) + \Phi(\alpha\sqrt{2\theta}) - 1 \right)$$
(5.112)

and the conclusion follows.

Consequently,

$$\int_{-\alpha}^{\beta} e^{-\left(1+\sqrt{d}\right)(1\pm\lambda)x^2} dx \sim \sqrt{\frac{\pi}{\left(1+\sqrt{d}\right)\left(1\pm\lambda\right)}}$$

Assume from the time being that

$$\int_0^1 I_d(s)ds \sim \int_{(1-\alpha)s^\star(d)}^1 I_d(s)ds, \quad \alpha \in (0,1)$$

so that

$$p(d+1) \sim \int_{(1-\alpha)s^{\star}(d)}^{1} I_{d}(s)ds$$
  

$$\sim \int_{(1-\alpha)s^{\star}(d)}^{(1+\beta)s^{\star}(d)} I_{d}(s)ds$$
  

$$= s^{\star}(d)I_{d}(s^{\star}(d)) \int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})x^{2}(1-h_{d}(s^{\star}(d)x))}dx$$
  

$$= s^{\star}(d)I_{d}(s^{\star}(d)) \cdot \sqrt{\frac{\pi}{1+\sqrt{d}}} \cdot \frac{\int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})x^{2}(1-h_{d}(s^{\star}(d)x))}dx}{\sqrt{\frac{\pi}{1+\sqrt{d}}}} (5.113)$$

It follows that

$$\frac{p(d+1)}{s^{\star}(d)I_d(s^{\star}(d))\cdot\sqrt{\frac{\pi}{1+\sqrt{d}}}} \sim \frac{\int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})x^2(1-h_d(s^{\star}(d)x))}dx}{\sqrt{\frac{\pi}{1+\sqrt{d}}}}$$
(5.114)

# From earlier inequalities we conclude that

$$\limsup_{d \to \infty} \frac{\int_{-\alpha}^{\beta} e^{-\left(1+\sqrt{d}\right)x^2\left(1-h_d(s^{\star}(d)x)\right)} dx}{\sqrt{\frac{\pi}{1+\sqrt{d}}}} \\ \leqslant \lim_{d \to \infty} \frac{\int_{-\alpha}^{\beta} e^{-\left(1+\sqrt{d}\right)\left(1-\lambda\right)x^2} dx}{\sqrt{\frac{\pi}{1+\sqrt{d}}}} = \frac{1}{\sqrt{1-\lambda}}$$
(5.115)

and

$$\frac{1}{\sqrt{1+\lambda}} = \lim_{d \to \infty} \frac{\int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})(1+\lambda)x^2} dx}{\sqrt{\frac{\pi}{1+\sqrt{d}}}} \\
\leqslant \liminf_{d \to \infty} \frac{\int_{-\alpha}^{\beta} e^{-(1+\sqrt{d})x^2(1-h_d(s^*(d)x))} dx}{\sqrt{\frac{\pi}{1+\sqrt{d}}}}$$
(5.116)

Finally,

$$\limsup_{d \to \infty} \frac{p(d+1)}{s^{\star}(d)I_d(s^{\star}(d)) \cdot \sqrt{\frac{\pi}{1+\sqrt{d}}}} \leq \frac{1}{\sqrt{1-\lambda}}$$

and

$$\frac{1}{\sqrt{1+\lambda}} \leq \liminf_{d \to \infty} \frac{p(d+1)}{s^{\star}(d)I_d(s^{\star}(d)) \cdot \sqrt{\frac{\pi}{1+\sqrt{d}}}}$$

Since  $\lambda$  is arbitrary in (0, 1) we conclude that

$$\lim_{d \to \infty} \frac{p_{\xi}(d+1)}{s^{\star}(d)I_d(s^{\star}(d)) \cdot \sqrt{\frac{\pi}{1+\sqrt{d}}}} = 1,$$

or equivalently,

$$p_{\xi}(d+1) \sim s^{\star}(d) I_d(s^{\star}(d)) \cdot \sqrt{\frac{\pi}{1+\sqrt{d}}}.$$

Substituting we find

$$p_{\xi}(d+1) \sim \sqrt{\frac{\pi}{e^2}} \frac{e^{-2\sqrt{d}}}{\sqrt{1+\sqrt{d}}} \sim \sqrt{\frac{\pi}{e^2}} \frac{e^{-2\sqrt{d}}}{\sqrt[4]{d}}$$

Part II

Games on Networks

## Chapter 6: Learning in Games

There has been a growing interest in game theory as a tool to solve complex engineering problems that involve interactions among (many) distributed agents or subsystems. As these agents are expected to interact many times over time, researchers proposed a variety of learning rules with desired properties, that helped the agents to determine their future strategies on the basis of the past information available to them. One example of such a desired property is convergence to some form of equilibrium that might be thought of as an approximation of a desired operating point for the entire system. Many researchers have worked on this interesting problem, and as a result many different types of learning rules have been already proposed in the literature, e.g., [74, 75, 78, 78, 79].

However, when casting an engineering problem in a game-theoretic framework there are a few important practical considerations that need to be taken into account. For instance, controllers in engineering systems are not necessarily synchronized; instead, they are often *event driven* and update their (control) actions as new measurements or observations become available. For this reason, guaranteeing acceptable performance under *asynchronous* operations is crucial in many engineering systems. Moreover, in settings where the system consists of many individual subsystems with their own individual controllers, the measurements available to a particular subsystem or a controller could reflect the *past* decisions taken by the individual controllers. For example, consider the setting of a wind farm composed of a number of wind turbines <sup>1</sup>. The sensor data available at a particular wind turbine could reflect past decisions taken by other wind turbines because the actions taken by the other wind turbines could affect the former with varying time delays.

Unfortunately, most of the existing studies on learning in games do not consider either delays experienced by payoff information or asynchronous updates of strategies by players. In our work we propose two classes of learning rules which could be thought as a first step towards bridging this gap between the current literature and the sound design of engineering systems on a game-theoretic framework. First, we consider the better-reply rule [85] in which the players aim to improve their immediate payoffs. This learning rule has been studied extensively due to its simplicity [73, 74, 95]. More precisely, the proposed rule requires the following: (i) if there is no strictly better reply (SBR), the agent stays with the same action, and (ii) if there exists at least one SBR, it switches to each SBR with positive probability and also sticks with the previous actions with positive probability (often called *inertia*). This inertia may, for instance, model the scenario where an agent waits at least one more period and tries the same action before switching/commiting to an SBR. Clearly, the learning rule is computationally inexpensive in that only a number of simple comparisons need to be made.

<sup>&</sup>lt;sup>1</sup>A wind turbine can control how much power it draws from the wind by varying its axial induction factor [77]

In Chapter 7 we introduce a general framework for modelling asynchronous updates of strategies by the players, possibly based on delayed or even outdated payoff information. Assuming that the payoff information available to the players is accurate, we prove that if all the agents update in accordance to the better-reply rule, under a set of mild technical conditions, the action profiles <sup>2</sup> played by the agents converges to a pure strategy Nash equilibrium (PSNE) almost surely [58] in a class of games, which we call *generalized weakly acyclic games* (GWAGs). We note that this almost sure convergence of action profiles takes place even when the players update their strategies in an asynchronous manner on the basis of delayed payoff information. Finally, we demonstrate that if the game is *not* generalized weakly acyclic, the better-reply rule in general cannot guarantee the almost sure convergence of action profiles even when the strategies are updated synchronously using current payoff information.

In many practical scenarios, erroneous decision-making by the individual agents (possibly controllers in an engineering system) is also a major concern along with delayed payoff information and asynchronous operations. Such erroneous decisionmaking could stem from faulty available payoff information and that is what we consider in Section 7.6. We show in fact that as the payoff information becomes more reliable, the probability that the agents play a PSNE under the better-reply rule tends to one over time. When the probability of error is sufficiently small, the set of action profiles that are played by the system most of the time are called the *stochastically stable states* [95]. Thus for the GBRR rules the set of stochastically

 $<sup>^2\</sup>mathrm{An}$  action profile specifies the actions played by all the agents in the system

stable states are a subset of the set of PSNE(s). However there are two shortcomings of the better reply rules – (i) a somewhat undesirable property is that under the exact payoff information setting the system converges almost surely to a PSNE depending on the system's initial conditions, and (ii) under the faulty payoff information setting the set of stochastically stable states are not easily characterisable. This gives us the reason to devise a learning rule which is not only robust with respect to delayed payoff information and asynchrony but also addresses the abovementioned shortcomings. This leads us to our next piece of work where we have proposed a new class of rules, which we call the class of monitoring rules.

In Chapter 8, we discuss a simplified representative version in the class of monitoring rules called the *reduced simple experimentation with monitoring* (RSEM) rule [86]. Under the RSEM rule, the agents can either be in an *explore* state or in a *converged* state – where the state dictates how actions are chosen at a particular time. While at the explore state the agent tries out every action with a positive probability in an effort to play the action which cannot be improved any further. Once such an action is played the agent switches to the converged state with a positive probability. At the converged state, the agent simply keeps playing the previous action as long as 'playing conditions remain unchanged', i.e., the payoff remains the same and no better reply exists. As promised earlier, in Chapter 8, we show convergence of the action profile to a PSNE under asynchronous updates and payoff information delays, firstly, in the class of GWAGs, and also in games satisfying a 'payoff interdependence assumption' [79, 86, 87, 96]. When the game satisfies the 'payoff interdependence assumption', the RSEM rule ensures almost sure convergence to the set of PSNE(s) independent of the initial conditions. Also, under the setting of erroneous determination of the SBRs, i.e., erroneous estimation of the set of better replies, the set of stochastically stable states are well characterisable. In particular, they are those PSNE(s) such that when it is in effect in the system, it is least likely that any of the agents see a better reply. It is worth pointing out that the results for the RSEM rule have been shown for erroneous determination of the SBRs but an accurate received payoff for the played action. Future work is in order to consider the situation of erroneous payoff for the played action as well.

Until now, we argued erroneous decision-making by the agents to be a result of faulty available payoff information. However there might be practical situations where the agents receive correct payoff information yet implement non-equilibrium actions either by mistake or maybe because its system has been compromised. Consider the model described earlier where the agents correspond to controllers driving their own (sub)-systems which in turn are a part of a much bigger system. There is a possibility that the controller decides on a certain action but its own (sub)-system cannot implement it either due to mistakes or maybe because its system has been compromised and the controller has lost command on its (sub)-system.

Such a scenario could be potentially detrimental to a system which is already at an equilibrium. A moment of thought suggests that the algorithm should allow brief changes in the received payoff information and instead be responsive to long term changes. A simple implementation of this idea leads to a generalization of the RSEM rule where the agents at the converged state allow at most a fixed number of successive changes in its payoff information before permanently moving away from an equilibrium action. This rule is formally introduced in Chapter 9 as the *simple* experimentation with monitoring (SEM) rule. We show that when the agents follow the SEM rule, the action profiles selectively converge to resilient PSNE(s) that can tolerate deviations by potentially multiple agents. By controlling the number of successive changes in the payoff information that can be allowed by the agents as a tunable parameter, either (i) PSNEs with certain desired resilience or (ii) the most resilient PSNE(s) could be reached.

This portion of the thesis is organized as follows: In Chapter 7, we first introduce the framework for considering asynchronous updates and delayed payoff information, and also present the results under better-reply rules. In the same chapter we also introduce a setting for considering faulty payoff information and investigate convergence under the better-reply rules. In the next chapter, we introduce the RSEM rule and present convergence results under delays and asynchrony. We also distinguish our findings from that obtained under the better-reply rules under faulty payoff information. In Chapter 9, we generalize the RSEM rule to ensure selective convergence to *resilient* PSNEs under erroneous decision-making due to faulty implementation of intended actions. But first, in the following section we present an outline of our main contributions before doing a brief literature review.

### 6.1 Summary of Contributions

Our main contributions can be summarized as follows:

1. We propose a general framework for considering delayed payoff information

and asynchronous updates by agents. To the best of our knowledge, this work is the first one that introduces a general game-theoretic framework for modeling more realistic engineering systems with delays and asynchrony.

- 2. We demonstrate that a simple and intuitive rule such as the better-reply rule can ensure almost sure convergence to the set of PSNEs even under asynchronous updates and delays in a class of games, which we call the generalized weakly acyclic games. Not only that, we also show the probability that the players have not converged to a PSNE *decays geometrically* with time, thereby proving that the expected convergence time is finite.
- 3. Motivated by practical scenarios, we model erroneous decision-making by agents due to faulty payoff information. Under the better-reply rule, we show that as the payoff information becomes more reliable, the probability that the players play a PSNE tends to one over time.
- 4. For the better reply rules, under the setting where there are no errors, the system converges to a PSNE which could depend on its initial conditions. Also, under the aforementioned setting of faulty available payoff information, we are unable to characterise the set of stochastically stable equilibria. To resolve these issues, we propose the RSEM rule:
  - (a) Firstly, the RSEM rule is shown to ensure almost sure convergence of the action profiles to a PSNE under delays and asynchrony for GWAGs and also in games that satisfy a 'payoff interdependence assumption'.

Moreover, under both the settings we show that the probability that the agents have not converged to a PSNE decays geometrically with time. When the game satisfies the payoff interdependence assumption the system converges to any PSNE independently of the initial conditions.

- (b) Under the setting of faulty determination of SBRs, the stochastically stable states under the RSEM rule are those PSNE(s) which make it least likely for any of the agents to see a better reply when it is in effect.
- 5. While erroneous decision-making could occur due to faulty available payoff information, it could also be a result of faulty implementation of an intended action. Firstly, we consider a model to account for such a scenario. Next, we show that the RSEM rule can be generalized to ensure that the system reaches PSNE(s) which are resilient to deviations by potentially multiple agents.

# 6.2 Related Literature

There is already a large volume of literature on learning in games with findings of varying nature, e.g., [39, 43, 52, 56, 63, 69, 95, 96]. For this reason, we only provide a very brief summary of a limited number of studies, and refer an interested reader to the references therein for additional studies. First, a popular early learning procedure is fictitious play [45, 55, 56, 82, 83]; players form beliefs regarding the opponents' players, for instance, based on the empirical frequencies of their plays, and pick optimal strategies with respect to their beliefs. The convergence of strategy profile to equilibria is proven only in somewhat restrictive settings, such as zero-sum games and potential games.

Bayesian learning has drawn attention of many researchers for quite some time: In [66], Jordan studied the Bayesian learning processes for normal form games and convergence to a Nash equilibrium under mild assumptions on the prior distribution over payoff functions. In [67], he also studied far-sighted Bayesian learning and its convergence to a Nash equilibrium in repeated game settings. Another related work by Kalai and Lehrer [68] showed that, if players in a repeated game start with subjective beliefs about their opponents' strategies and their beliefs are "compatible" with their true strategies, i.e., the players' prior beliefs assign positive probability to all strategy profiles that will be chosen with positive probability according to the true strategies, Bayesian learning leads to accurate prediction of future play of the game. Moreover, they illustrated that if the players know their payoff functions, the convergence to a Nash equilibrium occurs.

Researchers also considered non-Bayesian learning. For instance, Foster and Vohra [50] studied the problem of forecasting and pointed out the benefits of randomized forecasting against an oblivious or adaptive adversary. In their follow-up study [49], they demonstrated that when the players in a normal form game use a learning rule with calibrated forecast of the other players' plays and each player plays myopically with respect to the forecast distribution, the limit points of the sequence of plays are correlated equilibria. This was put forth as an alternative to Aumann's proof that the common prior assumption and rationality imply a correlated equilibrium. Interestingly, they also showed the converse of the finding also holds in the sense that, for every correlated equilibrium, there exists some calibrated learning rule that would result in the equilibrium being played in the limit.

Another class of learning rules is based on *regrets*. Regrets capture the additional payoff a player could have received by playing a different action. There are several regret-based learning rules, e.g., [57, 59, 62], which guarantee convergence to Nash equilibria, correlated equilibria or Hannan set in an appropriate sense. Hart and Mas-Colell in [59] introduced a simple adaptive procedure called the regret *matching* where players switch from their current plays to others with probabilities proportional to their regrets. They showed that the empirical distribution of plays converges to a correlated equilibrium if all the players follow the procedure. In a slightly different informational setting, Hart and Mas-Colell proposed a reinforcement learning-based technique to estimate regrets, which they called the *modified* regret matching, with a similar convergence property. A summary of existing convergence results for generalized regret matching can be found in [61]. Regret testing is another regret-based uncoupled learning rule, which was first introduced by Foster and Young [51]. Germano and Lugosi [57] showed that when all players adopt this scheme, the (mixed) strategy profile of the players converges to a Nash equilibrium of the stage game almost surely.

Oftentimes, real world problems exhibit special structures which could be leveraged while designing learning rules. In this context many of the existing learning rules target games with special structures, such as identical interest games, potential games (PGs), and weakly acyclic games (WAGs). Although an arbitrary game is not guaranteed to possess a PSNE, one exists for PGs because a maximizer of the potential function is a PSNE [88]. This led to further research with problems being formulated as PGs. For example, Arslan et al. [41] model the autonomous vehicle-target assignment problem as a PG. They present two learning rules and demonstrate their convergence to PSNEs. Marden et al. [76] study large-scale games with many players with large strategy spaces. They generalize the notion of PGs and propose a learning rule that guarantees convergence to a PSNE in the class of games they consider, with applications to congestion games. The WAGs were first studied in a systemic manner in [95]. Since then, there has been considerable interest in WAGs. For instance, Marden et al. [73] establish the relations between cooperative control problems (e.g., consensus problem) and game theoretic models. In addition, they propose the better reply with inertia dynamics and apply it to a class of games, which they call *sometimes weakly acyclic games*, to address timevarying objective functions and action sets. In another related study [74], Marden et al. proposed regret-based dynamics that achieve almost sure convergence to a strict Nash equilibrium in weakly acyclic games.

More recently, researchers aimed to design so-called payoff-based learning rules (also known as 'completely uncoupled dynamics') with provable convergence to efficient Nash equilibria (in an appropriate sense) in potential games [78] and weakly acyclic games [75]. Pradelski and Young extended the result to general games satisfying what they called an *interdependence* assumption, a weaker form of which has been actually used in our work. Related to these studies, Marden et al. [79] proposed learning rules that seek Pareto optimal strategy profile, i.e., a maximizer of aggregate payoff. This result was sharpened by Menon and Baras [80] to show that, under some conditions, the probability that the strategy profile lies in the set of aggregate payoff maximizers converges to one over time. Marden and Shamma [78] studied the effects of asynchrony in log-linear learning using independent revision processes and showed that, in potential games with sufficiently small probability of players revising their strategies at each time  $t \in \mathbb{N}$ , only the maximizers of potential functions can be stochastically stable. We also point out an interesting study by Hart and Mansour [64] on the communication complexity of uncoupled equilibrium procedures. Not surprisingly, they showed that any pure Nash equilibrium procedure has communication complexity that grows exponentially with the number of players.

### Chapter 7: Class of Better reply rules

In this chapter, we consider a simple and intuitive class of learning rules called the better-reply rules. Along with identifying a new class of games – generalized weakly acyclic games which contains well known classes of games such as weakly acyclic games, potential games and identical interest games (see Section 6.2 for practical significance of these classes of games), we also show convergence results of the action profiles under the better-reply rules in the context of these games. We draw connections between the better-reply rules and the GWAGs, and argue that the GWAGs is a class of games naturally associated with the better-reply rules. Keeping in mind our original motivation to better model practical engineering concerns in a game-theoretic setting, we outline a framework for considering asynchronous updates of strategies by the agents based on payoff information with potentially time-varying delays. We demonstrate that when all agents update their strategies according to the better-reply rules, the action profile converges to a pure-strategy Nash equilibrium with probability 1 (or almost surely) if the game is a GWAG. As discussed in Chapter 6, we take our attempt at modelling practical scenarios one step further, and consider the setting of erroneous decision-making by the agents due to faulty payoff information.

### 7.1 Stage and Repeated Games

In this section, we first describe the strategic-form stage game and the infinitely repeated game we adopt for analysis. A word on notation: An inequality between two action profiles is an element-wise inequality.

Finite stage game: Let  $\mathcal{P} := \{1, 2, ..., n\}$  be the finite set of agents. The pure action or strategy space of agent  $i \in \mathcal{P}$  and the joint strategy space of all agents are denoted by  $\mathcal{A}_i = \{1, 2, ..., A_i\}$  and  $\mathcal{A} := \prod_{i \in \mathcal{P}} \mathcal{A}_i$ , respectively. We assume that the strategy spaces  $\mathcal{A}_i$  are finite for all  $i \in \mathcal{P}$ . The payoff function of agent i is given by  $U_i : \mathcal{A} \to \mathbb{R} := (-\infty, \infty)$ . Hence, the strategic-form finite (stage) game is given by  $\mathbb{G} := (\mathcal{P}, \{\mathcal{A}_i, i \in \mathcal{P}\}, \{U_i, i \in \mathcal{P}\}).$ 

A mixed strategy of agent  $i \in \mathcal{P}$  is a probability distribution  $\mathbf{p}_i = (p_i(a_i), a_i \in \mathcal{A}_i) \in \Delta(\mathcal{A}_i)$ , where  $\Delta(\mathcal{A}_i)$  denotes the probability simplex over  $\mathcal{A}_i$ ; agent  $i \in \mathcal{P}$  chooses action  $a_i \in \mathcal{A}_i$  with probability  $p_i(a_i)$ . A pure strategy is a special case where the probability distribution is concentrated on a single action.

A strategy profile is a collection of strategies, one strategy for each agent. Throughout the remainder of the thesis, a strategy profile refers to a pure strategy profile unless stated otherwise. Furthermore, we find it convenient to differentiate the (mixed) strategy profile from the actions played by the agents (according to the mixed strategy profile). For this reason, we refer to the set of actions played by the agents as an *action profile*.

Given a strategy profile  $\mathbf{a} = (a_1, a_2, \dots, a_n) \in \mathcal{A}, \mathbf{a}_{-i}$  denotes the strategy

profile of all the agents other than agent *i*, i.e.,  $\mathbf{a}_{-i} = (a_1, \ldots, a_{i-1}, a_{i+1}, \ldots, a_n)$ . Similarly, given a subset of agents  $I \subsetneq \mathcal{P}$ ,  $\mathbf{a}_I$  is the strategy profile comprising strategies picked by the agents in *I*. We say that  $\mathbf{a}^* \in \mathcal{A}$  is a pure-strategy Nash equilibrium if, for every agent  $i \in \mathcal{P}$ ,

$$U_i(a_i^{\star}, \mathbf{a}_{-i}^{\star}) \ge U_i(a_i, \mathbf{a}_{-i}^{\star}) \text{ for all } a_i \in \mathcal{A}_i \setminus \{a_i^{\star}\}.$$

$$(7.1)$$

The pure-strategy Nash equilibrium is *strict* if the inequality in (7.1) is strict for all agents  $i \in \mathcal{P}$ . Hereafter a Nash equilibrium refers to a pure-strategy Nash equilibrium. We denote the set of Nash equilibria of  $\mathbb{G}$  by  $\mathcal{A}_{NE}$ .

**Repeated game:** We employ a repeated game setting to model the repeated interactions among agents. In an (infinitely) repeated game, the above stage game G is repeated at each time  $t \in \mathbb{N} := \{1, 2, \ldots\}$ . At time t, agent  $i \in \mathcal{P}$  chooses its action  $a_i(t)$  according to some mixed strategy  $\mathbf{p}_i(t) \in \Delta(\mathcal{A}_i)$ . We denote the action profile played by the agents at time t by  $\mathbf{a}(t) = (a_i(t), i \in \mathcal{P})$ . However, unlike in a traditional repeated game, we do not necessarily assume that the payoffs agents receive at time t depend on  $\mathbf{a}(t)$ .

The agents are allowed to revise their strategies based on (the history of) payoffs in a repeated game. In this chapter we describe two simple better-reply rules (that were first studied in [85]) which ensures almost sure convergence under asynchronous updates of strategies by agents based on their delayed payoff information in a class of games we call generalized weakly acyclic games described in the following section.

### 7.2 Generalized weakly acyclic games

In order to define the GWAGs [85], we first need to introduce the notion of generalized better reply paths (GBRPs). A GBRP is a sequence of action profiles  $(\mathbf{a}^1, \mathbf{a}^2, \dots, \mathbf{a}^L)$  such that, for every  $1 \leq \ell \leq L-1$ , there exists a set of agents  $I_\ell \subseteq \mathcal{P}$ that satisfies

i. 
$$a_i^{\ell} \neq a_i^{\ell+1}$$
 and  $U_i(\mathbf{a}^{\ell}) < U_i(a_i^{\ell+1}, \mathbf{a}_{-i}^{\ell})$  for all  $i \in I_{\ell}$ , and

ii. 
$$a_i^{\ell} = a_i^{\ell+1}$$
 for all  $i \in \mathcal{P} \setminus I_{\ell}$ .

These conditions mean that a GBRP consists of *transitions* from one action profile to another action profile, in each of which a set of agents that can achieve a higher payoff via unilateral deviation switch their actions *simultaneously*, while the remaining agents stay with their previous actions. A better reply path (BRP) used to define WAGs [73,74] is a special case of GBRPs with  $|I_{\ell}| = 1$  for all  $\ell = 1, 2, ..., L - 1$ .

**Definition 7.1.** A game is a GWAG if (i) the set of PSNEs is nonempty and (ii) for every action profile  $\mathbf{a}^* \in \mathcal{A} \setminus \mathcal{A}_{NE}$ , there exists a GBRP  $(\mathbf{a}^1, \mathbf{a}^2, \dots, \mathbf{a}^L)$  such that  $\mathbf{a}^1 = \mathbf{a}^*$  and  $\mathbf{a}^L \in \mathcal{A}_{NE}$ .

It is clear from the definition that WAGs are special cases of GWAGs where only BRPs are allowed, i.e., only a single agent is allowed to switch or deviate at a time. Due to this constraint, we suspect that there is a large class of games that are GWAGs, but not WAGs as illustrated by the following simple game.



Figure 7.1: An example of a GWAG that is not a WAG. (a) Game in normal form,(b) all possible unilateral deviations and simultaneous deviations

◇ Example of a GWAG that is not a WAG: Consider a three-agents game shown in Fig 7.1 in normal form. All three agents have the identical action space  $\{0, 1\}$ . The unique PSNE of the game is  $\mathbf{a}^{NE} = (1, 1, 1)$ . The solid red arrows in Fig. 7.1(b) indicate all possible *unilateral* deviations that would improve the payoff of the agent that deviates. From the figure, we can verify that there does not exist a BRP from any action profile that is not the PSNE or  $\mathbf{a}^{\dagger} = (0, 1, 1)$ ; the PSNE is reachable only from  $\mathbf{a}^{\dagger}$ , and it is not possible to reach either the PSNE or  $\mathbf{a}^{\dagger}$  from any other action profile. For this reason, this game is *not* a WAG.

On the other hand, this game is a GWAG. To see this, note that both agents 2 and 3 have an incentive to deviate at  $\mathbf{a}^* = (0, 0, 0)$ . Therefore, from action profile  $\mathbf{a}^*$ , we can find a GBRP given by  $(\mathbf{a}^1, \mathbf{a}^2, \mathbf{a}^3) = (\mathbf{a}^*, \mathbf{a}^{\dagger}, \mathbf{a}^{NE})$  with  $I_1 = \{2, 3\}$  and  $I_2 = \{1\}$ . Since it is possible to reach  $\mathbf{a}^*$  from other remaining action profiles, we can construct a GBRP from them as well. Therefore, this example illustrates that the GWAGs give rise to a strictly larger class of games than WAGs.

In Section 7.4, we draw a relationship between a class of better reply algorithms and the class of GWAGs. However, next we state the following lemma which provides an alternate definition of a GWAG.

Lemma 7.1. A game  $\mathbb{G}$  is a GWAG if and only if there exists a potential function  $\phi : \mathcal{A} \to \mathbb{R}$  such that, for every  $\mathbf{a} \notin \mathcal{A}_{NE}$ , there exists a subset of agents  $I(\mathbf{a}) \subseteq \mathcal{P}$ and  $a'_i \neq a_i$  for all  $i \in I(\mathbf{a})$  such that  $U_i(a'_i, \mathbf{a}_{-i}) > U_i(\mathbf{a})$  and  $\phi(\mathbf{a}'_{I(\mathbf{a})}, \mathbf{a}_{-I(\mathbf{a})}) > \phi(\mathbf{a})$ . **Proof.** Sufficiency ('if'): Pick  $\mathbf{a}^0 \notin \mathcal{A}_{NE}$ . Then, there exists some subset  $I(\mathbf{a}^0) \subseteq \mathcal{P}$  with action profile  $\mathbf{a}'_{I(\mathbf{a}^0)} \neq \mathbf{a}^0_{I(\mathbf{a}^0)}^1$  such that  $U_i(a'_i, \mathbf{a}^0_{-i}) > U_i(\mathbf{a}^0)$  for all  $i \in I(\mathbf{a}^0)$ and  $\phi(\mathbf{a}^1) > \phi(\mathbf{a}^0)$ , where  $\mathbf{a}^1 = (\mathbf{a}'_{I(\mathbf{a}^0)}, \mathbf{a}^0_{-I(\mathbf{a}^0)})$ .

If  $\mathbf{a}^1 \notin \mathcal{A}_{NE}$ , we can repeat this process and construct a sequence  $\mathbf{a}^0, \mathbf{a}^1, \ldots$ , until we have an action profile that belongs to  $\mathcal{A}_{NE}$ . Since  $\phi(\mathbf{a}^{\ell}) < \phi(\mathbf{a}^{\ell+1})$  for all  $\ell = 0, 1, \ldots$  and there are finitely many action profiles, this process will terminate after a finite number of iterations and the final action profile, say  $\mathbf{a}^M$ , must be a PSNE. It is an easy exercise to verify that  $(\mathbf{a}^0, \mathbf{a}^1, \ldots, \mathbf{a}^M)$  is a GBRP from its construction. Since we can find a GBRP that leads to a PSNE for any non-PSNE action profile  $\mathbf{a}^0$ , the game is a GWAG.

Necessity ('only if'): Define the length of a GBRP to be the number of action profiles in the sequence. For each  $\mathbf{a} \notin \mathcal{A}_{NE}$ , find a shortest GBRP to a PSNE. If we represent all PSNEs in  $\mathcal{A}_{NE}$  using a single node  $\mathbf{a}^{NE}$ , then these shortest GBRPs from all non-PSNE action profiles to  $\mathbf{a}^{NE}$  give rise to a spanning tree rooted at  $\mathbf{a}^{NE}$ , which we denote by  $\mathcal{T}$ .

Suppose that we assign a value  $\phi_0$  to all PSNEs (represented by  $\mathbf{a}^{NE}$  in  $\mathcal{T}$ ). Let  $\phi : \mathcal{A} \to \mathbb{R}$  be a potential function, where  $\phi(\mathbf{a}) = \phi_0 - d(\mathbf{a})$  and  $d(\mathbf{a})$  is the hop distance of  $\mathbf{a}$  to  $\mathbf{a}^{NE}$  in  $\mathcal{T}$ . Then, from the construction of the spanning tree  $\mathcal{T}$ , it is clear that, for any action profile  $\mathbf{a} \notin \mathcal{A}_{NE}$ , there exists a subset  $I \subseteq \mathcal{P}$  and  $\mathbf{a}'_I \neq \mathbf{a}_I$ such that  $U_i(a'_i, \mathbf{a}_{-i}) > U_i(\mathbf{a})$  for all  $i \in I$  and  $\phi(\mathbf{a}'_I, \mathbf{a}_{-I}) > \phi(\mathbf{a})$ .

<sup>&</sup>lt;sup>1</sup>The inequality between two action profiles is an element-wise inequality throughout the thesis.

#### 7.3 Proposed Update Rules - Generalized Better Reply Rules

**Payoff information for updates** – Let  $S = \{B, I, W\}$  and, for each agent  $i \in \mathcal{P}$ , define its *classification* mapping  $C_i : \mathcal{A}_i \times \mathcal{A} \to S$ , where

$$C_i(a'_i, \mathbf{a}) = \begin{cases} B & \text{if } U_i(a'_i, \mathbf{a}_{-i}) > U_i(\mathbf{a}), \\ I & \text{if } U_i(a'_i, \mathbf{a}_{-i}) = U_i(\mathbf{a}), \\ W & \text{otherwise.} \end{cases}$$

The payoff information on which agent *i* bases its decision lies in  $\mathcal{I}_i = S^{A_i}$ . The interpretation is that, if the strategy profile generating the payoff information is  $\mathbf{a} \in \mathcal{A}$ , the payoff information agent *i* has available is  $(\mathcal{C}_i(a'_i, \mathbf{a}), a'_i \in \mathcal{A}_i)$ . To simplify notation, for fixed  $\mathbf{a} \in \mathcal{A}$ , we denote the payoff information vector  $(\mathcal{C}_i(a'_i, \mathbf{a}), a'_i \in \mathcal{A}_i)$  by  $\mathcal{C}_i(\mathbf{a})$ .

*Remark:* Clearly, the payoff information we assume is less restrictive than the payoff vector  $(U_i(a'_i; \mathbf{a}_{-i}), a'_i \in \mathcal{A}_i)$  assumed in [57, 59], but is more stringent than that of the completely uncoupled payoff-based learning rules, e.g., [79, 80, 87]. We argue that in some scenarios, even though agents cannot determine the exact payoffs for the strategies not played, they might be able to determine which strategies could have yielded higher (or smaller) payoffs than the previously chosen strategy.

For example, consider an interdependent security game in which each agent chooses a combination of security measures from a set of available security measures for its own protection [72], e.g., cybersecurity measures including incoming-traffic monitoring and intrusion detection systems. In this case, based on the number of successful attacks a agent suffers, the resulting losses, as well as the costs of various security measures, it might be able to determine if another combination of security measures would have achieved a lower overall cost.

We find it convenient to define following mappings we use in the subsequent sections: For each agent  $i \in \mathcal{P}$ , let  $BR_i : \mathcal{A} \to 2^{\mathcal{A}_i}$  and  $IR_i : \mathcal{A} \to 2^{\mathcal{A}_i}$ , where

$$BR_i(\mathbf{a}) = \{a'_i \in \mathcal{A}_i \mid \mathcal{C}_i(a'_i, \mathbf{a}) = B\} \text{ and}$$
$$IR_i(\mathbf{a}) = \{a'_i \in \mathcal{A}_i \mid \mathcal{C}_i(a'_i, \mathbf{a}) = I\}.$$

Clearly,  $BR_i(\mathbf{a})$  is the set of better replies for agent *i* given the strategy profile of the other agents  $\mathbf{a}_{-i}$ .

In Sections 7.4 and 7.5, we assume that the payoff information available to the agents, i.e.,  $C_i(\tilde{\mathbf{a}}^i(t))$ , is accurate. In other words, the agents can correctly determine  $BR_i(\tilde{\mathbf{a}}^i(t))$  and  $IR_i(\tilde{\mathbf{a}}^i(t))$ . In Section 7.6, we relax this assumption and consider the scenario where agents cannot perfectly determine these sets and, as a result, make mistakes.

In this section, we describe the better-reply rules we study. Recall that  $\mathcal{T}^i = \{T_k^i, k \in \mathbb{N}\} \subseteq \mathbb{N}$  is the update time sequence of agent *i*.

At time t = 1, the agents choose their initial action profile  $\mathbf{a}(1)$  according to some distribution  $\mathcal{G}$  over  $\mathcal{A}$ . Subsequently, agents revise their strategies according to the following rule: Fix  $\underline{\epsilon} > 0$  and  $\beta_i : S^{A_i} \to \Delta(\mathcal{A}_i), i \in \mathcal{P}$ . The mappings  $\beta_i(\mathbf{c}_i) = (\beta_i(a_i; \mathbf{c}_i), a_i \in \mathcal{A}_i)$  are used to determine the mixed strategy to be employed when there is a better reply.

#### **GBRR-I** Rule

- if  $BR_i(\tilde{\mathbf{a}}^i(t)) = \emptyset$ 
  - choose  $a_i(t) = a_i(t-1)$ ;
- else (i.e.,  $BR_i(\tilde{\mathbf{a}}^i(t)) \neq \emptyset$ )
  - choose  $a_i(t) = a_i$  with probability  $\beta_i(a_i; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t))) \ge \underline{\epsilon}$  for each  $a_i \in BR_i(\tilde{\mathbf{a}}^i(t))$ , and  $a_i(t) = a_i(t-1)$  with the remaining probability;

Note that when there is no better reply, i.e.,  $BR_i(\tilde{\mathbf{a}}^i(t)) = \emptyset$ , agent *i* plays the same action  $a_i(t-1)$  at time *t*. On the other hand, if there exists at least one better reply, agent *i* (a) picks each of the better replies with probability at least  $\underline{\epsilon}$  or (b) continues to play the same action  $a_i(t-1)$  with probability  $1 - \sum_{a_i \in BR_i(\tilde{\mathbf{a}}^i(t))} \beta(a_i; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t)))$ . Thus, it plays a mixed strategy.

Throughout the chapter, we assume

$$\max_{i\in\mathcal{P}}\left(\max_{\mathbf{a}\in\mathcal{A}}\sum_{a_{i}'\in BR_{i}(\mathbf{a})}\beta_{i}(a_{i}';\mathcal{C}_{i}(\mathbf{a}))\right)=:\mu<1.$$

This guarantees that, even when the set of better replies is nonempty, the agent continues to play the same action  $a_i(t-1)$  with strictly positive probability.

In the GBRR-I rule, the same action  $a_i(t-1)$  is chosen at time t whenever  $BR_i(\tilde{\mathbf{a}}^i(t)) = \emptyset$ . In the following slightly modified rule, which we call the GBRR-II rule, we relax this constraint and allow additional exploration among best responses. Assume  $\varepsilon^* > 0$  and let  $\vartheta_i : S^{A_i} \to \Delta(\mathcal{A}_i), i \in \mathcal{P}$ . The role of the mappings  $\vartheta_i(\mathbf{c}_i) = (\vartheta_i(a_i; \mathbf{c}_i), a_i \in \mathcal{A}_i), i \in \mathcal{P}$ , is similar to that of  $\beta_i, i \in \mathcal{P}$ , and they determine the mixed strategies to be played by the agents when there are multiple best responses.

### **GBRR-II Rule:** Set $\theta_i$ = TRUE at time t = 1.

• if  $BR_i(\tilde{\mathbf{a}}^i(t)) = \emptyset$ 

- if  $|IR_i(\tilde{\mathbf{a}}^i(t))| = 1$  (i.e.,  $IR_i(\tilde{\mathbf{a}}^i(t)) = \{a_i(t-1)\}$ )
  - \* choose  $a_i(t) = a_i(t-1)$ ;

- else (i.e., 
$$|IR_i(\tilde{\mathbf{a}}^i(t))| > 1)$$

\* if 
$$\theta_i = \text{TRUE}$$

 $\circ \text{ choose each action } a_i \in IR_i(\tilde{\mathbf{a}}^i(t)) \text{ with probability } \vartheta(a_i; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t))) \ge \\ \varepsilon^{\star} (\text{with } \sum_{a_i \in IR_i(\tilde{\mathbf{a}}^i(t))} \vartheta(a_i; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t))) = 1); \qquad (*) \\ \diamond \text{ set } \theta_i = \text{FALSE}; \\ \ast \text{ else (i.e., } \theta_i = \text{FALSE}) \\ \diamond \text{ choose } a_i(t) = a_i(t-1) ;$ 

• else (i.e.,  $BR_i(\tilde{\mathbf{a}}^i(t)) \neq \emptyset$ )

- choose  $a_i(t) = a_i$  with probability  $\beta(a_i; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t))) \ge \underline{\epsilon}$  for each  $a_i \in BR_i(\tilde{\mathbf{a}}^i(t))$ , and  $a_i(t) = a_i(t-1)$  with the remaining probability;

```
- set \theta_i = TRUE;
```

The key difference between the two update rules is that GBRR-II permits the agent to explore among the best responses in some cases even when there is no better reply (case (\*)).

These rules are quite intuitive; an agent picks every better reply with positive probability (without knowing the exact payoffs achieved by those better replies). But, if no better reply is found, the agent either keeps playing the same action (GBRR-I) or selects one of the best responses (GBRR-II).

Modeling mixed strategies in GBRRs – For the completeness of exposition and analysis, we describe how we model mixed strategies exercised by the agents in GBRRs: For every  $i \in \mathcal{P}$ , define  $\mathbf{u}^i := (u_t^i, t \in \mathbb{N}^+)$  and  $\mathbf{v}^i := (v_t^i, t \in \mathbb{N}^+)$  to be two sequences of independent uniform random variables over (0, 1]. We assume that  $(\mathbf{u}^i, \mathbf{v}^i), i \in \mathcal{P}$ , are mutually independent.

Suppose that agent *i* revises its strategy at time  $t \in \mathcal{T}^i$  and  $BR_i(\tilde{\mathbf{a}}^i(t)) \neq \emptyset$ . agent *i* first orders the better replies, for example, by increasing index. We denote the *ordered* set of better replies by  $OBR_i(\tilde{\mathbf{a}}^i(t)) = \{a_i^{(1)}, a_i^{(2)}, \ldots, a_i^{(m^*)}\}$ , where  $m^*$ is the number of better replies. It then chooses

$$a_i(t) = \begin{cases} a_i^{(l)} & \text{if } u_t^i \in (\Lambda(l-1), \Lambda(l)), \ l = 1, 2, \dots, m^*, \\ a_i(t-1) & \text{otherwise}, \end{cases}$$

where  $\Lambda(l) := \sum_{l'=1}^{l} \beta\left(a_i^{(l')}; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t))\right), \ l = 1, \dots, m^*.$ 

Under GBRR-II, agent *i*'s mixed strategy in case (\*) is handled in an analogous manner. agent *i* first orders the best responses in  $IR_i(\tilde{\mathbf{a}}^i(t))$ , which we denote by  $OIR_i(\tilde{\mathbf{a}}^i(t)) = \{a_i^{(1)}, a_i^{(2)}, \dots, a_i^{(m^{\dagger})}\}$ , where  $m^{\dagger}$  is the number of best responses. Then, it picks  $a_i(t) = a_i^{(l)}, l = 1, 2, \dots, m^{\dagger}$ , if  $v_t^i \in (\Upsilon(l-1), \Upsilon(l)]$ , where

$$\Upsilon(l) := \sum_{l'=1}^{l} \vartheta \left( a_i^{(l')}; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t)) \right), \ l = 1, \dots, m^{\dagger}.$$

We would like to point out one observation that we make use of in our analy-

sis. Because mixed strategies are implemented using mutually independent uniform random variables, given a mixed strategy selected by agent i at some time t, the action picked by the agent is conditionally independent of the past history.

Convergent sequences of strategy profiles – For each  $t \in \mathbb{N}$ , let  $\mathcal{A}^t = \prod_{t'=1}^t \mathcal{A}$ . Similarly,  $\mathcal{A}^{\infty} = \prod_{t'=1}^{\infty} \mathcal{A}$ . When the agents revise their strategies according to fixed update rules described above, together with the revision processes  $\mathbb{N}_i, i \in \mathcal{P}$ , and (the distribution of) initial action profile  $\mathbf{a}(1)$ , the update rules  $\mathbf{S} = (S_i, i \in \mathcal{P})$  induces a distribution over  $\mathcal{A}^{\infty}$  (with a suitable  $\sigma$ -field  $\tilde{\mathcal{F}}$  on  $\mathcal{A}^{\infty}$ ); let  $\mathbf{\vec{a}} = (\mathbf{a}(t), t \in \mathbb{N})$  be the sequence of action profiles played by the agents.

We first define some subsets of  $\mathcal{A}^{\infty}$ , which are of interest to us. For each  $t \in \mathbb{N}$ , let

$$\mathcal{A}_t^{\infty} := \{ \bar{\mathbf{a}} \in \mathcal{A}^{\infty} \mid \bar{\mathbf{a}}(t') = \mathbf{a}^{\star} \forall t' \ge t \text{ for some } \mathbf{a}^{\star} \in \mathcal{A}_{NE} \}.$$

Note that these sets are increasing in t, i.e.,  $\mathcal{A}_t^{\infty} \subsetneq \mathcal{A}_{t+1}^{\infty}$  for all  $t \in \mathbb{N}$ . In addition, define

$$\mathcal{A}^{\infty}_{\star} := \bigcup_{t \in \mathbb{N}} \mathcal{A}^{\infty}_{t}$$
$$= \{ \bar{\mathbf{a}} \in \mathcal{A}^{\infty} \mid \bar{\mathbf{a}}(t') = \mathbf{a}^{\star} \ \forall \ t' \ge T^{\star} \text{ for some } \mathbf{a}^{\star} \in \mathcal{A}_{NE} \text{ and } T^{\star} < \infty \}.$$

Simply put,  $\mathcal{A}_t^{\infty}$  (resp.  $\mathcal{A}_{\star}^{\infty}$ ) is the set of strategy profile sequences that converge to a Nash equilibrium by time t (resp. at some finite time).

#### 7.4 Convergence under GBRR – Synchronous Update Case

In this section, we start with a simpler scenario in which every agent receives the payoff information  $C_i(\mathbf{a}(t-1))$  at each  $t \in \mathbb{N}^+$ , and updates its strategy according to a GBRR rule. In other words,  $\mathcal{T}^i = \mathbb{N}$  for all  $i \in \mathcal{P}$  and  $\tilde{\mathbf{a}}^i(t) = \mathbf{a}(t-1)$  for all  $i \in \mathcal{P}$  and  $t \in \mathbb{N}^+$ . We will consider more general settings in the following section.

Let  $\vec{\mathbf{a}} = (\mathbf{a}(t), t \in \mathbb{N})$  be the sequence of action profiles selected by the agents using a GBRR rule (either GBRR-I or GBRR-II). The following theorem guarantees the *almost sure convergence* (or *convergence with probability 1*) of the played action profiles  $\mathbf{a}(t), t \in \mathbb{N}$ , to a Nash equilibrium as  $t \to \infty$  [58]. Its proof can be found in [85] and is omitted here. Instead we will prove the convergence results under the asynchronous update scenario.

**Theorem 7.2.** Suppose that the game  $\mathbb{G}$  is generalized weakly acyclic. Then, for any fixed initial distribution  $\boldsymbol{\nu} \in \Delta(\mathcal{A})$ , we have  $\mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}^{\infty}_{\star}\right] = 1$ .

In addition to the almost sure convergence of action profiles, the following theorem states that the probability that the action profile has not converged to a Nash equilibrium by time  $t \in \mathbb{N}$  decays geometrically with t.

**Theorem 7.3.** Suppose that the game  $\mathbb{G}$  is generalized weakly acyclic. Then, there exist  $C < \infty$  and  $\zeta \in (0, 1)$  such that, regardless of the initial distribution  $\boldsymbol{\nu} \in \Delta(\mathcal{A})$ ,  $\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_t^{\infty}\right] \leq \min(1, C \cdot \zeta^t)$  for all  $t \in \mathbb{N}$ .

Theorem 7.3 implies that the expected convergence time  $\mathbb{E}[N_{conv}]$  is finite, where  $N_{conv} = \inf\{t \in \mathbb{N} \mid \vec{\mathbf{a}} \in \mathcal{A}_t^{\infty}\}$  is the time it takes for the agents to reach a Nash equilibrium and remain there afterward.

Before we proceed, let us comment on the convergence rate, which is captured by  $\zeta$ . Recall from Section 7.1 that, for  $\mathbf{a} \notin \mathcal{A}_{NE}$ , the length of a shortest generalized better reply path  $h^*(\mathbf{a})$  to a Nash equilibrium is  $L^*(\mathbf{a})$ . The parameter  $\zeta$ in Theorem 7.3 is shaped by  $\max_{\mathbf{a} \in \mathcal{A} \setminus \mathcal{A}_{NE}} L^*(\mathbf{a})$ . A possible interpretation of this observation is as follows: One can view the length of a generalized better reply path from strategy profile  $\mathbf{a}$  to a Nash equilibrium as a measure of how "difficult" it is to reach the Nash equilibrium from the action profile following the generalized better reply path, and a shortest generalized better reply path offers the "easiest" path to a Nash equilibrium. Hence, the lower bound on convergence rate captured by  $\zeta$ is influenced by the "most difficult" strategy profile from which the agents need to find the "easiest" path to a Nash equilibrium.

**Theorem 7.4.** Suppose that the game  $\mathbb{G}$  is not generalized weakly acyclic. Then, there exists at least one strategy profile  $\mathbf{a}' \notin \mathcal{A}_{NE}$  such that, under the GBRR-I rule,

$$\mathbb{P}\left[\mathbf{a}(t) \notin \mathcal{A}_{NE} \text{ for all } t \in \mathbb{N} \mid \mathbf{a}(1) = \mathbf{a}'\right] = 1$$

**Proof.** Since the game is assumed to be not generalized weakly acyclic, there exists a strategy profile  $\mathbf{a}^+ \notin \mathcal{A}_{NE}$  with no generalized better reply path to a Nash equilibrium. From the description of the GBRR-I rule, it is clear that any admissible transition from a strategy profile  $\mathbf{a}^1$  to another strategy profile  $\mathbf{a}^2$  ( $\mathbf{a}^1 \neq \mathbf{a}^2$ ) constitutes a generalized better reply path ( $\mathbf{a}^1, \mathbf{a}^2$ ). Therefore, if  $\mathbf{a}(1) = \mathbf{a}^+$ , it is not possible to reach any Nash equilibrium under the GBRR-I rule.

Note that

$$\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}^{\infty}_{\star} \mid \mathbf{a}(1) = \mathbf{a}'\right] \ge \mathbb{P}\left[\mathbf{a}(t) \notin \mathcal{A}_{NE} \; \forall \; t \in \mathbb{N} \mid \mathbf{a}(1) = \mathbf{a}'\right]$$

and, hence,

$$\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}^{\infty}_{\star} \mid \mathbf{a}(1) = \mathbf{a}'\right] = 1.$$

To rephrase it, conditional on  $\{\mathbf{a}(1) = \mathbf{a}'\}$ , the probability that  $\mathbf{a}(t), t \in \mathbb{N}$ , converge to a Nash equilibrium is equal to zero.

Theorems 7.2 and 7.4 together state that, if  $\boldsymbol{\nu}(\mathbf{a}') > 0$  for all  $\mathbf{a}' \in \mathcal{A}$ , i.e., every strategy profile is selected with positive probability at time t = 1, the GBRR-I rule guarantees the almost sure convergence of action profiles to a Nash equilibrium *if and only if* the game is generalized weakly acyclic.

A similar, but somewhat weaker result holds for the GBRR-II rule: We call a sequence of strategy profiles  $(\mathbf{a}^1, \mathbf{a}^2, \dots, \mathbf{a}^L)$  a *weak* generalized better reply path if, for every  $\ell \in \{1, \dots, L-1\}$ , there exists a subset of agents  $I'_{\ell} \subseteq \mathcal{P}$  such that

i. 
$$a_i^{\ell} \neq a_i^{\ell+1}$$
 and  $U_i(\mathbf{a}^{\ell}) \leq U_i(a_i^{\ell+1}, \mathbf{a}_{-i}^{\ell})$  for all  $i \in I'_{\ell}$ , and

ii.  $a_i^{\ell} = a_i^{\ell+1}$  for all  $i \in \mathcal{P} \setminus I'_{\ell}$ .

A game is said to be generalized weakly  $\operatorname{acyclic}^+$  if, for every  $\mathbf{a}' \notin \mathcal{A}_{NE}$ , there exists a weak generalized better reply path  $(\mathbf{a}^1, \mathbf{a}^2, \dots, \mathbf{a}^L)$  with  $\mathbf{a}^1 = \mathbf{a}'$  and  $\mathbf{a}^L \in \mathcal{A}_{NE}$ . One can show that if the game is not generalized weakly  $\operatorname{acyclic}^+$ , there exists at least one strategy profile  $\mathbf{a}^{\dagger} \notin \mathcal{A}_{NE}$  such that  $\mathbb{P}[\mathbf{a}(t) \notin \mathcal{A}_{NE} \forall t \in \mathbb{N} \mid \mathbf{a}(1) =$  $\mathbf{a}^{\dagger}] = 1$  when the agents employ the GBRR-II rule for revising strategies. *Remark:* The class of generalized weakly acyclic<sup>+</sup> games is strictly larger than that of generalized weakly acyclic games and there are generalized weakly acyclic<sup>+</sup> games that are not generalized weakly acyclic, for which GBRR-II guarantees the almost sure convergence of action profiles to a Nash equilibrium. In this sense, GBRR-II possesses a somewhat stronger convergence property than GBRR-I.

# 7.5 Convergence under GBRR – Asynchronous Update Case

In the previous section, we assumed that all agents update their strategies at every  $t \in \mathbb{N}^+$  based on the action profile played at time t - 1. As already indicated in Chapter 6, in some settings this assumption might not hold. In this section, we extend the model in two directions: We allow (i) asynchronous updates of strategies by the agents and (ii) time-varying delays experienced by payoff information available to the agents. Note that the latter implies that the agents might base their decisions on outdated payoff information at times. We mention that there are some related studies that examine the effects of delays in evolutionary games, e.g., [94] and asynchronous distributed computation of average values, e.g., [48,93].

### 7.5.1 Model with asynchronous updates and delays

There are many different ways in which one can capture and model asynchronous operation and delays in payoff information. In this subsection, we describe the model we assume for our study. For concreteness, we explain it using an example.



Figure 7.2: Example – two interacting systems controlled by separate agents.

Consider the system in Figure 7.2. There are two *interacting* systems, each of which is controlled by a agent (controller); for each agent i, i = 1, 2, a corresponding system i generates payoffs for agent i. Obviously, the case in which a single system provides payoffs to all agents is a special case of this more general setting.

In this scenario, depending on the structure of the system, new strategies adopted by agents might experience delays before taking effect at the systems. In addition, the payoffs to agents (e.g., the current states of the systems) may not be observable to the agents instantaneously. In other words, they can only observe delayed past payoffs.

To model this scenario, we introduce two types of delays – (i) forward delays and (ii) feedback delays. A forward delay refers to the amount of time that elapses before a agent's strategy goes into effect at a system after it is adopted. On the other hand, a feedback delay is the amount of time it takes for payoff information to become available to a agent after it is generated. These are illustrated in Figure 7.3 for the example in Figure 7.2. In the figure, after agent i, i = 1, 2, updates its



Figure 7.3: Forward and feedback delays in the system.

strategy, the forward delay experienced before system i sees the new strategy is shown as a (red or green) solid arrow (with label 'A'), whereas the *additional* forward delay to the other system appears as a dotted arrow. The feedback delay experienced by payoff information generated by a system before the corresponding agent observes it is shown as a (blue or purple) solid arrow (with label 'F').

We shall model these delays using random variables as follows. First, recall that the agents choose their initial action profile  $\mathbf{a}(1) = (a_i(1), i \in \mathcal{P})$  according to some distribution  $\boldsymbol{\nu}$  and the sequence of agent *i*'s update times, namely  $\mathcal{T}^i$ , is determined by a discrete-time counting process. Denote the inter-update times of agent *i* by  $U_k^i := T_{k+1}^i - T_k^i$ ,  $k \in \mathbb{N}$ .
Each inter-update time  $U_k^i$  is given by a sum of two random variables  $-X_k^i$ and  $Y_k^i$ . The random variable  $X_k^i$  models the forward delay experienced by the kth action of agent i chosen at time  $T_k^i$ , i.e.,  $a_i(T_k^i)$ ; we assume that  $a(T_k^i)$  goes into effect at system i at time  $T_k^i + X_k^i =: R_k^i$  after incurring a forward delay of  $X_k^i$ , at which point system i generates payoff information for agent i. This payoff information undergoes a delay of  $Y_k^i$  and is observed by agent i at the next update time  $T_{k+1}^i(=R_k^i+Y_k^i)$ . Throughout this section, we assume that  $X_k^i \ge 0$  and  $Y_k^i \ge 1$ for all  $i \in \mathcal{P}$  and  $k \in \mathbb{N}$ . For notational simplicity, we denote the pair  $(X_k^i, Y_k^i)$  by  $Z_k^i$  and the sequences  $\{Z_k^i, k \in \mathbb{N}\}$  and  $\{R_k^i, k \in \mathbb{N}\}$  by  $\mathcal{Z}^i$  and  $\mathcal{R}^i$ , respectively.

Similarly, the actions chosen by agent i at time  $T_k^i$  might not be seen by another system  $j, j \neq i$ , immediately; instead, it takes effect at system j at time  $T_k^i + V_k^{i,j} =: R_k^{i,j}$ , where  $V_k^{i,j}$  is the forward delay that action  $a_i(T_k^i)$  experiences before influencing system j. In the example shown in Figure 7.2, once the new action  $a_i^{\text{new}}$ of agent i starts affecting system i, through the interaction between the systems, the other system will also be affected by  $a_i^{\text{new}}$  possibly after some additional delay. In this case, we will have  $X_k^i \leq V_k^{i,j}$ . For every  $i, j \in \mathcal{P}, i \neq j$ , let  $\mathcal{V}^{i,j} := \{V_k^{i,j}, k \in \mathbb{N}\}$ and  $\mathcal{R}^{i,j} := \{R_k^{i,j}, k \in \mathbb{N}\}$ .

In order to complete the model, we need to take care of initial conditions. Because the actions chosen by the agents at time t = 1 might not take effect at the systems right away, we impose the following initial conditions: For each system i, we assume that there is some strategy profile  $\bar{\mathbf{a}}^i = (\bar{a}^i_j, j \in \mathcal{P}) \in \mathcal{A}$  that is in place at the system. In other words, for each  $i \in \mathcal{P}$ , system i sees action  $\bar{a}^i_i$  from agent itill  $R^i_1 = 1 + X^i_1$  and  $\bar{a}^i_j$  from agent  $j, j \neq i$ , till  $R^{j,i}_1 = 1 + V^{j,i}_1$ . Although we can easily handle scenarios in which these strategy profiles are random, here we assume that they are deterministic for the ease of exposition.

Remark: The settings studied in the previous section can be viewed as a special case of the general settings described in this section with  $X_k^i = V_k^{i,j} = 0$  and  $Y_k^i = 1$  (so that  $U_k^i = X_k^i + Y_k^i = 1$ ) for all  $i, j \in \mathcal{P}$  ( $i \neq j$ ) and  $k \in \mathbb{N}$ . To rephrase it, not only do the agents update strategies synchronously, there is no forward delay (and the strategies take effect immediately) and the feedback delay is always one.

In order to make progress, we assume that  $\mathbb{N}_i, \mathbb{Z}^i$ , and  $\mathcal{V}^{i,j}$  satisfy the following assumptions.

- A1 The random variables  $\{\mathcal{Z}^i, \mathcal{V}^{i,j}, j \neq i\}$  for different agents  $i \in \mathcal{P}$  are mutually independent.
- A2. For every  $i \in \mathcal{P}$ ,  $\mathbb{P}\left[U_k^i < \infty, V_k^{i,j} < \infty \text{ for all } j \neq i \text{ and } k \in \mathbb{N}\right] = 1.$
- A3.  $\mathbb{P}\left[R_k^{i,j} < R_{k+1}^{i,j} \text{ for all } k \in \mathbb{N}\right] = 1 \text{ for all } i, j \in \mathcal{P}, j \neq i.$
- A4. Let  $\mathbf{H}_{k}^{i} := \left( (Z_{\ell}^{i}, V_{\ell}^{i,j}, j \neq i), 1 \leq \ell \leq k \right)$ . There exist  $\eta > 0$  and  $\Delta_{\eta} < \infty$  such that, for all  $i \in \mathcal{P}, k \in \mathbb{N}$  and  $\varphi \in \mathbb{Z}_{+}$ ,

$$\mathbb{P}\left[U_{k+1}^{i} \leqslant \Delta_{\eta} + \varphi, V_{k+1}^{i,j} \leqslant \Delta_{\eta} + \varphi \text{ for all } j \neq i \mid U_{k+1}^{i} \geqslant \varphi, \mathbf{H}_{k}^{i}\right] \geqslant \eta$$

A5. There exist  $\delta > 0$  and  $\Delta_{\delta} < \Delta_{\eta}$  such that, for all  $i \in \mathcal{P}$  and  $k \in \mathbb{N}$ ,

$$\mathbb{P}\left[X_{k+1}^{i} \leqslant \Delta_{\delta} \mid \mathbf{H}_{k}^{i}\right] \ge \delta$$
(7.2)

and

$$\mathbb{P}\left[V_{k+1}^{i,j} \ge \Delta_{\delta} \text{ for all } j \neq i \mid \mathbf{H}_{k}^{i}\right] \ge \delta.$$
(7.3)

A6. For all  $0 \leq u \leq u' < \infty$ ,

$$\mathbb{P}\left[X_{k+1}^{i} \leqslant x \mid U_{k+1}^{i} \leqslant u, \mathbf{H}_{k}^{i}\right] \ge \mathbb{P}\left[X_{k+1}^{i} \leqslant x \mid U_{k+1}^{i} \leqslant u', \mathbf{H}_{k}^{i}\right]$$
for all  $x \in \mathbb{Z}_{+}$ ,

i.e., the conditional distribution of  $X_{k+1}^i$  is stochastically larger given  $\{U_{k+1}^i \leq u'\}$  than  $\{U_{k+1}^i \leq u\}$  [91].

Assumption A2 ensures that every agent updates infinitely many times with probability 1. Assumption A3 guarantees that agents see the effects of new actions in the same order they were adopted. Assumption A4 essentially implies that the distributions of random variables  $\{X_k^i, V_k^{i,j}, j \neq i\}, k \in \mathbb{N}$ , do not have a heavy tail. Assumption A5 means that, with positive probability, agent *i*'s action affects its own payoff no later than those of other agents.

Although these assumptions are technical in nature, we feel that they are not restrictive and are likely to hold in many cases of practical interest. For example, when the strategy update times are given by suitable delayed renewal processes with constant forward delays (with  $X_k^i \leq V_k^{i,j}$  for all  $j \neq i$ ), the above assumptions hold.

# 7.5.2 Convergence of action profiles with asynchronous updates and delays

Let  $\vec{\mathbf{a}} = (\mathbf{a}(t), t \in \mathbb{N})$  be the sequence of action profiles played by the agents using a GBRR rule. The following two theorems suggest that when the game is generalized weakly acyclic and a GBRR rule is employed for updating strategies, neither payoff information delays nor asynchronous updates of strategies among the agents prevent their action profiles from reaching a Nash equilibrium with probability 1, under Assumptions A1 through A6.

**Theorem 7.5.** Suppose that the game  $\mathbb{G}$  is generalized weakly acyclic and that Assumptions A1 through A6 hold. Then, for any initial distribution  $\boldsymbol{\nu} \in \Delta(\mathcal{A})$ ,  $\mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}^{\infty}_{\star}\right] = 1.$ 

**Proof.** A proof of the theorem is provided in Section 7.7.

**Theorem 7.6.** Suppose that the game  $\mathbb{G}$  is generalized weakly acyclic and that Assumptions A1 through A6 are in place. Then, there exist  $C' < \infty$  and  $\overline{\zeta} \in (0,1)$ such that, for any initial distribution  $\boldsymbol{\nu} \in \Delta(\mathcal{A})$ ,  $\mathbb{P}[\mathbf{\vec{a}} \notin \mathcal{A}_t^{\infty}] \leq \min(1, C' \cdot \overline{\zeta}^t)$  for all  $t \in \mathbb{N}$ .

Theorem 7.6 follows directly from Corollary 2 in the proof of Theorem 7.5 as explained in Section 7.7.

*Remark:* Theorem 7.5 and 7.6 are established under Assumptions A1 through A6 for generalized weakly acyclic games. However, as we explain in Section 7.7, for weakly acyclic games, they hold under Assumptions A1 through A4.

## 7.6 Case with Erroneous Payoff Information

In the previous sections, we assumed that accurate payoff information  $C_i(\tilde{\mathbf{a}}^i(t))$ is available to the agents for updates. Although this is a reasonable assumption in some cases, there are other scenarios, in which agents might not be able to reliably determine better replies. To address this partially, we replace it with a somewhat weaker assumption that the agents are able to make more reliable determinations over time, for instance, by observing its own payoff history.

Clearly, when the agents make mistakes with the determinations of  $C_i(\tilde{\mathbf{a}}^i(t))$ , it is unlikely that the strategy profile will converge to a Nash equilibrium and remain there forever with probability 1. Instead, we will show that, under some conditions, the probability that the agents play a Nash equilibrium at time  $t \in \mathbb{N}$  goes to one over time. Although we suspect that our result can be generalized to settings with asynchronous updates studied in the previous section under suitable assumptions, here we only consider settings with synchronous updates studied in Section 7.4.

## 7.6.1 Preliminaries

Modeling unreliable payoff information – We assume that, for all  $t-1 \in$ N, given the action profile  $\mathbf{a}(t-1)$ , (a) the events of having false payoff information at time t are conditionally independent of the past and (b) their probabilities depend only on the action profile  $\mathbf{a}(t-1)$ . This is modeled as follows: First, the probabilities of erroneous classification of actions are given by mappings  $\mathbf{q}_t^i : \mathcal{A} \times \mathcal{A}_i \to \Delta(\mathcal{S})$ ,  $t \in \mathbb{N}^+$ . The interpretation is that (i)  $\mathbf{q}_t^i(\mathbf{a}, a_i') := (q_t^i(\varsigma; \mathbf{a}, a_i'), \varsigma \in \mathcal{S})$  is a probability distribution over  $\mathcal{S}$  and (ii) conditional on the event  $\{\mathbf{a}(t-1) = \mathbf{a}\}$ , the probability of action  $a_i' \in \mathcal{A}_i$  being classified as  $\varsigma \in \mathcal{S}$  by agent i at time t is equal to  $q_t^i(\varsigma; \mathbf{a}, a_i')$ .

Second, for each agent  $i \in \mathcal{P}$ , we define an array of independent uniform(0,1] random variables  $\mathbf{W}^i = {\mathbf{w}_t^i, t \in \mathbb{N}^+}$ , where  $\mathbf{w}_t^i = (w_{t,a}^i, a \in \mathcal{A}_i)$  with  $w_{t,a}^i \sim$  uniform(0,1]. To facilitate the exposition, we order the elements in S as follows:  $\varsigma_1 = B, \varsigma_2 = I$  and  $\varsigma_3 = W$ . Then, action  $a'_i \in \mathcal{A}_i$  is classified as  $\varsigma_l, l = 1, 2, 3$ , by agent *i* at time  $t \in \mathbb{N}^+$  if

$$w_{t,a_i'}^i \in \left(\sum_{l'=1}^{l-1} q_t^i(\varsigma_{l'}; \mathbf{a}(t-1), a_i'), \sum_{l'=1}^l q_t^i(\varsigma_{l'}; \mathbf{a}(t-1), a_i')\right].$$

We note that, because the uniform random variables are mutually independent, given the action profile  $\mathbf{a}(t-1)$ , each action is incorrectly classified by a agent independently of each other at time t and also of the past history. This allows us to model the evolution of the action profile as a Markov process in our analysis.

Assumption 7.1. For every  $i \in \mathcal{P}$  and  $t \in \mathbb{N}^+$ ,  $\mathbf{q}_t^i(\mathbf{a}, a_i') > \mathbf{0} := \begin{bmatrix} 0 & 0 & 0 \end{bmatrix}^T$  for all  $\mathbf{a} \in \mathcal{A}$  and  $a_i' \in \mathcal{A}_i$ .

**Nonhomogeneous Markov chain** – Before presenting our main results, we first introduce some terminologies we borrow from [53]. Once the mappings  $\mathbf{q}_{t}^{i}, i \in \mathcal{P}$  and  $t \in \mathbb{N}^{+}$ , and  $\boldsymbol{\beta}_{i}$  and  $\boldsymbol{\vartheta}_{i}, i \in \mathcal{P}$ , are fixed, the evolution of action profile  $\{\mathbf{a}(t), t \in \mathbb{N}\}$  can be modeled as a *nonhomogeneous* (discrete-time) Markov chain [44], where the transition matrix at time  $t \in \mathbb{N}$ , denoted by  $\mathbf{P}(t)$ , is determined by the mappings  $\boldsymbol{\beta}_{i}$  and  $\boldsymbol{\vartheta}_{i}, i \in \mathcal{P}$ , and  $\mathbf{q}_{t}^{i}, i \in \mathcal{P}$  and  $t \in \mathbb{N}^{+}$ . Because we assume  $\mathbf{q}_{t}^{i}(\mathbf{a}, a_{i}') > \mathbf{0}$ , every action  $a_{i}'$  of agent i will be classified as a better reply with positive probability at every  $t \in \mathbb{N}^{+}$  and, consequently, we have  $P_{\mathbf{a}^{1}, \mathbf{a}^{2}}(t) > 0$  for all  $\mathbf{a}^{1}, \mathbf{a}^{2} \in \mathcal{A}$ .

In addition to the nonhomogeneous Markov chain  $\{\mathbf{a}(t), t \in \mathbb{N}\}$ , we define, for each  $t \in \mathbb{N}$ , a time homogeneous (discrete-time) Markov chain  $\mathbb{X}^t = \{\mathbf{x}^t(n), n \in \mathbb{N}\}$ with a common state space  $\mathcal{A}$  and a transition matrix  $\mathbf{P}^t = \mathbf{P}(t)$ . Since  $P_{\mathbf{a}^1, \mathbf{a}^2}^t > 0$  for all  $\mathbf{a}^1, \mathbf{a}^2 \in \mathcal{A}$  as mentioned above, the Markov chain  $\mathbb{X}^t$  is ergodic with a unique stationary distribution [58, 89], which we denote by  $\boldsymbol{\mu}^t = (\boldsymbol{\mu}^t(\mathbf{a}), \mathbf{a} \in \mathcal{A})$ . We call the Markov chain in which the agents make no mistake (i.e., the model assumed in Section 7.4) an *unperturbed* Markov chain. We denote the transition matrix of the unperturbed Markov chain by  $\mathbf{P}^0$ .

Assumption 7.2. There exists a decreasing, positive sequence  $(\epsilon_t, t \in \mathbb{N})$  such that (i)  $\lim_{t\to\infty} \epsilon_t = 0$  and (ii) for every  $i \in \mathcal{P}$ ,  $\mathbf{a} \in \mathcal{A}$ ,  $a'_i \in \mathcal{A}_i \setminus \{a_i\}$ , and  $\varsigma \neq C_i(a'_i, \mathbf{a})$ , there are constants  $c_i(\mathbf{a}, a'_i, \varsigma) > 0$  and  $\gamma_i(\mathbf{a}, a'_i, \varsigma) > 0$  satisfying

$$\lim_{t \to \infty} \frac{q_t^i(\varsigma; \mathbf{a}, a_i')}{\epsilon_t^{\gamma_i(\mathbf{a}, a_i', \varsigma)}} = c_i(\mathbf{a}, a_i', \varsigma),$$

i.e.,  $\mathbf{q}_t^i(\varsigma; \mathbf{a}, a_i') \sim c_i(\mathbf{a}, a_i', \varsigma) \cdot \epsilon_t^{\gamma_i(\mathbf{a}, a_i', \varsigma)}$ .

First, Assumption 7.2 implies that the probability of agent *i* making an erroneous determination goes to zero as  $t \to \infty$ , i.e.,

$$\limsup_{t \to \infty} \left( \max_{i \in \mathcal{P}} \left( \max_{a'_i \in \mathcal{A}_i, \mathbf{a} \in \mathcal{A}} \left( \sum_{\varsigma \neq \mathcal{C}_i(a'_i, \mathbf{a})} q_t^i(\varsigma; \mathbf{a}, a'_i) \right) \right) \right) = 0$$

Second, it states that, for all sufficiently large t, the transition probabilities  $P_{\mathbf{a}^1,\mathbf{a}^2}(t)$ can be well approximated using finite sums of power functions of  $\epsilon_t$ . In other words,  $P_{\mathbf{a}^1,\mathbf{a}^2}(t) \sim \sum_{k=1}^{K} c_k \cdot (\epsilon_t)^{r_k}$ , where  $c_k \in \mathbb{R}$  and  $r_k \ge 0$ . As a result, for all  $\mathbf{a}^1, \mathbf{a}^2 \in \mathcal{A}$ ,  $\mathbf{a}^1 \neq \mathbf{a}^2$ , there is  $r(\mathbf{a}^1, \mathbf{a}^2) \ge 0$  such that

$$0 < \lim_{t \to \infty} \frac{P_{\mathbf{a}^1, \mathbf{a}^2}(t)}{\epsilon_t^{r(\mathbf{a}^1, \mathbf{a}^2)}} < \infty,$$

and we can find  $0 < \underline{\xi} < \overline{\xi} < \infty$  such that, for all sufficiently large t,

$$\underline{\xi} \cdot \epsilon_t^{r(\mathbf{a}^1, \mathbf{a}^2)} < P_{\mathbf{a}^1, \mathbf{a}^2}(t) < \overline{\xi} \cdot \epsilon_t^{r(\mathbf{a}^1, \mathbf{a}^2)}.$$
(7.4)

**Resistance of paths** – A path from a strategy profile  $\mathbf{a}^1$  to another strategy profile  $\mathbf{a}^2$  is a sequence of strategy profiles  $(\mathbf{a}^{(1)}, \mathbf{a}^{(2)}, \dots, \mathbf{a}^{(M)})$  with  $\mathbf{a}^{(1)} = \mathbf{a}^1, \mathbf{a}^{(M)} =$  $\mathbf{a}^2$  and  $\mathbf{a}^{(\ell)} \neq \mathbf{a}^{(\ell+1)}$  for all  $\ell = 1, 2, \dots, M - 1$ .

**Definition 7.1.** The resistance of a path  $h(\mathbf{a}^1 \to \mathbf{a}^2) := (\mathbf{a}^{(1)} = \mathbf{a}^1, \mathbf{a}^{(2)}, \dots, \mathbf{a}^{(M)} = \mathbf{a}^2)$  from  $\mathbf{a}^1$  to  $\mathbf{a}^2$  is equal to  $r_p(h(\mathbf{a}^1 \to \mathbf{a}^2)) = \sum_{\ell=1}^{M-1} r(\mathbf{a}^{(\ell)}, \mathbf{a}^{(\ell+1)})$ . The resistance from  $\mathbf{a}^1$  to  $\mathbf{a}^2$ , denoted by  $\rho(\mathbf{a}^1, \mathbf{a}^2)$ , is defined to be the smallest resistance among the paths from  $\mathbf{a}^1$  to  $\mathbf{a}^2$ , i.e.,  $\rho(\mathbf{a}^1, \mathbf{a}^2) := \inf\{r_p(h(\mathbf{a}^1 \to \mathbf{a}^2)) \mid h(\mathbf{a}^1 \to \mathbf{a}^2)\}$  is a path from  $\mathbf{a}^1$  to  $\mathbf{a}^2\}$ .

**Definition 7.2.** Given a subset  $\tilde{\mathcal{A}} \subset \mathcal{A}$ , its co-radius is given by  $\tau(\tilde{\mathcal{A}}) = \max_{\mathbf{a}' \in \mathcal{A} \setminus \tilde{\mathcal{A}}} (\min_{\mathbf{a} \in \tilde{\mathcal{A}}} \rho(\mathbf{a}', \mathbf{a})).$ 

The co-radius  $\tau(\tilde{\mathcal{A}})$  is the maximum resistance that must be overcome to reach some strategy profile in  $\tilde{\mathcal{A}}$  from a strategy profile outside  $\tilde{\mathcal{A}}$ . Thus, it measures how "easy" it is for the agents to reach a strategy profile in  $\tilde{\mathcal{A}}$  starting from any strategy profile outside  $\tilde{\mathcal{A}}$ .

Define  $\kappa := \min_{\mathbf{a}^* \in \mathcal{A}_{NE}} \tau(\{\mathbf{a}^*\})$ , i.e., the smallest co-radius among all Nash equilibria. Based on the above observation,  $\kappa$  indicates how "easily" the agents can reach a Nash equilibrium starting from *any* strategy profile.

Minimum resistance W-tree – Construct a directed graph  $\mathcal{G} = (\mathcal{V}, \mathcal{E})$ , where  $\mathcal{V} = \mathcal{A}$  and the (directed) edge set  $\mathcal{E}$  contains all possible one-step transitions in the Markov chain  $\mathbb{X}^t$ , i.e.,  $(\mathbf{a}_1, \mathbf{a}_2) \in \mathcal{A} \times \mathcal{A}$  with  $P_{\mathbf{a}_1, \mathbf{a}_2}^t > 0$ . Recall that, from Assumption 7.1, we have  $P_{\mathbf{a}^1, \mathbf{a}^2}(t) > 0$  for all  $\mathbf{a}^1, \mathbf{a}^2 \in \mathcal{A}$ . Hence, the edge set  $\mathcal{E} = \mathcal{A} \times \mathcal{A}$ . Given a subgraph  $\mathcal{G}' = (\mathcal{V}', \mathcal{E}')$  of  $\mathcal{G}$ , where  $\mathcal{V}' \subseteq \mathcal{V}$  and  $\mathcal{E}' \subseteq \mathcal{E}$ , its resistance is defined as the sum of the resistances of all edges in the subgraph [53], and we denote it by  $\pi(\mathcal{G}')$ . In other words,  $\pi(\mathcal{G}') = \sum_{e \in \mathcal{E}'} r(e)$ , where r(e) is the resistance of (directed) edge e.

For every  $\mathbf{a} \in \mathcal{A}$ , let  $\mathcal{G}_T(\mathbf{a})$  be the set of *trees* in  $\mathcal{G}$  rooted at  $\mathbf{a}$ , such that there exists a path from every  $\mathbf{a}' \in \mathcal{A}$  to  $\mathbf{a}$ . The trees in  $\mathcal{G}_T(\mathbf{a})$  are called Wtrees rooted at  $\mathbf{a}$ . A minimum resistance W-tree in  $\mathcal{G}_T(\mathbf{a})$  is denoted by  $\Gamma^*(\mathbf{a})$ , i.e.,  $\Gamma^*(\mathbf{a}) \in \arg\min_{\Gamma(\mathbf{a})\in\mathcal{G}_T(\mathbf{a})} \pi(\Gamma(\mathbf{a}))$ . Let  $\pi_{\min}(\mathbf{a}) := \pi(\Gamma^*(\mathbf{a})) = \min_{\Gamma(\mathbf{a})\in\mathcal{G}_T(\mathbf{a})} \pi(\Gamma(\mathbf{a}))$ .

## 7.6.2 Main results

First, we state an auxiliary result that follows directly from Theorem 7.2 (which implies that Nash equilibria are absorbing states of the unperturbed Markov chain) and [95, Theorem 4].

**Theorem 7.7.** As  $t \to \infty$ ,  $\mu^t \to \mu^0$ , where  $\mu^0$  is a stationary distribution of the unperturbed Markov chain. In addition,  $\mu^0(\mathbf{a}^*) > 0$  if and only if (i)  $\mathbf{a}^* \in \mathcal{A}_{NE}$  and (ii)  $\pi_{\min}(\mathbf{a}^*) = \min_{\mathbf{a}' \in \mathcal{A}_{NE}} \pi_{\min}(\mathbf{a}')$ .

A Nash equilibrium  $\mathbf{a}^*$  with  $\mu^0(\mathbf{a}^*) > 0$  is said to be *stochastically stable* [53]. We denote the set of stochastically stable Nash equilibria by  $\mathcal{A}_{SS} \subseteq \mathcal{A}_{NE}$ .

We now prove that, under some conditions, the nonhomogeneous Markov chain  $\{\mathbf{a}(t), t \in \mathbb{N}\}\$  is strongly ergodic [44]; the strong ergodicity of the nonhomogeneous Markov chain means that its distribution converges to  $\boldsymbol{\mu}^0$  as  $t \to \infty$ . Thus, it tells us  $\mathbb{P}[\mathbf{a}(t) \in \mathcal{A}_{SS}] \to 1$  as  $t \to \infty$ , which in turn implies  $\lim_{t\to\infty} \mathbb{P}[\mathbf{a}(t) \in \mathcal{A}_{NE}] = 1$ 

because  $\mathcal{A}_{SS} \subseteq \mathcal{A}_{NE}$ .

Before we proceed, we introduce the assumption under which we establish the strong ergodicity of the nonhomogeneous Markov chain.

Assumption 7.3. For every  $i \in \mathcal{P}$ ,  $\mathbf{a} \in \mathcal{A}$ ,  $a'_i \in \mathcal{A}_i \setminus \{a_i\}$ , and  $\varsigma \neq \mathcal{C}_i(a'_i, \mathbf{a})$ , there exist some function  $f \in C^{\infty}$  and  $\epsilon^* > 0$  such that (i)  $q^i_t(\varsigma; \mathbf{a}, a'_i) = f(\epsilon_t)$  if  $0 < \epsilon_t < \epsilon^*$  and (ii)  $f(\epsilon) \sim \alpha \cdot \epsilon^{\beta}$  for some  $\alpha, \beta > 0$ . In addition,  $f'(\epsilon) \sim \alpha' \cdot \epsilon^{\beta'}$  for some  $\alpha' > 0$  and  $\beta' \in \mathbb{R}$ .

The first part of Assumption 7.3 is simply rehashing of Assumption 7.2. The assumption essentially states that both the probabilities of false classifications and their derivatives with respect to  $\epsilon_t$  asymptotically behave like power functions of  $\epsilon_t$ . Thus, for sufficiently large t, both the transition probabilities  $P_{\mathbf{a}^1,\mathbf{a}^2}(t)$  and their derivatives can be well approximated using *finite* sums of power functions of  $\epsilon_t$ . In addition, the assumption implies that the probability of correct classification of actions tends to one over time.

**Theorem 7.8.** Suppose that Assumptions 7.2 and 7.3 hold with  $\sum_{t \in \mathbb{N}} \epsilon_t^{\kappa} = \infty$ . Then, the nonhomogeneous Markov chain  $\{\mathbf{a}(t), t \in \mathbb{N}\}$  is strongly ergodic with limiting distribution  $\boldsymbol{\mu}^0$ . Consequently, it satisfies  $\lim_{t\to\infty} \mathbb{P}[\mathbf{a}(t) \in \mathcal{A}_{SS}] = 1$ .

**Proof.** A proof of Theorem 7.8 is provided in Section 7.9.

The condition  $\sum_{t \in \mathbb{N}} \epsilon_t^{\kappa} = \infty$  in Theorem 7.8 reveals the following interesting observation. When  $\kappa$  is larger, it is more difficult for agents to reach a Nash equilibrium in the worst case. This demands that the erroneous classification probabilities

decrease *slower*. The intuition behind this is that greater erroneous classification probabilities make it "easier" for the played action profile to transition from one strategy profile to another, thereby facilitating the exploration of new strategy profiles, including Nash equilibria.

*Remark:* Suppose that Assumption 7.2 does not hold and agents continue to make mistakes with probability that does not vanish. However, if the probability of making a mistake is sufficiently small, our results can be modified to demonstrate that, for large  $t \in \mathbb{N}$ , the probability that the chosen action profile  $\mathbf{a}(t)$  is a Nash equilibrium is close to one. Hence, even though the probability of playing a Nash equilibrium does not reach one as  $t \to \infty$  in this case, the probability will be close to one.

A question that arises is whether anything more can be said about the set of stochastically stable Nash equilibria  $\mathcal{A}_{SS}$ . For any  $\mathbf{a}^* \in \mathcal{A}_{NE}$ , let the states from which we can reach  $\mathbf{a}^*$  through zero resistance paths (GBRPs) be denoted as  $D_{\mathbf{a}^*} \subseteq \mathcal{A}$ , i.e.,

$$D_{\mathbf{a}^*} = \{ \mathbf{a} \in \mathcal{A} \mid \rho(\mathbf{a}, \mathbf{a}^*) = 0 \}$$

This represents the domain of attraction of a particular PSNE. By the GWAG assumption,  $\cup_{\mathbf{a}^* \in \mathcal{A}_{NE}} D_{\mathbf{a}^*} = \mathcal{A}$ , i.e., we can reach at least one PSNE from any non-equilibrium action profile by following a GBRP. Choose any  $\mathbf{a}' \in \mathcal{A}_{NE}$ . The minimum resistance W-tree  $\Gamma^*(\mathbf{a}')$  contains resistance paths from other PSNE(s) and non-PSNE action profiles. For any action profile in  $D_{\mathbf{a}'}$ , by definition we can construct zero resistance paths to  $\mathbf{a}'$ . However for reaching  $\mathbf{a}'$  from any action profile

outside of  $D_{\mathbf{a}'}$ , perturbations due to the error model is required. Therefore, nothing more can be inferred about  $\mathcal{A}_{SS}$  than already stated. It would in general depend both on the error model and the structure of the game itself. This is a major reason for considering the class of RSEM rules which seem to exhibit more favourable properties in similar settings, in particular, the domain of attraction of each PSNE is better characterisable.

## 7.7 Proof of Theorem 7.5

Before we present the proof, we introduce the following notation. Consider a mapping  $\mathcal{L} : \mathcal{A}^* \to \mathbb{Z}_+ := \{0, 1, 2, \ldots\}$ , where  $\mathcal{A}^*$  denotes the Kleene star on  $\mathcal{A}$ . Given a sequence of L action profiles, say  $(\mathbf{a}^1, \mathbf{a}^2, \ldots, \mathbf{a}^L), \mathcal{L}((\mathbf{a}^1, \mathbf{a}^2, \ldots, \mathbf{a}^L)) = L-1$ gives us its length, i.e., the number of transitions in it.

Since the game G is generalized weakly acyclic, for every strategy profile  $\mathbf{a} \notin \mathcal{A}_{NE}$ , there exists at least one generalized better reply path that starts with  $\mathbf{a}$  and leads to a Nash equilibrium. For each strategy profile  $\mathbf{a} \in \mathcal{A}$ , we choose a generalized better reply path with the shortest length (according to the mapping  $\mathcal{L}$ ) and denote it by  $\mathbf{p}(\mathbf{a})$ . Clearly, for a Nash equilibrium  $\mathbf{a}^* \in \mathcal{A}_{NE}$ , we have  $\mathbf{p}(\mathbf{a}^*) = (\mathbf{a}^*)$  and  $\mathcal{L}(\mathbf{p}(\mathbf{a}^*)) = 0$ .

Although it is not necessary, in order to facilitate the exposition, we assume that the shortest generalized better reply paths  $\{\mathbf{p}(\mathbf{a}), \mathbf{a} \in \mathcal{A}\}$  satisfy the following consistency condition: Suppose that a non-Nash equilibrium strategy profile  $\mathbf{a}^1$ appears in the generalized better reply path,  $\mathbf{p}(\mathbf{a}^2)$ , of another action profile  $\mathbf{a}^2$ . Then, the subsequence in  $\mathbf{p}(\mathbf{a}^2)$  that starts with  $\mathbf{a}^1$  is identical to  $\mathbf{p}(\mathbf{a}^1)$ . Such generalized better reply paths can be constructed easily in a manner similar to Dijkstra's algorithm, starting with  $\mathcal{A}_{NE}$  [42].

We introduce two mappings:  $\Phi : \mathcal{A} \setminus \mathcal{A}_{NE} \to \mathcal{A}$  and  $I : \mathcal{A} \setminus \mathcal{A}_{NE} \to 2^{\mathcal{P}}$ , where  $\Phi(\mathbf{a})$  denotes the second strategy profile in  $\mathbf{p}(\mathbf{a})$  following  $\mathbf{a}$  and  $I(\mathbf{a}) = \{i \in \mathcal{P} \mid a_i \neq \Phi_i(\mathbf{a})\}$ , i.e., the set of agents that change their strategies going from strategy profile  $\mathbf{a}$  to  $\Phi(\mathbf{a})$ .

We prove Theorem 7.5 with the help of several lemmas, whose proofs are deferred to the following section. The first lemma states that, if the action profile at time t is not a Nash equilibrium, the action profile will reach  $\Phi(\mathbf{a}(t))$  within  $3\Delta_{\eta}$ periods with positive probability, where  $\Delta_{\eta}$  is the constant in Assumption A4.

**Lemma 7.9.** Under the assumptions in Theorem 7.5, there exists  $q_1 > 0$  such that, for every  $\mathbf{a} \notin \mathcal{A}_{NE}$  and  $t \in \mathbb{N}$ ,

$$\mathbb{P}\left[\mathbf{a}(t+3\Delta_{\eta}) = \Phi(\mathbf{a}(t)) \mid \mathbf{a}(t) = \mathbf{a}\right] \ge q_1.$$
(7.5)

Repeating Lemma 7.9, we readily obtain the following corollary that tells us that, starting at  $\mathbf{a}(t)$  at time t, we reach a Nash equilibrium within a finite number of periods with positive probability.

**Corollary 1.** Suppose that the assumptions in Theorem 7.5 hold. Then, for any  $\mathbf{a} \notin \mathcal{A}_{NE}$  and every  $t \in \mathbb{N}$ , we have

$$\mathbb{P}\left[\mathbf{a}\left(t+3\Delta_{\eta}\mathcal{L}(\mathbf{p}(\mathbf{a}))\right)\in\mathcal{A}_{NE}\mid\mathbf{a}(t)=\mathbf{a}\right]\geqslant q_{1}^{\mathcal{L}(\mathbf{p}(\mathbf{a}))}.$$
(7.6)

The next lemma states that, once the agents reach a Nash equilibrium  $\mathbf{a}^*$ ,

there is positive probability that they will remain at the Nash equilibrium  $\mathbf{a}^*$  for good.

**Lemma 7.10.** Under the assumptions in Theorem 7.5, there exists  $q_2 > 0$  such that, for all  $\mathbf{a}^* \in \mathcal{A}_{NE}$  and  $t \in \mathbb{N}$ , we have

$$\mathbb{P}\left[\mathbf{a}(t') = \mathbf{a}^* \text{ for all } t' \ge t+1 \mid \mathbf{a}(t) = \mathbf{a}^*\right] \ge q_2.$$
(7.7)

Lemmas 7.9 and 7.10 yield the following corollary. Let  $L_{\max} := \max_{\mathbf{a} \in \mathcal{A} \setminus \mathcal{A}_{NE}} \mathcal{L}(\mathbf{p}(\mathbf{a})).$ 

**Corollary 2.** Suppose that the assumptions in Theorem 7.5 are true. Then, for any  $\mathbf{a} \in \mathcal{A}$ , we have

$$\mathbb{P}\left[\mathbf{a}(t') \in \mathcal{A}_{NE} \text{ for all } t' \ge t + 3\Delta_{\eta} L_{\max} \mid \mathbf{a}(t) = \mathbf{a}\right] \ge q_1^{L_{\max}} \cdot q_2.$$
(7.8)

Define  $\bar{L} := 3\Delta_{\eta}L_{\max}$  and  $\bar{q} := q_1^{L_{\max}} \cdot q_2$ . Corollary 2 implies that, for all  $n \in \mathbb{N}$ ,  $\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_{n\bar{L}+1}^{\infty}\right] \leq (1-\bar{q})^n$ . Recall that  $\mathcal{A}_t^{\infty}$  increases with t. Consequently,  $\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_t^{\infty}\right]$ decreases with t and  $\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_{t+1}^{\infty}\right] \leq (1-\bar{q})^{[t/\bar{L}]} \leq \tilde{C} \cdot \tilde{\zeta}^t$ , where  $\tilde{C} = (1-\bar{q})^{-1}$  and  $\tilde{\zeta} = (1-\bar{q})^{1/\bar{L}} < 1$ . Because  $\mathcal{A}_{\star}^{\infty} = \cup_{t \in \mathbb{N}} \mathcal{A}_t^{\infty}$ , we have  $\mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}_{\star}^{\infty}\right] \geq \mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}_t^{\infty}\right]$  for all  $t \in \mathbb{N}$ . Finally,  $\lim_{t\to\infty} \mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}_t^{\infty}\right] = 1 - \lim_{t\to\infty} \mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_t^{\infty}\right] = 1$ , yielding the desired result  $\mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}_{\star}^{\infty}\right] = 1$ .

*Remark:* As mentioned in Section 7.5, Theorems 7.5 and 7.6 hold under less restrictive assumptions for weakly acyclic games. This is because Lemma 7.9 is true under Assumptions A1 through A4 for weakly acyclic games and the proof of Lemma 7.10 presented in Section 7.8.2 requires only Assumptions A1 through A4.

## 7.8 Proofs of Lemmas introduced in Section 7.7

## 7.8.1 Proof of Lemma 7.9

For each  $i \in \mathcal{P}$ , we define  $k_{\max}^i : \mathbb{N} \to \mathbb{N}$ , where  $k_{\max}^i(t) = \max\{k \in \mathbb{N} \mid T_k^i \leq t\}$ . In addition, for every  $i \in \mathcal{P}$  and  $t \in \mathbb{N}$ ,  $d_i^F(t) = X_{k_{\max}^i(t)}^i$  and  $d_i^R(t) = Y_{k_{\max}^i(t)}^i$ . Similarly, for all  $i, j \in \mathcal{P}, i \neq j$ , and  $t \in \mathbb{N}, d_{i,j}^-(t) = V_{k_{\max}^i(t)}^{i,j}$  and  $d_{i,j}^+(t) = V_{k_{\max}^i(t)+1}^{i,j}$ . For notational convenience, we denote the interval  $\{t + (\ell - 1)\Delta_{\eta} + 1, \dots, t + \ell \cdot \Delta_{\eta}\}$  by  $\overline{T}_{\ell}(t), \ell \in \mathbb{N}$ , where  $\Delta_{\eta}$  is the constant in Assumption A4 in Section 7.5.

Next, we define following events. Note that  $T_{k_{\max}^i(t)}^i$  (resp.  $T_{k_{\max}^i(t)+1}^i$ ) denotes the last time by time t (resp. the first time after time t) at which agent i updates its strategy.

$$\mathcal{E}_{0} = \{ T^{i}_{k^{i}_{\max}(t)+1} \leqslant t + \Delta_{\eta} \text{ for all } i \in \mathcal{P} \\$$
  
and  $T^{i}_{k^{i}_{\max}(t)} + d^{-}_{i,j}(t) \leqslant t + \Delta_{\eta} \text{ for all } i, j \in \mathcal{P}, i \neq j \}$ 

 $\begin{aligned} \mathcal{E}_1 &= \{\mathcal{T}^i \cap \overline{T}_2(t) \neq \emptyset \text{ for all } i \in \mathcal{P}\}, \text{ i.e., all agents update at least once during} \\ &\text{ the interval } \overline{T}_2(t) \end{aligned}$   $\begin{aligned} \mathcal{E}_2 &= \{a_i(t') = a_i(t), t' = t + 1, \dots, T^i_{k^i_{\max}(t+2\Delta_\eta)+1} - 1 \text{ for all } i \in \mathcal{P}\} \\ \mathcal{E}_3 &= \{d^F_i(t+2\Delta_\eta) \leqslant t + 2\Delta_\eta + \Delta_\delta - T^i_{k^i_{\max}(t+2\Delta_\eta)} \text{ for all } i \in \mathcal{P}\} \end{aligned}$   $\begin{aligned} \mathcal{E}_4 &= \{\mathcal{T}^i \cap \overline{T}_3(t) \neq \emptyset \text{ for all } i \in \mathcal{P}\} \\ \mathcal{E}_5 &= \{a_i(T^i_{k^i_{\max}(t+2\Delta_\eta)+1}) = \Phi_i(\mathbf{a}(t)) \text{ for all } i \in \mathcal{P}\} \end{aligned}$   $\begin{aligned} \mathcal{E}_6 &= \{d^+_{i,j}(t+2\Delta_\eta) \geqslant \Delta_\delta \text{ for all } i, j \in \mathcal{P}, i \neq j\} \\ \mathcal{E}_7 &= \{a_i(t') = \Phi_i(\mathbf{a}(t)) \text{ for all } i \in \mathcal{P} \text{ and } t' = T^i_{k^i_{\max}(t+2\Delta_\eta)+1} + 1, \dots, t + 3\Delta_\eta\} \end{aligned}$ 

Event  $\mathcal{E}_0$  ensures that (a) all agents update at least once during the interval  $\overline{T}_1(t)$  and (b) action profile  $\mathbf{a}(t)$  starts affecting the payoffs of all agents by time  $t + \Delta_\eta$ . The second event  $\mathcal{E}_1$  demands that all the agents revise their strategies at least once during the interval  $\overline{T}_2(t)$ . Events  $\mathcal{E}_2$  and  $\mathcal{E}_5$  together require every agent  $i \in \mathcal{P}$  to continue playing the same action  $a_i(t)$  till  $T_{k_{\max}^i(t+2\Delta_\eta)+1}^i - 1$  and then switch to  $\Phi_i(\mathbf{a}(t))$  at time  $T_{k_{\max}^i(t+2\Delta_\eta)+1}^i$ . In order to make sure that event  $\mathcal{E}_5$  is feasible, events  $\mathcal{E}_3$  and  $\mathcal{E}_6$  ensure that, at the first time agents update their strategies during the interval  $\overline{T}_3(t)$ , they will see the payoff information in response to  $\mathbf{a}(t)$  and, as a result, choose  $\Phi_i(\mathbf{a}(t))$  with positive probability. Finally, event  $\mathcal{E}_7$  demands that the agents continue to play strategy  $\Phi_i(\mathbf{a}(t))$  till  $t + 3\Delta_\eta$ .

We can lower bound the conditional probability in (7.5) in Lemma 7.9 as follows. First,

$$\mathbb{P}\left[\mathbf{a}(t+3\Delta_{\eta}) = \Phi(\mathbf{a}(t)) \mid \mathbf{a}(t) = \mathbf{a}\right]$$
  
$$\geq \mathbb{P}\left[\mathbf{a}(t+3\Delta_{\eta}) = \Phi(\mathbf{a}(t)), \mathcal{E}_{i}, i = 0, 1, \dots, 7 \mid \mathbf{a}(t) = \mathbf{a}\right]$$
(7.9)

We rewrite (7.9) as a product of conditional probabilities.

$$(7.9) = \mathbb{P}\left[\mathbf{a}(t+3\Delta_{\eta}) = \Phi(\mathbf{a}(t)) \mid \mathbf{a}(t) = \mathbf{a}, \mathcal{E}_i, i = 0, 1, \dots, 7\right]$$
(7.10)

$$\times \mathbb{P}\left[\mathcal{E}_7 \mid \mathbf{a}(t) = \mathbf{a}, \mathcal{E}_i, i = 0, 1, \dots, 6\right]$$
(7.11)

$$\times \mathbb{P}\left[\mathcal{E}_5 \cap \mathcal{E}_6 \mid \mathbf{a}(t) = \mathbf{a}, \mathcal{E}_i, i = 0, \dots, 4\right]$$
(7.12)

$$\times \mathbb{P}\left[\mathcal{E}_3 \cap \mathcal{E}_4 \mid \mathbf{a}(t) = \mathbf{a}, \mathcal{E}_i, i = 0, 1, 2\right]$$
(7.13)

$$\times \mathbb{P}\left[\mathcal{E}_2 \mid \mathbf{a}(t) = \mathbf{a}, \mathcal{E}_i, i = 0, 1\right]$$
(7.14)

$$\times \mathbb{P}\left[\mathcal{E}_0 \cap \mathcal{E}_1 \mid \mathbf{a}(t) = \mathbf{a}\right]$$
(7.15)

We now lower bound the terms in (7.10) through (7.15). First, it is clear from the definitions of the events  $\mathcal{E}_i$ , i = 0, 1, ..., 7, that (7.10) is equal to one. Since  $1-\mu$  is a lower bound to the probability with which a agent with a better reply plays the same action played at the previous time, (7.11)  $\geq (1-\mu)^{n(\Delta_{\eta}-1)}$  and (7.14)  $\geq (1-\mu)^{2\Delta_{\eta}n}$ . From Assumption A5 (more precisely, (7.3)), we have  $(7.12) \geq (\underline{\epsilon} \cdot (1-\mu) \cdot \delta)^n$ , where  $\underline{\epsilon}$  is a positive lower bound to the probability with which a agent chooses a better reply. By Assumptions A4 through A6, (7.13)  $\geq (\eta \cdot \delta)^n$ . Finally, Assumption A4 implies (7.15)  $\geq \eta^{2n}$ . From these lower bounds, the inequality in (7.5) holds with  $q_1 := ((1-\mu)^{3\Delta_{\eta}} \cdot \underline{\epsilon} \cdot \delta^2 \cdot \eta^3)^n$ .

# 7.8.2 Proof of Lemma 7.10

We first define the following three events.

$$\mathcal{E}'_{0} = \{ T^{i}_{k^{i}_{\max}(t)+1} \leq t + \Delta_{\eta} \text{ for all } i \in \mathcal{P} \\$$
  
and  $T^{i}_{k^{i}_{\max}(t)} + d^{-}_{i,j}(t) \leq t + \Delta_{\eta} \text{ for all } i, j \in \mathcal{P}, i \neq j \}$   
$$\mathcal{E}'_{1} = \{ \mathcal{T}^{i} \cap \overline{T}_{2}(t) \neq \emptyset \text{ for all } i \in \mathcal{P} \}$$
  
$$\mathcal{E}'_{2} = \{ \mathbf{a}(t') = \mathbf{a}^{*} \text{ for all } t' = t + 1, \dots, t + 2\Delta_{\eta} \}$$

First, events  $\mathcal{E}'_0$  and  $\mathcal{E}'_2$  ensure that the payoffs of all agents are determined by strategy profile  $\mathbf{a}^*$  by time  $t + \Delta_\eta$ . Note that it also implies that every agent updates its strategy at least once during the interval  $\overline{T}_1(t)$  because  $T^i_{k^i_{\max}(t)+1} \leq$  $t + \Delta_\eta$ . Second, event  $\mathcal{E}'_1$  demands the agents to revise their strategies at least once during the interval  $\overline{T}_2(t)$  as well. Finally, event  $\mathcal{E}'_2$  requires that the agents play the equilibrium strategies till  $t + 2\Delta_\eta$ . Note that, if the events  $\mathcal{E}'_0$  through  $\mathcal{E}'_2$  occur, all payoff information will be produced on the basis of strategy profile  $\mathbf{a}^*$  after time  $t + \Delta_\eta$  and, as a result, no agent will change its action after time  $t + 2\Delta_\eta$ . This is because, for every agent  $i \in \mathcal{P}$ , we have  $\tilde{\mathbf{a}}^i_{-i}(T^i_{k^i_{\max}(t+2\Delta)+1}) = \mathbf{a}^*_{-i}$ . Once this happens, no agent will ever see a better reply after  $t + 2\Delta_\eta$  and, consequently, will not deviate from its equilibrium action, and the strategy profile will remain  $\mathbf{a}^*$  for good. Thus,

$$\mathbb{P}\left[\mathbf{a}(t') = \mathbf{a}^* \text{ for all } t' \ge t+1 \mid \mathbf{a}(t) = \mathbf{a}^*, \mathcal{E}'_i, i = 0, 1, 2\right] = 1.$$

Putting things together, we get

$$\mathbb{P}\left[\mathbf{a}(t') = \mathbf{a}^* \text{ for all } t' \ge t+1 \mid \mathbf{a}(t) = \mathbf{a}^*\right]$$
$$\ge \mathbb{P}\left[\mathbf{a}(t') = \mathbf{a}^* \text{ for all } t' \ge t+1 \mid \mathbf{a}(t) = \mathbf{a}^*, \mathcal{E}'_i, i = 0, 1, 2\right]$$
$$\times \mathbb{P}\left[\mathcal{E}'_0 \cap \mathcal{E}'_1 \cap \mathcal{E}'_2 \mid \mathbf{a}(t) = \mathbf{a}^*\right]$$
$$= \mathbb{P}\left[\mathcal{E}'_0 \cap \mathcal{E}'_1 \cap \mathcal{E}'_2 \mid \mathbf{a}(t) = \mathbf{a}^*\right]$$

From Assumption A4, we get the following lower bound.

$$\mathbb{P}\left[\mathcal{E}'_0 \cap \mathcal{E}'_1 \cap \mathcal{E}'_2 \mid \mathbf{a}(t) = \mathbf{a}^*\right] \ge \eta^n \cdot \eta^n (1-\mu)^{2n\Delta_\eta} = \left(\eta \cdot (1-\mu)^{\Delta_\eta}\right)^{2n} > 0.$$

Hence, the lemma follows with  $q_2 := (\eta \cdot (1-\mu)^{\Delta_{\eta}})^{2n}$ .

# 7.9 Proof of Theorem 7.8

From [53, Lemma 3.1, p. 177], we know that the stationary distribution  $\mu^t = (\mu^t(\mathbf{a}), \mathbf{a} \in \mathcal{A})$  is given by

$$\mu^{t}(\mathbf{a}) = \frac{Q_{\mathbf{a}}(t)}{\sum_{\mathbf{a}'\in\mathcal{A}} Q_{\mathbf{a}'}(t)}, \ \mathbf{a}\in\mathcal{A},$$
(7.16)

where  $Q_{\mathbf{a}'}(t) = \sum_{T \in \mathcal{T}(\mathbf{a}')} \left( \prod_{(\mathbf{a}^1, \mathbf{a}^2) \in T} P_{\mathbf{a}^1, \mathbf{a}^2}(t) \right)$ . Consider the derivative of  $\mu^t(\mathbf{a})$  with respect to  $\epsilon_t$ .

$$\left(\mu^{t}(\mathbf{a})\right)' = \frac{Q'_{\mathbf{a}}(t)\sum_{\mathbf{a}'\in\mathcal{A}}Q_{\mathbf{a}'(t)} - Q_{\mathbf{a}}(t)\sum_{\mathbf{a}'\in\mathcal{A}}Q'_{\mathbf{a}'}(t)}{\left(\sum_{\mathbf{a}'\in\mathcal{A}}Q_{\mathbf{a}'}(t)\right)^{2}}$$
(7.17)

By Assumption 7.3 and the comment following it, for all sufficiently large t, the numerator of (7.17) behaves like a finite sum of power functions of  $\epsilon_t$ . Therefore, there exists finite  $T^*$  such that, for all  $t \ge T^*$  and all  $\mathbf{a} \in \mathcal{A}$ , the sign of  $(\mu^t(\mathbf{a}))'$  remains the same. Based on this observation, we can partition the set  $\mathcal{A}$  into  $\mathcal{A}^-$  and  $\mathcal{A}^+ = \mathcal{A} \setminus \mathcal{A}^-$ , where  $\mathcal{A}^- = \{\mathbf{a} \in \mathcal{A} \mid (\boldsymbol{\mu}^t(\mathbf{a}))' \le 0 \text{ for all } t \ge T^*\}.$ 

Theorem 8.3 of [44, p. 242] tells us that the nonhomogeneous Markov chain  $\{\mathbf{a}(t), t \in \mathbb{N}\}$  is strongly ergodic if it is weakly ergodic and the following inequality holds.

$$\sum_{t\in\mathbb{N}}\left|\left|\boldsymbol{\mu}^{t}-\boldsymbol{\mu}^{t+1}\right|\right|_{1}<\infty$$
(7.18)

First, the following lemma tells us that, under the conditions in Theorem 7.8, the Markov chain is weakly ergodic. Its proof can be found in Section 7.10.

**Lemma 7.11.** Suppose that Assumption 7.2 holds with  $\sum_{t \in \mathbb{N}} \epsilon_t^{\kappa} = \infty$ . Then, the nonhomogeneous Markov chain  $\{\mathbf{a}(t), t \in \mathbb{N}\}$  is weakly ergodic.

Second, we can show (7.18) as follows. We first break the summation into two summations.

$$\sum_{t \in \mathbb{N}} \left\| \boldsymbol{\mu}^{t} - \boldsymbol{\mu}^{t+1} \right\|_{1} = \sum_{t=1}^{T^{*}} \left\| \boldsymbol{\mu}^{t} - \boldsymbol{\mu}^{t+1} \right\|_{1} + \sum_{t>T^{*}} \left\| \boldsymbol{\mu}^{t} - \boldsymbol{\mu}^{t+1} \right\|_{1}$$
(7.19)

Applying the inequality  $||\boldsymbol{\mu}^t - \boldsymbol{\mu}^{t+1}||_1 \leq 2$  to the first term in (7.19) and the definition of  $||\cdot||_1$  norm for the summands in the second term, we obtain

$$(7.19) \leq 2T^* + \sum_{t>T^*} \left( \sum_{\mathbf{a}\in\mathcal{A}} |\mu^t(\mathbf{a}) - \mu^{t+1}(\mathbf{a})| \right)$$
$$= 2T^* + \sum_{t>T^*} \left( \sum_{\mathbf{a}\in\mathcal{A}^-} \left( \mu^{t+1}(\mathbf{a}) - \mu^t(\mathbf{a}) \right) \right)$$
$$+ \sum_{t>T^*} \left( \sum_{\mathbf{a}\in\mathcal{A}^+} \left( \mu^t(\mathbf{a}) - \mu^{t+1}(\mathbf{a}) \right) \right)$$
$$= 2T^* + \sum_{\mathbf{a}\in\mathcal{A}^-} \left( \mu^0(\mathbf{a}) - \mu^{T^*+1}(\mathbf{a}) \right)$$
$$+ \sum_{\mathbf{a}\in\mathcal{A}^+} \left( \mu^{T^*+1}(\mathbf{a}) - \mu^0(\mathbf{a}) \right)$$
$$\leq 2T^* + 2 < \infty,$$
(7.20)

where the second equality in (7.20) follows from Theorem 7.7, i.e.,  $\mu^t \to \mu^0$  as  $t \to \infty$ .

# 7.10 Proof of Lemma 7.11

Let us first introduce some notation and terminology we use in the proof. For notational ease, we denote the product  $\mathbf{P}(m) \times \mathbf{P}(m+1) \times \cdots \times \mathbf{P}(n-1)$ of transition matrices by  $\mathbf{P}(m, n)$ . Given a stochastic matrix  $\mathbf{P}$ , its Dobrushin's ergodic coefficient [44, p. 235] is given by

$$\delta(\mathbf{P}) = 1 - \min_{\mathbf{a}^1, \mathbf{a}^2 \in \mathcal{A}} \left( \sum_{\mathbf{a} \in \mathcal{A}} \min\left(P_{\mathbf{a}^1, \mathbf{a}}, P_{\mathbf{a}^2, \mathbf{a}}\right) \right).$$
(7.21)

Theorem 8.2 of [44, p. 241] states that the nonhomogeneous Markov chain  $\{\mathbf{a}(t), t \in \mathbb{N}\}$  is weakly ergodic if

$$\sum_{n \in \mathbb{N}} \left( 1 - \delta(\mathbf{P}(t_n, t_{n+1})) \right) = \infty.$$
(7.22)

For each  $\mathbf{a}^* \in \mathcal{A}_{NE}$  and  $\mathbf{a} \in \mathcal{A}$ , let  $\ell(\mathbf{a}, \mathbf{a}^*) = \min\{\mathcal{L}(h(\mathbf{a} \to \mathbf{a}^*)) \mid r_p(h(\mathbf{a} \to \mathbf{a}^*)) = \rho(\mathbf{a}, \mathbf{a}^*)\}$ , i.e., the shortest length of the paths  $h(\mathbf{a} \to \mathbf{a}^*)$  with the least resistance  $r_p(h) = \rho(\mathbf{a}, \mathbf{a}^*)$ , and define  $\ell_{\max}(\mathbf{a}^*) := \max_{\mathbf{a} \in \mathcal{A}} \ell(\mathbf{a}, \mathbf{a}^*)$ . Then, the bounds in eq. (7.4) tell us that, for all  $\mathbf{a} \in \mathcal{A}$  and all sufficiently large t,

$$P^{m}_{\mathbf{a},\mathbf{a}^{*}}(t) > \underline{\xi}^{m} \cdot \epsilon^{\tau(\{\mathbf{a}^{*}\})}_{t+m} \text{ for all } m \ge \ell_{\max}(\mathbf{a}^{*}).$$

$$(7.23)$$

Let  $\ell_{\max} = \max_{\mathbf{a}^* \in \mathcal{A}_{NE}} \ell_{\max}(\mathbf{a}^*)$ , and consider the sequence  $(t_k, k \in \mathbb{N})$ , where  $t_k = k \cdot \ell_{\max}$ . First, from (7.23), for all distinct  $\mathbf{a}^1, \mathbf{a}^2 \in \mathcal{A}$  and all sufficiently large  $k \in \mathbb{N}$ , we have the following bound.

$$\sum_{\mathbf{a}\in\mathcal{A}} \min\left(P_{\mathbf{a}^{1},\mathbf{a}}(t_{k},t_{k+1}),P_{\mathbf{a}^{2},\mathbf{a}}(t_{k},t_{k+1})\right)$$

$$\geq \sum_{\mathbf{a}^{*}\in\mathcal{A}_{NE}} \min\left(P_{\mathbf{a}^{1},\mathbf{a}^{*}}(t_{k},t_{k+1}),P_{\mathbf{a}^{2},\mathbf{a}^{*}}(t_{k},t_{k+1})\right)$$

$$\geq \underline{\xi}^{\ell_{\max}} \sum_{\mathbf{a}^{*}\in\mathcal{A}_{NE}} \epsilon_{t_{k+1}}^{\tau(\{\mathbf{a}^{*}\})},$$
(7.24)

where the second inequality follows from (7.23).

Using the expression for Dobrushin's coefficient in (7.21), we obtain

$$\sum_{n \in \mathbb{N}} \left( 1 - \delta(\mathbf{P}(t_n, t_{n+1})) \right)$$
$$= \sum_{n \in \mathbb{N}} \min_{\mathbf{a}^1, \mathbf{a}^2 \in \mathcal{A}} \left( \sum_{\mathbf{a} \in \mathcal{A}} \min\left( P_{\mathbf{a}^1, \mathbf{a}}(t_n, t_{n+1}), P_{\mathbf{a}^2, \mathbf{a}}(t_n, t_{n+1}) \right) \right).$$
(7.25)

In light of (7.24), it is clear that (7.25) diverges if  $\sum_{n \in \mathbb{N}} \left( \sum_{\mathbf{a}^* \in \mathcal{A}_{NE}} \epsilon_{t_{n+1}}^{\tau(\mathbf{a}^*)} \right)$  diverges.

We state the following lemma without a proof, which is straightforward.

**Lemma 7.12.** Suppose that  $(\alpha(n), n \in \mathbb{N})$  is a decreasing, positive sequence with  $\sum_{n \in \mathbb{N}} \alpha(n) = \infty$ . Then, for any  $k \in \mathbb{N}$  and  $\ell \in \mathbb{Z}_+$ , we have  $\sum_{n \in \mathbb{N}} \alpha(k \cdot n + \ell) = \infty$ .

First, because  $\kappa = \min_{\mathbf{a}^* \in \mathcal{A}_{NE}} \tau(\{\mathbf{a}^*\})$ 

$$\sum_{n \in \mathbb{N}} \left( \sum_{\mathbf{a}^* \in \mathcal{A}_{NE}} \epsilon_{t_{n+1}}^{\tau(\{\mathbf{a}^*\})} \right) \geqslant \sum_{n \in \mathbb{N}} \epsilon_{t_{n+1}}^{\kappa}.$$
(7.26)

Second, recall that  $\sum_{t\in\mathbb{N}} \epsilon_t^{\kappa} = \infty$  by the assumption in Lemma 7.11. Thus, by Lemma 7.12, we get  $\sum_{n\in\mathbb{N}} \epsilon_{t_{n+1}}^{\kappa} = \infty$ . Together with (7.26), this implies that (7.25) diverges, completing the proof of (7.22).

# Chapter 8: Class of Monitoring Rules

In the previous chapter we considered the class of better reply rules which ensures convergence to a pure strategy Nash equilibrium in a large class of games, identified as the generalized weakly acyclic games, under asynchronous updates and payoff information delays. Moreover, we also showed convergence to a PSNE under faulty available payoff information, causing players to make erroneous decisions occasionally. Under the latter setting of erroneous payoff information, when the better reply rules are followed by the agents, the PSNE(s) that are stable in the *long run* or are *stochastically stable* would in general depend both on the error model and the structure of the game itself. In this chapter we consider another class of learning rules that are also robust to delays and asynchrony like the better reply rules. However, under a (partial) erroneous payoff information setting where the SBRs are estimated with errors, the stochastically stable states are better characterisable.

The learning rule in its simplest form is described in the first section, followed by the convergence results under the learning rule for the asynchronous update setting introduced in Section 7.5. Later, we consider the faulty payoff information setting introduced in Section 7.6 and contrast the convergence results from that obtained for the better-reply rules. It is worth mentioning that we adopt the strategic-form repeated game setting described in Section 7.1.

## 8.1 Proposed Update Rule

In this section, we present our proposed learning rule, called the Reduced Simple Experimentation with Monitoring (RSEM). Under the RSEM update rule, at each time  $t \in \mathbb{N}$ , every agent is at one of two possible states – Explore (E) or Converged (C). We denote the state of agent  $i \in \mathcal{P}$  at time t by  $s_i(t) \in \Psi_0 := \{E, C\}$ . Let  $\Psi := \Psi_0^n$ , and the state vector  $\mathbf{s}(t) := (s_i(t), i \in \mathcal{P}) \in \Psi$ . The rule governing the update of an agent's state will be explained shortly.

Recall that the mappings  $C_i$  and  $BR_i$  introduced in the previous chapter are the classification and the better reply mappings for agent *i* respectively. Also, recall that the update time sequence for player *i* is denoted as  $\mathcal{T}^i = \{T_k^i, k \in \mathbb{N}\} \subseteq \mathbb{N}$ . In the synchronous update case with no delays we have  $\mathcal{T}^i = \mathbb{N}$  for all  $i \in \mathcal{P}$  and  $\tilde{\mathbf{a}}^i(t) = \mathbf{a}(t-1)$  for all  $i \in \mathcal{P}$  and  $t \in \mathbb{N}^+$ , where  $\tilde{\mathbf{a}}^i(t) \in \mathcal{A}$  is the strategy profile responsible for generating the payoff information observed by player *i* at time *t*.

**Payoff Information for updates** – We assume that the payoff information available to agent *i* at time  $T_k^i$ ,  $k \ge 2$ , is of the form  $((\mathcal{C}_i(a_i, \tilde{\mathbf{a}}^i(T_k^i)), a_i \in \mathcal{A}_i); U_i(\tilde{\mathbf{a}}^i(T_k^i)))$ , compactly denoted as  $(\mathcal{C}_i(\tilde{\mathbf{a}}^i(T_k^i)); U_i(\tilde{\mathbf{a}}^i(T_k^i)))$ . In other words, as assumed in the previous chapter, agent *i* knows the actions that would lead to higher payoff with respect to the played action given the action profile of the other agents in effect (at the agent *i*'s system) at the time the payoff feedback is generated, i.e.,  $R_{k-1}^i$ .

## 8.1.1 Action updates

In this subsection, we first describe how the agents choose their actions according to the payoff feedback they receive at the times of updates.

Initially, at time  $T_1^i = 1$ , the agents choose their actions according to some (joint) distribution  $\mathcal{G}$  over  $\mathcal{A}$ . For  $k \ge 2$ , agent  $i \in \mathcal{P}$  updates its action at time  $T_k^i$  according to the rule provided below: We assume that agent i continues to play action  $a_i(T_k^i)$  between  $T_k^i$  and  $T_{k+1}^i - 1$ , i.e.,  $a_i(t) = a_i(T_k^i)$  for all  $t \in \{T_k^i, \ldots, T_{k+1}^i - 1\}$ . Fix  $\delta \in (0, 1/(\max_{i \in \mathcal{P}} |\mathcal{A}_i|))$  and  $\mathcal{\beta}_i : S^{A_i} \to \Delta(\mathcal{A}_i), i \in \mathcal{P}$ . The mappings  $\mathcal{\beta}_i(\mathbf{c}_i) =$  $(\beta_i(a_i; \mathbf{c}_i), a_i \in \mathcal{A}_i)$  are used to determine the mixed strategy to be employed given the payoff information.

## Action Selection Rule:

For k = 2, 3, ...,

• if  $s_i(T_k^i) = E$ 

- choose  $a_i(T_k^i) = a_i$  with probability  $\beta_i(a_i; \mathcal{C}_i(\tilde{\mathbf{a}}^i(T_k^i))) \ge \delta$  for all  $a_i \in \mathcal{A}_i$ 

• else (i.e.,  $s_i(T_k^i) = C$ )

$$- \text{ set } a_i(T_k^i) = a_i(T_k^i - 1);$$

It is clear that, under RSEM, an agent may choose a new action only if it is at state E. Otherwise, it continues to play the same action employed at the previous time.

Modeling mixed strategies in RSEM rule – As was done for the GBRR rules, we describe how we model mixed strategies exercised by the players under the RSEM rule: For every  $i \in \mathcal{P}$ , define  $\mathbf{u}^i := (u_t^i, t \in \mathbb{N}^+)$  to be a sequence of independent uniform random variables over (0, 1].

Suppose that player *i* revises its strategy at time  $t \in \mathcal{T}^i$  and  $s_i(t) = E$ . It then chooses

$$a_i(t) = \ell, \text{ if } u_t^i \in (\Lambda(\ell - 1), \Lambda(\ell)], \ \ell = 1, 2, \dots, A_i$$

$$(8.1)$$

where  $\Lambda(\ell) := \sum_{\ell'=1}^{\ell} \beta\left(a_i^{(\ell')}; \mathcal{C}_i(\tilde{\mathbf{a}}^i(t))\right), \ \ell = 1, \dots, A_i.$ 

# 8.1.2 State dynamics

As explained in the previous subsection, under the RSEM rule, the state of an agent plays a key role in its action selection. Hence, the dynamics of  $\mathbf{s}(t), t \in \mathbb{N}$ , play a major role in the algorithm. In this subsection, we explain how the agents update their states based on the received payoff feedback.

At time t = 1, we assume that all agents are at state E, i.e.,  $\mathbf{s}(1) = (E, E, \dots, E)$ . Agent i first updates its state right after it receives new payoff information at  $T_k^i$ ,  $k \in \mathbb{N}$ , following which it chooses an action.

The state of agent *i* at time  $T_k^i$  depends on (i) the payoff information vector  $C_i(\tilde{\mathbf{a}}^i(T_k^i))$  if  $s_i(T_{k-1}^i) = E$  and (ii) the payoffs obtained at time  $T_{k-1}^i$  and  $T_k^i$  if  $s_i(T_{k-1}^i) = C$ . Note that at time  $T_2^i$  only case S1 is applicable because at time  $T_1^i = 1$  all the agents are in the Explore state. Therefore case S2 in the state update rule can be effective only for  $k \ge 3$ .

State Update Rule: Fix  $p \in (0, 1)$ .

For k = 2, 3, ...

- S1. if  $s_i(T_{k-1}^i) = E$ 
  - if  $BR_i(\tilde{\mathbf{a}}^i(T_k^i)) \neq \emptyset$ , then  $s_i(T_k^i) = E$

• else (i.e., 
$$BR_i(\tilde{\mathbf{a}}^i(T_k^i)) = \emptyset$$
)

$$s_i(T_k^i) = \begin{cases} E & \text{with probability } p \\ C & \text{with probability } 1 - p \end{cases}$$

S2. else (i.e.,  $s_i(T_{k-1}^i) = C$ )

• if 
$$BR_i(\tilde{\mathbf{a}}^i(T_k^i)) \neq \emptyset$$
, then  $s_i(T_k^i) = E$ 

• else ( i.e., 
$$BR_i(\tilde{\mathbf{a}}^i(T_k^i)) = \emptyset$$
 )

- if 
$$U_i(\tilde{\mathbf{a}}^i(T_k^i)) \neq U_i(\tilde{\mathbf{a}}^i(T_{k-1}^i))$$
, then  $s_i(T_k^i) = E$   
- else ( i.e.,  $U_i(\tilde{\mathbf{a}}^i(T_k^i)) = U_i(\tilde{\mathbf{a}}^i(T_{k-1}^i))$  ), then  $s_i(T_k^i) = C$ 

In a nutshell, agent *i* transitions to or remains at state *E* if either  $BR_i(\tilde{\mathbf{a}}^i(T_k^i)) \neq \emptyset$  or  $U_i(\tilde{\mathbf{a}}^i(T_k^i)) \neq U_i(\tilde{\mathbf{a}}^i(T_{k-1}^i))$ . Note that when an agent is at the converged state, seeing a better reply is equivalent to a change in the payoff information vector  $C_i(\tilde{\mathbf{a}}^i(\cdot))$ . The second condition means that although there are no better replies, the received payoff has changed from the last time. In such a scenario, the agent prefers to transition to the Explore state.

The RSEM rule and the GBRR rules– In the RSEM rule when the agents are in the explore state, all the actions are chosen with positive probability. In the converged state, they play their previous action as long as the received payoff remains the same and there are no better replies. While the GBRR-I rule only allows agents to experiment among their better replies and, if there are no better replies, then the previous action is played. It is easy to see that the class of RSEM rules are more exploratory since in the RSEM rule the agent is allowed to choose actions that are not among the better replies. Hence, if we have a sequence of action profiles generated by the GBRR rule  $\vec{\mathbf{a}} = (\mathbf{a}(t), t \in \mathbb{N})$ , it can also be generated under the RSEM rule.

#### 8.2 Convergence under RSEM

Let  $\vec{\mathbf{a}} = (\mathbf{a}(t), t \in \mathbb{N})$  be the sequence of action profiles played by the players using the RSEM rule. First, we state the convergence results under the RSEM rule for the asynchronous update setting described in Section 7.5.1 when the game is generalized weakly acyclic.

**Theorem 8.1.** Suppose that the game  $\mathbb{G}$  is generalized weakly acyclic and that Assumptions A1 through A6 hold. Then, for any initial distribution  $\mathcal{G} \in \Delta(\mathcal{A})$ , under the RSEM rule,

$$\mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}^{\infty}_{\star}\right] = 1.$$

**Proof.** From the discussion in Section 8.1, we conclude that the agents under the RSEM rule can also follow the GBRPs. Thus, the proof follows from arguments

similar to that for the GBRR rule in Section 7.7.

**Theorem 8.2.** Suppose that the game  $\mathbb{G}$  is generalized weakly acyclic and that Assumptions A1 through A6 are in place. Then, there exist  $C_{rsem} < \infty$  and  $\zeta_{rsem} \in$ (0,1) such that, for any initial distribution  $\mathcal{G} \in \Delta(\mathcal{A})$  under the RSEM rule

$$\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_t^{\infty}\right] \leqslant \min(1, C_{rsem} \cdot \zeta_{rsem}^t)$$

for all  $t \in \mathbb{N}$ .

The above-mentioned results not only hold for games that are generalized weakly acyclic, but also for games that satisfy the following interdependence assumption.

Assumption 8.1. (Interdependence Assumption) For every  $\mathbf{a} = (a_i, i \in \mathcal{P}) \in \mathcal{A}$  and  $J \subsetneq \mathcal{P}$ , there exist an agent  $i \notin J$  and  $\mathbf{a}_J^* \in \prod_{j \in J} \mathcal{A}_j$  such that  $U_i(\mathbf{a}) \neq U_i(\mathbf{a}_J^*, \mathbf{a}_{-J})$ .

Assumption 8.1 simply states that, given any action profile and a strict subset J of the agents, we can find another agent  $i \notin J$  whose payoff would change if the agents in J changed their actions to  $\mathbf{a}_{J}^{*}$ . Put differently, it implies that it is not possible to partition the set of agents into two subsets that do not interact with each other. The interdependence assumption has been used in the literature, for instance, to prove the convergence of action profile to efficient equilibria or Pareto optimal point [79,87,96]. Also, it is worth noting that there is no clear relationship between the class of GWAGs and the class of games satisfying the interdependence assumption. In general, there could be games that are GWAGs but do not satisfy the interdependence assumption and vice versa. Under this interdependence assumption, we can show that, if all agents update their actions according to the RSEM

rule, the action profile converges almost surely to a PSNE. However, we can have the desired convergence result under a weaker form of interdependence assumption.

Assumption 8.2. (Weak Interdependence Assumption) For every  $\mathbf{a} = (a_i, i \in \mathcal{P}) \in \mathcal{A}$  and  $J \subsetneq \mathcal{P}$  such that  $BR_i(\mathbf{a}) = \emptyset$  for all  $i \notin J$ , then there exist an agent  $i^* \notin J$ and  $\mathbf{a}_J^* \in \prod_{j \in J} \mathcal{A}_j$  such that either  $U_{i^*}(\mathbf{a}) \neq U_{i^*}(\mathbf{a}_J^*, \mathbf{a}_{-J})$  or  $BR_{i^*}(\mathbf{a}_J^*, \mathbf{a}_{-J}) \neq \emptyset$ .

Assumption 8.2 implies that for any action profile  $\mathbf{a} \in \mathcal{A}$  and a strict subset  $J \subsetneq \mathcal{P}$  where all the agents outside the set J see no better reply, there exists an action profile  $\mathbf{a}_J^*$  for the agents in the set J such that when the action profile  $(\mathbf{a}_J^*, \mathbf{a}_{-J})$  is adopted at least one agent outside the set J either sees a better reply or a change in its received payoff.

The following two theorems suggest that when the game satisfies the weak interdependence assumption and has a non-empty set of PSNE(s), and a RSEM rule is employed for updating strategies, the action profile reaches a Nash equilibrium under payoff information delays and asynchronous updates with probability 1, under Assumptions A1 through A4.

**Theorem 8.3.** Suppose that the game  $\mathbb{G}$  satisfies Assumption 8.2 and has a nonempty set of PSNE(s) denoted by  $\mathcal{A}_{NE}$  and that Assumptions A1 through A4 hold. Then, for any initial distribution  $\mathcal{G} \in \Delta(\mathcal{A})$ ,

$$\mathbb{P}\left[\vec{\mathbf{a}} \in \mathcal{A}^{\infty}_{\star}\right] = 1.$$

**Proof.** A proof of the theorem is provided in Section 8.3.

In addition to the almost sure convergence of the action profile, we can establish that the probability that the action profile has not converged to a PSNE decays geometrically with time t.

**Theorem 8.4.** Suppose that the game  $\mathbb{G}$  satisfies Assumption 8.2 and has a nonempty set of PSNE(s) denoted by  $\mathcal{A}_{NE}$  and that Assumptions A1 through A4 are in place. Then, there exist  $\tilde{C}_{rsem} < \infty$  and  $\tilde{\zeta}_{rsem} \in (0, 1)$  such that, for any initial distribution  $\mathcal{G} \in \Delta(\mathcal{A})$ ,

$$\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_t^{\infty}\right] \leqslant \min(1, \tilde{C}_{rsem} \cdot \tilde{\zeta}_{rsem}^t).$$

Theorem 8.4 follows directly from Corollary 4 in the proof of Theorem 8.3 in Section 8.3.

*Remark:* While both Theorem 8.1 and 8.3 ensure convergence to the set of PSNE almost surely, it is important to make the distinction that starting from any initial action profile  $\mathbf{a}(1)$  under the interdependence assumption the agents can converge to any PSNE, which is not always true for the case of GWAGs. This fact will become essential under various erroneous settings when we characterise the set of stochastically stable equilibria in this chapter and also in the next chapter.

# 8.3 Proof of Theorem 8.3

First, we define some notation: Let  $\mathcal{Z} := \mathcal{S} \times \mathcal{A}$  and denote the pair  $(\mathbf{s}(t), \mathbf{a}(t)) \in \mathcal{Z}$  by  $\mathbf{z}(t), t \in \mathbb{N}$ . Define  $\mathbf{s}^{\star} \in \mathcal{S}$  to be the state vector in which every agent is at state C, i.e.,  $\mathbf{s}^{\star} = (C, \ldots, C)$ , and  $\mathcal{Z}_{NE} = \{(\mathbf{s}^{\star}, \mathbf{a}^{\star}) \in \mathcal{Z} \mid \mathbf{a}^{\star} \in \mathcal{A}_{NE}\}.$ 

The theorem will be proved with the help of several lemmata we introduce. Their proofs are provided in Sections 8.4.1 through 8.4.3. For any  $\mathbf{s} \in \mathcal{S}$ ,

$$C(\mathbf{s}) = \{i \in \mathcal{P} \mid s_i = C\} \text{ and } E(\mathbf{s}) = \{i \in \mathcal{P} \mid s_i = E\}.$$

The first lemma states that, if the action profile at time t is not a PSNE, then even if all agents are at state C at time t, there is positive probability that at least one agent will transition to state E after  $3\Delta_{\eta}$  periods (where  $\Delta_{\eta}$  is the constant introduced in Assumption A4 in Section 7.5.1).

**Lemma 8.5.** For every  $\mathbf{a} \notin \mathcal{A}_{NE}$  and  $t \in \mathbb{N}$ ,

$$\mathbb{P}\left[E(\mathbf{s}(t+3\Delta_{\eta}))\neq\emptyset\mid\mathbf{z}(t)=(\mathbf{s}^{\star},\mathbf{a})\right]\geqslant\zeta_{0}>0.$$
(8.2)

The second lemma shows that, if there is at least one agent at state E at time  $t \in \mathbb{N}$ , there is positive probability that the number of agents at state E will increase after a finite number of periods.

**Lemma 8.6.** For every  $r \in \{1, 2, ..., n-1\}$ , there exists  $0 < D_1 \leq 4\Delta_\eta$  such that, for every  $t \in \mathbb{N}$  and  $\mathbf{z} = (\mathbf{s}, \mathbf{a}) \in \mathcal{Z}$  with  $|E(\mathbf{s})| = r$ , we have

$$\mathbb{P}\left[\left|E(\mathbf{s}(t+D_1))\right| \ge r+1 \mid \mathbf{z}(t) = \mathbf{z}\right] \ge \varsigma_r > 0.$$
(8.3)

The following corollary now follows from Lemmas 8.5 and 8.6, by repeatedly applying Lemma 8.6 until all agents switch to state E.

**Corollary 3.** There exists  $0 < D \leq 4n\Delta_{\eta}$  such that, for all  $\mathbf{z} \in \mathcal{Z} \setminus \mathcal{Z}_{NE}$  and  $t \in \mathbb{N}$ ,

$$\mathbb{P}\left[|E(\mathbf{s}(t+D))| = n \mid \mathbf{z}(t) = \mathbf{z}\right] \ge \mu > 0.$$
(8.4)

The final lemma has two parts; first, it states that, if all agents are at state E at some time t, then for any  $\mathbf{z}^* \in \mathcal{Z}_{NE}$ , there is positive probability that they will reach  $\mathbf{z}^*$  within  $4\Delta_{\eta}$  periods. Second, if the agents are at some  $\mathbf{z}^* \in \mathcal{Z}_{NE}$  at time t, with positive probability they will remain at  $\mathbf{z}^*$  for good.

Lemma 8.7. (i) Suppose that  $\mathbf{z}(t) = \mathbf{z} = (\mathbf{s}, \mathbf{a})$  where  $|E(\mathbf{s})| = n$ , i.e., all agents are at state E. Then, for all  $\mathbf{z}^* = (\mathbf{s}^*, \mathbf{a}^*) \in \mathcal{Z}_{NE}$ , we have

$$\mathbb{P}\left[\mathbf{z}(t+4\Delta_{\eta})=\mathbf{z}^{\star} \mid \mathbf{z}(t)=\mathbf{z}\right] \ge \rho_1 > 0.$$

(ii) For every  $\mathbf{z}^{\star} \in \mathcal{Z}_{NE}$ ,

$$\mathbb{P}\left[\mathbf{z}(t') = \mathbf{z}^{\star} \text{ for all } t' \ge t \mid \mathbf{z}(t) = \mathbf{z}^{\star}\right] \ge \rho_2 > 0.$$

The following corollary is a consequence of the above lemmas.

**Corollary 4.** There exist  $0 < \tilde{D} \leq 4(n+1)\Delta_{\eta}$  such that, for all  $\mathbf{z} \in \mathcal{Z} \setminus \mathcal{Z}_{NE}$ ,  $\mathbf{z}^{\star} \in \mathcal{Z}_{NE}$  and  $t \in \mathbb{N}$ , we have

$$\mathbb{P}\left[\mathbf{z}(t') = \mathbf{z}^{\star} \text{ for all } t' \ge t + \tilde{D} \mid \mathbf{z}(t) = \mathbf{z}\right] \ge \tilde{\mu} > 0.$$
(8.5)

Comparing Corollary 4 with Corollary 2 from the previous chapter, we observe that unlike the better reply rules the RSEM allows convergence to any PSNE starting from any initial action profile.

We now proceed with the proof of Theorem 8.3. Lemma 8.5 shows that, if the action profile is not a PSNE at time t, then after a finite number of periods, at least one agent will be at state E. Lemma 8.6 then claims that, whenever there is at least one agent at state E, after finitely many periods, all agents will be state E(Corollary 3). Once all agents are at state E, Lemma 8.7 asserts that they can reach any  $\mathbf{z}^{\star} \in \mathcal{Z}_{NE}$  with positive probability after a finite number of periods and stay there forever. Finally, Corollary 4 implies that, for all  $n \in \mathbb{N}$ ,  $\mathbb{P}\left[\vec{\mathbf{a}} \notin \mathcal{A}_{n\tilde{D}+1}^{\infty}\right] \leq (1-\tilde{\mu})^n$ . The rest of the proof follows using arguments similar to that of the proof for Theorem 7.5.

## 8.4 Proof of Lemmata for Theorem 8.3

#### 8.4.1 Proof of Lemma 8.5

Because  $\mathbf{a} \notin \mathcal{A}_{NE}$ , when the agents adopt  $\mathbf{a}$ , there is at least one agent, say agent  $i^*$ , with an incentive to deviate from  $a_{i^*}$  when the action profile  $\mathbf{a}$  is in effect at the systems of all agents. We will prove that the state of agent i will transition to E with positive probability after  $3\Delta_{\eta}$  periods.

We recall some of the notation introduced in the previous chapter: For each  $i \in \mathcal{P}$ , we define  $k_{\max}^i : \mathbb{N} \to \mathbb{N}$ , where  $k_{\max}^i(t) = \max\{k \in \mathbb{N} \mid T_k^i \leq t\}$ . Similarly, for all  $i, j \in \mathcal{P}, j \neq i$ , and  $t \in \mathbb{N}, d_{i,j}^-(t) = V_{k_{\max}^i(t)}^{i,j}$  and  $d_{i,j}^+(t) = V_{k_{\max}^i(t)+1}^{i,j}$ , where  $V_k^{i,j}$ , as defined in Section 7.5, is the forward delay that action  $a_i(T_k^i)$  experiences before influencing system j. For notational convenience, we denote the interval  $\{t + (\ell - 1)\Delta_{\eta} + 1, \ldots, t + \ell \cdot \Delta_{\eta}\}$  by  $\overline{T}_{\ell}(t), \ell \in \mathbb{N}$ , where  $\Delta_{\eta}$  is the constant in Assumption A4 in Section 7.5.

Next, we define the following events. Note that  $T^i_{k^i_{\max}(t)}$  (resp.  $T^i_{k^i_{\max}(t)+1}$ ) denotes the last time by time t (resp. the first time after time t) at which player i updates its strategy.

$$\mathcal{E}_{1} = \{T_{k_{\max}^{i}(t)}^{i} + d_{i,j}^{-}(t) \leq t + \Delta_{\eta} \text{ for all } i, j \in \mathcal{P}, j \neq i\}$$

$$\mathcal{E}_{2} = \{\mathcal{T}^{i^{*}} \cap \overline{T}_{\ell}(t) \neq \emptyset, \ \ell = 2, 3\}$$

$$\mathcal{E}_{3} = \{a_{i}(t') = a_{i} \text{ for all } i \in E(\mathbf{s}(t')), \ t' = t + 1, \dots, t + 3\Delta_{\eta}\}$$

$$\mathcal{E}_{4} = \{i^{*} \in E(\mathbf{s}(t'+1)) \text{ if } i^{*} \in E(\mathbf{s}(t')) \text{ for all } t' = t, \dots, t + 3\Delta_{\eta} - 1\}$$

The event  $\mathcal{E}_1$  implies that the action profile at time t is seen by all systems by time  $t + \Delta_{\eta}$ . The second event  $\mathcal{E}_2$  simply states that agent  $i^*$  updates its action at least once in each of the intervals  $\overline{T}_2$  and  $\overline{T}_3$ . Event  $\mathcal{E}_3$  requires any agent at state E between t + 1 and  $t + 3\Delta_{\eta}$  to choose the same action it did at time t (which will happen with strictly positive probability). Finally, the fourth event  $\mathcal{E}_4$  demands that the agent  $i^*$ , once it switches to state E (which will happen by time  $t + 3\Delta_{\eta}$  if the events  $\mathcal{E}_1$  through  $\mathcal{E}_3$  take place because  $\mathbf{a} \notin \mathcal{A}_{NE}$ ), remain at state E till time  $t + 3\Delta_{\eta}$ .

We use the following lower bound to complete the proof.

$$\mathbb{P}\left[E(\mathbf{a}(t+3\Delta_{\eta}))\neq\emptyset\mid\mathbf{z}(t)=(\mathbf{s}^{\star},\mathbf{a})\right]$$
  

$$\geq \mathbb{P}\left[E(\mathbf{a}(t+3\Delta_{\eta}))\neq\emptyset,\mathcal{E}_{1},\mathcal{E}_{2},\mathcal{E}_{3},\mathcal{E}_{4}\mid\mathbf{z}(t)=(\mathbf{s}^{\star},\mathbf{a})\right]$$
  

$$=\mathbb{P}\left[E(\mathbf{a}(t+3\Delta_{\eta}))\neq\emptyset\mid\mathcal{E}_{1},\mathcal{E}_{2},\mathcal{E}_{3},\mathcal{E}_{4},\mathbf{z}(t)=(\mathbf{s}^{\star},\mathbf{a})\right]$$
  

$$\times\mathbb{P}\left[\mathcal{E}_{1},\mathcal{E}_{2},\mathcal{E}_{3},\mathcal{E}_{4}\mid\mathbf{z}(t)=(\mathbf{s}^{\star},\mathbf{a})\right]$$
(8.6)

From the explanations of the events  $\mathcal{E}_1$  through  $\mathcal{E}_4$  above, it is clear that for every player  $i \in \mathcal{P}$ , we have  $\tilde{\mathbf{a}}_{-i}^i \left( T_{k_{\max}^i(t+2\Delta_\eta)+1}^i \right) = \mathbf{a}_{-i}$ . When this happens it is clear that at least one agent, namely agent  $i^*$  will see a better reply after  $t + 2\Delta_\eta$  and move to state E. Hence, the first conditional probability in (8.6) is one. Hence, if we can show that the second conditional probability is also positive, the lemma is proved.

First, we rewrite  $\mathbb{P}[\mathcal{E}_1, \mathcal{E}_2, \mathcal{E}_3, \mathcal{E}_4 \mid \mathbf{z}(t) = (\mathbf{s}^{\star}, \mathbf{a})]$  as follows.

$$\mathbb{P}\left[\mathcal{E}_{1}, \mathcal{E}_{2}, \mathcal{E}_{3}, \mathcal{E}_{4} \mid \mathbf{z}(t) = (\mathbf{s}^{\star}, \mathbf{a})\right]$$
$$= \mathbb{P}\left[\mathcal{E}_{1}, \mathcal{E}_{2} \mid \mathbf{z}(t) = (\mathbf{s}^{\star}, \mathbf{a})\right] \cdot \mathbb{P}\left[\mathcal{E}_{3}, \mathcal{E}_{4} \mid \mathbf{z}(t) = (\mathbf{s}^{\star}, \mathbf{a}), \mathcal{E}_{1}, \mathcal{E}_{2}\right].$$
(8.7)

The first term in (8.7) is lower bounded by  $\eta^{n+2}$  from Assumption A4. Bounding the second term in (8.7),

where the last step follows by noting that agents in the Explore state experiment with every action with a probability of at least  $\delta$ , while the step before that is a result of the fact that under the RSEM rule every agent remains at the explore state with a probability of at least p (recall that when an agent sees a better reply it continues at the explore state and when it sees no better replies it moves to the converged state with probability 1 - p). Putting together, we have the following lower bound

$$\mathbb{P}\left[E(\mathbf{a}(t+3\Delta_{\eta}))\neq\emptyset\mid\mathbf{z}(t)=(\mathbf{s}^{\star},\mathbf{a})\right] \ge (\delta^{n}\cdot p)^{3\Delta_{\eta}}\cdot\eta^{n+2}=:\zeta_{0}>0.$$

#### 8.4.2 Proof of Lemma 8.6

We consider two cases.

c1. There exists at least one agent  $i^+ \in C(\mathbf{s}(t))$  such that  $U_{i^+}(\tilde{\mathbf{a}}^{i^+}(T_k^{i^+})) \neq U_{i^+}(\mathbf{a}(t))$ or  $BR_{i^+}(\mathbf{a}(t)) \neq \emptyset$ . In other words, either the payoff received by agent  $i^+$  at the last time of update is different from what it would receive if its system
generated the payoff in response to the current action profile at time t, or the present action profile allows a better response for agent  $i^+$ .

c2. There is no such agent, i.e., for all  $i \in C(\mathbf{s}(t))$ , we have  $U_i(\tilde{\mathbf{a}}^i(T_k^i)) = U_i(\mathbf{a}(t))$ and  $BR_i(\mathbf{a}(t)) = \emptyset$ .

**Case c1:** First, we define the following events.

$$\mathcal{E}'_{1} = \{T^{i}_{k^{i}_{\max}(t)} + d^{-}_{i,j}(t) \leq t + \Delta_{\eta} \text{ for all } i, j \in \mathcal{P}, i \neq j\}$$
$$\mathcal{E}'_{2} = \{\mathcal{T}^{i^{+}} \cap \overline{T}_{\ell}(t) \neq \emptyset, \ \ell = 2, 3\}$$
$$\mathcal{E}'_{3} = \{a_{i}(t') = a_{i} \text{ for all } i \in E(\mathbf{s}(t')), \ t' = t + 1, \dots, t + 3\Delta_{\eta}\}$$
$$\mathcal{E}'_{4} = \{E(\mathbf{s}(t')) \subseteq E(\mathbf{s}(t'+1)) \text{ for all } t' = t, \dots, t + 3\Delta_{\eta} - 1\}$$

Using a similar argument used in Section 8.4.1, we obtain

$$\mathbb{P}\left[|E(\mathbf{s}(t+3\Delta_{\eta}))| \ge r+1 \mid \mathbf{z}(k) = \mathbf{z}\right]$$
  
$$\ge \mathbb{P}\left[|E(\mathbf{s}(t+3\Delta_{\eta}))| \ge r+1, \mathcal{E}'_{1}, \mathcal{E}'_{2}, \mathcal{E}'_{3}, \mathcal{E}'_{4} \mid \mathbf{z}(k) = \mathbf{z}\right]$$
  
$$= \mathbb{P}\left[|E(\mathbf{s}(t+3\Delta_{\eta}))| \ge r+1 \mid \mathcal{E}'_{1}, \mathcal{E}'_{2}, \mathcal{E}'_{3}, \mathcal{E}'_{4}, \mathbf{z}(t) = \mathbf{z}\right] \cdot \mathbb{P}\left[\mathcal{E}'_{1}, \mathcal{E}'_{2}, \mathcal{E}'_{3}, \mathcal{E}'_{4} \mid \mathbf{z}(t) = (\mathbf{z})\right]$$

Event  $\mathcal{E}'_4$  ensures that the set of exploring agents is non-decreasing, i.e., no exploring agents goes to the Converged state in the interval  $(t, t + 3\Delta_\eta]$ . Also the event  $\mathcal{E}'_3$  does not allow any agent to change its action. Therefore, if events  $\mathcal{E}'_1$ through  $\mathcal{E}'_4$  take place, by time  $t + 3\Delta_\eta$  agent  $i^+$  would have received the payoff information corresponding to the action profile  $\mathbf{a}(t)$  and switched its state to E, leading to an increase in the number of exploring agents. Thus, the first conditional probability in (8.9) is equal to one. Decomposing the second term in (8.9), we obtain

$$\mathbb{P}\left[\mathcal{E}_{1}^{\prime}, \mathcal{E}_{2}^{\prime}, \mathcal{E}_{3}^{\prime}, \mathcal{E}_{4}^{\prime} \mid \mathbf{z}(t) = \mathbf{z}\right]$$
$$= \mathbb{P}\left[\mathcal{E}_{1}^{\prime}, \mathcal{E}_{2}^{\prime} \mid \mathbf{z}(t) = \mathbf{z}\right] \cdot \mathbb{P}\left[\mathcal{E}_{3}^{\prime} \mid \mathbf{z}(t) = \mathbf{z}, \mathcal{E}_{1}^{\prime}, \mathcal{E}_{2}^{\prime}\right] \cdot \mathbb{P}\left[\mathcal{E}_{4}^{\prime} \mid \mathbf{z}(t) = \mathbf{z}, \mathcal{E}_{1}^{\prime}, \mathcal{E}_{2}^{\prime}, \mathcal{E}_{3}^{\prime}\right]. \quad (8.10)$$

From Assumption A4, the first term in (8.10) can be lower bounded by  $\eta^{n+2}$ . The second term can be lower bounded by  $\delta^{3n\Delta_{\eta}}$  following similar arguments used in the proof of Lemma 8.5. Lower bounding the third term in (8.10),

$$\mathbb{P}\left[\mathcal{E}'_{4} \mid \mathbf{z}(t) = \mathbf{z}, \mathcal{E}'_{1}, \mathcal{E}'_{2}, \mathcal{E}'_{3}\right] = \prod_{\ell=t}^{t+3\Delta_{\eta}-1} \mathbb{P}\left[E(s(\ell)) \subseteq E(s(\ell+1)) \middle| E(s(m)) \subseteq E(s(m+1))\right]$$
  
for all  $m = t, \dots, \ell-1, \mathbf{z}(t) = \mathbf{z}, \mathcal{E}'_{1}, \mathcal{E}'_{2}, \mathcal{E}'_{3}\right]$   
$$\geqslant p^{3n\Delta_{\eta}}$$
(8.11)

Returning to (8.9), the second conditional probability is lower bounded by  $(\delta \cdot p)^{3n\Delta_{\eta}} \cdot \eta^{n+2} > 0.$ 

**Case c2:** Recall that, in this case, for every agent  $i \in C(\mathbf{s}(t))$ , we have  $U_i(\tilde{\mathbf{a}}^i(T_k^i)) = U_i(\mathbf{a}(t))$  and  $BR_i(\mathbf{a}(t)) = \emptyset$ . Now, Assumption 8.2 implies that there exists  $i' \notin E(\mathbf{s}(t)) := J_1$  and  $\mathbf{a}_{J_1}^*$  such that if the agents in  $J_1$  adopt the actions in  $\mathbf{a}_{J_1}^*$  while the other agents choose the same action stipulated by  $\mathbf{a}(t)$ , then either agent i''s payoff changes or it sees a better reply. To move an agent in the Converged state to the Explore state, we will be using the weak interdependence assumption and argue that the exploring agents can change their actions such that at least one of the agents in the Converged state see a change in their payoff or a better reply and start exploring. As we will prove, this implies that there is positive probability that agent i' will switch its state to E after  $4\Delta_\eta$  periods.

To this end, we define the following events

$$\mathcal{E}_{1}^{+} = \{T_{k_{\max}^{i}(t+\Delta_{\eta})}^{i} + d_{i,j}^{-}(t+\Delta_{\eta}) \leq t+2\Delta_{\eta} \text{ for all } i, j \in \mathcal{P}, i \neq j\}$$

$$\mathcal{E}_{2}^{+} = \{\mathcal{T}^{i'} \cap \overline{T}_{\ell}(t) \neq \emptyset, \ \ell = 3, 4\}$$

$$\mathcal{E}_{3}^{+} = \{E(\mathbf{s}(t')) \subseteq E(\mathbf{s}(t'+1)) \text{ for all } t' = t, \dots, t+4\Delta_{\eta} - 1\}$$

$$\mathcal{E}_{4}^{+} = \{\mathcal{T}^{i} \cap \overline{T}_{1}(t) \neq \emptyset \text{ for all } i \in J_{1}\}$$

$$\mathcal{E}_{5}^{+} = \{a_{i}(t') = \tilde{a}_{i} \text{ for all } i \in \mathcal{P} \text{ and } t' = T_{k_{\max}^{i}(t)+1}^{i}, \dots, t+4\Delta_{\eta}\}$$

where

$$\tilde{a}_i = \begin{cases} a_i(t) & \text{if } i \notin J_1, \\ \\ a_i^* & \text{if } i \in J_1. \end{cases}$$

The rest of the proof follows from a similar argument.

$$\mathbb{P}\left[|E(\mathbf{s}(t+4\Delta_{\eta}))| \ge r+1 \mid \mathbf{z}(t) = \mathbf{z}\right] \\
\ge \mathbb{P}\left[|E(\mathbf{s}(t+4\Delta_{\eta}))| \ge r+1, \mathcal{E}_{1}^{+}, \mathcal{E}_{2}^{+}, \mathcal{E}_{3}^{+}, \mathcal{E}_{4}^{+}, \mathcal{E}_{5}^{+} \mid \mathbf{z}(t) = \mathbf{z}\right] \\
= \mathbb{P}\left[|E(\mathbf{s}(t+3\Delta_{\eta}))| \ge r+1 \mid \mathcal{E}_{1}^{+}, \mathcal{E}_{2}^{+}, \mathcal{E}_{3}^{+}, \mathcal{E}_{4}^{+}, \mathcal{E}_{5}^{+}, \mathbf{z}(t) = \mathbf{z}\right] \\
\times \mathbb{P}\left[\mathcal{E}_{1}^{+}, \mathcal{E}_{2}^{+}, \mathcal{E}_{3}^{+}, \mathcal{E}_{4}^{+}, \mathcal{E}_{5}^{+} \mid \mathbf{z}(t) = \mathbf{z}\right].$$
(8.12)

Event  $\mathcal{E}_4^+$  ensures that all the agents in the Explore state update during the interval  $\overline{T}_1(t)$ , while event  $\mathcal{E}_5^+$  ensures that the exploring agents update their actions according to the action profile  $(\mathbf{a}_{-J_1}, \mathbf{a}_{J_1}^*)$  and keep it fixed. Event  $\mathcal{E}_1^+$  enforces that, under events  $\mathcal{E}_4^+$  and  $\mathcal{E}_5^+$ , the current action profile at time  $t + \Delta_\eta - (\mathbf{a}_{-J_1}, \mathbf{a}_{J_1}^*)$  takes effect in the system of agent i' by time  $t + 2\Delta_\eta$ . Finally, events  $\mathcal{E}_2^+$  and  $\mathcal{E}_3^+$  ensure that agent  $i^+$  reaches the Explore state by time  $t + 4\Delta_\eta$  and stays there. From the definitions of the above events, the first conditional probability is one because agent i' will have switched its state to E (from C) by time  $t + 4\Delta_{\eta}$  and the exploring agents would still be exploring if the events  $\mathcal{E}_1^+$  through  $\mathcal{E}_5^+$  take place.

Lower bounding the second term in (8.12),

$$\mathbb{P}\left[\mathcal{E}_{1}^{+}, \mathcal{E}_{2}^{+}, \mathcal{E}_{3}^{+}, \mathcal{E}_{4}^{+}, \mathcal{E}_{5}^{+} \mid \mathbf{z}(t) = \mathbf{z}\right]$$
$$= \mathbb{P}\left[\mathcal{E}_{4}^{+} \mid \mathbf{z}(t) = \mathbf{z}\right] \cdot \mathbb{P}\left[\mathcal{E}_{1}^{+}, \mathcal{E}_{2}^{+} \mid \mathcal{E}_{4}^{+}, \mathbf{z}(t) = \mathbf{z}\right]$$
(8.13)

$$\times \mathbb{P}\left[\mathcal{E}_{3}^{+} \mid \mathcal{E}_{1}^{+}, \mathcal{E}_{2}^{+}, \mathcal{E}_{4}^{+}, \mathbf{z}(t) = \mathbf{z}\right] \cdot \mathbb{P}\left[\mathcal{E}_{5}^{+} \mid \mathcal{E}_{1}^{+}, \mathcal{E}_{2}^{+}, \mathcal{E}_{3}^{+}, \mathcal{E}_{4}^{+}, \mathbf{z}(t) = \mathbf{z}\right]$$
(8.14)

Using Assumption A4, we lower bound the first and second term in (8.13) by  $\eta^r$  and  $\eta^{n+2}$  respectively. Since the event  $\mathcal{E}'_4$  defined in the previous case is similar to event  $\mathcal{E}^+_3$ , we can use a similar argument to lower bound the first term in (8.14) by  $p^{4n\Delta\eta}$ . A lower bound of  $\delta^{4n\Delta\eta}$  is obtained for the final term in (8.14) by following similar steps as in the proof for Lemma 8.5 (see (8.8)).

Therefore, the second conditional probability in (8.12) is lower bounded by  $-\eta^{n+2+r} (\delta \cdot p)^{4n\Delta_{\eta}} := \varsigma_r$ , which is smaller than the lower bound obtained in case c1

 $((\delta \cdot p)^{3n\Delta_{\eta}} \cdot \eta^{n+2}).$ 

# 8.4.3 Proof of Lemma 8.7

We first prove Lemma 8.7(i). First, define the following events.

$$\mathcal{E}_{1}^{\#} = \{\mathcal{T}^{i} \cap \overline{T}_{1}(t) \neq \emptyset \text{ for all } i \in \mathcal{P}\}$$

$$\mathcal{E}_{2}^{\#} = \{T_{k_{\max}^{i}(t+\Delta_{\eta})}^{i} + d_{i,j}^{-}(t) \leq t + 2\Delta_{\eta} \text{ for all } i, j \in \mathcal{P}, j \neq i\}$$

$$\mathcal{E}_{3}^{\#} = \{a_{i}(t') = a_{i}^{\star} \text{ for all } i \in \mathcal{P} \text{ and } t' = T_{k_{\max}^{i}(t)+1}^{i}, \dots, t + 4\Delta_{\eta}\}$$

$$\mathcal{E}_{4}^{\#} = \{\mathcal{T}^{i} \cap \overline{T}_{\ell}(t) \neq \emptyset \text{ for all } i \in \mathcal{P} \text{ and } \ell = 3, 4\}$$

$$\mathcal{E}_{5}^{\#} = \{s_{i}(T_{k_{\max}^{i}(t+3\Delta_{\eta})+1}^{i}) = C \text{ for all } i \in E(\mathbf{s}(t+3\Delta_{\eta}))\}$$

Events  $\mathcal{E}_1^{\#}, \mathcal{E}_2^{\#}, \mathcal{E}_3^{\#}$  together imply that all agents update their actions to  $a_i^{\star}$ during  $\overline{T}_1(t)$  and their actions go into effect by  $t + 2\Delta_{\eta}$ . Events  $\mathcal{E}_4^{\#}$  and  $\mathcal{E}_5^{\#}$  state that all agents update at least once during the intervals  $\overline{T}_3(t)$  and  $\overline{T}_4(t)$ , and switch their state to C. Hence, together these events mean that all agents are at state Cand adopt the PSNE  $\mathbf{a}^{\star}$  at time  $t + 4\Delta_{\eta}$ . Therefore,

$$\mathbb{P}\left[\mathbf{z}(t+4\Delta_{\eta})=\mathbf{z}^{\star} \mid \mathbf{z}(t)=\mathbf{z}\right]$$

$$\geq \mathbb{P}\left[\mathbf{z}(t+4\Delta_{\eta})=\mathbf{z}^{\star} \mid \mathcal{E}_{1}^{\#}, \mathcal{E}_{2}^{\#}, \mathcal{E}_{3}^{\#}, \mathcal{E}_{4}^{\#}, \mathcal{E}_{5}^{\#}, \mathbf{z}(t)=\mathbf{z}\right]$$

$$\times \mathbb{P}\left[\mathcal{E}_{1}^{\#}, \mathcal{E}_{2}^{\#}, \mathcal{E}_{3}^{\#}, \mathcal{E}_{4}^{\#}, \mathcal{E}_{5}^{\#} \mid \mathbf{z}(t)=\mathbf{z}\right].$$
(8.15)

As argued above, the first conditional probability in (8.15) is one. Lower bounding the second conditional probability,

$$\mathbb{P}\left[\mathcal{E}_{1}^{\#}, \mathcal{E}_{2}^{\#}, \mathcal{E}_{3}^{\#}, \mathcal{E}_{4}^{\#}, \mathcal{E}_{5}^{\#} \mid \mathbf{z}(t) = \mathbf{z}\right]$$
$$= \mathbb{P}\left[\mathcal{E}_{1}^{\#}, \mathcal{E}_{2}^{\#}, \mathcal{E}_{4}^{\#} \mid \mathbf{z}(t) = \mathbf{z}\right] \cdot \mathbb{P}\left[\mathcal{E}_{3}^{\#} \mid \mathbf{z}(t) = \mathbf{z}, \mathcal{E}_{1}^{\#}, \mathcal{E}_{2}^{\#}, \mathcal{E}_{4}^{\#}\right]$$
(8.16)

$$\times \mathbb{P}\left[\mathcal{E}_{5}^{\#} \mid \mathbf{z}(t) = \mathbf{z}, \mathcal{E}_{1}^{\#}, \mathcal{E}_{2}^{\#}, \mathcal{E}_{3}^{\#}, \mathcal{E}_{4}^{\#}\right].$$
(8.17)

From previous arguments the first two terms in (8.16) are lower bounded by  $\eta^{4n}$  and  $\delta^{4n\Delta_{\eta}}$  respectively. To lower bound (8.17), observe that conditioned on the events  $\{\mathcal{E}_{i}^{\#}, i = 1, 2, 3, 4\}$ , the agents in the Explore state at time  $t + 3\Delta_{\eta}$  all see an empty better reply set whenever they update in the interval  $\overline{T}_{4}(t)$ . Therefore, under the RSEM rule, (8.17) is lower bounded by  $(1 - p)^{n}$ . Collecting the lower bounds, the second conditional probability in (8.15) can be lower bounded by  $\eta^{4n} \cdot \delta^{4n\Delta_{\eta}} \cdot (1-p)^{n}$ . This completes the proof of Lemma 8.7(i).

Before we prove Lemma 8.7(ii), note that even though the agents are at  $\mathbf{z}^{\star} \in \mathcal{Z}_{NE}$  at time t, it is still possible for some agents to transition to state E due to the delays in the system. Hence, the conditional probability in the lemma is strictly positive as we will show, but is in general less than one.

First, we define the following events.

$$\mathcal{E}_{1}^{-} = \{\mathcal{T}^{i} \cap \overline{T}_{\ell}(t) \neq \emptyset \text{ for all } i \in \mathcal{P} \text{ and } \ell = 2, 3\}$$

$$\mathcal{E}_{2}^{-} = \{a_{i}(t') = a_{i}^{\star} \text{ for all } i \in \mathcal{P} \text{ and } t' = T_{k_{\max}^{i}(t)+1}^{i}, \dots, t + 3\Delta_{\eta}\}$$

$$\mathcal{E}_{3}^{-} = \{T_{k_{\max}^{i}(t)}^{i} + d_{i,j}^{-}(t) \leq t + \Delta_{\eta} \text{ for all } i, j \in \mathcal{P}, j \neq i\}$$

$$\mathcal{E}_{4}^{-} = \{s_{i}(T_{k_{\max}^{i}(t+2\Delta_{\eta})+1}^{i}) = C \text{ for all } i \in E(\mathbf{s}(t+\Delta_{\eta}))\}$$

Note that events  $\mathcal{E}_1^-$  through  $\mathcal{E}_4^-$  mean that all agents continue to play the action profile  $\mathbf{a}^*$  between time t and  $t + 3\Delta_\eta$  (and afterwards). All agents see  $\mathbf{a}^*$  after time  $t + \Delta_\eta$  (event  $\mathcal{E}_3^-$ ) and all agents update during the interval  $\overline{T}_2(t)$  (event  $\mathcal{E}_1^-$ ). Thus, when the agents update during the interval  $\overline{T}_3(t)$ , conditional on event  $\mathcal{E}_2^-$ , all agents will see the payoff in response to  $\mathbf{a}^*$ , i.e.,  $\tilde{\mathbf{a}}^i \left(T_{k_{\max}^i(t+2\Delta_\eta)+1}^i\right) = \mathbf{a}^*$ . Finally, the agents' states will have changed to C by time  $t + 3\Delta_\eta$  (event  $\mathcal{E}_4^-$ ) and, as a result, they will keep playing  $\mathbf{a}^{\star}$  after time  $t + 3\Delta_{\eta}$ .

Following essentially the same argument, we have

$$\mathbb{P}\left[\mathbf{z}(t') = \mathbf{z}^{\star} \text{ for all } t' \ge t \mid \mathbf{z}(t) = \mathbf{z}\right]$$
$$\ge \mathbb{P}\left[\mathbf{z}(t') = \mathbf{z}^{\star} \text{ for all } t' \ge t \mid \mathbf{z}(t) = \mathbf{z}, \mathcal{E}_{1}^{-}, \mathcal{E}_{2}^{-}, \mathcal{E}_{3}^{-}, \mathcal{E}_{4}^{-}\right]$$
$$\times \mathbb{P}\left[\mathcal{E}_{1}^{-}, \mathcal{E}_{2}^{-}, \mathcal{E}_{3}^{-}, \mathcal{E}_{4}^{-}, \mid \mathbf{z}(t) = \mathbf{z}\right],$$
(8.18)

where the first conditional probability in (8.18) is one. The second conditional probability can be lower bounded by  $\eta^{2n} \cdot \delta^{3n\Delta_{\eta}}(1-p)^n$ .

# 8.5 Case with Erroneous Payoff Information

Until now in this chapter, we assumed that accurate payoff information  $(C_i(\tilde{\mathbf{a}}^i(t)); U_i(\tilde{\mathbf{a}}^i(t)))$ is available to the players for updates. For reasons already discussed in Section 7.6, it would be helpful to relax the above setting by assuming that the agents make mistakes while determining the payoff information  $C_i(\tilde{\mathbf{a}}^i(t))$ . Once again for ease of exposition, we consider the setting of the synchronous update case, which we believe can be easily extended to the asynchronous update scenario.

## 8.5.1 Preliminaries

The setting is that of Section 7.6. Recall that the probabilities of erroneous classification of actions are given by mappings  $\mathbf{q}_t^i : \mathcal{A} \times \mathcal{A}_i \to \Delta(\mathcal{S})$ , where  $\mathbf{q}_t^i(\mathbf{a}, a_i') :=$  $(q_t^i(\varsigma; \mathbf{a}, a_i'), \varsigma \in \mathcal{S})$  is a probability distribution over  $\mathcal{S}$ . We assume that the mappings  $\{\mathbf{q}_t^i, i \in \mathcal{P}, t \in \mathbb{N}^+\}$  satisfy Assumptions 7.1 and 7.2 stated in the previous chapter. As we did in Section 7.6, we can model the evolution of joint states  $\{(\mathbf{s}(t), \mathbf{a}(t)), t \in \mathbb{N}\}\$  as a nonhomogeneous (discrete-time) Markov chain, where the transition matrix at time  $t \in \mathbb{N}$ , denoted by  $\mathbf{P}(t)$ , is determined by: (i)  $\boldsymbol{\beta}_i, i \in \mathcal{P}$ , and the parameters of the RSEM rule p and  $\delta$ , and (ii) the mapping  $\mathbf{q}_t^i, i \in \mathcal{P}$  and  $t \in \mathbb{N}^+$ . For each  $t \in \mathbb{N}$ , a time homogeneous (discrete-time) Markov chain  $\mathbb{X}^t$  can be defined with a common state space  $\mathcal{C} \subset \mathbb{Z}^{-1}$  and transition matrix  $\mathbf{P}^t = \mathbf{P}(t)$ . For each  $t \in \mathbb{N}$  the Markov chain  $\mathbb{X}^t$  will be ergodic, which follows from the fact that, given any initial state with  $\mathbf{z}(1) = ((E, \ldots, E), \mathbf{a})$  with  $\mathbf{a} \in \mathcal{A}$ , we will revisit the same initial state  $\mathbf{z}(1)$  w.p. 1. The MC  $\mathbb{X}^t$  will thus have an unique stationary distribution denoted by  $\boldsymbol{\mu}^t = (\boldsymbol{\mu}^t(\mathbf{z}), \ \mathbf{z} \in \mathcal{C})$ . In this context, the MC in which the players make correct payoff observation is called an unperturbed Markov chain.

As in Section 7.6 we can construct a minimum resistance W-tree rooted at any  $\mathbf{z} \in \mathcal{C}$  denoted by  $\Gamma^*(\mathbf{z})$  with resistance  $\pi_{\min}(\mathbf{z}) := \pi(\Gamma^*(\mathbf{z}))$ . From Theorem 7.7 we know that, a state will be stochastically stable if its corresponding W-tree has the least resistance among all the minimum resistance W-trees. Next, we present results that characterize the set of stochastically stable states  $\mathcal{C}_{SS} \subset \mathcal{C}$ .

### 8.5.2 Main result

**Theorem 8.8.** As  $t \to \infty$ ,  $\mu^t \to \mu^0$ , where  $\mu^0$  is a stationary distribution of the unperturbed Markov chain. In addition,  $\mu^0(\mathbf{z}^*) > 0$  if and only if (i)  $\mathbf{z}^* \in \mathcal{Z}_{NE}$ and (ii)  $\psi(\mathbf{a}^*) = \max_{\mathbf{a}' \in \mathcal{Z}_{NE}} \psi(\mathbf{a}')$ , where  $\psi(\mathbf{a}) = \min_{i \in \mathcal{P}} \min_{a'_i \in \mathcal{A}_i \setminus \{a_i\}} \gamma_i(\mathbf{a}, a'_i, B)$  for

<sup>&</sup>lt;sup>1</sup>There may be some states in  $\mathcal{Z}$  which are unreachable from the initial state  $\mathbf{z}(1)$ , and we do not include those unreachable states in the state space  $\mathcal{C}$ .

 $\mathbf{a} \in \mathcal{A}_{NE}.$ 

**Proof.** First we prove the first statement. For any  $\mathbf{z}' \in C \setminus \mathcal{Z}_{NE}$ , from Corollary 4 we know that there exists a zero resistance path from  $\mathbf{z}'$  to any joint state in  $\mathcal{Z}_{NE}$ . In other words according to the terminology introduced in Section 7.6, the domain of attraction of the joint state  $\mathbf{z}'$  is  $\mathcal{Z} \setminus \mathcal{Z}_{NE}$ . Also, it is clear that the minimum resistance path from any state in  $\mathcal{Z}_{NE}$  to any state in  $\mathcal{C} \setminus \mathcal{Z}_{NE}$  is strictly positive. Therefore, for all  $\mathbf{z}' \in \mathcal{C} \setminus \mathcal{Z}_{NE}$ ,

$$\pi_{\min}(\mathbf{z}') > \pi_{\min}(\mathbf{z}^*), \ \mathbf{z}^* \in \mathcal{Z}_{NE}$$

which is sufficient to prove the first statement.

To prove the second statement we once again use Corollary 4 to conclude that any state  $\mathbf{z}' \in \mathcal{C} \setminus \mathcal{Z}_{NE}$  will contribute zero resistance to the resistance tree  $\Gamma^*(\mathbf{z}^*)$  for any  $\mathbf{z}^* \in \mathcal{Z}_{NE}$ . Therefore, for  $\mathbf{z}^* \in \mathcal{Z}_{NE}$ 

$$\pi_{\min}(\mathbf{z}^*) = \sum_{\mathbf{z} \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}^*\}} \rho(\mathbf{z}, \mathbf{z}^*).$$
(8.19)

Choose  $\mathbf{z}'' = (\mathbf{s}^*, \mathbf{a}''), \mathbf{z}^* = (\mathbf{s}^*, \mathbf{a}^*) \in \mathcal{Z}_{NE}$ . The minimum resistance transition from the state  $\mathbf{z}''$  to a state where at least one agent is exploring, i.e., a joint state in  $\mathcal{C} \setminus \mathcal{Z}_{NE}$ , has a resistance of  $\psi(\mathbf{a}'')$ . Once a state in  $\mathcal{C} \setminus \mathcal{Z}_{NE}$  is reached, we know that a zero resistance path to  $\mathbf{z}^*$  exists, which implies that  $\rho(\mathbf{z}'', \mathbf{z})$  is simply  $\psi(\mathbf{a}'')$ . Returning to (8.19), we obtain

$$\pi_{\min}(\mathbf{z}^*) = \sum_{\mathbf{z}=(\mathbf{s}^*,\mathbf{a})\in\mathcal{Z}_{NE}\setminus\{\mathbf{z}^*\}} \psi(\mathbf{a}) = \sum_{\mathbf{z}=(\mathbf{s}^*,\mathbf{a})\in\mathcal{Z}_{NE}} \psi(\mathbf{a}) - \psi(\mathbf{a}^*)$$
(8.20)

which implies that

$$\mathcal{C}_{SS} = \arg\min_{\mathbf{z}^* \in \mathcal{Z}_{NE}} \pi_{\min}(\mathbf{z}^*) = \arg\max_{\mathbf{z}^* \in \mathcal{Z}_{NE}} \psi(\mathbf{a}^*).$$
(8.21)

For  $\mathbf{z} = (\mathbf{s}^*, \mathbf{a}) \in \mathcal{Z}_{NE}$ , by definition,  $\psi(\mathbf{a})$  is the resistance of the minimum resistance transition from the PSNE  $\mathbf{z}$  to any other state. For all time  $t \in \mathbb{N}^+$ , the probability of deviation from a PSNE joint state  $(\mathbf{s}^*, \mathbf{a})$  is given by  $\sum_{\mathbf{z}' \in \mathcal{C} \setminus \mathcal{Z}_{NE}} P_{\mathbf{z}, \mathbf{z}'}(t)$ . We know for each  $\mathbf{z} = (\mathbf{s}^*, \mathbf{a}) \in \mathcal{Z}_{NE}$  there exists a constant  $c(\mathbf{a}) > 0$  such that  $\sum_{\mathbf{z}' \in \mathcal{C} \setminus \mathcal{Z}_{NE}} P_{\mathbf{z}, \mathbf{z}'}(t) \sim c(\mathbf{a}) \epsilon_t^{\psi(\mathbf{a})}$ . Therefore for large t, Theorem 8.8 ensures stochastic stability of those PSNE(s) which under the error model has among the least probability to get destabilised.

From the expression of  $C_{SS}$  in (8.21), it is worth noting that the RSEM rule is such that the set of stochastically stable states is completely characterised by the error model  $\{\mathbf{q}_t^i, i \in \mathcal{P}, t \in \mathbb{N}^+\}$ . This is certainly not the case for the GBRR rules, where the set of stochastically stable states depend both on the structure of the game and the error model.

## Chapter 9: Monitoring Rules under Erroneous Execution

In the previous chapter we introduced the class of monitoring rules by looking at a simple version called the Reduced Simple Experimentation with Monitoring (RSEM) rule. We considered asynchronous updates by agents under delayed payoff information and errors due to faulty payoff information, more particularly, errors due erroneous estimation of SBRs. In this chapter we specifically consider errors due to faulty execution of intended actions. This requires a generalization of the simple RSEM by endowing an agent with several monitoring or alert states such that the agent/controller becomes resilient to occasional deviations from expected situations but is instead responsive to long term changes.

## 9.1 Resilience of PSNEs

As mentioned earlier, we are interested in designing a distributed learning rule that will allow the agents to target (more) resilient equilibria. To this end, we first define the *resilience* of a PSNE as follows.

Denote the set of PSNEs by  $\mathcal{A}_{NE} \subset \mathcal{A}$ . Let  $d : \mathcal{A} \times \mathcal{A} \to \mathbb{Z}_+ := \{0, 1, 2, \ldots\}$ be a mapping that measures the distance between two action profiles, where

$$d(\mathbf{a}^1, \mathbf{a}^2) = \sum_{i \in \mathcal{P}} \mathbf{1} \left[ a_i^1 \neq a_i^2 \right], \quad \mathbf{a}^1, \mathbf{a}^2 \in \mathcal{A}.$$
(9.1)

For each  $\tau \in \mathbb{Z}_+$ , let  $\mathcal{N}_\tau : \mathcal{A} \to 2^{\mathcal{A}}$ , where  $\mathcal{N}_\tau(\mathbf{a}) = \{\mathbf{a}' \in \mathcal{A} \mid d(\mathbf{a}, \mathbf{a}') \leq \tau\}$ ,  $\mathbf{a} \in \mathcal{A}$ . In other words,  $\mathcal{N}_\tau(\mathbf{a})$  is the set of action profiles whose distance from  $\mathbf{a}$  is at most  $\tau$  according to the distance measure in (9.1).

Recall from Chapter 7 that  $C_i$  and  $BR_i$  are the classification and better reply mappings respectively. The resilience of a PSNE is given by a mapping  $R : \mathcal{A}_{NE} \rightarrow \mathbb{Z}_+$ , where for  $\mathbf{a}^* \in \mathcal{A}_{NE}$ ,

$$R(\mathbf{a}^{\star}) = \max\{\tau \ge 0 \mid BR_i(a_i^{\star}, \mathbf{a}_{-i}') = \emptyset \text{ for all } i \in \mathcal{P} \text{ and } \mathbf{a}' \in \mathcal{N}_{\tau}(\mathbf{a}^{\star})\}.$$

It is clear from the definition that the resilience of a PSNE is the largest number of deviating agents the PSNE can tolerate before at least one agent finds an incentive to switch its action. When  $R(\mathbf{a}^{\star}) = K$ , we say that  $\mathbf{a}^{\star}$  is K-resilient. Define  $R_{\max}^{\star} := \max_{\mathbf{a}^{\star} \in \mathcal{A}_{NE}} R(\mathbf{a}^{\star})$  to be the maximum resilience among all PSNEs.

### 9.2 Proposed Update Rule

In this section, we present our proposed algorithm, called the Simple Experimentation with Monitoring (SEM), for seeking resilient PSNEs. This is an extension of the RSEM algorithm discussed in the previous chapter. In order to describe the algorithm, we first introduce some notation. At each time  $t \in \mathbb{N}$ , the state of agent  $i \in \mathcal{P}$  is denoted by  $s_i(t) \in \Psi'_0$ . The set  $\Psi'_0$  consists of (i) Converged (C), (ii) Explore (E), and (iii) Alert  $(L_1, \ldots, L_T)$ , where T is the number of alert states. We let  $\Psi' = \prod_{i \in \mathcal{P}} \Psi'_0$ , and the state vector at time  $t \in \mathbb{N}$  is given by  $\mathbf{s}(t) = (s_i(t), i \in \mathcal{P}) \in \Psi'$ . The rule governing the update of an agent's state will be explained shortly. • Action profile selected by agents vs. action profile adopted by agents: Since we are interested in scenarios where the agents intermittently make mistakes and execute incorrect actions, we distinguish (i) the action that is selected by agent *i* at time  $t \in \mathbb{N}$ , namely  $a_i(t)$ , according to the action update rule (described in Section 9.2.1) and (ii) the action that is adopted at time  $t \in \mathbb{N}$ , denoted by  $\tilde{a}_i(t)$ (Section 9.2.2). We denote the action profile selected by the agents at time  $t \in \mathbb{N}$  by  $\mathbf{a}(t)$ , and the action profile adopted at time t by  $\tilde{\mathbf{a}}(t)$ . Note that  $\mathbf{a}(t)$  and  $\tilde{\mathbf{a}}(t)$  may not be identical unless no agent makes a mistake at time t.

**Payoff Information for updates** – First, let us denote  $\hat{\mathbf{a}}_i(t) = (a_i(t), \tilde{\mathbf{a}}_{-i}(t))$ . Clearly,  $\hat{\mathbf{a}}_i(t)$  is the action profile that would be in effect at time t assuming agent i did not make any erroneous implementation of its selected action. We assume that the payoff information available to agent i at time  $t \ge 2$  is of the form –  $((\mathcal{C}_i(a_i, \hat{\mathbf{a}}_i(t-1)), a_i \in \mathcal{A}_i); U_i(\tilde{\mathbf{a}}(t-1))).$  Therefore, the payoff classification information evaluates which strategies would have yielded a higher or a lower payoff than the previously selected strategy  $a_i(t-1)$ . Clearly, it is sensible to use the payoff classification information with respect to the intended action  $a_i(t-1)$  irrespective of whether that action was actually implemented or not. We also assume that an agent knows whether or not an action was correctly implemented in the previous time instant. This makes sense in many practical scenarios where an agent can monitor if the intended action was actually implemented. For instance, returning to the example of distributed traffic routing considered in Chapter 6, it is reasonable to assume that a driver can comprehend in hindsight if it correctly followed the suggested route.

Now, in case there were an error in implementation by a particular agent i, i.e.,  $\tilde{a}_i(t-1) \neq a_i(t-1)$ , there might be scenarios where obtaining the aforementioned payoff classification information might be difficult for that particular agent. Although we do not consider such a setting here, we would however like to comment that the algorithm can be easily modified to account for such scenarios.

# 9.2.1 Action updates with no errors $-\mathbf{a}(t), t \in \mathbb{N}$

In this subsection, we first describe how the agents choose their actions according to the payoff feedback. At time t = 1, the agents choose their action profile  $\mathbf{a}(1)$  according to some joint distribution  $\mathcal{G}$ . Starting with t = 2, at time  $t \in \mathbb{N}$ , every agent updates its action using some update rule, which will be explained shortly. Fix  $\delta \in (0, 1/(\max_{i \in \mathcal{P}} |\mathcal{A}_i|))$  and  $\mathcal{\beta}_i : \mathcal{S}^{A_i} \to \Delta(\mathcal{A}_i), i \in \mathcal{P}$ .

#### Action Selection Rule:

For  $t = 2, 3, \ldots$ ,

• if  $s_i(t) = E$ 

- choose  $a_i(t) = a_i$  with probability  $\beta_i(a_i; \mathcal{C}_i(a_i, \hat{\mathbf{a}}_i(t-1)) \ge \delta$  for all  $a_i \in \mathcal{A}_i$ 

• else (i.e.,  $s_i(t) = C$  or  $L_\ell, \ \ell \in \mathcal{L} := \{1, 2, \dots, T\}$ )

 $- \text{ set } a_i(t) = a_i(t-1)$ 

Note that, under SEM, an agent may choose a new action only if it is at state E. Otherwise, it continues to play the same action employed at the previous time.

# 9.2.2 Adopted action profiles $-\tilde{\mathbf{a}}(t), t \in \mathbb{N}$

As mentioned earlier, we are interested in designing a distributed learning rule that will allow the agents to seek out resilient PSNEs under erroneous decisionmaking by the agents. Here, we describe one way to model these erroneous actions with the help of mutually independent Bernoulli processes<sup>1</sup>: For each agent  $i \in \mathcal{P}$ , let  $\mathcal{B}_i = \{B_i(t), t \in \mathbb{N}\}$  be a Bernoulli process with  $\mathbb{P}[B_i(t) = 1] = \epsilon$  for some  $\epsilon > 0$ . The random variable  $B_i(t)$ ,  $i \in \mathcal{P}$  and  $t \in \mathbb{N}$ , indicates whether or not system of agent *i* makes a mistake and executes an incorrect action at time *t* as follows. For each  $i \in \mathcal{P}$  and  $t \in \mathbb{N}_+$ ,

- E1. if  $B_i(t) = 0$ ,  $\tilde{a}_i(t) = a_i(t)$ ;
- E2. else (i.e.,  $B_i(t) = 1$ ),  $\mathbb{P}[\tilde{a}_i(t) = a_i] = g_i(a_i(t), a_i)$  independently of the past, where for each  $a_i \in \mathcal{A}_i$ ,  $g_i(a_i, \cdot)$  is some arbitrary distribution over  $\mathcal{A}_i \setminus \{a_i\}$  with  $g_i(a_i, a'_i) > 0$  for all  $a'_i \in \mathcal{A}_i \setminus \{a_i\}$ .

According to this setup, each agent adopts an erroneous action with probability  $\epsilon$  at each time  $t \in \mathbb{N}_+$ , independently of the past and other agents. As previously

<sup>&</sup>lt;sup>1</sup>While we use a specific model for introducing perturbations in the system as a reasonable approximation, identifying the correct or accurate faulty behavior is not the focus here and other perturbation models may be used instead.

mentioned, for each  $i \in \mathcal{P}$  and  $t = 2, 3, \ldots$ , we assume at time t agent i can observes  $B_i(t-1)$ . In other words, at a particular time t, agent i knows whether an action was correctly implemented or not in the previous time instant t-1.

## 9.2.3 State dynamics

As explained in the previous subsections, under SEM, agents' selections of actions depend on their states. Hence, the dynamics of the agents' states play a key role in the algorithm. In this subsection, we explain how the agents update their states based on the received payoff information.

For each  $i \in \mathcal{P}$  and  $t \in \mathbb{N}_+$  such that  $s_i(t) \in \{L_\ell, \ell \in \mathcal{L}\}$ , let

$$\nu_t^i = \max\{t' < t : s_i(t') = C\}$$
(9.2)

and  $U_i^*(t) = U_i(\tilde{\mathbf{a}}(\nu_t^i))$ . Note that the payoff information vector is determined by the actions *adopted* by other agents, not those *selected* by them. Hence, even when the agents select a PSNE at time t, i.e.,  $\mathbf{a}(t) \in \mathcal{A}_{NE}$ , the adopted action profile  $\tilde{\mathbf{a}}(t)$ and, hence, the payoff information vectors seen by the agents may differ from those of the selected PSNE.

At time t = 1, all agents are initially at state E, i.e.,  $\mathbf{s}(1) = (E, \ldots, E)$ . The state of agent i at time  $t \ge 2$  depends on its state at time t - 1 and the payoff information available at time t. Note that at time instant 2 only case S1 is applicable because at time instant 1 all the agents are in the Explore state. Therefore case S2 in the state update rule can be effective only for  $t \ge 3$ . Similarly, case S3 is applicable only for  $t \ge 4$ . Fix  $p \in (0, 1)$ .

# State Update Rule:

For t = 2, 3, ...,

S1. if  $s_i(t-1) = E$ 

• if 
$$BR_i(\hat{\mathbf{a}}_i(t-1)) \neq \emptyset$$
:  $s_i(t) = E$ 

• else (i.e., 
$$BR_i(\hat{\mathbf{a}}_i(t-1)) = \emptyset$$
):

$$s_i(t) = \begin{cases} E & \text{with probability } p \\ C & \text{with probability } 1 - p \end{cases}$$

S2. if  $s_i(t-1) = C$ 

• if 
$$BR_i(\hat{\mathbf{a}}_i(t-1)) \neq \emptyset$$
:  $s_i(t) = E$ 

 $\bullet$  else

- if 
$$B_i(t-1) = 0$$
 and  $U_i(\tilde{\mathbf{a}}(t-1)) \neq U_i^*(t-1)$ :  $s_i(t) = L_1$   
- else (i.e., (i)  $B_i(t-1) = 1$ , or (ii)  $B_i(t-1) = 0$  and  $U_i(\tilde{\mathbf{a}}(t-1)) = U_i^*(t-1)$ ):  $s_i(t) = C$ 

S3. if  $s_i(t-1) = L_\ell$ ,  $\ell = 1, 2, \dots, T-1$ 

- if  $BR_i(\hat{\mathbf{a}}_i(t-1)) \neq \emptyset$ :  $s_i(t) = E$
- $\bullet$  else

- if  $B_i(t-1) = 1$ :  $s_i(t) = L_\ell$ 

S4. if  $s_i(t-1) = L_T$ 

• if 
$$BR_i(\hat{\mathbf{a}}_i(t-1)) \neq \emptyset$$
:  $s_i(t) = E$ 

• else

- if 
$$B_i(t-1) = 1$$
:  $s_i(t) = L_T$   
- else if  $B_i(t-1) = 0$  and  $U_i(\tilde{\mathbf{a}}(t-1)) \neq U_i^*(t-1)$ :  $s_i(t) = E$   
- else (i.e.,  $B_i(t-1) = 0$  and  $U_i(\tilde{\mathbf{a}}(t-1)) = U_i^*(t-1)$ ):  $s_i(t) = C$ 

In essence, if the SBR set is nonempty, the agent always moves to state E. Also, when the agent is at an alert state,  $L_{\ell}$ ,  $\ell \in \mathcal{L}$ , if there were erroneous implementation of the selected action then the agent stays in the same state, otherwise if the selected action was carried out and the received payoff goes back to  $U_i^*(t-1)$ , i.e., the payoff it was expecting the last time it was at state C, it immediately returns to state C. Otherwise, even when the SBR set is empty, it moves to the next alert state. The state transition of an agent is summarized in Figure 9.1.

# 9.3 Main Result

We redefine some notation from the previous chapter: Let  $\mathcal{Z} := \Psi' \times \mathcal{A}$  and denote the pair  $(\mathbf{s}(t), \mathbf{a}(t)) \in \mathcal{Z}$  by  $\mathbf{z}(t), k \in \mathbb{N}$ . Define  $\mathbf{s}^* \in \Psi'$  to be the state vector



Figure 9.1: State transition diagram in SEM.

in which every agent is at state C, i.e,  $\mathbf{s}^{\star} = (C, \ldots, C)$ , and  $\mathcal{Z}_{NE} = \{(\mathbf{s}^{\star}, \mathbf{a}^{\star}) \in \mathcal{Z} \mid \mathbf{a}^{\star} \in \mathcal{A}_{NE}\}.$ 

We assume the game satisfies the weak payoff interdependence assumption (Assumption 8.2) introduced in the previous chapter, which essentially states that, for every action profile  $\mathbf{a} \in \mathcal{A}$  and subset of agents  $J \subsetneq \mathcal{P}$  such that all the agents outside the set J see no better reply, there exists an agent  $i^* \notin J$  and a choice of action profile  $\mathbf{a}_J^* \in \prod_{j \in J} \mathcal{A}_j$  such that under the action profile  $(\mathbf{a}_J^*, \mathbf{a}_{-J})$ , agent  $i^*$  either sees a better reply  $(BR_{i^*}(\mathbf{a}_J^*, \mathbf{a}_{-J}) \neq \emptyset)$  or a change in its payoff  $(U_{i^*}(\mathbf{a}) \neq U_{i^*}(\mathbf{a}_J^*, \mathbf{a}_{-J}))$  Assumption 9.1. There exists a decreasing, positive sequence  $(\epsilon_t, t \in \mathbb{N})$  such that  $\lim_{t\to\infty} \epsilon_t = 0.$ 

As in the previous chapter, we can model the evolution of joint states  $\{(\mathbf{s}(t), \mathbf{a}(t)), t \in \mathbb{N}\}$  as a nonhomogeneous (discrete-time) Markov chain, where the transition matrix at time  $t \in \mathbb{N}$  is denoted by  $\mathbf{P}(t)$ . For each  $t \in \mathbb{N}$ , a time homogeneous (discretetime) Markov chain  $\mathbb{X}^t$  can be defined with a common state space  $\mathcal{C} \subset \mathbb{Z}^{-2}$  and transition matrix  $\mathbf{P}^t = \mathbf{P}(t)$ . For reasons explained in the previous chapter, under Assumption 9.1, for every  $t \in \mathbb{N}$  the present Markov chain  $\mathbb{X}^t$  will also be ergodic with an unique stationary distribution denoted by  $\boldsymbol{\mu}^t = (\boldsymbol{\mu}^t(\mathbf{z}), \ \mathbf{z} \in \mathcal{C})$ .

The following main result tells us that, if all agents adopt the proposed SEM learning rule, one of the two statements holds. The proof of the theorem is provided in Section 9.4.

**Theorem 9.1.** Suppose that the game  $\mathbb{G}$  satisfies Assumption 8.2 and  $\mathcal{A}_{NE}$  is nonempty. Under Assumption 9.1, as  $t \to \infty$ ,  $\mu^t \to \mu^0$  where  $\mu^0$  is a stationary distribution of the unperturbed Markov chain.

(a) If 
$$R_{\max}^* < T$$
, then  $\mu^0(\mathbf{z}^*) > 0$  if and only if (i)  $\mathbf{z}^* \in \mathcal{Z}_{NE}$ , and (ii)  $R(\mathbf{a}^*) = R_{\max}^*$ 

(b) If  $R_{\max}^* \ge T$ , then  $\mu^0(\mathbf{z}^*) > 0$  if and only if (i)  $\mathbf{z}^* \in \mathcal{Z}_{NE}$ , and (ii)  $R(\mathbf{a}^*) \ge T$ .

The findings in Theorem 9.1 can be interpreted as follows. First, note that, when  $\epsilon > 0$ , due to the occasional erroneous actions, it is not possible to obtain

<sup>&</sup>lt;sup>2</sup>As we noted in the previous chapter, there may be some states in  $\mathcal{Z}$  which are unreachable from the initial state  $\mathbf{z}(1)$ , and we do not include those unreachable states in the state space  $\mathcal{C}$ .

almost sure convergence to any PSNE. However, our result demonstrates that it is possible to *guide* the agents to PSNEs with certain resilience properties. If the parameter T, i.e., the number of alert states, is strictly larger than the maximum resilience among all PSNEs, when all agents adopt the SEM rule, for small error probability  $\epsilon$ , the agents will spend a majority of time at the most resilient PSNE(s). On the other hand, if there are PSNEs whose resilience is at least T, then the agents will spend most of their time at the PSNEs that are at least T-resilient. Hence, the SEM rule offers a *tunable* knob, namely T, with which we can choose the desired resilience of PSNEs at which the agents will spend most of the time.

Elementary Markov chain theory allows us to assert that when  $\epsilon = 0$ ,  $\mathbf{a}(t)$  converges to a PSNE in  $\mathcal{A}_{NE}$  almost surely under Assumption 8.2. Theorem 9.1 goes one step further; it states that, even when the agents are not perfect and make sporadic mistakes in executing the selected actions, under the SEM rule, the agents will stay at the resilient PSNEs most of the time, and the subset of PSNEs at which they spend most time can be chosen via the tunable parameter T in the rule.

## 9.4 Proof of Theorem 9.1

We first state a lemma that will be used to prove the theorem. Recall that given a state vector  $\mathbf{s} \in \Psi'$ , the set of agents in the Explore state is denoted by  $E(\mathbf{s})$ .

**Lemma 9.2.** Suppose that the game  $\mathbb{G}$  satisfies Assumption 8.2 and  $\mathcal{A}_{NE} \neq \emptyset$ . When  $\epsilon = 0$ , under SEM, the following hold. i. For all  $\mathbf{z}^* \in \mathcal{Z}_{NE}$  and  $\mathbf{z} = (\mathbf{s}, \mathbf{a}) \in \mathcal{C}$  with  $\mathbf{s} \in \Psi'$  such that  $E(\mathbf{s}) \neq \emptyset$ , there exist  $0 < D_1 < \infty$  and  $0 < \rho_1 < 1$  such that

$$\mathbb{P}\left[\mathbf{z}(t+D_1) = \mathbf{z}^* \mid \mathbf{z}(t) = \mathbf{z}\right) \ge \rho_1.$$

ii. For all  $\mathbf{z} = (\mathbf{s}, \mathbf{a}^*) \in \mathcal{C}$  with  $\mathbf{a}^* \in \mathcal{A}_{NE}$  and  $\mathbf{s} \in \Psi' \setminus \{\mathbf{s}^*\}$ , there exist  $0 < D_2 < \infty$ and  $0 < \rho_2 < 1$  such that

$$\mathbb{P}\left[\mathbf{z}(t+D_2) = (\mathbf{s}^{\star}, \mathbf{a}^{\star}) \mid \mathbf{z}(t) = \mathbf{z}\right] \ge \rho_2.$$

iii. For all  $\mathbf{z}^* \in \mathcal{Z}_{NE}$  and  $\mathbf{z} = (\mathbf{s}, \mathbf{a}) \in \mathcal{C}$  with  $\mathbf{a} \notin \mathcal{A}_{NE}$ , there exist  $0 < D_3 < \infty$ and  $0 < \rho_3 < 1$  such that

$$\mathbb{P}\left[\mathbf{z}(t+D_3)=\mathbf{z}^* \mid \mathbf{z}(t)=\mathbf{z}\right] \ge \rho_3.$$

Lemma 9.2 also implies that  $\mathcal{Z}_{NE} \subset \mathcal{C}$ . It also leads to the following corollary.

**Corollary 9.3.** Suppose that the game  $\mathbb{G}$  satisfies Assumption 8.2 and  $\mathcal{A}_{NE} \neq \emptyset$ . Then, when  $\epsilon = 0$ , under the SEM rule, only the states in  $\mathcal{Z}_{NE}$  are absorbing states.

We now proceed with the proof of the theorem. Recall that  $\mathbb{X}^t$  denotes the MC with state space  $\mathcal{C} \subset \mathcal{Z}$ , in which each agent makes mistake with probability  $\epsilon_t$ , independently of each other. Let  $\mathbf{P}^t = [P_{\mathbf{z}_1, \mathbf{z}_2}^t; \mathbf{z}_1, \mathbf{z}_2 \in \mathcal{C}]$  denote the corresponding (one-step) transition matrix, where  $P_{\mathbf{z}_1, \mathbf{z}_2}^t$  is the transition probability from  $\mathbf{z}_1$  to  $\mathbf{z}_2$ .

Under Assumption 9.1, if for some  $t \in \mathbb{N}$ ,  $\mathbf{z}_1, \mathbf{z}_2 \in \mathcal{C}$ ,  $\mathbf{z}_1 \neq \mathbf{z}_2$ , and  $\epsilon_t > 0$ ,  $P_{\mathbf{z}_1, \mathbf{z}_2}(t) > 0$ , then there is  $r(\mathbf{z}_1, \mathbf{z}_2) \ge 0$  such that

$$0 < \lim_{t \to \infty} \frac{P_{\mathbf{z}_1, \mathbf{z}_2}(t)}{\epsilon_t^{r(\mathbf{z}_1, \mathbf{z}_2)}} < \infty.$$

This defines the resistance between two joint states in  $\mathcal{C}$ . Since the Markov chain  $\mathbb{X}^t$  is ergodic for all  $t \in \mathbb{N}$ , we know there exists at least one sequence of transitions between any two joint states in  $\mathcal{C}$ . Therefore, we can define path resistances and resistance of minimum resistance paths between any two joint states as in Section 7.6. Therefore we can proceed as in Section 8.5, and construct minimum resistance W-tree  $\Gamma^*(\mathbf{z})$  rooted at any  $\mathbf{z} \in \mathcal{C}$ , to characterize the set of stochastically stable states.

First, we can show that any state  $\mathbf{z} \in \mathcal{C} \setminus \mathcal{Z}_{NE}$  is not SS. In other words, only the absorbing states of the unperturbed MC, which belong to  $\mathcal{Z}_{NE}$  by Corollary 9.3, are SS from the definition of stochastic stability.

The theorem can be proved with the help of the following three claims, which we will prove shortly.

- C1. For any  $\mathbf{z}^* \in \mathcal{Z}_{NE}$ , the resistance of the minimum resistance  $\mathbf{z}^*$ -tree,  $\Gamma^*(\mathbf{z}^*)$ , is equal to that of the Steiner tree in  $\Gamma^*(\mathbf{z}^*)$  that connects only the states in  $\mathcal{Z}_{NE}$  to  $\mathbf{z}^*$ . In order to prove this, we argue that the edges not in the Steiner tree have zero resistance.
- C2. Fix any  $\mathbf{z}' = (\mathbf{s}^*, \mathbf{a}') \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}^*\}$  and let  $\mathbf{z}^{\dagger}$  be the first state in  $\mathcal{Z}_{NE}$  visited along the (directed) path from  $\mathbf{z}'$  to  $\mathbf{z}^*$ . We call  $\mathbf{z}^{\dagger}$  the *parent state* of  $\mathbf{z}'$  in

 $\Gamma^*(\mathbf{z}^*)$  and denote it by  $\eta(\mathbf{z}', \Gamma^*(\mathbf{z}^*))$ . Then, the resistance of the subpath from  $\mathbf{z}'$  to  $\mathbf{z}^{\dagger}$  is equal to  $\min\{R(\mathbf{a}'), T\} + 1$ .

C3. Suppose that  $\mathbf{z}'$  and  $\mathbf{z}''$  ( $\mathbf{z}', \mathbf{z}'' \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}^*\}$ ) have the same parent state, say  $\mathbf{z}^{\dagger}$ . Then, the intersection of the subpaths from  $\mathbf{z}'$  and  $\mathbf{z}''$  to  $\mathbf{z}^{\dagger}$  has zero resistance.

Together, these claims imply that

$$\pi(\Gamma^*(\mathbf{z}^*)) = \sum_{(\mathbf{s}^*, \mathbf{a}') \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}^*\}} \left( \min\{R(\mathbf{a}'), T\} + 1 \right).$$
(9.3)

The theorem then follows directly from (9.3).

Proof of C1 – First, define

$$\mathcal{C}_A = \{ (\mathbf{s}, \mathbf{a}) \in \mathcal{C} \mid \mathbf{a} \notin \mathcal{A}_{NE} \} \text{ and}$$
$$\mathcal{C}_B = \{ (\mathbf{s}, \mathbf{a}^*) \in \mathcal{C} \mid \mathbf{s} \neq \mathbf{s}^*, \ \mathbf{a}^* \in \mathcal{A}_{NE} \}.$$

Then,  $\{\mathcal{Z}_{NE}, \mathcal{C}_A, \mathcal{C}_B\}$  forms a partition of  $\mathcal{C}$ .

Fix  $\mathbf{z}^{\star} = (\mathbf{s}^{\star}, \mathbf{a}^{\star}) \in \mathcal{Z}_{NE}$ , and consider any minimum resistance  $\mathbf{z}^{\star}$ -tree,  $\Gamma^{*}(\mathbf{z}^{\star}) = (\mathcal{C}, \mathcal{E}^{\star})$ . Let  $\mathbf{S}(\Gamma^{*}(\mathbf{z}^{\star})) = (\tilde{\mathcal{V}}, \tilde{\mathcal{E}})$  denote the Steiner tree in  $\Gamma^{*}(\mathbf{z}^{\star})$  which connects the states in  $\mathcal{Z}_{NE} \setminus \{\mathbf{z}^{\star}\}$  to  $\mathbf{z}^{\star}$ . An example of this is shown in Fig. 9.2, where the states in  $\mathcal{Z}_{NE}$  appear as shaded circles, and the edges in the Steiner tree are shown as solid arrows.

We now argue that

$$\pi(\Gamma^*(\mathbf{z}^*)) = \pi(\mathbf{S}(\Gamma^*(\mathbf{z}^*))). \tag{9.4}$$



Figure 9.2: An example of a minimum resistance  $\mathbf{z}^{\star}$ -tree,  $\Gamma^{*}(\mathbf{z}^{\star})$ , and the corresponding Steiner tree  $\mathbf{S}(\Gamma^{*}(\mathbf{z}^{\star}))$ .

Let  $\mathcal{V}^c = \mathcal{C} \setminus \tilde{\mathcal{V}}$  and  $\mathcal{E}^c = \mathcal{E}^* \setminus \tilde{\mathcal{E}}$ , i.e., the set of edges in  $\Gamma^*(\mathbf{z}^*)$  which do not belong to the Steiner tree.

Since  $\{\mathcal{Z}_{NE}, \mathcal{C}_A, \mathcal{C}_B\}$  is a partition of  $\mathcal{C}$  and  $\mathcal{Z}_{NE} \subset \tilde{\mathcal{V}}$ , we have  $\mathcal{V}^c \subset \mathcal{C}_A \cup \mathcal{C}_B$ . Therefore, for any state  $\mathbf{z} \in \mathcal{V}^c$ , by Lemma 9.2-ii and -iii, there is a zero resistance path from  $\mathbf{z}$  to some  $\mathbf{z}^{\dagger} \in \mathcal{Z}_{NE}$ . Therefore, since  $\Gamma^*(\mathbf{z}^*)$  is assumed to be a minimum resistance  $\mathbf{z}^*$ -tree, the subpath from any state  $\mathbf{z} \in \mathcal{V}^c$  to the Steiner tree  $\mathbf{S}(\Gamma^*(\mathbf{z}^*))$ should have zero resistance; otherwise, we can find another subpath to the Steiner tree with zero resistance, which contradicts the assumption that  $\Gamma^*(\mathbf{z}^*)$  is a minimum resistance  $\mathbf{z}^*$ -tree. This observation means that only the edges in  $\tilde{\mathcal{E}}$  can have positive resistance and the equality in (9.4) follows.

Proof of C2 and C3 – Define

$$\mathcal{C}_C = \{ (\mathbf{s}, \mathbf{a}) \in \mathcal{C} \mid E(\mathbf{s}) \neq \emptyset \}.$$

We first introduce a lemma that will be used to prove the claims. For each  $\mathbf{z}' \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}^{\star}\}$ , we denote the first state in  $\mathcal{C}_{C}$  along the path from  $\mathbf{z}'$  to  $\mathbf{z}^{\star}$  in  $\Gamma^{*}(\mathbf{z}^{\star})$  by  $\zeta(\mathbf{z}', \Gamma^{*}(\mathbf{z}^{\star}))$ .

**Lemma 9.4.** For every  $\mathbf{z} \in \mathcal{Z}_{NE}$  and its minimum resistance tree  $\Gamma^*(\mathbf{z})$ , the following hold.

- i. The (directed) path from any  $\mathbf{z}' \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}\}$  to  $\mathbf{z}$  in  $\Gamma^*(\mathbf{z})$  includes a state in  $\mathcal{C}_C$ .
- ii. For any distinct  $\mathbf{z}'$  and  $\mathbf{z}''$  in  $\mathcal{Z}_{NE} \setminus \{\mathbf{z}\}$ , the directed subpath from  $\mathbf{z}'$  to  $\zeta(\mathbf{z}', \Gamma^*(\mathbf{z}))$ and that from  $\mathbf{z}''$  to  $\zeta(\mathbf{z}'', \Gamma^*(\mathbf{z}))$  are edge disjoint, i.e., the subpaths do not share any edge.
- iii. The resistance of the subpath from  $\mathbf{z}' = (\mathbf{s}^{\star}, \mathbf{a}') \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}\}$  to  $\zeta(\mathbf{z}', \Gamma^*(\mathbf{z}))$  is equal to  $\min\{R(\mathbf{a}'), T\} + 1$ .

**Proof.** Lemma 9.4-i follows from the observation that a transition from  $\mathbf{z}' \in \mathcal{Z}_{NE}$ to another  $\mathbf{z}'' \in \mathcal{Z}_{NE}$  requires some agents to change their *selected* actions. From the description of our algorithm, this is only possible when such agents first change their states from C to E. Lemma 9.4-ii a consequence of the observation that, while transitioning from some  $\mathbf{z}' = (\mathbf{s}^*, \mathbf{a}')$  to  $\zeta(\mathbf{z}', \Gamma^*(\mathbf{z}))$ , the *selected* action profile is fixed at  $\mathbf{a}'$  because the agents can change their *selected* actions only when they are at state E.

Recall that there are only two possible ways for an agent to reach the state E, starting at state C: i) It can move directly to E when it finds an SBR, or ii) it can travel through the sequence of alert states  $(L_{\ell}, \ell \in \mathcal{L})$  when it sees a change in observed payoff information. In the first case, from the definition of the resilience of a PSNE, for every  $\mathbf{z}' = (\mathbf{s}^{\star}, \mathbf{a}') \in \mathcal{Z}_{NE}$ , there exists (i)  $J \subset \mathcal{P}$  with  $|J| = R(\mathbf{a}') + 1$  and  $\mathbf{a}_{J}^{+} \in \prod_{i \in J} \mathcal{A}_{i}$  such that, if the agents in J switch their actions to  $\mathbf{a}_{J}^{+}$ , at least one agent sees an SBR. By definition of the resilience of a PSNE, it is clear that the resistance of such a transition is  $R(\mathbf{a}') + 1$ .

In the second case, using Assumption 8.2, we can find (a) an agent, say i, and action  $a_i^+ \in \mathcal{A}_i$  and (b) another agent  $j \neq i$  such that if agent i switches its action to  $a_i^+$  (from  $a_i'$ ), then agent j sees different payoff information. Therefore, if agent i repeats this T + 1 times (by mistake), then agent j will switch its state to E (through the chain of alert states). The total resistance along this sequence of transitions is equal to T + 1. From these two cases, the resistance of the (directed) path from  $\mathbf{z}'$  to  $\zeta(\mathbf{z}', \Gamma^*(\mathbf{z}))$  is equal to min $\{R(\mathbf{a}'), T\} + 1$ .

Consider a state  $\mathbf{z}' = (\mathbf{s}^{\star}, \mathbf{a}') \in \mathcal{Z}_{NE} \setminus \{\mathbf{z}^{\star}\}$ . Lemma 9.2-i tells us that there exists a path from  $\zeta(\mathbf{z}', \Gamma^*(\mathbf{z}^{\star}))$  to  $\mathbf{z}^{\star}$  with zero resistance. This observation and Lemma 9.4-ii prove the claim C3. In addition, the (directed) subpath from  $\mathbf{z}'$  to

 $\eta(\mathbf{z}', \Gamma^*(\mathbf{z}^*))$  has all of its resistance in the subpath from  $\mathbf{z}'$  to  $\zeta(\mathbf{z}', \Gamma^*(\mathbf{z}^*))$ . Together with Lemma 9.4-iii, this observation proves the claim C2.

# 9.5 Proof of Lemma 9.2

Before proving Lemma 9.2, we prove an intermediate result which claims that starting from any initial state with  $E(\mathbf{s})$  non-empty, we can reach a state where all the agents are exploring (within a finite number of steps and with positive probability). Note that this is similar to Corollary 3 from the previous chapter. While we had proven a similar statement for the RSEM rule, we now prove the statement for the more general SEM rule.

Lemma 9.5. Suppose that the game  $\mathbb{G}$  satisfies Assumption 8.2. Then, under the SEM rule with  $\epsilon = 0$ , there exist  $0 < D_e < \infty$  and  $0 < \rho_e < 1$  such that for all t > 0,  $\mathbf{z} = (\mathbf{s}, \mathbf{a}) \in \mathcal{C}$  with  $E(\mathbf{s}) \neq \emptyset$ ,

$$\mathbb{P}\left[s_i(t+D_e) = E \text{ for all } i \in \mathcal{P} \mid \mathbf{z}(t) = \mathbf{z}\right] \ge \rho_e.$$
(9.5)

**Proof.** First we introduce the following notation: Given a state vector  $\mathbf{s} \in \mathcal{S}$ ,

$$L(\mathbf{s}) = \{i \in \mathcal{P} \mid s_i = L_\ell \text{ for some } \ell \in \mathcal{L} \},$$
$$C(\mathbf{s}) = \{i \in \mathcal{P} \mid s_i = C \}.$$

By definition,  $L(\mathbf{s}) \cup E(\mathbf{s}) \cup C(\mathbf{s}) = \mathcal{P}$ . The proof is a two step process. First we show that with positive probability and in finite time  $t^{\#}$  we reach a state  $\mathbf{z}(t+t^{\#}) = \mathbf{z}^{\#} = (\mathbf{s}^{\#}, \mathbf{a}^{\#})$ , such that  $E(\mathbf{s}^{\#}) \neq \emptyset$ ,  $L(\mathbf{s}^{\#}) = \emptyset$  and  $L(\mathbf{s}(t+t^{\#}+1)) = \emptyset$ . Then, we show that the rest of the agents in the converged state can be driven to the explore state in finite time with positive probability.

Let us define some events: For any time  $\kappa > 0, K > 0, J \subseteq \mathcal{P}$  and  $\bar{\mathbf{a}} \in \mathcal{A}$ ,

$$\mathcal{E}_1(\kappa, K) = \{ E(\mathbf{s}(\ell)) \supseteq E(\mathbf{s}(\ell-1)) \text{ for all } \ell = \kappa + 1, \dots, \kappa + K \},$$
$$\mathcal{E}_2(\bar{\mathbf{a}}, \kappa, K) = \{ a_i(\ell) = \bar{a}_i \text{ for } i \in E(\mathbf{s}(\ell)), \text{ for all } \ell = \kappa, \dots, \kappa + K \},$$
$$\mathcal{E}_3(J, \kappa) = \{ s_i(\kappa) = E, i \in J \}.$$

Fix  $\mathbf{z} = (\mathbf{s}, \mathbf{a}) \in \mathcal{C}$  satisfying  $E(\mathbf{s}) \neq \emptyset$ . Given the state and action configuration at time  $t, L(\mathbf{s}(t+1))$  is deterministic. Consider the case where  $L(\mathbf{s}(t+1)) \neq \emptyset$ . For  $i \in L(\mathbf{s}(t+1))$  it is clear that  $U_i(\tilde{\mathbf{a}}(t)) \neq U_i^*(t)$ . Therefore if the agents who are in the explore state, i.e.,  $E(\mathbf{s}(t+1))$  continue playing their previous actions, agent i would reach state E at least within the next T periods. If agent i reached state Ebefore T periods, we require it to continue playing its previous action for the rest of the time so that the remaining agents in  $L(\mathbf{s}(t+1))$  (if any) also reach state E. Therefore, we have

$$\mathbb{P}\left[E(\mathbf{s}(t+T+1)) = E(\mathbf{s}) \cup L(\mathbf{s}(t+1)) | \mathbf{z}(t) = \mathbf{z}\right]$$
  

$$\geq \mathbb{P}\left[\mathcal{E}_{1}(t,T+1) \cap \mathcal{E}_{2}(\mathbf{a},t,T+1) | \mathbf{z}(t) = \mathbf{z}\right]$$
  

$$= \mathbb{P}\left[\mathcal{E}_{1}(t,T+1) | \mathbf{z}(t) = \mathbf{z}\right] \cdot \mathbb{P}\left[\mathcal{E}_{2}(\mathbf{a},t,T+1) | \mathcal{E}_{1}(t,T+1), \mathbf{z}(t) = \mathbf{z}\right].$$
(9.6)

Lower bounding the first term in (9.6),

$$\mathbb{P}\left[\mathcal{E}_{1}(t,T+1)|\mathbf{z}(t)=\mathbf{z}\right] \\
= \prod_{\ell=t+1}^{t+T+1} \mathbb{P}\left[E(\mathbf{s}(\ell)) \supseteq E(\mathbf{s}(\ell-1))\Big|E(\mathbf{s}(m)) \supseteq E(\mathbf{s}(m-1))\right] \\
\text{for all } m = t+1,\ldots,\ell-1, \mathbf{z}(t) = \mathbf{z}\right] \\
\geqslant \delta^{|E(\mathbf{s}) \cup L(\mathbf{s}(t+1))|(T+1)} \tag{9.7}$$

where the final step follows by noting that in the Explore state each agent plays a particular (desired) action with probability at least  $\delta$ . Moving to the second term in (9.6)

$$\mathbb{P}\left[\mathcal{E}_{2}(\mathbf{a},t,T+1)\big|\mathbf{z}(t) = \mathbf{z}, \mathcal{E}_{1}(t,T+1)\right] \\
= \prod_{\ell=t}^{t+T+1} \mathbb{P}\left[a_{i}(\ell) = a_{i} \text{ for } i \in E(\mathbf{s}(\ell))\big|a_{i}(m) = a_{i} \text{ for } i \in E(\mathbf{s}(m)), \\
m = t, \dots, \ell - 1, \ \mathbf{z}(t) = \mathbf{z}\right] \\
\ge p^{\left|E(\mathbf{s}) \cup L(\mathbf{s}(t+1))\big|(T+1)},$$
(9.8)

where the final bound follows from the fact that each agent at state E will remain in the same state with probability at least p. Collecting the lower bounds and returning to (9.6),

$$\mathbb{P}\left[E(\mathbf{s}(t+T+1)) = E(\mathbf{s}) \cup L(\mathbf{s}(t+1)) \middle| \mathbf{z}(t) = \mathbf{z}\right] \ge (\delta p)^{\left|E(\mathbf{s}) \cup L(\mathbf{s}(t+1))\right|(T+1)}.$$
 (9.9)

Define the set  $J_1 = E(\mathbf{s}) \cup L(\mathbf{s}(t+1))$ . By time  $t_1 = t + T + 1$  we have all the agents either in the explore state or the converged state. Next, we look at events that will force the agents in the converged state to the explore state. By Assumption 8.2, there exists at least one player  $i_1 \notin J_1$  such that with a choice of action  $\mathbf{a}'_{J_1} \in \mathcal{A}_{J_1}$ either

$$U_{i_1}(\mathbf{a}'_{J_1}, \mathbf{a}_{-J_1}) \neq U_{i_1}(\mathbf{a}_{J_1}, \mathbf{a}_{-J_1})$$

or  $BR_{i_1}(\mathbf{a}'_{J_1}, \mathbf{a}_{-J_1}) \neq \emptyset$ . In case there are multiple possible action configurations then choose one using some convention, e.g., lexical.

Denote the set of agents who see a better reply due to the agents in  $J_1$  changing their actions to  $\mathbf{a}'_{J_1}$  by  $B_1^{BR}$ ; while  $B_1^{PI}$  be the agents observing a change in payoff without seeing a better reply. From Assumption 8.2 we know that  $B_1 \equiv B_1^{BR} \cup B_1^{PI} \neq \emptyset$ .

Agents in  $B_1^{BR}$  will transition to the explore state in one time step, whereas the agents in  $B_1^{PI}$  needs T + 1 time periods to transition through the chain of alert states before it can explore. For simplicity, we ensure that the action configuration  $\mathbf{a}_1^{\dagger} = (\mathbf{a}_{J_1}', \mathbf{a}_{-J_1})$  is continued for T + 1 consecutive time steps. We have the following bound

$$\mathbb{P}\left[\mathcal{E}_{3}\left(J_{1}\cup B_{1},t_{1}+T+1\right)\left|\mathbf{z}(t)=\mathbf{z},\mathcal{E}_{3}(J_{1},t_{1})\right]\right]$$
  
$$\geq \mathbb{P}\left[\mathcal{E}_{1}(t_{1},T+1)\cap\mathcal{E}_{2}(\mathbf{a}_{1}^{\dagger},t_{1},T+1)\left|\mathbf{z}(t)=\mathbf{z},\mathcal{E}_{3}(J_{1},t_{1})\right]\right]$$
  
$$\geq (\delta p)^{|J_{1}\cap B_{1}|(T+1)},$$
(9.10)

where the final step follows from similar arguments used to obtain the bound (9.9). Set  $J_2 = J_1 \cap B_1 \supseteq J_1 \cup \{i_1\}$ ; and repeat the same argument noting that  $L(\mathbf{s}(t_1 + T+1)) = \emptyset$  if the events described above occur.

Suppose the following sequence of sets are defined  $\{J_1, \ldots, J_{\alpha}\}$  such that  $J_{\alpha} =$ 

 $\mathcal{P}$  and the following sequence of times are defined  $\{t_1, t_2, \ldots, t_{\alpha}\}$ , where

$$t_j = t + j(T+1), \quad j = 1, ..., \alpha.$$

The time  $t_j$  denotes the time by which all the agents in the set  $J_j$  are in state E if the proper sequence of events is followed. The worst case scenario would be when  $\alpha = n$ . Note that the sequence of sets  $\{J_1, \ldots, J_{\alpha-1}\}$  are not random sets but depends only on the initial state  $\mathbf{z}(t) = (\mathbf{s}, \mathbf{a})$ . Next we move on to show the lower bound as proposed in (9.5), using (9.9) and (9.10). Observing that the event  $\mathcal{E}_3(\mathcal{P}, t_\alpha)$  implies that all the agents are in the Explore state by time  $t_\alpha$ , we obtain

$$\mathbb{P}\left[\mathcal{E}_{3}(\mathcal{P}, t_{\alpha}) | \mathbf{z}(k) = \mathbf{z}\right]$$

$$\geq \mathbb{P}\left[\mathcal{E}_{3}(J_{\ell}, t_{\ell}), \quad \ell = 1, \dots, \alpha | \mathbf{z}(t) = \mathbf{z}\right]$$

$$= \prod_{\ell=1}^{\alpha} \mathbb{P}\left[\mathcal{E}_{3}(J_{\ell}, t_{\ell}) | \mathbf{z}(t) = \mathbf{z}, \mathcal{E}_{3}(J_{m}, k_{m}) \text{ for } m = 1, \dots, \ell - 1\right]$$

$$\geq (\delta p)^{(|J_{1}|+\ldots+|J_{\alpha}|)(T+1)} \geq (\delta p)^{\frac{n(n+1)}{2}(T+1)}.$$
(9.11)

Therefore,  $D_e = n(T+1)$  and set  $\rho_e$  as  $(\delta p)^{\frac{n(n+1)}{2}(T+1)}$ .

Having proved the intermediate result Lemma 9.5, we show the proof for Lemma 9.2. Throughout the proof, for any  $\mathbf{s} \in \Psi'$ , the set of agents in the Explore, Converged and the Alert states are denoted as  $E(\mathbf{s}), C(\mathbf{s})$  and  $L(\mathbf{s})$  respectively. Proof of Lemma 9.2(i) – Fix  $\mathbf{z} = (\mathbf{s}, \mathbf{a}) \in \mathcal{C}$  with  $E(\mathbf{s}) \neq \emptyset$ . By virtue of Lemma 9.5, there exists  $0 < D_e < \infty$  and  $0 < \rho_e < 1$  such that

$$\mathbb{P}\left[s_i(t+D_e) = E, \text{ for all } i \in \mathcal{P} | \mathbf{z}(t) = \mathbf{z}\right] \ge \rho_e.$$
(9.12)

Fix  $\mathbf{a}^* \in \mathcal{A}_{NE}$  and let  $t_e = t + D_e$ . Define the following events

$$\mathcal{E}_1 = \{ s_i(t_e) = E \text{ for all } i \in \mathcal{P} \}$$
$$\mathcal{E}_2 = \{ a_i(t_e) = a_i^* \text{ for all } i \in \mathcal{P} \}$$

Let  $D_1 = D_e + 1$ . Then, we have

$$\mathbb{P}\left[\mathbf{z}(t+D_1) = \mathbf{z}^* \mid \mathbf{z}(t) = \mathbf{z}\right] = \mathbb{P}\left[\mathbf{z}(t+D_1) = \mathbf{z}^*, \mathcal{E}_1, \mathcal{E}_2 \mid \mathbf{z}(k) = \mathbf{z}\right]$$
$$= \mathbb{P}\left[\mathbf{z}(t+D_1) = \mathbf{z}^* \mid \mathcal{E}_1, \mathcal{E}_2, \mathbf{z}(k) = \mathbf{z}\right]$$
(9.13)

$$\times \mathbb{P}\left[\mathcal{E}_1 \mid \mathbf{z}(k) = \mathbf{z}\right] \cdot \mathbb{P}\left[\mathcal{E}_2 \mid \mathcal{E}_1, \mathbf{z}(k) = \mathbf{z}\right] \qquad (9.14)$$

The term (9.13) corresponds to the probability that all the agents go to the Converged state conditioned on the events that at time  $t_e$  all the agents reached the Explore state and played the equilibrium action corresponding to  $\mathbf{a}^*$ . Hence, at time  $D_1$  none of the agents will see an SBR, and as a result the conditional probability in (9.13) can be lower bounded by  $(1-p)^n$ . From (9.12), the first term in (9.14) is lower bounded by  $\rho_e > 0$ . Further, the second conditional probability in (9.14) which corresponds to a particular action profile being played by agents in the Explore state, is lower bounded by  $\delta^n$  from the description of the SEM algorithm in Section 9.2. Putting the bounds together, we obtain

$$\mathbb{P}\left[\mathbf{z}(t+D_1) = \mathbf{z}^* \mid \mathbf{z}(t) = \mathbf{z}\right] \ge \rho_e[\delta(1-p)]^n.$$
(9.15)

Proof of Lemma 9.2(ii) – Fix  $\mathbf{z} = (\mathbf{s}, \mathbf{a}^*) \in \mathcal{C}$  with  $\mathbf{a}^* \in \mathcal{A}_{NE}$  and  $\mathbf{s} \in \Psi' \setminus \{\mathbf{s}^*\}$ . Within one time interval, the agents in  $E(\mathbf{s})$  could move (with positive probability) to the converged state because they see no better reply, which yields

$$\mathbb{P}\left[s_i(t+1) = C \text{ for all } i \in E(\mathbf{s}) \mid \mathbf{z}(t) = \mathbf{z}\right] \ge (1-p)^{|E(\mathbf{s})|}.$$
(9.16)

For those agents in  $L(\mathbf{s})$  who still continue to see a change in payoff information due to the action configuration at time t, they move to the next alert state or the explore state if they were at the final alert state. Otherwise they move to the converged state. Therefore, we now look at events which ensure that agents in  $E(\mathbf{s}(t+1))$  play their equilibrium action and move to the converged state; and agents in  $L(\mathbf{s}(t+1))$  traverse the chain of alert states – move to the explore state – play their equilibrium action and finally end up in the converged state. Let  $D_2 = T + 1$ . Therefore,

$$\mathbb{P}\left[\mathbf{z}(t+D_2) = \mathbf{z}^* \mid s_i(t+1) = C \text{ for all } i \in E(\mathbf{s}), \mathbf{z}(t) = \mathbf{z}\right]$$
$$\geq \left[\delta(1-p)\right]^{n-|E(\mathbf{s})|}, \tag{9.17}$$

where we assume all the agents in  $L(\mathbf{s}(t+1))$  are in mood  $L_1$  as a worst case scenario, and we use the fact that under the conditioning event,  $C(\mathbf{s}(t+1)) = E(\mathbf{s})$ . Combining (9.16) and (9.17) we obtain

$$\mathbb{P}\left[\mathbf{z}(t+D_2) = \mathbf{z}^* \mid \mathbf{z}(t) = \mathbf{z}\right] \ge \left[\delta(1-p)\right]^n.$$

Proof of Lemma 9.2(iii) – Fix  $\mathbf{z} = (\mathbf{s}, \mathbf{a}) \in \mathcal{C}$  such that  $\mathbf{a} \notin \mathcal{A}_{NE}$ , then

$$\mathbb{P}\left[E(\mathbf{s}(t+1))\neq\emptyset|\mathbf{z}(t)=\mathbf{z}\right]=1.$$

Once, the number of agents in state E becomes positive we are back to Lemma 9.2(i).

## Chapter 10: Future work

In this part of the thesis we study two classes of learning rules that ensure convergence to PSNE(s) under delayed payoff information and asynchronous decisionmaking.

For the class of better reply rules, we consider faulty payoff information in that the agents cannot estimate the better reply set reliably. We show that the set of stochastically stable states, i.e., the states that are played most of the time as the payoff information becomes more reliable, are a subset of the set of PSNE(s). However due to difficulty in characterising the set of stochastically stable states we consider the class of monitoring rules. We first look at the setting where the erroneous decision-making of the agents are due to occasional misclassification of better replies while the payoff for the played action is assumed to be accurate. For this scenario, we are able to characterise the set of stochastically stable states as the PSNE(s) which when in effect would make it less likely for any of the agents to see a better reply. As a future work, we would like to formulate a more realistic setting where not only the payoff classification information is spurious but the payoff for the played action is also not accurate. Under such a setting, we would like to explore if the set of stochastically stable states can be characterised when the agents follow
the monitoring rules.

Since the existence of PSNE is not always guaranteed in strategic-form games, we would like to explore other learning rules that might yield convergence (in an appropriate sense) to possibly other kinds of equilibria (mixed Nash equilibria, correlated equilibria) under delayed payoff information and asynchronous settings.

## Bibliography

- R. Albert and A.L. Barabási, "Statistical mechanics of complex systems," Review of Modern Physics 74 (2002), pp. 47-97.
- [2] A.L. Barabási and R. Albert, "Emergence of scaling in random networks," Science 286 (1999), pp. 509-512.
- [3] A. Barrat, M. Barthelemy and A. Vespignani, Dynamical processes on Complex Networks, Cambridge University Press, Cambridge (U.K.), 2008.
- [4] G. Bianconi and A.L. Barabási, "Competition and multiscaling in evolving networks," *Europhysics Letters* 54.4 (2001): 436.
- [5] P. Billingsley, Convergence of Probability Measures, John Wiley & Sons, New York (NY), 1968.
- [6] M. Boguñá and R. Pastor-Storras, "Class of correlated random networks with hidden variables," *Physical Review* E 68 (2003), 036112.
- [7] B. Bollobás, *Random Graphs*, Second Edition, Cambridge Studies in Advanced Mathematics, Cambridge University Press, Cambridge (UK), 2001.
- [8] T. Britton, M. Deijfen and A. Martin-Löf, "Generating simple random graphs with prescribed degree distribution," *Journal of Statistical Physics* 124.6 (2006): 1377-1397.
- [9] G. Caldarelli, A. Capocci, P. De Los Rios and M.A. Muñoz, "Scale-free networks from varying vertex intrinsic fitness," *Physical Review Letters* 89 (2002), 258702.
- [10] R. Cohen and S. Havlin, Complex Networks: Structure, Robustness and Function, Cambridge University Press, Cambridge (U.K.), 2010.

- [11] H.A. David and H.N. Nagaraja, Order Statistics, 3rd Edition, Wiley Series in Probability and Statistics, John Wiley & Sons, Hoboken (NJ), 2003.
- [12] P. Diaconis, S. Holmes and S. Janson, "Threshold graph limits and random threshold graphs," *Internet Mathematics* 5 (2008), pp. 267-320
- [13] M. Draief and L. Massoulié, Epidemics and Rumours in Complex Networks, London Mathematical Society Lecture Note Series 369, Cambridge University Press, Cambridge (UK), 2010.
- [14] S.N. Dorogovstev and J.F.F. Mendes, "Evolution of networks," Advances in Physics 51 (2002), pp. 1097-1187.
- [15] R. Durrett, Random Graph Dynamics, Cambridge Series in Statistical and probabilistic mathematics, Cambridge University Press, Cambridge (U.K.), 2007.
- [16] P. Embrechts, C. Klüppelberg and T. Mikosch, Modelling Extremal Events for Insurance and Finance, Stochastic Modelling and Applied Probability, Springer-Verlag, New York (NY), 1997.
- [17] P. Erdős and A. Rényi, "On the evolution of random graphs," Publ. Math. Inst. Hung. Acad. Sci 5 (1960), pp. 17-61.
- [18] G. Ergün and G. J. Rodgers, "Growing random networks with fitness," *Physica A: Statistical Mechanics and its Applications* 303.1 (2002): 261-272.
- [19] L. Eschenauer and V.D. Gligor, "A key-management scheme for distributed sensor networks," in Proc. 9th ACM Conf. Comput. Commun. Security, Washington, DC, Nov. 2002, pp. 41-47.
- [20] D. Estrin, R. Govindan, J. Heidemann, S. Kumar, "Next century challenges: Scalable coordination in sensor networks," in *Proc. of the ACM/IEEE International Conference on Mobile Computing and Networking*, 1999, pp. 263-270.
- [21] W. Feller, An introduction to probability theory and its applications, Volume II, John Wiley & Sons, 2008.
- [22] A. Fujihara, Y. Ide, N. Konno, N. Masuda, H. Miwa and M. Uchida, "Limit theorems for the average distance and the degree distribution of the threshold network model," *Interdisciplinary Information Sciences* 15 (2003), pp. 361-366.

- [23] S. Ghadge et al. "A statistical construction of power-law networks," International Journal of Parallel, Emergent and Distributed Systems 25.3 (2010): 223-235.
- [24] E. Godehardt and J. Jaworski, "Two models of random intersection graphs for classification," in *Studies in Classification*, *Data Analysis and Knowledge Organization*, O. Optiz and M. Schwaiger, Eds. Berlin, Germany: Springer-Verlag, 2003, vol. 22, pp. 67-82.
- [25] E. Godehardt, J. Jaworski, and K. Rybarczyk, "Random intersection graphs and classification," in *Studies in Classification*, *Data Analysis and Knowledge Organization*, H. Lens and R. Decker, Eds. Berlin, Germany: Springer-Verlag, 2007, vol. 33, pp. 67-74.
- [26] M.O. Jackson, Social and Economic Networks, Princeton University Press, Princeton (NJ), 2008.
- [27] S. Janson, T. Luczak and A. Ruciński, *Random Graphs*, Wiley-Interscience Series in Discrete Mathematics and Optimization, John Wiley & Sons, 2000.
- [28] A. M. Makowski and O. Yağan, "Scaling laws for connectivity in random threshold graph models with non-negative fitness variables," *IEEE Journal on Selected Areas in Communications* **JSAC–31** (2013), Special Issues on Emerging Technologies in Communications (Area 4: Social Networks).
- [29] P. Marbach, "A lower-bound on the number of rankings required in rec- ommender systems using collaborative filtering," in *Proc. 42nd Annu. Conf. Inf. Sci. Syst.*, Princeton, NJ, Mar. 2008, pp. 292-297.
- [30] M.E.J. Newman, "The structure and function of complex networks," SIAM Review 45 (2003), pp. 167-256.
- [31] M.E.J. Newman, Networks: An Introduction, Oxford University Press, Oxford (U.K.), 2010.
- [32] M.E.J. Newman, A.L. Barabási and D.J. Watts, Editors, The Structure and Dynamics of Networks, Princeton University Press, Princeton (NJ), 2006.
- [33] S. Pal and A.M. Makowski, "On the asymptotics of degree distributions," 53rd IEEE Conference on Decision and Control (CDC 2015), Osaka (Japan), December 2015.
- [34] M.D. Penrose, Random Geometric Graphs, Oxford Studies in Probability 5, Oxford University Press, New York (NY), 2003.

- [35] G.J. Pottie, W.J. Kaiser, Wireless integrated network sensors, Commun. ACM 43 (5) (2000) 51-58.
- [36] S. Ross, Stochastic Processes, J. Wiley & Sons, New York (NY), 1984.
- [37] V.D.P. Servedio and G. Caldarelli, "Vertex intrinsic fitness: How to produce arbitrary scale-free networks," *Physical Review E* **70** (2004), 056126
- [38] O. Yağan and A.M. Makowski, "Zero-one laws for connectivity in random key graphs." *IEEE Transactions on Information Theory* **IT-58** (2012), pp. 2983-2999.
- [39] C. Alós-Ferrer and N. Netzer, "The logit-response dynamics," Games and Economic Behavior, 68(2):413-427, Mar. 2010.
- [40] J. Antoniou and A. Pitsillides, *Game Theory in Communication Networks:* Cooperative Resolution of Interactive Networking Scenarios, CRC Press, 2012.
- [41] G. Arslan, J.R. Marden, and J. S. Shamma, "Autonomous vehicle-target assignment: A game-theoretical formulation," *Journal of Dynamic Systems, Mea*surement, and Control, 129(5):584-596, Apr. 2007.
- [42] D. Bertsekas and R. Gallager, *Data Networks*, 2nd ed., Prentice Hall, 1992.
- [43] L.E. Blume, "The statistical mechanics of strategic interaction," Games and Economic Behavior, 5(3):387-424, Jul. 1993.
- [44] P. Brémaud, Markov Chains: Gibbs Fields, Monte Carlo Simulation, and Queues, Springer, 1999.
- [45] G.W. Brown, "Iterative solution of games by fictitious play," Activity Analysis of Production and Allocation, edited by T.C. Koopmans, Wiley, 1951.
- [46] D.E. Charilas and A.D. Panagopoulos, "A survey on game theory applications in wireless networks," *Computer Networks*, 54(18):3421-3430, Dec. 2010.
- [47] S.H. Chun and R.J. La, "Secondary spectrum trading auction-based framework for spectrum allocation and profit sharing," *IEEE/ACM Trans. of Net*working, 21(1):176-189, Feb. 2013.
- [48] L. Fang, and P.J. Antsaklis, "Information consensus of asynchronous discretetime multi-agent systems," Proceedings of American Control Conference, Portland (OR), Jun. 2005.

- [49] D.P. Foster and R.V. Vohra, "Calibrated learning and correlated equilibrium," Games and Economic Behavior, 21(1-2): 40-55, Oct. 1997.
- [50] D.P. Foster and R.V. Vohra, "Asymptotic calibration," *Biometrika*, 85.2:379-390, 1998.
- [51] D.P. Foster and H.P. Young, "Regret testing: A simple payoff-based procedure for learning Nash equilibrium," University of Pennsylvania and Johns Hopkins University (mimeo), 2003.
- [52] D.P. Foster and H.P. Young, "Learning, hypothesis testing, and Nash equilibrium," *Games and Economic Behavior*, 45(1):73-96, Oct. 2003.
- [53] M.I. Freidlin and A.D. Wentzell, Random Perturbations of Dynamical Systems, second ed., Springer, 1998.
- [54] C.S. Fisk, "Game theory and transportation systems modelling," Transportation Research Part B: Methodological, 18B(4/5):301-313, 1984.
- [55] D. Fudenberg and D.K. Levine, The Theory of Learning in Games, The MIT Press, 1998.
- [56] D. Fudenberg and D.K. Levine, "Conditional universal consistency," Games and Economic Behavior, 29(1-2):104-130, Oct. 1999.
- [57] F. Germano and G. Lugosi, "Global Nash convergence of Foster and Young's regret testing," *Games and Economic Behavior*, 60(1):135-154, Jul. 2007.
- [58] G.R. Grimmett and D.R. Stirzaker, *Probability and Random Processes*, third edition, Oxford University Press, 2001.
- [59] S. Hart and A. Mas-Colell, "A simple adaptive procedure leading to correlated equilibrium," *Econometrica* 68(5):1127-1150, Sep. 2000.
- [60] S. Hart and A. Mas-Colell, "Regret-based continuous-time dynamics," Games and Economic Behavior, 45(2):375-394, Nov. 2003.
- [61] S. Hart, "Adaptive heuristics," *Econometrica* 73.5: 1401-1430, 2005.
- [62] S. Hart and A. Mas-Colell, "A reinforcement procedure leading to correlated equilibrium," *Economics Essays*, ed. by G. Debreu, W. Neuefeind and W. Trockel, pp.181-200, Springer Berlin Heidelberg, 2001.

- [63] S. Hart and A. Mas-Colell, "Uncoupled dynamics do not lead to Nash equilibrium," *American Economic Review*, 93(5):1830-1836, Dec. 2003.
- [64] S. Hart and Y. Mansour, "How long to equilibrium? The communication complexity of uncoupled equilibrium procedures," *Games and Economic Behavior*, 69(1):107-126, May 2010.
- [65] Y. Hollander and J.N. Prashker, "The applicability of non-cooperative game theory in transport analysis," *Transportation*, 33(5):481-496, Sep. 2006.
- [66] J.S. Jordan, "Bayesian learning in normal form games," Games and Economic Behavior 3(1): 60-81, Feb. 1991.
- [67] J.S. Jordan, "Bayesian learning in repeated games," Games and Economic Behavior 9(1): 8-20, Apr. 1995.
- [68] E. Kalai and E. Lehrer, "Rational learning leads to Nash equilibrium," Econometrica: Journal of the Econometric Society: 1019-1045, 1993.
- [69] M. Kandori, G.J. Mailath and R. Rob, "Learning, mutation, and long run equilibria in games," *Econometrica: Journal of the Econometric Society*: 29-56, 1993.
- [70] S. Kim, *Game Theory Applications in Network Design*, Information Science Reference, 2014.
- [71] R.J. La and V. Anantharam, "Optimal routing control: repeated game approach," *IEEE Trans. on Automatic Control*, 47(3):437-450, Mar. 2002.
- [72] H. Kunreuther and G. Heal, "Interdependent security," Journal of Risk and Uncertainty, 26(2/3):231-249, 2003.
- [73] J.R. Marden, G. Arslan and J.S. Shamma, "Connections between cooperative control and potential games illustrated on the consensus problem," Proceedings of European Control Conference (ECC), Kos (Greece), Jul. 2007.
- [74] J.R. Marden, G. Arslan and J.S. Shamma, "Regret based dynamics: convergence in weakly acyclic games," Proceedings of the 6th International Conference on Autonomous Agents and Multiagent Systems (AAMAS), Honolulu (HI), May 2007.

- [75] J.R. Marden, H.P. Young, G. Arslan and J.S. Shamma, "Payoff-based dynamics for multiplayer weakly acyclic games," SIAM Journal on Control and Optimization, 48(1):373-396, 2009.
- [76] J.R. Marden, G. Arslan and J.S. Shamma, "Joint strategy fictitious play with inertia for potential games," *IEEE Trans. on Automatic Control*, 54(2):208-220, Feb. 2009.
- [77] J.R. Marden, S.D. Ruben and L.Y. Pao, "A model-free approach to wind farm control using game theoretic methods," *IEEE Transactions on Control Systems Technology*, 21.4 (2013): 1207-1214.
- [78] J.R. Marden and J.S. Shamma, "Revisiting log-linear learning: Asynchrony, completeness and payoff-based implementation," *Games and Economic Behavior*, 75(2):788-808, Jul. 2012.
- [79] J.R. Marden, H.P. Young and L.Y. Pao, "Achieving pareto optimality through distributed learning," Proceedings of the 51st IEEE Conference on Decision and Control (CDC), Maui (HI), Dec. 2012.
- [80] A. Menon, and J.S. Baras, "A distributed learning algorithm with bit-valued communications for multi-agent welfare optimization," Proceedings of the 52nd IEEE Conference on Decision and Control (CDC), Florence (Italy), Dec. 2013.
- [81] J. Miękisz, M. Matuszak, and J. Poleszczuk, "Stochastic stability in threeplayer games with time delays," *Dynamic Games and Applications* 4.4 (2014): 489-498.
- [82] P. Milgrom and J. Roberts, "Adaptive and sophisticated learning in normal form games," *Games and Economic Behavior*, 3(1):82-100, Febb. 1991.
- [83] D. Monderer and L.S. Shapley, "Fictitious play property for games with identical interests," *Journal of Economic Theory*, 68:258-265, 1996.
- [84] D. Monderer and L.S. Shapley, "Potential games," Games and Economic Behavior, 14(1):124-143, May 1996.
- [85] S. Pal and R.J. La, "A simple learning rule in games and its convergence to pure-strategy Nash equilibria," Proceedings of American Control Conference, Chicago (IL), Jul. 2015.
- [86] S. Pal and R.J. La, "A simple learning rule with monitoring leading to Nash Equilibrium under delays," International Conference on Game Theory, Stony Brook (NY), Jul. 2015.

- [87] B.S.R. Pradelski and H.P. Young, "Learning efficient Nash equilibria in distributed systems," *Games and Economic Behavior*, 75(2):882-897, Jul. 2012.
- [88] R.W. Rosenthal, "A class of games possessing pure-strategy Nash equilibria," International Journal of Game Theory, 2(1):65-67, 1973.
- [89] S.M. Ross, *Stochastic Processes*, 2nd edition, Wiley, 1995.
- [90] M.J. Sanjari and G.B. Gharehpetian, "Game-theoretic approach to cooperative control of distributed energy resources in islanded microgrid considering voltage and frequency stability," *Neural Computing and Applications*, 25(2):343-351, Aug. 2014.
- [91] M. Shaked and J.G. Shanthikumar, *Stochastic Orders*, Springer Series in Statistics, Springer, 2007.
- [92] J.S. Shamma, Cooperative Control of Distributed Multi-Agent Systems, Wiley, 2008.
- [93] F. Xiao, and L. Wang, "State consensus for multi-agent systems with switching topologies and time-varying delays," *International Journal of Control* 79(10):1277-1284, 2006.
- [94] T. Yi, and W. Zuwang, "Effect of time delay and evolutionarily stable strategy," Journal of Theoretical Biology 187(1):111-116, 1997.
- [95] H.P. Young, "The evolution of conventions," Econometrica: Journal of the Econometric Society, 61(1):57-84, Jan. 1993.
- [96] H.P. Young, "Learning by trial and error," *Games and Economic Behavior* 65(2):626-643, Mar. 2009.